

VERLAG DER TECHNISCHEN UNIVERSITÄT GRAZ

### **Christian Stelzer-Landauer**

## Approximation of Dirac Operators with Delta-Shell Potentials in the Norm Resolvent Sense

#### **CES 47**

**MONOGRAPHIC SERIES TU GRAZ** COMPUTATION IN ENGINEERING AND SCIENCE



**Christian Stelzer-Landauer** 

#### Approximation of Dirac Operators with Delta-Shell Potentials in the Norm Resolvent Sense

Monographic Series TU Graz Computation in Engineering and Science

#### Series Editors

G. Brenn	Institute of Fluid Mechanics and Heat Transfer
G. A. Holzapfel	Institute of Biomechanics
W. von der Linden	Institute of Theoretical and Computational Physics
M. Schanz	Institute of Applied Mechanics
O. Steinbach	Institute of Applied Mathematics

#### Monographic Series TU Graz

Computation in Engineering and Science

**Christian Stelzer-Landauer** 

# Approximation of Dirac Operators with Delta-Shell Potentials in the Norm Resolvent Sense

This work is based on the dissertation "Approximation of Dirac Operators with  $\delta$ -shell Potentials in the Norm Resolvent Sense", presented at Graz University of Technology, Institute of Applied Mathematics in 2024.

Supervision / Assessment:

Jussi Behrndt (Graz University of Technology) Albert Mas (Universitat Politècnica de Catalunya) Matěj Tušek (Czech Technical University in Prague)

Cover	Verlag der Technischen Universität Graz
Cover photo	Vier-Spezies-Rechenmaschine
	by courtesy of the Gottfried Wilhelm Leibniz Bibliothek
	Niedersächsische Landesbibliothek Hannover
Printed by	Buchschmiede (DATAFORM Media GmbH)

2025 Verlag der Technischen Universität Graz www.tugraz-verlag.at

Print ISBN 978-3-99161-052-6

E-Book ISBN 978-3-99161-053-3 DOI 10.3217/978-3-99161-052-6



This work is licensed under the Creative Commons Attribution 4.0 International (CC BY 4.0) license. https://creativecommons.org/licenses/by/4.0/deed.en

This CC license does not apply to the cover, third party material (attributed to other sources) and content noted otherwise.

#### Abstract

The present thesis is devoted to the approximation of Dirac operators with  $\delta$ -shell potentials supported on the boundary of a two or three-dimensional  $C^2$ -domain. These singular potentials are used as idealized replacements for potentials which are strongly localized in a neighbourhood of the support of the  $\delta$ -shell potential and they often simplify the spectral analysis. To justify the usage of such potentials it is essential to prove that Dirac operators with  $\delta$ -shell potentials can be approximated by Dirac operators with strongly localized potentials in a way which transfers the spectral properties. The most important contribution of this thesis is the establishment of conditions for the convergence of Dirac operators with strongly localized potentials in the norm resolvent sense. This type of convergence implies that the spectrum of the Dirac operator with  $\delta$ -shell potential can be completely characterized by the spectra of the approximating operators and vice versa. In the special case of electrostatic and Lorentz scalar  $\delta$ -shell potentials an explicit convergence condition is provided. Furthermore, counterexamples which imply the sharpness of this condition are also presented.

#### Zusammenfassung

Das Ziel dieser Dissertation ist es, Dirac Operatoren mit  $\delta$ -Potentialen, welche auf dem Rand eines zwei- oder dreidimensionalen  $C^2$ -Gebietes definiert sind, zu approximieren. Derartige  $\delta$ -Potentiale werden als Idealisierung von regulären Potentialen gesehen, welche stark in der Umgebung des Trägers des  $\delta$ -Potentials lokalisert sind. Um eine Verwendung von solchen Potentialen rechtzufertigen, muss gezeigt werden, dass Dirac Operatoren mit  $\delta$ -Potentialen durch Dirac Operatoren mit stark lokalisierten Potentialen auf eine Weise angenähert werden können, welche auch spektrale Eigenschaften überträgt. Der wichtigste Beitrag dieser Arbeit zur aktuellen Forschung ist die Angabe von Bedingungen für die Konvergenz im Normresolventensinn. Konvergenz in diesem Sinn impliziert, dass das Spektrum des Dirac Operatoren mit  $\delta$ -Potential vollständig durch die Spektren der approximierenden Operatoren charakterisiert werden kann und umgekehrt dasselbe gilt. Für den Spezialfall von elektrostatischen und Lorentz-skalaren  $\delta$ -Potentialen wird eine explizite Konvergenzbedinung angegeben. Durch passende Gegenbeispiele wird zusätzlich gezeigt, dass die Bedingung auch scharf ist.

#### Contents

1	Intr	oduction	3
	1.1	Description of the problem	3
	1.2	State of the art	4
	1.3	Rigorous definition of various objects	6
	1.4	New results and structure of this thesis	8
<b>2</b>	Pre	liminaries	13
	2.1	Notations	13
	2.2	Special $C^2$ -surfaces and corresponding Sobolev spaces	17
	2.3	Tubular neighbourhoods of special $C^2$ -surfaces	21
	2.4	Bochner spaces	27
	2.5	Norm resolvent convergence	32
	2.6	Invertibility of bounded operators	35
3	The	e free Dirac operator and perturbed Dirac operators	37
	3.1	The free Dirac operator and associated integral operators	37
	3.2	Dirac operators with regular potentials	44
	3.3	Dirac operators with $\delta$ -shell potentials $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	46
4	Nor	m resolvent convergence of Dirac operators with general strongly	
	loca	lized potentials	51
	4.1	A resolvent formula for Dirac operators with strongly localized potentials	52
	4.2	The shift operator	54
	4.3	Convergence of $A_{\varepsilon}(z)$ , $B_{\varepsilon}(z)$ and $C_{\varepsilon}(z)$	59
	4.4	Properties of the limit operators $A_0(z)$ , $B_0(z)$ and $C_0(z)$	66
	4.5	Convergence conditions for Dirac operators with general strongly lo-	
		calized potentials	73
<b>5</b>	An	explicit convergence condition for Dirac operators with strongly	
	loca	lized electrostatic and Lorentz scalar potentials	77
	5.1	Analysis of $I + B_{\varepsilon}(z)Vq$ for rotated $C_{b}^{2}$ -graphs	78
		5.1.1 Hyperplanes and constant interaction strengths	80
		5.1.2 General rotated $C_b^2$ -graphs	92
	5.2	Main results	108
6	Cou	Interexamples	119

<b>7</b>	Cor	sequences of the approximation results	133
	7.1	Approximation of Dirac operators with $\delta$ -shell potentials that induce	
		confinement	133
	7.2	Spectra of Dirac operators with strongly localized potentials	144
8	Cor	vergence of Dirac operators with semilocal potentials	149
	8.1	General interactions	150
	8.2	An explicit condition for electrostatic and Lorentz scalar interactions	153
$\mathbf{A}$	ppen	dix A. Partitions of unity	160
$\mathbf{A}$	ppen	dix B. Proof of the estimate (4.34)	163
$\mathbf{A}$	ppen	dix C. Additional results for Section 5.1.2 and Section 8.2	176
Re	efere	nces	181

#### Acknowledgment

I want to make use of the opportunity to thank the following people, without whom this thesis would not have been possible. First, I would like to thank my supervisor Prof. Jussi Behrndt as well as Dr. Markus Holzmann for their guidance and support throughout this process. I would also like to express my gratitude towards my colleagues Georg Stenzel and Prof. Petr Siegl for taking their time to proofread the introduction and to give feedback. Moreover, I would like to thank Nicolas Weber for interesting conversations which led to improvements within this thesis. Thanks also to Dr. Albert Mas and Dr. Matěj Tušek for kindly agreeing to evaluate this thesis. Lastly, I must thank my family, especially my wife, for their unwavering support and understanding while writing and completing this thesis.

Thank you all.

#### 1 Introduction

#### 1.1 Description of the problem

Differential operators coupled with singular potentials are frequently used in mathematical physics. Such singular potentials model regular potentials which have very large values in the vicinity of a set of measure zero and small values everywhere else. In contrast to regular potentials, they cannot be represented by functions and have to be described by distributions. Nonetheless, the spectral analysis of differential operators with singular potentials often simplifies substantially and may even reduce to an explicitly solvable problem.

To justify the replacement of regular potentials by singular potentials, approximation results are necessary, i.e. if H is a differential operator and S is a singular potential, one has to show that H+S can be approximated by  $H+V_{\varepsilon}$  as  $\varepsilon \to 0$  in a suitable sense, where  $(V_{\varepsilon})_{\varepsilon>0}$  is a family of regular potentials which converges in the distributional sense to S. Here, "suitable sense" means in particular that the convergence should relate the spectra as well as the associated spectral projections of H+S and  $H+V_{\varepsilon}$ . In the context of self-adjoint unbounded operators, appropriate notions of convergence are the strong resolvent convergence and the norm resolvent convergence, which means that  $(H + V_{\varepsilon} - i)^{-1}$  converges to  $(H + S - i)^{-1}$  as  $\varepsilon \to 0$  in the strong sense or in the operator norm, respectively.

In this thesis we focus on the approximation of Dirac operators with  $\delta$ -shell potentials. We start by explaining Dirac operators. They were introduced by Paul Dirac in 1929 and are used to describe spin 1/2 particles in a quantum mechanical framework. Moreover, in contrast to Schrödinger operators, they also comply with the theory of relativity. The free Dirac operator without any potential has the following form in natural units:

$$H = -i\sum_{j=1}^{\theta} \alpha_j \partial_j + m\beta, \qquad \operatorname{dom} H = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N) \subset L^2(\mathbb{R}^{\theta}; \mathbb{C}^N).$$
(1.1)

Here,  $\theta \in \{1, 2, 3\}$  denotes the space dimension,  $m \in \mathbb{R}$  describes the mass of a particle and  $\alpha_1, \ldots, \alpha_{\theta}, \beta \in \mathbb{C}^{N \times N}, N = 2 \lceil \frac{\theta}{2} \rceil$ , are the Dirac matrices introduced in Definition 3.1 for  $\theta \in \{2, 3\}$ . For  $\theta \in \{1, 2\}$  one has  $\alpha_j = \sigma_j, j \in \{1, \ldots, \theta\}$ , and

 $\beta = \sigma_3$ , with the Pauli matrices

$$\sigma_1 = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \sigma_2 = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \sigma_3 = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}.$$

Adding a symmetric matrix-valued potential S to H allows one to also model the influence of external fields, e.g. electrostatic, Lorentz scalar or magnetic fields. In this thesis we are interested in the case  $S = V\delta_{\Sigma}$ , where  $\delta_{\Sigma}$  is the singular  $\delta$ -shell potential supported on a  $C^2$ -smooth hypersurface  $\Sigma \subset \mathbb{R}^{\theta}$  and V is a symmetric matrix-valued function on  $\Sigma$ . If  $V = \eta I_N + \tau \beta + \lambda i(\alpha \cdot \nu)\beta + \omega(\alpha \cdot \nu)$ , where  $\nu$ is the unit normal vector on  $\Sigma$  and  $\alpha \cdot \nu = \sum_{j=1}^{\theta} \alpha_j \nu_j$ , we call the scalar functions  $\eta, \tau, \lambda$  and  $\omega$  the electrostatic, Lorentz scalar, anomalous magnetic and magnetic interaction strengths, respectively.

The main aim of this thesis is to study the norm resolvent convergence of the operators  $H_{V_{\varepsilon}} = H + V_{\varepsilon}$ , where  $(V_{\varepsilon})_{\varepsilon>0}$  is a family of regular potentials which converges for  $\varepsilon \to 0$  in the distributional sense to  $V\delta_{\Sigma}$ .

#### 1.2 State of the art

For Schrödinger operators, which are the nonrelativistic counterparts of Dirac operators, the literature regarding such approximation results is extensive. It is well-known that in the one-dimensional setting, where  $\Sigma$  is either a single point or a countable set of points, Schrödinger operators with  $\delta$ -potentials can be approximated in the norm resolvent sense by Schrödinger operators with strongly localized potentials; see for instance [1] and the references therein. In two and three dimensions this problem was also considered for various choices of  $\Sigma$ ; see [2, 30, 31, 58, 68]. Furthermore, in [7] norm resolvent convergence was proven for a general class of  $C^2$ -smooth hypersurfaces in the multidimensional setting.

Approximation problems for one-dimensional Dirac operators with  $\Sigma = \{0\}$  were first considered in 1989 by Šeba in [67]. He investigated Dirac operators with potentials of the form  $V_{\varepsilon} = Vh_{\varepsilon}$ , where  $V = \eta I_2$ ,  $\eta \in \mathbb{R}$ , or  $V = \tau\beta$ ,  $\tau \in \mathbb{R}$ , and  $(h_{\varepsilon})_{\varepsilon>0}$  is a suitable family of functions converging for  $\varepsilon \to 0$  in the distributional sense to  $\delta_0$ , which is the  $\delta$ -potential supported in  $\{0\}$ . In this setting he was able to show that  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense to the operator  $H_{\tilde{V}\delta_0}$ . Here,  $\tilde{V} = \tilde{\eta}I_2$  or  $\tilde{V} = \tilde{\tau}\beta$ , where  $\tilde{\eta}$  and  $\tilde{\tau}$  are rescaled interaction strengths which depend nonlinearly on  $\eta$  and  $\tau$ , respectively. This rescaling does not appear in the case of Schrödinger operators and had already been observed a few years prior in various physics papers, see e.g. [22, 52, 53], when comparing the solutions of the Dirac eigenvalue equation with  $\delta$ potentials and strongly localized potentials. Furthermore, Šeba and the authors of [52, 53] related this phenomenon to Klein's paradox. In the nineties Hughes showed in [40, 41, 42] that  $H_{Vh_{\varepsilon}}$  converges in the strong resolvent sense to  $H_{\widetilde{V}\delta_0}$  for selfadjoint matrices  $V \in \mathbb{C}^{2\times 2}$ . Moreover, she was also able to find an explicit formula for the rescaling of V to  $\widetilde{V}$ ; cf. [42, Theorem 1 and Theorem 2]. In our terminology this formula is given by

$$\widetilde{V} = 2\alpha_1 \sin\left(\frac{\alpha_1 V}{2}\right) \cos\left(\frac{\alpha_1 V}{2}\right)^{-1}$$
(1.2)

provided that  $\cos\left(\frac{\alpha_1 V}{2}\right)$  is an invertible matrix. Finally, in 2020, Tušek extended in [72] the works of Šeba and Hughes by proving norm resolvent convergence in the onedimensional setting for a large class of self-adjoint interaction matrices  $V \in \mathbb{C}^{2\times 2}$ .

In the multidimensional setting the literature is less complete than in the onedimensional case and so far there only exist results on strong resolvent convergence. In this setting one defines based on a matrix-valued function V given on  $\Sigma$ , for  $\varepsilon > 0$ a potential  $V_{\varepsilon}$  which is supported in an  $\varepsilon$ -neighbourhood of  $\Sigma$  and converges for  $\varepsilon \to 0$ in the distributional sense to  $V\delta_{\Sigma}$ . In 2018 Mas and Pizzichillo considered this problem in three dimensions in [51], where  $\Sigma$  was assumed to be a compact C<sup>2</sup>-surface. Inspired by the methods used in [7] for the approximation of Schrödinger operators with  $\delta$ -shell potentials, they were able to show strong resolvent convergence in the case of purely electrostatic and purely Lorentz scalar interactions, if the interaction strengths satisfy a nonexplicit smallness condition. Moreover, they observed a similar rescaling of V to V as known from the one-dimensional counterpart. For the special case where  $\Sigma$  is the sphere, the same authors considered the convergence of the eigenvalues in [50]. Recently, the two-dimensional case with  $\Sigma$  being a smooth closed curve was considered for the first time in [24] by Cassano, Lotoreichik, Mas and Tušek. In this paper the authors established strong resolvent convergence of  $H_{V_{\varepsilon}}$  to  $H_{\widetilde{V}\delta_{\Sigma}}$  for interaction matrices of the type  $V = \eta I_N + \tau \beta + \lambda i (\alpha \cdot \nu) \beta$  without any smallness assumption. Behrndt, Holzmann and Tušek showed an analogous statement in the case where  $\Sigma$  is a straight line in [18]. Furthermore, in [74], Zreik transferred the methods from [24] to the three-dimensional setting and showed that  $H_{V_{\varepsilon}}$  converges in the strong resolvent sense to  $H_{\widetilde{V}\delta_{\Sigma}}$  for combinations of electrostatic and Lorentz scalar interaction strengths.

A different approach to approximate Dirac operators with  $\delta$ -shell potentials, which goes back to [23, 67], is via so-called nonlocal potentials; see also [34]. In one dimension such potentials are given by  $\underline{V}_{\varepsilon} = (\cdot, h_{\varepsilon})_{L^2(\mathbb{R})} V h_{\varepsilon}$  and they also converge in the distributional sense to  $V\delta_0$ . However, in contrast to the classical strongly localized potentials,  $H_{\underline{V}_{\varepsilon}}$  converges in the norm resolvent sense to  $H_{V\delta_0}$ , i.e. no rescaling of the interaction matrix is necessary. Tušek and Heriban took up this idea in [35] and considered the norm resolvent convergence of Dirac operators with such potentials in the multidimensional case. It turned out that in the mentioned case no rescaling is necessary either, but in contrast to the one-dimensional case, the limit operator is not a Dirac operator with a local  $\delta$ -shell potential but rather a Dirac operator with a so-called nonlocal  $\delta$ -shell potential.

#### 1.3 Rigorous definition of various objects

Before we state and discuss the main results of this thesis, let us fix some necessary notations. We assume that  $\Omega_+ \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , is a possibly unbounded  $C^2$ -smooth domain (according to Definition 2.1) and we set  $\Sigma := \partial \Omega_+$  and  $\Omega_- := \mathbb{R}^{\theta} \setminus \overline{\Omega_+}$ . Moreover, we denote the unit outward normal vector field of  $\Omega_+$  by  $\nu$ . For a function  $u : \mathbb{R}^{\theta} \to \mathbb{C}^N$  we write  $u_{\pm} := u \upharpoonright \Omega_{\pm}$  and we denote the Dirichlet trace operator by  $\mathbf{t}_{\Sigma}^{\pm} : H^1(\Omega_{\pm}; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)$ , where  $H^s$  are the  $L^2$ -based Sobolev spaces. Recall that  $\alpha_1, \ldots, \alpha_{\theta}, \beta \in \mathbb{C}^{N \times N}$  are the Dirac matrices defined in Definition 3.1. To shorten notation, we make use of the abbreviations

$$\alpha \cdot \nabla := \sum_{j=1}^{\theta} \alpha_j \partial_j \text{ and } \alpha \cdot x := \sum_{j=1}^{\theta} \alpha_j x_j, \quad x = (x_1, \dots, x_{\theta}) \in \mathbb{C}^{\theta}.$$

For  $m \in \mathbb{R}$  and  $\widetilde{V} \in L^{\infty}(\Sigma; \mathbb{C}^{N \times N})$  such that  $\widetilde{V} = \widetilde{V}^*$ , i.e.  $\widetilde{V}(x_{\Sigma}) = (\widetilde{V}(x_{\Sigma}))^*$  for a.e.  $x_{\Sigma} \in \Sigma$ , we introduce the operator

$$H_{\widetilde{V}\delta_{\Sigma}}u := (-i(\alpha \cdot \nabla) + m\beta)u_{+} \oplus (-i(\alpha \cdot \nabla) + m\beta)u_{-},$$
  

$$\operatorname{dom} H_{\widetilde{V}\delta_{\Sigma}} := \left\{ u \in H^{1}(\Omega_{+}; \mathbb{C}^{N}) \oplus H^{1}(\Omega_{-}; \mathbb{C}^{N}) :$$

$$i(\alpha \cdot \nu)(\boldsymbol{t}_{\Sigma}^{+}u_{+} - \boldsymbol{t}_{\Sigma}^{-}u_{-}) + \frac{\widetilde{V}}{2}(\boldsymbol{t}_{\Sigma}^{+}u_{+} + \boldsymbol{t}_{\Sigma}^{-}u_{-}) = 0 \right\} \subset L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}).$$

$$(1.3)$$

This is a rigorous realization of the formal operator

$$H + \widetilde{V}\delta_{\Sigma} = -i(\alpha \cdot \nabla) + m\beta + \widetilde{V}\delta_{\Sigma},$$

which was studied under various assumptions on  $\widetilde{V}$  and the interaction support  $\Sigma$ in [4, 5, 8, 9, 13, 16, 18, 19, 24, 59, 60]. The operator exhibits different behaviours depending on the properties of  $\widetilde{V}$ . To illustrate this fact, let us consider  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$ with  $\widetilde{\eta}, \widetilde{\tau} \in \mathbb{R}$ . It is well-known that  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint if  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 \neq 4$ . This case is referred to as the *noncritical* case. In the *critical* case, i.e.  $\widetilde{d} = 4$ ,  $H_{\widetilde{V}\delta_{\Sigma}}$  is not closed and only essentially self-adjoint; see [12, 13, 19, 56]. Not only this, by [12, 13, 17, 18, 19, 20] the spectral properties also change drastically in the critical case. We conclude the discussion of  $H_{\widetilde{V}\delta_{\Sigma}}$  by mentioning that if  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = -4$ , then  $H_{\widetilde{V}\delta_{\Sigma}}$  splits into the orthogonal sum of two operators acting in  $L^2(\Omega_+; \mathbb{C}^N)$  and  $L^2(\Omega_-; \mathbb{C}^N)$ ; cf. Proposition 3.15 (ii). This implies that the particle described by  $H_{\widetilde{V}\delta_{\Sigma}}$  cannot cross  $\Sigma$  and thus stays confined either in  $\Omega_+$  or in  $\Omega_-$ . Hence, we say that  $H_{\widetilde{V}\delta_{\Sigma}}$  induces confinement in this case.

It is the main goal of the present thesis to show that  $H_{\tilde{V}\delta_{\Sigma}}$  can be approximated in the norm resolvent sense by Dirac operators with strongly localized potentials. To introduce the latter operators, we define the map

$$\iota: \Sigma \times \mathbb{R} \to \mathbb{R}^{\theta}, \quad \iota(x_{\Sigma}, t) := x_{\Sigma} + t\nu(x_{\Sigma}), \quad (x_{\Sigma}, t) \in \Sigma \times \mathbb{R},$$

and for  $\varepsilon \in (0, \infty)$  we set  $\Omega_{\varepsilon} := \iota(\Sigma \times (-\varepsilon, \varepsilon))$ , which is the so-called *tubular neighbour*hood of  $\Sigma$ . Furthermore, for  $\varepsilon_{tub} > 0$  sufficiently small the map  $\iota \upharpoonright \Sigma \times (-\varepsilon_{tub}, \varepsilon_{tub})$ is injective; cf. Proposition 2.12. To define the above mentioned strongly localized potentials, we choose

$$q \in L^{\infty}((-1,1);\mathbb{R})$$
 with  $\int_{-1}^{1} q(s) \, ds = 1$ 

and

$$V \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$$
 such that  $V = V^*$ 

where  $W^1_{\infty}$  denotes the first order  $L^{\infty}$ -based Sobolev space. Since  $\iota \upharpoonright \mathbb{R} \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$ is injective, we can define for  $\varepsilon \in (0, \varepsilon_{\text{tub}})$ 

$$V_{\varepsilon}(x) := \begin{cases} \frac{1}{\varepsilon} V(x_{\Sigma}) q\left(\frac{t}{\varepsilon}\right), & x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}, \\ 0, & x \notin \Omega_{\varepsilon}, \end{cases}$$
(1.4)

and for  $m \in \mathbb{R}$  and  $\varepsilon \in (0, \varepsilon_{tub})$  the operator

$$H_{V_{\varepsilon}}u := -i(\alpha \cdot \nabla)u + m\beta u + V_{\varepsilon}u, \quad \mathrm{dom}\, H_{V_{\varepsilon}} := H^1(\mathbb{R}^{\theta}; \mathbb{C}^N).$$

Note that  $H_{V_{\varepsilon}} = H + V_{\varepsilon}$  is self-adjoint in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  as  $V_{\varepsilon} = V_{\varepsilon}^* \in L^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})$  and the free Dirac operator H from (1.1) is self-adjoint; cf. Proposition 3.3. Moreover, the sequence  $V_{\varepsilon}$  converges to  $V\delta_{\Sigma}$  as  $\varepsilon \to 0$  in the sense of distributions by construction.

Recall from Section 1.2 that the expected limit operator is not  $H_{V\delta_{\Sigma}}$  but rather  $H_{\widetilde{V}\delta_{\Sigma}}$ , where V has been rescaled to  $\widetilde{V}$ . Provided that  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ , the rescaling is given by

$$\widetilde{V} = 2(\alpha \cdot \nu) \sin\left(\frac{(\alpha \cdot \nu)V}{2}\right) \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1},\tag{1.5}$$

where analytic functions of matrices are defined via the corresponding power series; cf. (1.2) for the one-dimensional counterpart of this formula. If  $\eta, \tau \in W^1_{\infty}(\Sigma; \mathbb{R})$  and  $V = \eta I_N + \tau \beta$ , then the rescaling can be simplified to

$$\widetilde{V} = \frac{2 \tan\left(\frac{\sqrt{d}}{2}\right)}{\sqrt{d}} V, \qquad d = \eta^2 - \tau^2.$$

In particular,

$$\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta \quad \text{with} \quad (\widetilde{\eta}, \widetilde{\tau}) = \frac{2\tan\left(\frac{\sqrt{d}}{2}\right)}{\sqrt{d}}(\eta, \tau)$$
 (1.6)

in this special case, which is prevalent in literature; see [18, 24, 74] for analogous rescalings.

#### 1.4 New results and structure of this thesis

Having established some necessary notations in Section 1.3, we are now in the position to discuss the main new contributions of this thesis. The results, which are partially included in the preprints [14, 15], are presented in a manner that also elucidates the structure of this dissertation. After the preliminary Chapters 2–3, we find in Chapter 4 abstract conditions which guarantee the norm resolvent convergence of  $H_{V_{\varepsilon}}$  to  $H_{\tilde{V}\delta_{\Sigma}}$  for  $\varepsilon \to 0$ , where  $V_{\varepsilon}$  is the strongly localized potential based on the interaction matrix V from (1.4) and  $\tilde{V} = \tilde{V}(V)$  is the rescaled interaction matrix given by (1.5); see Theorem 4.15. In Corollary 4.16 we apply this theorem and show that if  $\|V\|_{W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})}$  is sufficiently small, then the operator  $H_{\tilde{V}\delta_{\Sigma}}$  is self-adjoint and  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$  for  $\varepsilon \to 0$ . Theorem 4.15 extends the current literature in the following three aspects:

- (i) Instead of strong resolvent convergence, we prove the norm resolvent convergence of the approximating family, which has not been established in the multidimensional situation so far. This type of convergence ensures that the spectrum of the limit operator  $H_{\tilde{V}\delta_{\Sigma}}$  can be completely characterized by the spectra of the approximating operators and it also implies the convergence of the related spectral projections.
- (ii) Instead of bounded curves in  $\mathbb{R}^2$  or bounded surfaces in  $\mathbb{R}^3$ , we treat a general class of bounded and unbounded interaction supports  $\Sigma$  which we call *special*  $C^2$ -surfaces. This class of surfaces can be described by finitely many rotated graphs of  $C^2$ -functions with bounded derivatives; see Definition 2.1. In particular, this class includes graphs of  $C^2$ -functions with bounded derivatives and boundaries of bounded  $C^2$ -domains.
- (iii) Instead of considering only electrostatic, Lorentz scalar, and anomalous magnetic interactions (which can be described by three real-valued functions), we allow general symmetric  $2 \times 2$  or  $4 \times 4$  matrix-valued functions as interaction strengths in dimensions two or three, respectively, and provide an explicit formula for rescaling when passing to the limit.

The key idea for the improvement from strong resolvent convergence to norm resolvent convergence mentioned in (i) is the following: We factorize the resolvent differences of the free Dirac operator and Dirac operators with strongly localized potentials and instead of viewing the individual factors as operators in  $L^2$ -spaces, we study their convergence properties as operators between various (Bochner-)Sobolev spaces.

In Chapter 5 we focus on interaction matrices having the form  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ , where  $C_b^1(\Sigma; \mathbb{R})$  is the set of all real-valued bounded  $C^1$ -smooth functions on  $\Sigma$  which have bounded first derivatives. The interaction strengths  $\eta$  and  $\tau$  are used to model electrostatic and Lorentz scalar interactions, respectively, and they are the most common interaction types in the literature of Dirac operators with  $\delta$ -shell potentials; see for instance [4, 9, 13, 19, 60]. In this setting we prove in Theorem 5.20 that the abstract conditions from Theorem 4.15 for norm resolvent convergence simplify to the explicit condition

$$\sup_{x_{\Sigma} \in \Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}, \qquad d = \eta^2 - \tau^2.$$
(1.7)

Inspired by the last paragraph of [24, Section 8] we then add a strongly localized magnetic potential to  $H_{V_{\varepsilon}}$ ; more precisely, we choose  $V = \eta I_N + \tau \beta + \pi (\alpha \cdot \nu)$ . It turns out that in this case  $H_{V_{\varepsilon}}$  also converges in the norm resolvent sense; see Theorem 5.21. However, by the specific choice of  $\pi$  as the magnetic interaction strength, the magnetic term disappears when rescaling. Hence, we end up with a limit operator  $H_{\tilde{V}\delta_{\Sigma}}$  which is again a Dirac operator with  $\delta$ -shell potential and only electrostatic and Lorentz scalar interactions, where the rescaling of  $\eta$  and  $\tau$  is different than in the case  $V = \eta I_N + \tau \beta$ . Using this result and Theorem 5.20 we can formulate Corollary 5.22, which states that every Dirac operator with a given  $\delta$ -shell potential  $\tilde{V}\delta_{\Sigma}, \tilde{V} = \tilde{\eta}I_N + \tilde{\tau}\beta, \tilde{\eta}, \tilde{\tau} \in C_b^1(\Sigma; \mathbb{R})$ , and  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2$  fulfilling

$$\sup_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| < 4 \quad \text{or} \quad \inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| > 4 \tag{1.8}$$

can be approximated by a sequence of Dirac operators with strongly localized potentials. In the case of constant interaction strengths this is particularly interesting, as it implies that every Dirac operator with a  $\delta$ -shell potential and constant electrostatic and Lorentz scalar interaction strengths satisfying  $|\tilde{d}| \neq 4$  can be approximated in the norm resolvent sense by Dirac operators with strongly localized potentials.

In Chapter 6 we show that the condition (1.7) is in fact optimal. We do this by providing suitable counterexamples. To discuss the counterexamples in more detail we assume  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ . By (1.6) we have

$$\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = 4 \tan\left(\frac{\sqrt{d}}{2}\right).$$

We distinguish between the critical  $(\tilde{d} = 4)$  and the noncritical  $(\tilde{d} \neq 4)$  case. Note that if  $d < \frac{\pi^2}{4}$ , i.e. (1.7) is fulfilled, then  $\tilde{d} < 4$  and hence  $\tilde{V}$  is noncritical. Now, let us assume that  $d \ge \frac{\pi^2}{4}$ , i.e. (1.7) is not fulfilled. If  $\tilde{d} = 4$ , then  $H_{\tilde{V}\delta_{\Sigma}}$  is not closed and only essentially self-adjoint. Hence,  $H_{V_{\varepsilon}}$  cannot converge in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$ . Moreover, as also the spectral properties change in the critical case, we are able to show in Theorem 6.1 that  $H_{V_{\varepsilon}}$  does not even converge to the closure of  $H_{\tilde{V}\delta_{\Sigma}}$  in the norm resolvent sense if  $\Sigma$  is compact and  $C^{\infty}$ -smooth. Furthermore, if  $d \ge \frac{\pi^2}{4}$  and  $\tilde{d} \ne 4$  we show in Theorem 6.7 that  $H_{V_{\varepsilon}}$  does not converge to  $H_{\tilde{V}\delta_{\Sigma}}$ in the norm resolvent sense under the geometric assumption that  $\Sigma$  contains a flat part.

Chapter 7 is split into two parts. In the first part we approximate Dirac operators which induce confinement and in the second part we present various spectral implications of the approximation results. Let us start by explaining the first part in more detail. From (1.8) we conclude that the confinement case, i.e.  $\tilde{d} = -4$ , is not included in our approximation results. To approximate Dirac operators with  $\delta$ shell potentials that induce confinement, we use the following approach: We choose  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  such that  $\sup_{x_{\Sigma} \in \Sigma} d(x_{\Sigma}) < 0$ . Moreover, we assume that  $f: (0, \varepsilon_{\text{tub}}) \to (0, \infty)$  is a suitable scaling function with  $f(\varepsilon) \to \infty$  for  $\varepsilon \to 0$ ; the exact conditions are given in (7.2). Then, Theorem 7.4 states that  $H_{f(\varepsilon)V_{\varepsilon}}$ converges in the norm resolvent sense to  $H_{\widetilde{V}\delta_{\Sigma}}$ , where

$$\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$$
  $(\widetilde{\eta}, \widetilde{\tau}) = \frac{2}{\sqrt{|d|}}(\eta, \tau).$ 

This immediately shows that  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 = -4$ , i.e.  $\tilde{V}$  induces indeed confinement. After considering the approximation of Dirac operators with  $\delta$ -shell potentials which induce confinement, we deal in Section 7.2 with the discrete and essential spectrum of  $H_{V_{\varepsilon}}$ . In particular, we find conditions which guarantee the existence of discrete eigenvalues in various situations.

Finally, in Chapter 8, we introduce so-called *semilocal* potentials. Recall from Section 1.2 that one can use nonlocal potentials to approximate Dirac operators with  $\delta$ -shell potentials without any rescaling in the one-dimensional setting. As already mentioned, this approach does not work for  $\theta \in \{2, 3\}$ . This leads us to the definition of semilocal potentials which allow approximations of Dirac operators with  $\delta$ -shell potentials supported on  $\Sigma$  without any rescaling. For  $V = V^* \in W^1_{\infty}(\Sigma, \mathbb{C}^{N \times N})$ ,  $q \in L^{\infty}((-1,1); \mathbb{R})$  with  $\int_{-1}^{1} q(t) dt = 1$  and  $\varepsilon \in (0, \varepsilon_{tub})$  we define

$$\underline{V_{\varepsilon}}: L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}), \\
(\underline{V_{\varepsilon}}u)(x) := \begin{cases} \frac{1}{\varepsilon}V(x_{\Sigma})q\left(\frac{t}{\varepsilon}\right)\int_{-1}^{1}u(\iota(x_{\Sigma}, \varepsilon s))q(s) \\ & \cdot \det(I - s\varepsilon W(x_{\Sigma}))\,ds, \quad x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}, \\
0, & x \notin \Omega_{\varepsilon}. \end{cases}$$

Here W denotes the Weingarten map associated with  $\Sigma$  and plays only a secondary role since it is scaled with  $\varepsilon$ . We call such potentials semilocal since they behave nonlocally with respect to the variable t in the normal direction of  $\Sigma$  and locally with respect to the variable  $x_{\Sigma} \in \Sigma$ . Similarly as in the local setting, however without any rescaling, we are able to show the norm resolvent convergence of  $H_{V_{\varepsilon}}$ to  $H_{V\delta_{\Sigma}}$  for general interaction matrices under abstract conditions; see Theorem 8.3. For interaction matrices of the form  $V = \eta I_N + \tau \beta$ ,  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ , we show in Theorem 8.9 that the simple condition

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < 4, \qquad d = \eta^2 - \tau^2,$$

guarantees the norm resolvent convergence.

Before concluding the introduction, it should be mentioned that in the process of writing this thesis, the language tool DeepL was consulted for stylistic and grammatical improvements concerning the English language.

#### 2 Preliminaries

In this chapter we provide necessary preliminary results. We start by introducing various notations and conventions in Section 2.1. Then, we define in Section 2.2 special  $C^2$ -surfaces which are roughly speaking subsets of unions of rotated  $C_b^2$ -graphs. These surfaces are important because we consider  $\delta$ -shell potentials and integral operators on such surfaces in the main parts of this thesis; cf. Section 3.1 and Section 3.3. After introducing and constructing Sobolev spaces on these special  $C^2$ -surfaces, we study tubular neighbourhoods of such surfaces in Section 2.3. In Section 2.4 we provide elementary definitions and results for Bochner spaces, which turn out to be very useful in the Chapters 4–7, where we consider functions with values on Sobolev spaces on special  $C^2$ -surfaces. In Section 2.5 we deal with the norm resolvent convergence of unbounded self-adjoint operators and its spectral implications. Finally, in Section 2.6, we collect conditions for the invertibility of bounded operators, which is crucial for handling resolvent formulas.

#### 2.1 Notations

In this section we provide a list with frequently used notations and conventions throughout this thesis.

- (i) By  $\theta \in \{2, 3\}$  we denote the space dimension and we set N = 2 for  $\theta = 2$  and N = 4 for  $\theta = 3$ .
- (ii) The symbol  $\Omega_+$  denotes an open subset of  $\mathbb{R}^{\theta}$  such that its boundary  $\Sigma$  is a special  $C^2$ -surface as in Definition 2.1. In this case we set  $\Omega_- := \mathbb{R}^{\theta} \setminus \overline{\Omega_+}$  and if u is a function defined on  $\mathbb{R}^{\theta}$ , then we define  $u_{\pm} := u \upharpoonright \Omega_{\pm}$ .
- (iii) For a topological vector space  $\mathcal{X}$  the expression  $\mathcal{X}'$  denotes its dual space; i.e. the space of all continuous linear functionals defined on  $\mathcal{X}$ . Moreover, for  $x \in \mathcal{X}$ and  $x' \in \mathcal{X}'$  we introduce the bilinear duality product

$$_{\mathcal{X}'}\langle x', x \rangle_{\mathcal{X}} := x'(x).$$

If the space  $\mathcal{X}$  is equipped with a continuous antilinear conjugation operation  $\mathcal{X} \ni x \to \overline{x} \in \mathcal{X}$  fulfilling  $\overline{\overline{x}} = x$  for all  $x \in \mathcal{X}$ , then we introduce the sesquilinear duality product

$$\langle x', x \rangle_{\mathcal{X}' \times \mathcal{X}} := {}_{\mathcal{X}'} \langle x', \overline{x} \rangle_{\mathcal{X}}.$$

Furthermore, if  $\mathcal{Y}$  is also a topological vector space with a continuous antilinear conjugation operation and  $\mathcal{A}$  is a linear bounded operator mapping from  $\mathcal{X}$  to  $\mathcal{Y}$ , then the antidual operator  $\mathcal{A}' : \mathcal{Y}' \to \mathcal{X}'$  (which is again linear and bounded) is defined by the relationship

$$\langle y', \mathcal{A}x \rangle_{\mathcal{Y}' \times \mathcal{Y}} = \langle \mathcal{A}'y', x \rangle_{\mathcal{X}' \times \mathcal{X}} \quad \forall x \in \mathcal{X}, y' \in \mathcal{Y}'.$$

- (iv) Let  $\mathcal{H}$  and  $\mathcal{G}$  be Hilbert spaces and  $\mathcal{A}$  be a linear operator mapping from  $\mathcal{H}$  to  $\mathcal{G}$ . The domain, kernel and range of  $\mathcal{A}$  are denoted by dom  $\mathcal{A}$ , ker  $\mathcal{A}$  and ran  $\mathcal{A}$ , respectively. The norm and the scalar product (which is antilinear with respect to the second argument) in  $\mathcal{H}$  are expressed by  $\|\cdot\|_{\mathcal{H}}$  and  $(\cdot, \cdot)_{\mathcal{H}}$ . If  $\mathcal{A}$  is bounded and everywhere defined, then we write  $\|\mathcal{A}\|_{\mathcal{H}\to\mathcal{G}}$  for its operator norm. If  $\mathcal{H} = \mathcal{G}$  and  $\mathcal{A}$  is closed, then the resolvent set, the spectrum and the point spectrum of  $\mathcal{A}$  are denoted by  $\rho(\mathcal{A})$ ,  $\sigma(\mathcal{A})$  and  $\sigma_p(\mathcal{A})$ , respectively. Furthermore, if the domain of  $\mathcal{A}$  is dense in  $\mathcal{H}$ , then the adjoint of  $\mathcal{A}$  is denoted by  $\mathcal{A}^*$  and if  $\mathcal{A} = \mathcal{A}^*$ , then  $\sigma_{\text{ess}}(\mathcal{A})$  and  $\sigma_{\text{disc}}(\mathcal{A})$  are the essential and discrete spectrum of  $\mathcal{A}$ .
- (v) Let  $\mathcal{H}$  and  $\mathcal{G}$  be Hilbert spaces, J be a countable index set,  $\mathcal{A}_j, j \in J$ , and  $\mathcal{A}$  be bounded linear operators mapping from  $\mathcal{H}$  to  $\mathcal{G}$ . If for every  $u \in \mathcal{H}$  and  $\delta > 0$ exists a finite index set  $J_{\delta,u} \subset J$  such that  $\left\|\sum_{j \in J'} \mathcal{A}_j u - \mathcal{A}u\right\|_{\mathcal{H}} < \delta$  for all finite index sets J' with  $J' \supset J_{\delta,u}$ , then we say that the family  $(\mathcal{A}_j)_{j \in J}$  is strongly summable and set  $\sum_{j \in J}^{\mathrm{st.}} \mathcal{A}_j := \mathcal{A}$ . Furthermore, if  $\widetilde{\mathcal{H}}$  and  $\widetilde{\mathcal{G}}$  are also Hilbert spaces and  $\mathcal{B} : \mathcal{G} \to \widetilde{\mathcal{G}}$  and  $\mathcal{C} : \widetilde{\mathcal{H}} \to \mathcal{H}$  are bounded linear operators, then the family  $(\mathcal{B}\mathcal{A}_j\mathcal{C})_{j \in J}$  is strongly summable and  $\sum_{j \in J}^{\mathrm{st.}} \mathcal{B}\mathcal{A}_j\mathcal{C} = \mathcal{B}(\sum_{j \in J}^{\mathrm{st.}} \mathcal{A}_j)\mathcal{C}$ .
- (vi) The expression  $[\cdot, \cdot]$  denotes the commutator of two operators.
- (vii) To denote sets of functions, we use symbols having the structure S(A; B), where S reflects the properties of the functions and A and B denote the domain and codomain of the functions, respectively. For example  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ ,  $C^1(\Sigma; \mathbb{R})$ ,  $W^1(\mathbb{R}; \mathbb{C}^{N \times N})$ , etc. Moreover, if S(A) is not specified otherwise, then we set  $S(A) := S(A; \mathbb{C})$ . For example  $L^{\infty}((-1, 1)) := L^{\infty}((-1, 1); \mathbb{C})$ .
- (viii) If  $n \in \mathbb{N}$  and  $U \subset \mathbb{R}^n$  is open, then  $H^r(U)$  and  $W^r_{\infty}(U)$  denote the  $L^2$  and  $L^{\infty}$ based Sobolev spaces of order r, respectively; cf. [54, Chaper 3]. Moreover, if  $\Sigma$  is the boundary of a  $C^2$ -domain, then  $H^r(\Sigma)$  and  $W^r_{\infty}(\Sigma)$  are Sobolev spaces on the boundary  $\Sigma$ ; cf. Section 2.2. Vector and matrix-valued Sobolev spaces are defined in the natural way, i.e. component-wise.
  - (ix) If  $k, n \in \mathbb{N}$  and  $U \subset \mathbb{R}^n$  is an open set, then we write  $C_b^k(U)$  for the space which contains all  $f \in C^k(U)$  such that f and all partial derivatives of f up to order k are bounded. Moreover, we set  $C_b^{\infty}(U) = \bigcap_{k=1}^{\infty} C_b^k(U)$ . If  $\Sigma$  is a  $C^2$ -smooth hypersurface and  $k \leq 2$ , then the space  $C_b^k(\Sigma)$  is defined via local

coordinates. The corresponding spaces of vector and matrix-valued functions are defined component-wise.

- (x) If  $n \in \mathbb{N}$  and  $U \subset \mathbb{R}^n$  is open, then  $C_0^{\infty}(U)$  denotes the set of all compactly supported  $C^{\infty}$ -functions on U. Furthermore, the set  $C_0^{\infty}(\overline{U})$  contains all  $C^{\infty}$ functions on U which have an extension to a function in  $C_0^{\infty}(\mathbb{R}^n)$ . Again, the corresponding spaces of vector and matrix-valued functions are defined component-wise.
- (xi) The usual  $L^2((-1,1))$ -based Bochner space of  $\mathcal{H}$ -valued functions is denoted by  $L^2((-1,1);\mathcal{H})$ ; cf. Section 2.4. For  $\mathcal{H} = H^r(S;\mathbb{C}^N)$ , where S is either equal to  $\Sigma$  or equal to  $\mathbb{R}^{\theta-1}$ , we write  $\mathcal{B}^r(S)$  instead of  $L^2((-1,1);H^r(S;\mathbb{C}^N))$ . We also write  $\|\cdot\|_r$  for the norm in  $\mathcal{B}^r(S)$ . In a similar way, we define

$$\begin{aligned} \|\cdot\|_{r \to r'} &:= \|\cdot\|_{\mathcal{B}^{r}(S) \to \mathcal{B}^{r'}(S)}, \\ \|\cdot\|_{r \to \mathcal{H}} &:= \|\cdot\|_{\mathcal{B}^{r}(S) \to \mathcal{H}}, \\ \|\cdot\|_{\mathcal{H} \to r'} &:= \|\cdot\|_{\mathcal{H} \to \mathcal{B}^{r'}(S)}. \end{aligned}$$

- (xii) If  $(\mathcal{O}, \mathscr{A}, \lambda)$  denotes a measure space, and  $f : \mathcal{O} \to \mathbb{C}$  is an integrable function, then  $\int_{\mathcal{O}} f(t) d\lambda(t)$  denotes the integral of f. In the case that  $\lambda$  is the Lebesgue measure we simplify this notation to  $\int_{\mathcal{O}} f(t) dt$ .
- (xiii) Following [54, Appendix B], we call  $(\mathcal{H}_0, \mathcal{H}_1)$  a compatible pair, if  $\mathcal{H}_0$  and  $\mathcal{H}_1$ are two Hilbert spaces which are continuously embedded in a bigger Hausdorff topological vector space. In this situation, one can construct with the K-method (or various other methods, see [25, 26, 43, 54], which yield the same spaces with equivalent norms) a family of Hilbert spaces  $[\mathcal{H}_0, \mathcal{H}_1]_{\tau}, \tau \in (0, 1)$ , such that  $\mathcal{H}_0 \cap \mathcal{H}_1 \subset [\mathcal{H}_0, \mathcal{H}_1]_{\tau} \subset \mathcal{H}_0 + \mathcal{H}_1$  for all  $\tau \in (0, 1)$ . Assume that  $(\mathcal{G}_0, \mathcal{G}_1)$  is another compatible pair of Hilbert spaces. Then, for two bounded operators  $\mathcal{A}_0: \mathcal{H}_0 \to \mathcal{G}_0$  and  $\mathcal{A}_1: \mathcal{H}_1 \to \mathcal{G}_1$  such that  $\mathcal{A}_0 u = \mathcal{A}_1 u$  for all  $u \in \mathcal{H}_0 \cap \mathcal{H}_1$ , there exists by [54, Theorem B.2] a unique bounded linear operator

$$\mathcal{A}_{ au}: [\mathcal{H}_0, \mathcal{H}_1]_{ au} o [\mathcal{G}_0, \mathcal{G}_1]_{ au}$$

such that  $\mathcal{A}_0 u = \mathcal{A}_1 u = \mathcal{A}_\tau u$  for all  $u \in \mathcal{H}_0 \cap \mathcal{H}_1$ . Moreover, its norm can be estimated by

$$\|\mathcal{A}_{\tau}\|_{[\mathcal{H}_0,\mathcal{H}_1]_{\tau} \to [\mathcal{G}_0,\mathcal{G}_1]_{\tau}} \leq \|\mathcal{A}_0\|_{\mathcal{H}_0 \to \mathcal{G}_0}^{1-\tau}\|\mathcal{A}_1\|_{\mathcal{H}_1 \to \mathcal{G}_1}^{\tau}.$$

(xiv) The application of a holomorphic function to a matrix (or a matrix-valued function) A is defined via the associated power series, whenever it converges. This implies for two holomorphic functions f, g that f(A)g(A) = (fg)(A).

- (xv) The symbol  $|\cdot|$  is used for the absolute value, the Euclidean vector norm or the Frobenius norm of a number, a vector or a matrix, respectively. We write  $\langle \cdot, \cdot \rangle$  for the Euclidean scalar product in  $\mathbb{C}^n$ ,  $n \in \mathbb{N}$ , which is antilinear in the second argument.
- (xvi) For  $v = (v_1, \ldots, v_n)^T \in \mathbb{C}^n$ ,  $n \in \mathbb{N}$ , we simply write  $v = (v_1, \ldots, v_n)$ . Similarly, we use the notation  $v = (v', v_n)$  for the vector  $v = (v'^T, v_n)^T$  with  $v' \in \mathbb{C}^{n-1}$ and  $v_n \in \mathbb{C}$ . Moreover, we set  $v[j] = v_j$  for  $j \in \{1, \ldots, n\}$ . Analogously, if  $A \in \mathbb{C}^{n \times n}$ , then  $A[j, k], j, k \in \{1, \ldots, n\}$ , denotes the (j, k)-th entry of the matrix A.
- (xvii) The expression  $\mathcal{F}$  denotes the Fourier transform in  $\mathbb{R}^{\theta-1}$ . Moreover,  $\mathcal{F}_1$  and  $\mathcal{F}_2$  denote the partial Fourier transforms in  $\mathbb{R}^{\theta}$  with respect to the first  $\theta 1$  variables and the  $\theta$ -th variable, respectively. These transforms are given for  $\psi \in \mathcal{S}(\mathbb{R}^{\theta-1})$  and  $u \in \mathcal{S}(\mathbb{R}^{\theta})$  by

$$\mathcal{F}\psi(\xi') = \frac{1}{\sqrt{(2\pi)^{\theta-1}}} \int_{\mathbb{R}^{\theta-1}} \psi(x') e^{-i\langle x',\xi'\rangle} dx', \quad \xi' \in \mathbb{R}^{\theta-1},$$
  
$$\mathcal{F}_1 u(\xi) = \frac{1}{\sqrt{(2\pi)^{\theta-1}}} \int_{\mathbb{R}^{\theta-1}} u(x',\xi_\theta) e^{-i\langle x',\xi'\rangle} dx', \quad \xi = (\xi',\xi_\theta) \in \mathbb{R}^{\theta},$$
  
$$\mathcal{F}_2 u(\xi) = \frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} u(\xi',x_\theta) e^{-ix_\theta\xi_\theta} dx_\theta, \qquad \xi = (\xi',\xi_\theta) \in \mathbb{R}^{\theta},$$

and can be uniquely extended to continuous operators in  $\mathcal{S}'(\mathbb{R}^{\theta-1})$  and  $\mathcal{S}'(\mathbb{R}^{\theta})$ , where  $\mathcal{S}$  denotes the space of tempered distributions; cf. [63, Chapter IX]. Moreover, the application of the Fourier transform to vector and matrix-valued functions or distributions is defined component-wise. The complete Fourier transform in  $\mathbb{R}^{\theta}$  is given by  $\mathcal{F}_{1,2} := \mathcal{F}_1 \mathcal{F}_2 = \mathcal{F}_2 \mathcal{F}_1$ .

- (xviii) The letter C > 0 always denotes a generic constant which may change inbetween lines.
  - (xix) The branch of the square root is fixed by  $\operatorname{Im} \sqrt{w} > 0$  for  $w \in \mathbb{C} \setminus [0, \infty)$ .
  - (xx) The tangens cardinalis tanc :  $\mathbb{C} \setminus \{n\pi + \frac{\pi}{2} : n \in \mathbb{Z}\} \to \mathbb{C}$  is defined by

$$\operatorname{tanc}(w) := \begin{cases} \frac{\operatorname{tan}(w)}{w}, & w \in \mathbb{C} \setminus \left(\{0\} \cup \{n\pi + \frac{\pi}{2} : n \in \mathbb{Z}\}\right), \\ 1, & w = 0. \end{cases}$$

For  $x \in \mathbb{R} \setminus \{0\}$  the equation  $\operatorname{tanc}(ix) = \frac{\operatorname{tanh}(x)}{x}$  is valid.

(xxi) The expression  $SO(\theta)$  denotes the rotation matrices in  $\mathbb{R}^{\theta \times \theta}$ , i.e. the real  $\theta \times \theta$  orthogonal matrices which have determinant one.

#### 2.2 Special C<sup>2</sup>-surfaces and corresponding Sobolev spaces

We introduce and study in this section, which is based on [14, Section 2.1], socalled *special*  $C^2$ -surfaces in  $\mathbb{R}^{\theta}$ . Moreover, by a partition of unity and suitable parametrizations we are able to define Sobolev spaces on such surfaces.

**Definition 2.1.** Let  $\Omega_+ \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , be an open set and  $\Sigma := \partial \Omega_+$ . Then, we call  $\Sigma$  a special  $C^2$ -surface if there exist open sets  $W_1, \ldots, W_p \subset \mathbb{R}^{\theta}$ , mappings  $\zeta_1, \ldots, \zeta_p \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R})$ , rotation matrices  $\kappa_1, \ldots, \kappa_p \in SO(\theta)$ , and an  $\varepsilon_{\Sigma} > 0$  such that the following is true:

- (i)  $\Sigma \subset \bigcup_{l=1}^{p} W_l$ .
- (ii) If  $x \in \partial \Omega_+ = \Sigma$ , then there exists  $l \in \{1, \ldots, p\}$  such that  $B(x, \varepsilon_{\Sigma}) \subset W_l$ .
- (iii)  $W_l \cap \Omega_+ = W_l \cap \Omega_l$ , where  $\Omega_l = \{\kappa_l(x', x_\theta) : x_\theta < \zeta_l(x'), (x', x_\theta) \in \mathbb{R}^\theta\}$ , for  $l \in \{1, \ldots, p\}$ .

Furthermore, in this case we define the sets  $\Sigma_l := \partial \Omega_l = \{\kappa_l(x', \zeta_l(x')) : x' \in \mathbb{R}^{\theta-1}\}$ and  $\Omega_- := \mathbb{R}^{\theta} \setminus \overline{\Omega_+}$ , denote the unit normal vector field at  $\Sigma$  that is pointing outwards of  $\Omega_+$  by  $\nu$  and use the expression  $\sigma$  for the  $(\theta - 1)$ -dimensional Hausdorff measure restricted to  $\Sigma$ .

One can check easily that compact  $C^2$ -hypersurfaces and  $C_b^2$ -graphs are special  $C^2$ surfaces. This class is essentially the intersection of the hypersurfaces described by [70, Chapter VI, Section 3.3] and [7, Definition 2.1 and Hypothesis 2.3]. The assumptions in [70] guarantee us on the one hand the existence of suitable trace and extension theorems, see Proposition 2.3 and Proposition 2.6, and on the other hand the assumptions in [7] imply that the  $\varepsilon$ -tubular neighbourhood can be identified with the set  $\Sigma \times (-\varepsilon, \varepsilon)$ ; see Proposition 2.12. Both are essential in the later parts of this thesis.

Next, we introduce Sobolev spaces on special  $C^2$ -surfaces. We recall that Sobolev spaces on open sets are defined as in [54, Chapter 3]. To transfer the definitions to  $\Sigma$ , we choose the partition of unity  $\varphi_1, \ldots, \varphi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  for  $\Sigma$  subordinate to  $W_1, \ldots, W_p$  from Proposition A.4. We define for  $l \in \{1, \ldots, p\}$ 

$$\varkappa_l : \mathbb{R}^{\theta-1} \to \Sigma_l, \qquad \varkappa_l(x') := \kappa_l(x', \zeta_l(x')), \tag{2.1}$$

and for  $\psi \in L^2(\Sigma; \mathbb{C}^N)$  we write

$$\psi_{\Sigma_l}(x') := \begin{cases} \varphi_l(\varkappa_l(x'))\psi(\varkappa_l(x')), & \varkappa_l(x') \in \Sigma, \\ 0, & \varkappa_l(x') \notin \Sigma. \end{cases}$$

Then,  $\psi_{\Sigma_l} \in L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and  $\psi(x_{\Sigma}) = \sum_{l=1, x_{\Sigma} \in \Sigma_l}^p \psi_{\Sigma_l}(\varkappa_l^{-1}(x_{\Sigma}))$  for  $x_{\Sigma} \in \Sigma$ . As usual, we introduce for  $r \in [0, 2]$ 

$$H^{r}(\Sigma; \mathbb{C}^{N}) := \left\{ \psi \in L^{2}(\Sigma; \mathbb{C}^{N}) : \psi_{\Sigma_{l}} \in H^{r}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N}) \text{ for all } l = 1, \dots, p \right\}$$
(2.2)

and endow this space with the scalar product

$$\langle \phi, \psi \rangle_{H^r(\Sigma;\mathbb{C}^N)} = \sum_{l=1}^p \langle \phi_{\Sigma_l}, \psi_{\Sigma_l} \rangle_{H^r(\mathbb{R}^{\theta-1};\mathbb{C}^N)}, \qquad \phi, \psi \in H^r(\Sigma;\mathbb{C}^N).$$

Sobolev spaces  $H^r(\Sigma; \mathbb{C}^N)$  with  $r \in [-2, 0)$  are defined by duality. Furthermore, setting  $U_l := \varkappa_l^{-1}(\Sigma \cap W_l)$  allows us to define for  $\mathcal{V} \in \{\mathbb{C}; \mathbb{C}^{N \times N}\}$  and  $k \in \{1, 2\}$  the Sobolev space

$$W^k_{\infty}(\Sigma; \mathcal{V}) := \left\{ F \in L^{\infty}(\Sigma; \mathcal{V}) : (F \circ \varkappa_l) \upharpoonright U_l \in W^k_{\infty}(U_l; \mathcal{V}) \text{ for all } l = 1, \dots, p \right\}$$

and equip it with the norm

$$||F||_{W^k_{\infty}(\Sigma;\mathcal{V})} := \max_{l \in \{1,\dots,p\}} ||(F \circ \varkappa_l) \upharpoonright U_l||_{W^k_{\infty}(U_l;\mathcal{V})}, \quad F \in W^k_{\infty}(\Sigma;\mathcal{V}).$$

Since the Sobolev spaces on  $\Sigma$  are defined via Sobolev spaces on open sets, one can check that  $H^r(\Sigma; \mathbb{C}^N)$  is a Hilbert space and  $W^k_{\infty}(\Sigma; \mathcal{V})$  is a Banach space. We state useful properties of the just introduced Sobolev spaces on  $\Sigma$  in the next proposition.

**Proposition 2.2.** Let  $\mathcal{V} \in \{\mathbb{C}, \mathbb{C}^{N \times N}\}$ ,  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1,  $k \in \{1, 2\}$  and  $r, r_1, r_2 \in [-2, 2]$ . Then, the following statements hold:

- (i) If  $r = (1 \tau)r_1 + \tau r_2$ , then  $H^r(\Sigma; \mathbb{C}^N) = [H^{r_1}(\Sigma; \mathbb{C}^N), H^{r_2}(\Sigma; \mathbb{C}^N)]_{\tau}$ .
- (ii) If  $0 \leq r_1 \leq r_2$ , then  $H^{r_2}(\Sigma; \mathbb{C}^N)$  is densely contained in  $H^{r_1}(\Sigma; \mathbb{C}^N)$ .
- (iii) If  $|r| \leq k$ ,  $\psi \in H^r(\Sigma; \mathbb{C}^N)$  and  $F \in W^k_{\infty}(\Sigma; \mathcal{V})$ , then  $F\psi \in H^r(\Sigma; \mathbb{C}^N)$  and

 $\|F\psi\|_{H^r(\Sigma;\mathbb{C}^N)} \le C \|F\|_{W^k_\infty(\Sigma;\mathcal{V})} \|\psi\|_{H^r(\Sigma;\mathbb{C}^N)},$ 

where C > 0 does not depend on  $\psi$  and F.

(iv) If  $F, G \in W^k_{\infty}(\Sigma; \mathcal{V})$ , then  $FG \in W^k_{\infty}(\Sigma; \mathcal{V})$  and

 $\|FG\|_{W^k_{\infty}(\Sigma;\mathcal{V})} \le C \|F\|_{W^k_{\infty}(\Sigma;\mathcal{V})} \|G\|_{W^k_{\infty}(\Sigma;\mathcal{V})},$ 

where C > 0 does not depend on F and G.

(v) If  $I \subset \mathbb{R}$  is an open interval,  $F \in C_b^1(I; \mathcal{V})$ ,  $G \in W^1_{\infty}(\Sigma)$  and  $G(\Sigma) \subset I$ , then  $F \circ G \in W^1_{\infty}(\Sigma; \mathcal{V})$  and

 $\|F \circ G\|_{W^{1}_{\infty}(\Sigma;\mathcal{V})} \leq C \|F\|_{W^{1}_{\infty}(I;\mathcal{V})} \|G\|_{W^{1}_{\infty}(\Sigma)},$ 

where C > 0 does not depend on F and G.

Proof. We start by showing (i). This can be found in [54, Theorem B.11] for the case that  $\Sigma$  is compact. The proof in [54] works as follows. First, the case of graphs is considered and then a partition of unity  $\varphi_1, \ldots, \varphi_p \in C_0^{\infty}(\mathbb{R}^{\theta})$  for  $\Sigma$  subordinate to  $W_1, \ldots, W_p$  and functions  $\chi_1, \ldots, \chi_p \in C_0^{\infty}(\mathbb{R}^{\theta})$  which fulfil supp  $\chi_l \subset W_l$  and  $\phi_l \chi_l = \varphi_l$  for  $l \in \{1, \ldots, p\}$  are used to transfer the results to compact hypersurfaces. However, the proof does not change if the functions  $\varphi_1, \ldots, \varphi_p$  and  $\chi_1, \ldots, \chi_p$  are assumed to be in  $C_b^{\infty}(\mathbb{R}^{\theta})$  instead of  $C_0^{\infty}(\mathbb{R}^{\theta})$  and hence Proposition A.4 shows that (i) is also valid if  $\Sigma$  is a special  $C^2$ -surface.

Next, let us consider (ii). By [54, the lines above eq. (3.22)]  $H^{r_1}(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  is densely contained in  $H^{r_2}(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ . Using this knowledge and the definition of  $H^r(\Sigma;\mathbb{C}^N)$ in (2.2) via the partition of unity  $\varphi_1, \ldots, \varphi_p$  from Proposition A.4 one concludes that also  $H^{r_2}(\Sigma;\mathbb{C}^N)$  is densely contained in  $H^{r_1}(\Sigma;\mathbb{C}^N)$ .

To prove (iii), we first assume that r is a nonnegative integer, then by applying the product rule for weak derivatives, see [29, Section 4.2.2], we get

$$\|F\psi\|_{H^{r}(\Sigma;\mathbb{C}^{N})}^{2} = \sum_{l=1}^{p} \|(F \circ \varkappa_{l})\psi_{\Sigma_{l}}\|_{H^{r}(U_{l};\mathbb{C}^{N})}^{2}$$
  
$$\leq C \sum_{l=1}^{p} \|F \circ \varkappa_{l}\|_{W_{\infty}^{k}(U_{l};\mathcal{V})}^{2} \|\psi_{\Sigma_{l}}\|_{H^{r}(U_{l};\mathbb{C}^{N})}^{2}$$
  
$$\leq C \|F\|_{W_{\infty}^{k}(\Sigma;\mathcal{V})}^{2} \sum_{l=1}^{p} \|\psi_{\Sigma_{l}}\|_{H^{r}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})}^{2}$$
  
$$= C \|F\|_{W_{\infty}^{k}(\Sigma;\mathcal{V})}^{2} \|\psi\|_{H^{r}(\Sigma;\mathbb{C}^{N})}^{2}.$$

If  $r \in [0, k]$ , then the result follows by interpolation from (i) and Section 2.1 (xiii). If  $r \in [-k, 0)$ , then  $F\psi$  is defined by (anti-)duality via

$$\langle F\psi,\gamma\rangle_{H^r(\Sigma;\mathbb{C}^N)\times H^{-r}(\Sigma;\mathbb{C}^N)} := \langle \psi,F^*\gamma\rangle_{H^r(\Sigma;\mathbb{C}^N)\times H^{-r}(\Sigma;\mathbb{C}^N)} \qquad \forall \gamma \in H^{-r}(\Sigma;\mathbb{C}^N).$$

Thus, the result is a consequence of the case where r is positive.

Items (iv) and (v) follow from the definition of  $W^k_{\infty}(\Sigma; \mathcal{V})$  via Sobolev spaces on open sets and the product rule and chain rule for weak derivatives; see again [29, Section 4.2.2].

Next, we formulate a suitable trace theorem for special  $C^2$ -surfaces.

**Proposition 2.3.** Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , be a special C<sup>2</sup>-surface as in Definition 2.1 and  $r \in (\frac{1}{2}, \frac{5}{2})$ . Then, the following is true:

(i) There exists a unique bounded and surjective operator

1

$$t_{\Sigma}^{\pm}: H^r(\Omega_{\pm}; \mathbb{C}^N) \to H^{r-1/2}(\Sigma; \mathbb{C}^N)$$

such that  $\mathbf{t}_{\Sigma}^{\pm} u = u \upharpoonright \Sigma$  for all  $u \in H^r(\Omega_{\pm}; \mathbb{C}^N) \cap C(\overline{\Omega_{\pm}}; \mathbb{C}^N)$ .

(ii) There exists a unique bounded and surjective operator

$$\boldsymbol{t}_{\Sigma}: H^{r}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to H^{r-1/2}(\Sigma; \mathbb{C}^{N})$$

such that  $\mathbf{t}_{\Sigma} u = u \upharpoonright \Sigma$  for all  $u \in H^r(\mathbb{R}^{\theta}; \mathbb{C}^N) \cap C(\mathbb{R}^{\theta}; \mathbb{C}^N)$ .

Proof. Item (i) follows from [21, Theorem 8.7] and [48, Theorem 2]. Moreover, defining  $\mathbf{t}_{\Sigma} u := \mathbf{t}_{\Sigma}^+ u_+$  for  $u \in H^1(\mathbb{R}; \mathbb{C}^N)$  shows that there exists a bounded surjective operator satisfying  $\mathbf{t}_{\Sigma} u = u \upharpoonright \Sigma$  for  $u \in H^r(\mathbb{R}^\theta; \mathbb{C}^N) \cap C(\mathbb{R}^\theta; \mathbb{C}^N)$ . The operator  $\mathbf{t}_{\Sigma}$  is unique since the set  $H^r(\mathbb{R}^\theta; \mathbb{C}^N) \cap C(\mathbb{R}^\theta; \mathbb{C}^N)$  is dense  $H^r(\mathbb{R}^\theta; \mathbb{C}^N)$ ; see for instance [54, the text above eq. (3.22)]. Thus, also (ii) is true.  $\Box$ 

**Remark 2.4.** For  $u \in H^r(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N) = H^r(\Omega_+; \mathbb{C}^N) \oplus H^r(\Omega_-; \mathbb{C}^N)$  we define  $t_{\Sigma}^{\pm} u := t_{\Sigma}^{\pm} u_{\pm}$ , where  $u_{\pm} := u \upharpoonright \Omega \pm$ .

**Corollary 2.5.** Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1,  $\nu$  be the unit normal vector field pointing outwards of  $\Omega_+$ ,  $f, g \in H^1(\Omega_{\pm}; \mathbb{C}^N)$  and  $j \in \{1, \ldots, \theta\}$ . Then,

$$(\partial_j f, g)_{L^2(\Omega_{\pm};\mathbb{C}^N)} + (f, \partial_j g)_{L^2(\Omega_{\pm};\mathbb{C}^N)} = \pm (\nu[j] \boldsymbol{t}_{\Sigma}^{\pm} f, \boldsymbol{t}_{\Sigma}^{\pm} g)_{L^2(\Sigma;\mathbb{C}^N)},$$

where  $\nu[j]$  denotes the *j*-th component of  $\nu$ .

Proof. If  $f, g \in C_0^{\infty}(\overline{\Omega_{\pm}}; \mathbb{C}^N)$ , then the statement is a consequence of the divergence theorem. The divergence theorem for Lipschitz domains with compact boundaries is given by [54, Theorem 3.34]. However, as  $f, g \in C_0^{\infty}(\overline{\Omega_{\pm}}; \mathbb{C}^N)$ , i.e. they have compactly supported  $C^{\infty}$ -extensions to  $\mathbb{R}^{\theta}$ , the divergence theorem for compact boundaries is also applicable in our setting. According to [54, the text below (3.23)]  $C_0^{\infty}(\overline{\Omega_{\pm}}; \mathbb{C}^N)$  is dense in  $H^1(\Omega_{\pm}; \mathbb{C}^N)$ . Thus, the result follows from the continuity of the trace operator.

Under our geometric assumptions we not only have a suitable trace theorem, additionally, according to the upcoming proposition there also exists an extension operator extending functions from  $H^r(\Omega_{\pm}; \mathbb{C}^N)$  to functions in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ .

**Proposition 2.6.** Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1. Then, there exists a bounded extension operator (called Stein's extension operator)  $E^{\pm} : L^2(\Omega_{\pm}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  which satisfies  $(Eu_{\pm}) \upharpoonright \Omega_{\pm} = u_{\pm}$  for all functions  $u \in L^2(\Omega_{\pm}; \mathbb{C}^N)$  and also acts as a bounded operator from  $H^r(\Omega_{\pm}; \mathbb{C}^N)$  to  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ for all  $r \ge 0$ . Proof. If  $r \in \mathbb{N}_0$ , then the proof follows from [70, Chapter VI, Section 3.1, Theorem 5] since  $\Omega_{\pm}$  satisfies the conditions in [70, Chapter VI, Section 3.3]. If  $r \notin \mathbb{N}_0$ , then interpolation, see [54, Theorem B.7 and Theorem B.8] and Section 2.1 (xiii), yields the claim.

#### 2.3 Tubular neighbourhoods of special $C^2$ -surfaces

In this section, which contains parts of [14, Section 2.1 and Appendix A], we study tubular neighbourhoods of  $\Sigma$ . This is important, as we define in (4.3) so-called strongly localized potentials on such sets.

**Definition 2.7.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface,  $\nu$ ,  $\Sigma_l$  and  $\Omega_l$  be as in Definition 2.1, and  $\varkappa_l$  be defined by (2.1). Then, we set

$$\iota: \Sigma \times \mathbb{R} \to \mathbb{R}^{\theta}, \qquad \iota(x_{\Sigma}, t) := x_{\Sigma} + t\nu(x_{\Sigma}),$$

and for  $\varepsilon > 0$  we call  $\Omega_{\varepsilon} := \iota(\Sigma \times (-\varepsilon, \varepsilon))$  the  $(\varepsilon$ -)tubular neighbourhood of  $\Sigma$ . Moreover, we define for  $l \in \{1, \ldots, p\}$  the function

$$\iota_l: \mathbb{R}^{\theta-1} \times \mathbb{R} \to \mathbb{R}^{\theta}, \qquad \iota_l(x', t) := \varkappa_l(x') + t\nu_l(\varkappa_l(x')),$$

where  $\nu_l$  denotes the unit normal vector field on  $\Sigma_l$  pointing outwards of  $\Omega_l$ .

Note that  $\iota(\varkappa_l(x'), t) = \iota_l(x', t)$  for all  $x' \in \varkappa_l^{-1}(\Sigma), t \in \mathbb{R}$  and  $l \in \{1, \ldots, p\}$ , and

$$\nu_l(\varkappa_l(x')) = \frac{\kappa_l(-\nabla\zeta_l(x'), 1)}{\sqrt{1 + |\nabla\zeta_l(x')|^2}} \qquad \forall x' \in \mathbb{R}^{\theta - 1},$$

with  $\zeta_l$  from Definition 2.1. From now on we write  $\nu_l(x')$  instead of  $\nu_l(\varkappa_l(x'))$  for  $x' \in \mathbb{R}^{\theta-1}$  in order to simplify notation.

Before we study the maps  $\iota$  and  $\iota_l$  in detail, we provide a useful variant of the mean value theorem for vector and matrix-valued functions.

**Lemma 2.8.** Let  $k, l, n \in \mathbb{N}$ ,  $U \subset \mathbb{R}^n$  be an open set and  $A \in C_b^1(U; \mathbb{C}^{k \times l})$ . If  $x, y \in U$  and the line segment connecting x and y is contained in U, then

$$|A(x) - A(y)| \le \sup_{\mu \in [0,1]} \left( \sum_{j=1}^{n} |(\partial_j A)(x + \mu(y - x))|^2 \right)^{1/2} |x - y| \le \sqrt{n} \sup_{\mu \in [0,1], j \in \{1,\dots,n\}} |(\partial_j A)(x + \mu(y - x))| |x - y|.$$

In particular, if  $U = \mathbb{R}^n$  and l = 1, then

$$|A(x) - A(y)| \le ||DA||_{L^{\infty}(\mathbb{R}^n; \mathbb{C}^{k \times n})} |x - y| \quad \forall x, y \in \mathbb{R}^n.$$

*Proof.* Recall that  $|\cdot|$  denotes, depending on the argument, the absolute value, the Euclidean vector norm, or the Frobenius matrix norm. The fundamental theorem of calculus and the Cauchy-Schwarz inequality lead to

$$\begin{split} |A(x) - A(y)| &= \left| \int_{0}^{1} \sum_{j=1}^{n} (\partial_{j}A)(x + \mu(y - x)) (x - y)[j] \, d\mu \right| \\ &\leq \int_{0}^{1} \sum_{j=1}^{n} |(\partial_{j}A)(x + \mu(y - x))||(x - y)[j]| \, d\mu \\ &\leq \sup_{\mu \in [0,1]} \sum_{j=1}^{n} |(\partial_{j}A)(x + \mu(y - x))||(x - y)[j]| \\ &\leq \sup_{\mu \in [0,1]} \left( \sum_{j=1}^{n} |(\partial_{j}A)(x + \mu(y - x))|^{2} \right)^{1/2} |x - y| \\ &\leq \sqrt{n} \sup_{\mu \in [0,1], j \in \{1, \dots, n\}} |(\partial_{j}A)(x + \mu(y - x))||x - y|. \end{split}$$

The estimate for the special case  $U = \mathbb{R}^n$  and l = 1 is an immediate consequence of the above estimate.

**Proposition 2.9.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1, and let  $\iota$  and  $\iota_l$ ,  $l \in \{1, \ldots, p\}$ , be as in Definition 2.7. Then, there exists an  $\varepsilon_{\iota} > 0$  and constants  $C_{\iota,1}, C_{\iota,2} > 0$  such that the following holds:

(i) For all  $x', y' \in \mathbb{R}^{\theta-1}$ ,  $t, s \in (-\varepsilon_{\iota}, \varepsilon_{\iota})$  and  $l \in \{1, \ldots, p\}$  we have  $C_{\iota,1}^{-1}(|x'-y'|+|t-s|) \leq |\iota_l(x',t) - \iota_l(y',s)| \leq C_{\iota,1}(|x'-y'|+|t-s|).$ 

(ii) For all  $x_{\Sigma}, y_{\Sigma} \in \Sigma$  and  $t, s \in (-\varepsilon_{\iota}, \varepsilon_{\iota})$  we have

$$C_{\iota,2}^{-1}(|x_{\Sigma} - y_{\Sigma}| + |t - s|) \le |\iota(x_{\Sigma}, t) - \iota(y_{\Sigma}, s)| \le C_{\iota,2}(|x_{\Sigma} - y_{\Sigma}| + |t - s|).$$

*Proof.* (i) Let  $x', y' \in \mathbb{R}^{\theta-1}$  and  $t, s \in (-\varepsilon_{\iota}, \varepsilon_{\iota})$  be fixed, where  $\varepsilon_{\iota} > 0$  is, at the moment, a fixed number. Using Definition 2.7, Lemma 2.8,  $\varkappa_l(x') = \kappa_l(x', \zeta_l(x'))$  and  $\varkappa_l(y') = \kappa_l(y', \zeta_l(y'))$  we find

$$\begin{aligned} |\iota_{l}(x',t) - \iota_{l}(y',s)| &\leq |\varkappa_{l}(x') - \varkappa_{l}(y')| + |t\nu_{l}(x') - s\nu_{l}(y')| \\ &\leq |\varkappa_{l}(x') - \varkappa_{l}(y')| + |t||\nu_{l}(x') - \nu_{l}(y')| + |t-s| \\ &\leq |x'-y'| + |\zeta_{l}(x') - \zeta_{l}(y')| + \varepsilon_{\iota}|\nu_{l}(x') - \nu_{l}(y')| + |t-s| \\ &\leq |x'-y'| + \|\nabla\zeta_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta-1})} |x'-y'| \\ &\quad + \varepsilon_{\iota}\|D\nu_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta-1})} + \varepsilon_{\iota}\|D\nu_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta\times(\theta-1)})} \big| (|x'-y'| + |t-s|). \end{aligned}$$

Now, the second estimate in (i) follows if we fix  $0 < \varepsilon_{\iota} \leq 1$  and choose

$$C_{\iota,1} \ge 1 + \max_{l \in \{1,...,p\}} \left( \|\nabla \zeta_l\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta-1})} + \|D\nu_l\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta\times(\theta-1)})} \right),$$

which is finite since we assumed in Definition 2.1 that  $\zeta_l \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R}), l \in \{1,\ldots,p\}$ . Next, we prove the first inequality in (i). We start by rewriting

$$|\iota_{l}(x',t) - \iota_{l}(y',s)|^{2} = |\varkappa_{l}(x') - \varkappa_{l}(y')|^{2} + 2\langle \varkappa_{l}(x') - \varkappa_{l}(y'), t\nu_{l}(x') - s\nu_{l}(y')\rangle + |t\nu_{l}(x') - s\nu_{l}(y')|^{2}.$$
(2.3)

We estimate all three terms on the right-hand side separately. For the first one, we find with  $\varkappa_l(x') = \kappa_l(x', \zeta_l(x')), \ \varkappa_l(y') = \kappa_l(y', \zeta_l(y'))$ , and as  $\kappa_l \in SO(\theta)$  that

$$|\varkappa_l(x') - \varkappa_l(y')|^2 = |x' - y'|^2 + |\zeta_l(x') - \zeta_l(y')|^2 \ge |x' - y'|^2.$$
(2.4)

Next, we consider the second term on the right-hand side of (2.3). We start by observing

$$\begin{aligned} \mathcal{I} &:= \left| \langle \varkappa_l(x') - \varkappa_l(y'), t\nu_l(x') \rangle \right| \\ &= \left| \frac{1}{\sqrt{1 + |\nabla \zeta_l(x')|^2}} \left\langle \kappa_l \begin{pmatrix} x' - y' \\ \zeta_l(x') - \zeta_l(y') \end{pmatrix}, t\kappa_l \begin{pmatrix} -\nabla \zeta_l(x') \\ 1 \end{pmatrix} \right\rangle \right| \\ &= \left| t \frac{\langle x' - y', -\nabla \zeta_l(x') \rangle + \zeta_l(x') - \zeta_l(y')}{\sqrt{1 + |\nabla \zeta_l(x')|^2}} \right|. \end{aligned}$$

The mean value theorem shows  $\zeta_l(x') - \zeta_l(y') = \langle x' - y', \nabla \zeta_l(x' + \mu(y' - x')) \rangle$  for some  $\mu \in [0, 1]$ . Using Lemma 2.8 the above expression can be further estimated by

$$\begin{aligned} \mathcal{I} &\leq \sup_{\mu \in [0,1]} \left| t \frac{\langle x' - y', \nabla \zeta_l(x') - \nabla \zeta_l(x' + \mu(y' - x')) \rangle}{\sqrt{1 + |\nabla \zeta_l(x')|^2}} \right| \\ &\leq \sup_{\mu \in [0,1]} \varepsilon_\iota |x' - y'| ||\mu(y' - x')| ||D\nabla \zeta_l||_{L^{\infty}(\mathbb{R}^{\theta - 1}; \mathbb{R}^{(\theta - 1) \times (\theta - 1)})} \\ &\leq \varepsilon_\iota |x' - y'|^2 ||D\nabla \zeta_l||_{L^{\infty}(\mathbb{R}^{\theta - 1}; \mathbb{R}^{(\theta - 1) \times (\theta - 1)})}. \end{aligned}$$

Similarly, one has

$$|\langle \varkappa_l(x') - \varkappa_l(y'), s\nu_l(y')\rangle| \le \varepsilon_\iota |x' - y'|^2 \|D\nabla\zeta_l\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{(\theta-1)\times(\theta-1)})},$$

and thus

$$2\langle \varkappa_{l}(x') - \varkappa_{l}(y'), t\nu_{l}(x') - s\nu_{l}(y')\rangle \\ \geq -4\varepsilon_{\iota} |x' - y'|^{2} \|D\nabla\zeta_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{(\theta-1)\times(\theta-1)})}.$$
(2.5)

To estimate the third term on the right-hand side in (2.3), we use Lemma 2.8 as well as  $(a-b)^2 \ge \frac{1}{2}a^2 - b^2$  for a, b > 0 and calculate

$$|t\nu_{l}(x') - s\nu_{l}(y')|^{2} = |(t - s)\nu_{l}(x') - s(\nu_{l}(y') - \nu_{l}(x'))|^{2}$$

$$\geq (|t - s| - |s(\nu_{l}(y') - \nu_{l}(x'))|)^{2}$$

$$\geq \frac{1}{2}|t - s|^{2} - s^{2}|\nu_{l}(y') - \nu_{l}(x')|^{2}$$

$$\geq \frac{1}{2}|t - s|^{2} - \varepsilon_{\iota}^{2}|x' - y'|^{2}||D\nu_{l}||^{2}_{L^{\infty}(\mathbb{R}^{\theta - 1};\mathbb{R}^{\theta \times (\theta - 1)})}.$$
(2.6)

By plugging (2.4)–(2.6) into (2.3) we obtain

$$\begin{aligned} |\iota_{l}(x',t) - \iota_{l}(y',s)|^{2} &\geq \frac{1}{2}|t-s|^{2} \\ &+ |x'-y'|^{2} \Big(1 - 4\varepsilon_{\iota} \|D\nabla\zeta_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{(\theta-1)\times(\theta-1)})} - \varepsilon_{\iota}^{2} \|D\nu_{l}\|_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta\times(\theta-1)})}^{2} \Big). \end{aligned}$$

As before we conclude from  $\zeta_l \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  that for  $\varepsilon_{\iota} > 0$  sufficiently small and  $C_{\iota,1} > 0$  sufficiently large the first inequality in (i) is also fulfilled.

(ii) We fix  $x_{\Sigma}, y_{\Sigma} \in \Sigma$  and  $t, s \in (-\varepsilon_{\iota}, \varepsilon_{\iota})$ . Let us first assume that  $x_{\Sigma}, y_{\Sigma} \in \Sigma_{l}$  for some  $l \in \{1, \ldots, p\}$ . Then, there exist  $x', y' \in \mathbb{R}^{\theta-1}$  such that  $x_{\Sigma} = \varkappa_{l}(x')$  and  $y_{\Sigma} = \varkappa_{l}(y')$ , and therefore  $\iota(x_{\Sigma}, t) = \iota_{l}(x', t)$  and  $\iota(y_{\Sigma}, s) = \iota_{l}(y', s)$ . In this case we see

$$|x_{\Sigma} - y_{\Sigma}| = \sqrt{|x' - y'|^2 + |\zeta_l(x') - \zeta_l(y')|^2}$$

and therefore combining

$$|x' - y'| \le |x_{\Sigma} - y_{\Sigma}| \le |x' - y'| \sqrt{1 + \|\nabla \zeta_l\|_{L^{\infty}(\mathbb{R}^{\theta - 1}; \mathbb{R}^{\theta - 1})}^2}$$

with (i) yields (ii). It remains to consider the case where  $x_{\Sigma}, y_{\Sigma} \in \Sigma$  and there is no  $l \in \{1, \ldots, p\}$  such that  $x_{\Sigma}, y_{\Sigma} \in \Sigma_l$ . Then, (ii) and (iii) from Definition 2.1 imply  $|x_{\Sigma} - y_{\Sigma}| \ge \varepsilon_{\Sigma}$ , where  $\varepsilon_{\Sigma}$  is the number specified in Definition 2.1. We choose  $\varepsilon_{\iota} \le \varepsilon_{\Sigma}/6$ . Then,  $|x_{\Sigma} - y_{\Sigma}| \ge 6\varepsilon_{\iota}, |t\nu(x_{\Sigma}) - s\nu(y_{\Sigma})| \le 2\varepsilon_{\iota}$  and  $|t - s| \le 2\varepsilon_{\iota}$  yield

$$|\iota(x_{\Sigma},t) - \iota(y_{\Sigma},s)| \le |x_{\Sigma} - y_{\Sigma}| + 2\varepsilon_{\iota} \le \frac{4}{3}|x_{\Sigma} - y_{\Sigma}| \le \frac{4}{3}(|x_{\Sigma} - y_{\Sigma}| + |t-s|)$$

and

$$\frac{1}{2} (|x_{\Sigma} - y_{\Sigma}| + |t - s|) \leq \frac{|x_{\Sigma} - y_{\Sigma}|}{2} + \varepsilon_{\iota} = \frac{|x_{\Sigma} - y_{\Sigma}|}{2} + 3\varepsilon_{\iota} - 2\varepsilon_{\iota}$$
$$\leq |x_{\Sigma} - y_{\Sigma}| - 2\varepsilon_{\iota} \leq |\iota(x_{\Sigma}, t) - \iota(y_{\Sigma}, s)|,$$

which imply (ii) also in this case.

Eventually, we state a useful consequence of Proposition 2.9.

**Corollary 2.10.** Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1 and let  $\varepsilon_{\iota}$  be as in Proposition 2.9. Then, the following holds:

- (i) For any  $x_{\Sigma} \in \Sigma$  and  $t \in (0, \varepsilon_{\iota})$  one has  $x_{\Sigma} + t\nu(x_{\Sigma}) \in \Omega_{-}$ .
- (ii) For any  $x_{\Sigma} \in \Sigma$  and  $t \in (-\varepsilon_{\iota}, 0)$  one has  $x_{\Sigma} + t\nu(x_{\Sigma}) \in \Omega_+$ .

Proof. We only show item (i), the proof of assertion (ii) follows along the same lines. We verify the claim by an indirect proof. Assume that there are  $x_{\Sigma} \in \Sigma$  and  $t \in (0, \varepsilon_{\iota})$ such that  $x_{\Sigma} + t\nu(x_{\Sigma}) \notin \Omega_{-}$ . Since  $\nu$  is pointing outwards of  $\Omega_{+}$ , we have for small  $\mu > 0$  that  $x_{\Sigma} + \mu t\nu(x_{\Sigma}) \in \Omega_{-}$ . By continuity, this implies that there exists  $\mu_{0} \in (0, 1]$ such that  $x_{\Sigma} + \mu_{0} t\nu(x_{\Sigma}) \in \Sigma$ . However, we obtain from Proposition 2.9 for all  $y_{\Sigma} \in \Sigma$ with a constant  $C_{\iota,2} > 0$  the inequality

$$|x_{\Sigma} + \mu_0 t \nu(x_{\Sigma}) - y_{\Sigma}| \ge C_{\iota,2}^{-1} \mu_0 t > 0;$$

this is a contradiction.

Proposition 2.9 shows that  $\iota$  is a bi-Lipschitz mapping on  $\Sigma \times (-\varepsilon_{\iota}, \varepsilon_{\iota})$ . In particular,  $\iota$  is injective on  $\Sigma \times (-\varepsilon_{\iota}, \varepsilon_{\iota})$  and thus  $\iota \upharpoonright \Sigma \times (-\varepsilon_{\iota}, \varepsilon_{\iota})$  is a bijection between  $\Omega_{\varepsilon_{\iota}}$  and  $\Sigma \times (-\varepsilon_{\iota}, \varepsilon_{\iota})$ . In Proposition 2.12 we show that we can also identify Lebesgue spaces on these sets with one another. Before we do so, let us introduce the Weingarten map (or shape operator) on  $\Sigma$ .

**Definition 2.11.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1 and denote for  $x_{\Sigma} = \varkappa_l(x') \in \Sigma$ ,  $l \in \{1, \ldots, p\}$ , the tangent hyperplane of  $\Sigma$  in the point  $x_{\Sigma}$  by the symbol  $T_{x_{\Sigma}} = \text{span} \{\partial_j \varkappa_l(x') : j = 1, \ldots, \theta - 1\}$ . The Weingarten map is the linear operator  $W(x_{\Sigma}) : T_{x_{\Sigma}} \to T_{x_{\Sigma}}$  defined by

$$W(x_{\Sigma})\partial_{j}\varkappa_{l}(x') = -\partial_{j}\nu(\varkappa_{l}(x')), \qquad j \in \{1, \dots, \theta - 1\}.$$

Using the chain rule and  $|\nu(x_{\Sigma})| = 1$  it is easy to show that  $W(x_{\Sigma})$  is well-defined, i.e. it is independent of the parametrization  $\varkappa_l$  and  $-\partial_j \nu(\varkappa_l(x')) \in T_{x_{\Sigma}}$ ; see also [47, Lemma 3.9].

Furthermore, we denote the matrix representation of  $W(x_{\Sigma})$  corresponding to the basis  $\{\partial_j \varkappa_l(x') : j = 1, \ldots, \theta - 1\}$  of  $T_{x_{\Sigma}}$  by  $L_l(x')$ . Then, the eigenvalues of  $W(x_{\Sigma})$  and  $L_l(x')$  coincide and therefore the expression

$$\det(I - tW(x_{\Sigma})) := \det(I_{\theta-1} - tL_l(x')), \qquad t \in \mathbb{R},$$

is well-defined. In the next proposition we state important properties of  $\iota$  and W, and identify  $L^1(\Omega_{\varepsilon})$  with  $L^1(\Sigma \times (-\varepsilon, \varepsilon))$ .

**Proposition 2.12.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1 and let  $\iota$  be as in Definition 2.7. Then, there exists an  $\varepsilon_{tub} \in (0, \varepsilon_{\iota})$  such that the following is true:

- (i)  $\iota \upharpoonright \Sigma \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$  is injective.
- (ii) There exists a C > 0 such that  $|1 \det(I \varepsilon W(x_{\Sigma}))| \leq C\varepsilon < 1/2$  for all  $x_{\Sigma} \in \Sigma$  and  $\varepsilon \in (-\varepsilon_{tub}, \varepsilon_{tub})$ .
- (iii) For  $\varepsilon \in (0, \varepsilon_{tub})$  one has  $u \circ (\iota \upharpoonright \Sigma \times (-\varepsilon, \varepsilon)) \in L^1(\Sigma \times (-\varepsilon, \varepsilon))$  if and only if  $u \in L^1(\Omega_{\varepsilon})$  and in this case

$$\int_{\Omega_{\varepsilon}} u(y) \, dy = \int_{-\varepsilon}^{\varepsilon} \int_{\Sigma} u(y_{\Sigma} + s\nu(y_{\Sigma})) \det(I - sW(y_{\Sigma})) \, d\sigma(y_{\Sigma}) \, ds.$$

*Proof.* Let  $\varepsilon_{\iota}$  be the number specified in Proposition 2.9 and let  $\varepsilon_{tub} \in (0, \varepsilon_{\iota})$ . Then, by Proposition 2.9 (ii) there exists  $C_{\iota,2} > 0$  such that

$$|\iota(x_{\Sigma},t) - \iota(y_{\Sigma},s)| \ge C_{\iota,2}^{-1}(|x_{\Sigma} - y_{\Sigma}| + |t-s|), \quad (x_{\Sigma},t), (y_{\Sigma},s) \in \Sigma \times (-\varepsilon_{\mathrm{tub}},\varepsilon_{\mathrm{tub}}).$$

Hence,  $\iota \upharpoonright \Sigma \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$  is injective, i.e. item (i) is true.

Next, we show assertion (ii). For this, we first define for  $l \in \{1, ..., p\}$  and  $x' \in \mathbb{R}^{\theta-1}$  the matrices

$$M_{l}(x') := \left( \langle \partial_{j} \varkappa_{l}(x'), \partial_{k} \varkappa_{l}(x') \rangle \right)_{j,k \in \{1,\dots,\theta-1\}}, \\ H_{l}(x') := \left( \langle \partial_{j} \varkappa_{l}(x'), -\partial_{k} \nu_{l}(x') \rangle \right)_{j,k \in \{1,\dots,\theta-1\}}$$

Then, we have for  $x' \in \mathbb{R}^{\theta-1}$  such that  $\varkappa_l(x') = x_{\Sigma} \in \Sigma$  and  $j, k \in \{1, \ldots, \theta-1\}$ 

$$H_{l}(x')[j,k] = \langle \partial_{j}\varkappa_{l}(x'), W(x_{\Sigma})\partial_{k}\varkappa_{l}(x') \rangle$$
  
$$= \sum_{n=1}^{\theta-1} \langle \partial_{j}\varkappa_{l}(x'), L_{l}(x')[n,k]\partial_{n}\varkappa_{l}(x') \rangle$$
  
$$= \sum_{n=1}^{\theta-1} M_{l}(x')[j,n]L_{l}(x')[n,k]$$
  
$$= (M_{l}(x')L_{l}(x'))[j,k].$$

Moreover, using the definition of  $\varkappa_l(x')$  one concludes  $M_l(x') = I_{\theta-1} + \nabla \zeta_l(x') \nabla \zeta_l(x')^T$ . The inverse of  $M_l(x')$  is given by  $I_{\theta-1} - (1 + |\nabla \zeta(x')|^2)^{-1} \nabla \zeta_l(x') \nabla \zeta_l(x')^T$ . Hence,

$$L_{l}(x') = \left(I_{\theta-1} - (1 + |\nabla\zeta_{l}(x')|^{2})^{-1}\nabla\zeta_{l}(x')\nabla\zeta_{l}(x')^{T}\right)H_{l}(x').$$
(2.7)

Now, recall that  $\det(I - \varepsilon W(x_{\Sigma})) = \det(I_{\theta-1} - \varepsilon L_l(x'))$  for  $x_{\Sigma} = \varkappa_l(x') \in \Sigma$ . Expressing the determinant as the product of the eigenvalues one verifies the equation
$1-\det(I_{\theta-1}-\varepsilon L_l(x')) = \varepsilon P_l(\varepsilon)$ , where  $P_l$  is a polynomial in  $\varepsilon$  with coefficients depending continuously on the entries of  $L_l(x')$ . Since  $\zeta_l \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  by Definition 2.1, equation (2.7) and the definition of  $H_l(x')$  imply  $\sup_{l \in \{1,...,p\}, x' \in \varkappa_l^{-1}(\Sigma)} |L_l(x')| < \infty$ . This shows that (ii) holds if  $\varepsilon_{tub} > 0$  is chosen sufficiently small.

Finally, the claim in (iii) follows from [7, Proposition 2.6] since (i), (ii), and Proposition 2.9 (i) show that  $\Sigma$  satisfying Definition 2.1 also fulfils [7, Hypothesis 2.3].

### 2.4 Bochner spaces

Bochner spaces play an essential role in this dissertation. They allow us to consider a certain class of integral operators, which are classically viewed as operators acting in  $L^2(\Sigma \times (-1,1); \mathbb{C}^N)$ , cf. [51, Section 3], as operators acting in the Bochner space  $L^2((-1,1); L^2(\Sigma; \mathbb{C}^N))$ . Moreover, by considering restrictions and extensions of these operators to  $L^2((-1,1); H^r(\Sigma; \mathbb{C}^N))$  we can also incorporate Sobolev regularity in our analysis. With this motivation in mind, we study Bochner spaces and operators acting in Bochner spaces in the current section, which is an extended version of [14, Section 2.2] and is based on [43, Chapter 1].

In this section we always assume that  $\mathcal{H}$  and  $\mathcal{G}$  are separable Hilbert spaces,  $(\mathcal{O}, \mathcal{A}, \lambda)$ and  $(\mathcal{P}, \mathcal{B}, \mu)$  are measure spaces with  $\sigma$ -finite measures, and  $\mathcal{L}(\mathcal{H}, \mathcal{G})$  is the set of bounded linear operators from  $\mathcal{H}$  to  $\mathcal{G}$ . Let us start by defining measurability for functions with values in Hilbert spaces.

**Definition 2.13.** We call  $f : \mathcal{O} \to \mathcal{H}$  (weakly) measurable if for all  $\varphi \in \mathcal{H}$  the mapping  $\mathcal{O} \ni t \mapsto (f(t), \varphi)_{\mathcal{H}}$  is measurable with respect to the measure  $\lambda$ . Furthermore, we call  $F : \mathcal{O} \to \mathcal{L}(\mathcal{H}, \mathcal{G})$  measurable if  $\mathcal{O} \ni t \mapsto F(t)h$  is measurable for all  $h \in \mathcal{H}$ .

Recall that a function  $f: \mathcal{O} \to \mathcal{H}$  is (strongly) measurable if f is  $\lambda$ -a.e. the pointwise limit of simple functions, and that in the present situation both notions of measurability coincide due to Pettis theorem; see [43, Theorem 1.1.20]. Moreover, if  $f: \mathcal{O} \to \mathcal{H}$ and  $F: \mathcal{O} \to \mathcal{L}(\mathcal{H}, \mathcal{G})$  are measurable, then the function  $\mathcal{O} \ni t \mapsto F(t)f(t) \in \mathcal{G}$  is measurable; see [43, Proposition 1.1.28].

**Definition 2.14.** We call a function  $f : \mathcal{O} \to \mathcal{H}$  simple if there exist  $n \in \mathbb{N}$ ,  $\psi_1, \ldots, \psi_n \in \mathcal{H}$  and  $\lambda$ -measurable sets  $\mathscr{A}_1, \ldots, \mathscr{A}_n \subset \mathcal{O}$  with finite measure such that  $f = \sum_{l=1}^n \chi_{\mathscr{A}_l} \psi_l$ , where  $\chi_{\mathscr{A}_l}$  denotes the characteristic function of the set  $\mathscr{A}_l$ ,  $l \in \{1, \ldots, n\}$ . In this case the Bochner integral of f is given by

$$\int_{\mathscr{O}} f(t) \, d\lambda(t) := \sum_{l=1}^n \lambda(\mathscr{A}_l) \psi_l.$$

Moreover, we call a measurable function  $f : \mathcal{O} \to \mathcal{H}$  Bochner integrable if there exists a sequence of simple functions  $(f_k)_{k \in \mathbb{N}}$  such that

$$\lim_{k \to \infty} \int_{\mathscr{O}} \|f - f_k\|_{\mathcal{H}} = 0.$$

In this case the sequence  $(\int_{\mathscr{O}} f_k(s) ds)_{k \in \mathbb{N}}$  converges and we define the integral of f (which does not depend on the particular choice of the sequence  $(f_k)_{k \in \mathbb{N}}$ ) by

$$\int_{\mathscr{O}} f(t) \, d\lambda(t) := \lim_{k \to \infty} \int_{\mathscr{O}} f_k(t) \, d\lambda(t).$$

By [43, Proposition 1.2.2] we have the following useful characterisation of Bochner integrability:

$$f$$
 is Bochner integrable  $\iff f$  is measurable and  $\int_{\mathscr{O}} ||f(t)|| d\lambda(t) < \infty.$  (2.8)

Furthermore, it follows directly from the definition of the Bochner integral, that if  $M \in \mathcal{L}(\mathcal{H}, \mathcal{G})$  and  $f : \mathcal{O} \to \mathcal{H}$  is integrable, then  $Mf : \mathcal{O} \to \mathcal{G}, t \mapsto Mf(t)$ , is integrable and there holds  $M \int_{\mathcal{O}} f(t) d\lambda(t) = \int_{\mathcal{O}} Mf(t) d\lambda(t)$ .

Many of the classical results from integration theory have natural extensions to the Bochner integration theory. Prominent examples of these extensions are Fubini's theorem and the dominated convergence theorem for Bochner integrals. They read as follows:

**Proposition 2.15** (Fubini's theorem, [43, Proposition 1.2.7]). Let  $f : \mathscr{O} \times \mathscr{P} \to \mathcal{H}$  be a Bochner-integrable function. Then, the following statements hold:

- (i) The function  $t \mapsto f(t, s)$  is Bochner integrable for  $\mu$ -a.e.  $s \in \mathscr{P}$ .
- (ii) The function  $s \mapsto f(t, s)$  is Bochner integrable for  $\lambda$ -a.e.  $t \in \mathcal{O}$ .
- (iii) The functions  $t \mapsto \int_{\mathscr{P}} f(t,s) d\mu(s)$  and  $s \mapsto \int_{\mathscr{O}} f(t,s) d\lambda(t)$  are Bochner integrable and

$$\int_{\mathscr{O}\times\mathscr{P}} f(t,s) \, d\lambda(t) d\mu(s) = \int_{\mathscr{P}} \left( \int_{\mathscr{O}} f(t,s) \, d\lambda(t) \right) d\mu(s)$$
$$= \int_{\mathscr{O}} \left( \int_{\mathscr{P}} f(t,s) \, d\mu(s) \right) d\lambda(t)$$

**Proposition 2.16** (Dominated convergence theorem, [43, Proposition 1.2.5]). Let the functions  $f_n : \mathcal{O} \to \mathcal{H}, n \in \mathbb{N}$ , be Bochner integrable. If a function  $f : \mathcal{O} \to \mathcal{H}$  as well as a nonnegative integrable function  $g : \mathcal{O} \to \mathbb{R}$  such that  $\lim_{n\to\infty} f_n = f \lambda$ -a.e. and  $\|f_n\|_{\mathcal{H}} \leq g \lambda$ -a.e. exist, then f is Bochner integrable and we have

$$\lim_{n \to \infty} \int_{\mathscr{O}} \|f(t) - f_n(t)\|_{\mathcal{H}} d\lambda(t) = 0.$$

In particular,

$$\lim_{n \to \infty} \int_{\mathscr{O}} f_n(t) \, d\lambda(t) = \int_{\mathscr{O}} f(t) \, d\lambda(t)$$

Having stated these fundamental results regarding the Bochner integral, we turn to the definition of Bochner  $L^2$ -spaces.

**Definition 2.17.** We define  $L^2(\mathcal{O}; \mathcal{H})$  as the space which contains all (equivalence classes of)  $\lambda$ -measurable functions  $f : \mathcal{O} \to \mathcal{H}$  such that

$$\int_{\mathscr{O}} \|f(t)\|_{\mathcal{H}}^2 d\lambda(t) < \infty.$$

Furthermore, we equip this space with the scalar product

$$\int_{\mathscr{O}} \left( f(t), g(t) \right)_{\mathcal{H}} d\lambda(t), \qquad f, g \in L^2(\mathscr{O}; \mathcal{H}).$$

It is not difficult to show that  $L^2(\mathscr{O}; \mathcal{H})$  is a Hilbert space; cf. [43, the comments below Definition 1.2.15]. The space  $L^2(\mathscr{O}; \mathcal{H})$  inherits many properties from  $\mathcal{H}$ . In this dissertation we are particularly interested in duality results, interpolation results, and in the case that  $\mathcal{H}$  itself is a  $L^2$ -space, i.e.  $\mathcal{H} = L^2(\mathscr{P}; \mathcal{G})$ . We summarize such results in the upcoming proposition.

**Proposition 2.18.** Suppose that  $L^2(\mathcal{O}; \mathcal{H})$  and  $L^2(\mathcal{O}; \mathcal{G})$  are Bochner spaces. Then, the following holds:

(i) Functionals defined by

$$L^{2}(\mathscr{O};\mathcal{H}) \ni g \mapsto \int_{\mathscr{O}} {}_{\mathcal{H}'} \langle f(t), g(t) \rangle_{\mathcal{H}} d\lambda(t), \qquad f \in L^{2}(\mathscr{O};\mathcal{H}'),$$

induce an isometric isomorphism between  $L^2(\mathcal{O}; \mathcal{H}')$  and the dual of  $L^2(\mathcal{O}; \mathcal{H})$ , *i.e.* 

$$L^2(\mathscr{O};\mathcal{H})'\simeq L^2(\mathscr{O};\mathcal{H}').$$

(ii) If G is a Hilbert space such that (H,G) is a compatible pair, cf. Section 2.1 (xiii), then also (L<sup>2</sup>(𝔅; H), L<sup>2</sup>(𝔅; G)) is a compatible pair and

$$L^{2}(\mathscr{O}; [\mathcal{H}, \mathcal{G}]_{\tau}) = \left[ L^{2}(\mathscr{O}; \mathcal{H}), L^{2}(\mathcal{O}; \mathcal{G}) \right]_{\tau}, \qquad \tau \in (0, 1),$$

with equivalent norms.

(iii) Let  $\mathcal{H} = L^2(\mathscr{P}; \mathcal{G})$ . In this case the identification of  $F \in L^2(\mathscr{O} \times \mathscr{P}; \mathcal{G})$  with the function  $f : t \mapsto F(t, \cdot)$  induces an isometric isomorphism between the spaces  $L^2(\mathscr{O} \times \mathscr{P}; \mathcal{G})$  and  $L^2(\mathscr{O}; \mathcal{H})$ . Furthermore, if  $f : \mathscr{O} \mapsto \mathcal{H}$  is Bochner integrable, then

$$\left(\int_{\mathscr{O}} f(t) \, d\lambda(t)\right)(s) = \int_{\mathscr{O}} f(t)(s) \, d\lambda(t) \qquad \text{for $\mu$-a.e. $s \in \mathscr{P}$}$$

*Proof.* The assertions (i), (ii) and (iii) follow from [43, Corollary 1.3.13 and Theorem 1.3.21], [43, Theorem 2.2.6 and Corollary C.4.2] and [43, Proposition 1.2.24 and Proposition 1.2.25], respectively.  $\Box$ 

After providing elementary statements about Bochner spaces, we turn to operators in these spaces. We start by introducing the most simple and natural classes of operators. Let  $\mathcal{Q} \in L^{\infty}(\mathcal{O})$  and  $\mathcal{A} \in \mathcal{L}(\mathcal{H}, \mathcal{G})$ . In this case we define

$$\mathcal{M}_{\mathcal{Q}} : L^{2}(\mathscr{O}; \mathcal{H}) \to L^{2}(\mathscr{O}; \mathcal{H}), \qquad (\mathcal{M}_{\mathcal{Q}}f)(t) := \mathcal{Q}(t)f(t), \mathcal{M}_{\mathcal{A}} : L^{2}(\mathscr{O}; \mathcal{H}) \to L^{2}(\mathscr{O}; \mathcal{G}), \qquad (\mathcal{M}_{\mathcal{A}}f)(t) := \mathcal{A}(f(t)).$$
(2.9)

Note that the norms  $\|\mathcal{M}_{\mathcal{Q}}\|_{L^{2}(\mathscr{O};\mathcal{H})\to L^{2}(\mathscr{O};\mathcal{H})}$  and  $\|\mathcal{M}_{\mathcal{A}}\|_{L^{2}(\mathscr{O};\mathcal{H})\to L^{2}(\mathscr{O};\mathcal{G})}$  are equal to  $\|\mathcal{Q}\|_{L^{\infty}(\mathcal{O})}$  and  $\|\mathcal{A}\|_{\mathcal{H}\to\mathcal{H}}$ , respectively. Moreover, from now on we identify  $\mathcal{M}_{\mathcal{Q}}$  and  $\mathcal{M}_{\mathcal{A}}$  with  $\mathcal{Q}$  and  $\mathcal{A}$ . In the case that  $\mathcal{O}$  is bounded the embedding

$$\mathfrak{J}: \mathcal{H} \to L^2(\mathscr{O}; \mathcal{H}), \qquad (\mathfrak{J}\varphi)(t) := \varphi,$$

$$(2.10)$$

and its adjoint

$$\mathfrak{J}^*: L^2(\mathscr{O}; \mathcal{H}) \to \mathcal{H}, \qquad \mathfrak{J}^*f = \int_{\mathcal{O}} f(t) dt$$
 (2.11)

are well-defined and bounded. After introducing these simple operators, we turn to so-called *decomposable* operators. To do so, let us assume that

$$M \in L^{\infty}(\mathscr{O}; \mathcal{L}(\mathcal{H}, \mathcal{G})) := \big\{ F : \mathscr{O} \to \mathcal{L}(\mathcal{H}, \mathcal{G}) \, \lambda \text{-measurable} : \|F\|_{\mathcal{H} \to \mathcal{G}} \in L^{\infty}(\mathscr{O}) \big\}.$$

Then, we define the operator  $\mathcal{M}: L^2(\mathscr{O}; \mathcal{H}) \to L^2(\mathscr{O}; \mathcal{G})$  through

$$(\mathcal{M}f)(t) := M(t)f(t) \quad \text{for } t \in \mathscr{O} \text{ and } f \in L^2(\mathscr{O}; \mathcal{H}).$$

$$(2.12)$$

Such operators are generalizations of multiplication operators and are usually considered in the context of direct integrals, see for instance [27] or [64, Section XIII.16], where the operators  $M(t), t \in \mathcal{O}$ , are called the fibers of  $\mathcal{M}$ . Similarly to before, we often identify  $\mathcal{M}$  with M. Next, we summarize the main properties of these operators in the following proposition. **Proposition 2.19.** Let  $M \in L^{\infty}(\mathcal{O}; \mathcal{L}(\mathcal{H}, \mathcal{G}))$  and  $\mathcal{M}$  be defined by (2.12). Then,  $\mathcal{M}$  is a well-defined bounded operator and its norm equals ess  $\sup_{t \in \mathcal{O}} ||M(t)||_{\mathcal{H} \to \mathcal{G}}$ . Moreover,  $\mathcal{M}$  is continuously invertible if and only if M(t) is continuously invertible for  $\lambda$ -a.e.  $t \in \mathcal{O}$  and  $M^{-1} \in L^{\infty}(\mathcal{O}; \mathcal{L}(\mathcal{G}, \mathcal{H}))$ . In this case there holds for every  $g \in L^2(\mathcal{O}; \mathcal{G})$  the equality  $(\mathcal{M}^{-1}g)(t) = (M(t))^{-1}g(t)$  for  $\lambda$ -a.e.  $t \in \mathcal{O}$ .

*Proof.* The proof follows from [27, Lemma 1.2 and Lemma 1.3].

Next, we consider operators induced by functions with values in the space of (unbounded) self-adjoint operators.

**Proposition 2.20** ([64, Theorem XIII.85]). Let A be a function mapping from  $\mathscr{O}$  into the space of (unbounded) self-adjoint operators in  $\mathcal{H}$ , such that  $(A + i)^{-1}$  is measurable. Moreover, let the operator  $\mathcal{A}$  acting in  $L^2(\mathscr{O}; \mathcal{H})$  be defined by

$$\operatorname{dom} \mathcal{A} := \left\{ f \in L^2(\mathscr{O}; \mathcal{H}) : f(t) \in \operatorname{dom} A(t) \text{ for } \lambda \text{-a.e. } t \in \mathscr{O} \text{ and the function} \\ \operatorname{defined} \operatorname{by} t \mapsto A(t)f(t) \text{ is in } L^2(\mathscr{O}; \mathcal{H}) \right\},$$
$$(\mathcal{A}f)(t) := A(t)f(t) \quad \text{for } \lambda \text{-a.e.} t \in \mathscr{O} \text{ and } f \in \operatorname{dom} \mathcal{A}.$$

Then, the operator  $\mathcal{A}$  is self-adjoint and  $z \in \sigma(\mathcal{A})$  if and only if for all  $\delta > 0$  the set  $\{t \in \mathcal{O} : \sigma(\mathcal{A}(t)) \cap (z - \delta, z + \delta) \neq \emptyset\}$  has positive measure.

As already mentioned in the beginning of this section we are particularly interested in the setting where  $L^2(\mathscr{O}; \mathcal{H}) = L^2((-1, 1); H^r(S; \mathbb{C}^N))$ , i.e.  $\mathscr{O} = (-1, 1), \mathscr{A}$  is the corresponding  $\sigma$ -algebra of Lebesgue measurable sets on  $(-1, 1), \lambda$  is the classical Lebesgue measure and  $\mathcal{H} = H^r(S; \mathbb{C}^N), r \in [-2, 2]$ , where S is either  $\mathbb{R}^{\theta-1}$  or  $S \subset \mathbb{R}^{\theta}$  is a special  $C^2$ -surface as in Definition 2.1. In order to shorten notation, we set  $\mathcal{B}^r(S) := L^2((-1, 1, ); H^r(S; \mathbb{C}^N))$  and use the conventions for the norm from Section 2.1 (xi) in this setting. We summarize important properties of these spaces in the following proposition.

**Proposition 2.21.** Let  $S = \mathbb{R}^{\theta-1}$  or  $S \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1. Then, the following is true:

(i) If  $\tau \in (0,1)$ ,  $r_1, r_2 \in [-2,2]$  and  $r = (1-\tau)r_1 + \tau r_2$ , then

$$\mathcal{B}^{r}(S) = \left[ \mathcal{B}^{r_1}(S), \mathcal{B}^{r_2}(S) \right]_{\tau}$$

and the corresponding norms are equivalent.

(ii) For  $r \in [0,2]$  there exists an isometric isomorphism between the dual space of  $\mathcal{B}^{r}(S)$  and  $\mathcal{B}^{-r}(S)$ , i.e.

$$\mathcal{B}^r(S)' \simeq \mathcal{B}^{-r}(S).$$

(iii) If  $0 \leq r_1 \leq r_2 \leq 2$ , then  $\mathcal{B}^{r_2}(S)$  is densely contained in  $\mathcal{B}^{r_1}(S)$ .

Proof. Item (i) and (ii) follow from Proposition 2.18, the definition of  $H^r(\Sigma; \mathbb{C}^N)$ in Section 2.2 and Proposition 2.2. It remains to prove (iii). We already know from Proposition 2.2 (ii) that  $H^{r_2}(S; \mathbb{C}^N)$  is densely contained in  $H^{r_1}(S; \mathbb{C}^N)$  if Sis a special  $C^2$ -surface. If  $S = \mathbb{R}^{\theta-1}$ , then this is well known and can be found for example in [54, the text above eq. (3.22)]. Now, let  $f \in \mathcal{B}^{r_1}(S)$  and  $\delta > 0$ . Then, according to [43, Lemma 1.2.19], there exists a simple function  $f_{\delta} = \sum_{l=1}^{n} \chi_{\mathscr{A}_l} \psi_l$  with  $\psi_1, \ldots, \psi_n \in H^{r_1}(S; \mathbb{C}^N)$  and measurable sets  $\mathscr{A}_1, \ldots, \mathscr{A}_n$  with finite measure such that  $\|f - f_{\delta}\|_{r_1} < \frac{\delta}{2}$ . Moreover, we can choose  $\widetilde{\psi}_l \in H^{r_2}(S; \mathbb{C}^N)$ ,  $l \in \{1, \ldots, n\}$ , such that

$$\|\psi_l - \widetilde{\psi}_l\|_{H^{r_1}(S;\mathbb{C}^N)} < \frac{1}{2n\delta \|\chi_{\mathscr{A}_l}\|_{L^2(-1,1)}}.$$

Thus,  $\widetilde{f}_{\delta} = \sum_{l=1}^{n} \chi_{\mathscr{A}_{l}} \widetilde{\psi}_{l} \in \mathcal{B}^{r_{2}}(S)$  and

$$\begin{split} \|f - f_{\delta}\|_{r_{1}} &\leq \|f - f_{\delta}\|_{r_{1}} + \|f_{\delta} - f_{\delta}\|_{r_{1}} \\ &\leq \frac{\delta}{2} + \sum_{l=1}^{n} \|(\widetilde{\psi}_{l} - \psi_{l})\chi_{\mathscr{A}_{l}}\|_{r_{1}} \\ &= \frac{\delta}{2} + \sum_{l=1}^{n} \|\psi_{l} - \widetilde{\psi}_{l}\|_{H^{r_{1}}(S;\mathbb{C}^{N})} \|\chi_{\mathscr{A}_{l}}\|_{L^{2}((-1,1))} \\ &< \delta, \end{split}$$

which finishes the proof.

#### 2.5 Norm resolvent convergence

In this section we study the convergence of unbounded self-adjoint operators via the concept of norm resolvent convergence. Throughout this section we assume that  $\mathcal{I} \subset \mathbb{R}$  and  $\mathcal{H}$  is a Hilbert space.

**Definition 2.22.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$ ,  $\omega_0 \in \overline{\mathcal{I}}$ and  $\mathcal{A}$  be a self-adjoint operator in  $\mathcal{H}$ . Then, we say that  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  (or simply  $\mathcal{A}_{\omega}$ ) converges for  $\omega \to \omega_0$  in the norm resolvent sense to  $\mathcal{A}$  if for all  $z \in \mathbb{C} \setminus \mathbb{R}$ 

$$\left\| (\mathcal{A}_{\omega} - z)^{-1} - (\mathcal{A} - z)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \to 0 \quad \text{for} \quad \omega \to \omega_0.$$

We start by stating a classical result which shows that  $\mathcal{A}_{\omega}$  converges in the norm resolvent sense to  $\mathcal{A}$  if  $(\mathcal{A}_{\omega} - z_0)^{-1}$  converges to  $(\mathcal{A} - z_0)^{-1}$  for only one  $z_0 \in \rho(\mathcal{A})$ .

**Proposition 2.23.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$ ,  $\omega_0 \in \overline{\mathcal{I}}$ and  $\mathcal{A}$  be a self-adjoint operator in  $\mathcal{H}$ . Moreover, assume that there exists a  $z_0 \in \mathbb{C}$ such that  $z_0 \in \rho(\mathcal{A}) \cap \rho(\mathcal{A}_{\omega})$  if  $|\omega - \omega_0|$  is sufficiently small and

$$\left\| (\mathcal{A}_{\omega} - z_0)^{-1} - (\mathcal{A} - z_0)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \to 0 \quad \text{for} \quad \omega \to \omega_0.$$

Then,  $\mathcal{A}_{\omega}$  converges for  $\omega \to \omega_0$  in the norm resolvent sense to  $\mathcal{A}$ .

*Proof.* The statement follows from [73, Satz 9.20 a)].

Next, we show that norm resolvent convergence is invariant with respect to bounded self-adjoint perturbations.

**Proposition 2.24.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$  which converges for  $\omega \to \omega_0 \in \overline{\mathcal{I}}$  in the norm resolvent sense to the self-adjoint operator  $\mathcal{A}$ ,  $\mathcal{K}$  be a bounded self-adjoint operator in  $\mathcal{H}$  and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then,

$$\begin{aligned} \left\| (\mathcal{A}_{\omega} + \mathcal{K} - z)^{-1} - (\mathcal{A} + \mathcal{K} - z)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \\ &\leq \left( 1 + |\operatorname{Im} z|^{-1} \|\mathcal{K}\|_{\mathcal{H} \to \mathcal{H}} \right)^2 \left\| (\mathcal{A}_{\omega} - z)^{-1} - (\mathcal{A} - z)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \quad \forall \omega \in \mathcal{I}. \end{aligned}$$

In particular,  $\mathcal{A}_{\omega} + \mathcal{K}$  converges for  $\omega \to \omega_0$  in the norm resolvent sense to  $\mathcal{A} + \mathcal{K}$ .

*Proof.* Let  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then,

$$(\mathcal{A}_{\omega} + \mathcal{K} - z)^{-1} - (\mathcal{A} + \mathcal{K} - z)^{-1}$$
  
=  $\left(I - (\mathcal{A} + \mathcal{K} - z)^{-1}\mathcal{K}\right)\left((\mathcal{A}_{\omega} - z)^{-1} - (\mathcal{A} - z)^{-1}\right)\left(I - \mathcal{K}(\mathcal{A}_{\omega} + \mathcal{K} - z)^{-1}\right)$ 

and therefore

$$\begin{aligned} \left\| (\mathcal{A}_{\omega} + \mathcal{K} - z)^{-1} - (\mathcal{A} + \mathcal{K} - z)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \\ &\leq \left( 1 + |\operatorname{Im} z|^{-1} \|\mathcal{K}\|_{\mathcal{H} \to \mathcal{H}} \right)^{2} \left\| (\mathcal{A}_{\omega} - z)^{-1} - (\mathcal{A} - z)^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \end{aligned}$$

converges to zero for  $\omega \to \omega_0$ .

In the next proposition we summarize important spectral implications of the norm resolvent convergence.

**Proposition 2.25.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$  which converges for  $\omega \to \omega_0 \in \overline{\mathcal{I}}$  in the norm resolvent sense to the self-adjoint operator  $\mathcal{A}$ . Then, there holds the following:

(i)  $\lim_{\omega\to\omega_0} \sigma(\mathcal{A}_{\omega}) = \sigma(\mathcal{A})$  is valid; this notation means on the one hand that the limit of every convergent sequence  $(\lambda_n)_{n\in\mathbb{N}}$  with  $\lambda_n \in \sigma(\mathcal{A}_{\omega_n})$  for a  $\omega_n \in \mathcal{I}$  and  $\omega_n \xrightarrow{n\to\infty} \omega_0$  is in  $\sigma(\mathcal{A})$ . On the other hand this means also that every  $\lambda \in \sigma(\mathcal{A})$  is the limit of such a sequence.

- (ii)  $\lim_{\omega \to \omega_0} \sigma_{\text{ess}}(\mathcal{A}_{\omega}) = \sigma_{\text{ess}}(\mathcal{A})$  in the same sense as described in (i).
- (iii) If  $a, b \in \rho(\mathcal{A})$  with a < b, then  $a, b \in \rho(\mathcal{A}_{\omega})$  for  $|\omega \omega_0|$  sufficiently small and there holds

 $\|P_{\mathcal{A}_{\omega}}(a,b) - P_{\mathcal{A}}(a,b)\|_{\mathcal{H} \to \mathcal{H}} \to 0 \quad \text{for} \quad \omega \to \omega_0,$ 

where  $P_{\mathcal{A}_{\omega}}$  and  $P_{\mathcal{A}}$  denote the spectral measures of  $\mathcal{A}_{\omega}$  and  $\mathcal{A}$ , respectively.

*Proof.* See [73, Satz 9.24].

Next, we present two consequences of the previous proposition.

**Proposition 2.26.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$  which converges for  $\omega \to \omega_0 \in \overline{\mathcal{I}}$  in the norm resolvent sense to the self-adjoint operator  $\mathcal{A}$ and assume that  $\mathcal{A}$  has at least  $M \in \mathbb{N}$  discrete eigenvalues counted with multiplicity. Then,  $\mathcal{A}_{\omega}$  has also at least M discrete eigenvalues counted with multiplicity for sufficiently small  $|\omega - \omega_0|$ .

Proof. The proof is based on the proofs of [8, Proposition 5.5] and [10, Theorem 2.7]. Let  $\lambda_1, \ldots, \lambda_{n_M}, n_M \in \mathbb{N}$ , be distinct discrete eigenvalues of  $\mathcal{A}$  such that the sum of their multiplicities equals M. Since,  $\lambda_1, \ldots, \lambda_{n_M}$ , are discrete eigenvalues, there exist  $a_1, \ldots, a_{n_M} \in \mathbb{R}$  and  $b_1, \ldots, b_{n_M} \in \mathbb{R}$  such that the intervals  $(a_j, b_j), j \in \{1, \ldots, n_M\}$ , are pairwise disjoint,  $\lambda_j \in (a_j, b_j)$  and  $[a_j, b_j] \setminus \{\lambda_j\} \subset \rho(\mathcal{A})$  for all  $j \in \{1, \ldots, n_M\}$ . In this case Proposition 2.25 implies that the spectral projections  $P_{\mathcal{A}_\omega}(a_j, b_j)$  converge in the operator norm to  $P_{\mathcal{A}}(a_j, b_j)$  for  $\omega \to \omega_0$ . Hence,

$$P_{\mathcal{A}_{\omega}}\Big(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\Big) = \sum_{j=1}^{n_{\mathcal{M}}}P_{\mathcal{A}_{\omega}}(a_{j},b_{j}) \xrightarrow{\omega \to \omega_{0}} \sum_{j=1}^{n_{\mathcal{M}}}P_{\mathcal{A}}(a_{j},b_{j}) = P_{\mathcal{A}}\Big(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\Big)$$

in the operator norm. Thus, [73, Satz 2.58] shows that the dimensions of the ranges of  $P_{\mathcal{A}_{\omega}}\left(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\right)$  and  $P_{\mathcal{A}}\left(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\right)$  coincide for sufficiently small  $|\omega - \omega_{0}|$ . Furthermore, there holds by construction dim $\left(\operatorname{ran} P_{\mathcal{A}}\left(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\right)\right) = M$ . Hence, the dimension of the range of  $P_{\mathcal{A}_{\omega}}\left(\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})\right)$  equals M if  $|\omega - \omega_{0}|$  is small enough. As  $M < \infty$ , [66, Proposition 8.11 (iv)] implies  $\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j}) \cap \sigma_{\mathrm{ess}}(\mathcal{A}_{\omega}) = \emptyset$  for sufficiently small  $|\omega - \omega_{0}|$ . Hence, all M eigenvalues in  $\bigcup_{j=1}^{n_{\mathcal{M}}}(a_{j},b_{j})$  are discrete eigenvalues for sufficiently small  $|\omega - \omega_{0}|$ .

**Proposition 2.27.** Let  $(\mathcal{A}_{\omega})_{\omega \in \mathcal{I}}$  be a family of self-adjoint operators in  $\mathcal{H}$  which converges for  $\omega \to \omega_0 \in \overline{\mathcal{I}}$  in the norm resolvent sense to the self-adjoint operator  $\mathcal{A}$  and assume that  $K \subset \mathbb{R}$  is a compact set. Then, there holds the following:

- (i) If  $\sigma(\mathcal{A}) \cap K = \emptyset$ , then  $\sigma(\mathcal{A}_{\omega}) \cap K = \emptyset$  for sufficiently small  $|\omega \omega_0|$ .
- (ii) If  $\sigma_{\text{ess}}(\mathcal{A}) \cap K = \emptyset$ , then  $\sigma_{\text{ess}}(\mathcal{A}_{\omega}) \cap K = \emptyset$  for sufficiently small  $|\omega \omega_0|$ .

Proof. We prove (i) by contraposition; the proof of (ii) follows along the same lines. Let us assume that there exist sequences  $(\omega_n)_{n\in\mathbb{N}}$  and  $(\lambda_n)_{n\in\mathbb{N}}$  such that  $\omega_n \xrightarrow{n\to\infty} \omega_0$ and  $\lambda_n \in \sigma(\mathcal{A}_{\omega_n}) \cap K$  for all  $n \in \mathbb{N}$ . Moreover, as K is compact it is no restriction to assume that  $(\lambda_n)_{n\in\mathbb{N}}$  converges to some  $\lambda \in K$ . Since  $\lambda_n \in \sigma(\mathcal{A}_n)$  for all  $n \in \mathbb{N}$ , Proposition 2.25 (i) implies  $\lambda \in \sigma(\mathcal{A})$ , which yields  $\lambda \in \sigma(\mathcal{A}) \cap K$ , i.e.  $\sigma(\mathcal{A}) \cap K \neq \emptyset$ .

### 2.6 Invertibility of bounded operators

We provide in this short section results regarding the invertibility of bounded operators. Throughout this section  $\mathcal{H}$  denotes a Hilbert space and  $\mathcal{L}(\mathcal{H})$  is the set of all bounded operators mapping from  $\mathcal{H}$  to  $\mathcal{H}$ .

**Proposition 2.28.** Let  $\mathcal{K}, \mathcal{T} \in \mathcal{L}(\mathcal{H})$  and assume  $\mathcal{T}^{-1} \in \mathcal{L}(\mathcal{H})$  as well as

$$\left\|\mathcal{T}^{-1}\right\|_{\mathcal{H}\to\mathcal{H}}\left\|\mathcal{K}\right\|_{\mathcal{H}\to\mathcal{H}}<1.$$

Then,  $(\mathcal{T} + \mathcal{K})^{-1} \in \mathcal{L}(\mathcal{H})$  and

$$\left\| (\mathcal{T} + \mathcal{K})^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \leq \frac{\left\| \mathcal{T}^{-1} \right\|_{\mathcal{H} \to \mathcal{H}}}{1 - \left\| \mathcal{T}^{-1} \right\|_{\mathcal{H} \to \mathcal{H}} \left\| \mathcal{K} \right\|_{\mathcal{H} \to \mathcal{H}}}.$$

*Proof.* This is a well-known result which follows for example from [44, Chapter IV, Theorem 1.16 and Remark 1.17].  $\Box$ 

Next, we provide a result which is known as Jacobson's lemma.

**Proposition 2.29.** Let  $\mathcal{T}, \mathcal{K} \in \mathcal{L}(\mathcal{H})$ . Then,  $\rho(\mathcal{KT}) \setminus \{0\} = \rho(\mathcal{TK}) \setminus \{0\}$  and for  $z \in \rho(\mathcal{KT}) \setminus \{0\}$  the formulas

$$(\mathcal{T}\mathcal{K}-z)^{-1}\mathcal{T} = \mathcal{T}(\mathcal{K}\mathcal{T}-z)^{-1}$$
$$(\mathcal{T}\mathcal{K}-z)^{-1} = \frac{1}{z} (\mathcal{T}(\mathcal{K}\mathcal{T}-z)^{-1}\mathcal{K}-I)$$

are valid.

*Proof.* The result about the resolvent sets can be found in [57, Proposition 2.1.8]. Furthermore, the formulas can be verified by applying  $\mathcal{TK} - z$  and using the identity  $(\mathcal{TK} - z)\mathcal{T} = \mathcal{T}(\mathcal{KT} - z)$ .

Finally, we present an invertibility result tailored to later applications in this thesis.

**Proposition 2.30.** Let  $\mathcal{A}, \mathcal{A}_0, \mathcal{K}_1, \mathcal{K}_2, \mathcal{T} \in \mathcal{L}(\mathcal{H})$ , and assume  $\mathcal{AT} = I + \mathcal{K}_1 + \mathcal{K}_2$ ,  $\mathcal{A}_0^{-1} \in \mathcal{L}(\mathcal{H})$  and

$$\left\|\mathcal{K}_{1} + (\mathcal{A}_{0} - \mathcal{A})\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right\|_{\mathcal{H} \to \mathcal{H}} < 1.$$
(2.13)

Then, a bounded right inverse of  ${\mathcal A}$  is given by

$$\left(\mathcal{T}-\mathcal{A}_0^{-1}\mathcal{K}_2\right)\left(I+\mathcal{K}_1+(\mathcal{A}_0-\mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2\right)^{-1}.$$

In particular, this right inverse of  $\mathcal{A}$  can be estimated by the expression

$$\frac{\left\|\mathcal{T} - \mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right\|_{\mathcal{H} \to \mathcal{H}}}{1 - \left\|\mathcal{K}_{1} + (\mathcal{A}_{0} - \mathcal{A})\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right\|_{\mathcal{H} \to \mathcal{H}}}.$$
(2.14)

*Proof.* By (2.13) we can apply Proposition 2.28 to  $I + \mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2$ . Hence,  $(I + \mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2)^{-1} \in \mathcal{L}(\mathcal{H})$  and its norm is bounded by

$$\frac{1}{1 - \left\| \mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A}) \mathcal{A}_0^{-1} \mathcal{K}_2 \right\|_{\mathcal{H} \to \mathcal{H}}}$$

Next, we calculate the product

$$\mathcal{A}\left(\mathcal{T}-\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)\left(I+\mathcal{K}_{1}+\left(\mathcal{A}_{0}-\mathcal{A}\right)\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)^{-1}$$
  
=  $\left(I+\mathcal{K}_{1}+\mathcal{K}_{2}-\mathcal{A}\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)\left(I+\mathcal{K}_{1}+\left(\mathcal{A}_{0}-\mathcal{A}\right)\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)^{-1}$   
=  $\left(I+\mathcal{K}_{1}+\left(\mathcal{A}_{0}-\mathcal{A}\right)\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)\left(I+\mathcal{K}_{1}+\left(\mathcal{A}_{0}-\mathcal{A}\right)\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)^{-1}$   
=  $I.$ 

Thus,  $(\mathcal{T} - \mathcal{A}_0^{-1} \mathcal{K}_2) (I + \mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A}) \mathcal{A}_0^{-1} \mathcal{K}_2)^{-1}$  is a right inverse of  $\mathcal{A}$  and its norm can be estimated by (2.14).

### 3 The free Dirac operator and perturbed Dirac operators

As mentioned in the introduction, Dirac operators model spin 1/2 particles subject to an external field which is modelled by a potential. In this chapter we provide elementary results for Dirac operators with various potentials. We start by dealing with the free Dirac operator, where the potential is set to zero, in Section 3.1. Then, we turn to Dirac operators with regular potentials, i.e. potentials which can be described by bounded operators. Lastly, in Section 3.3, we deal with Dirac operators with  $\delta$ -shell potentials, which are potentials that are only supported on a  $(\theta - 1)$ dimensional hypersurface in  $\mathbb{R}^{\theta}$ .

#### 3.1 The free Dirac operator and associated integral operators

In this section we lay the foundation for the study of Dirac operators. We start by defining the free Dirac operator. Afterwards, we study potential and boundary integral operators induced by the kernel of the resolvent of the free Dirac operator. Let us also mention that this section is based on [14, Section 2.3].

Before we introduce the free Dirac operator we define the so-called Dirac matrices.

**Definition 3.1.** The Pauli spin matrices are given by

$$\sigma_1 = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \quad \sigma_2 = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix}, \quad \sigma_3 = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}.$$

With their help the Dirac matrices  $\alpha_1, \ldots, \alpha_{\theta}, \beta \in \mathbb{C}^{N \times N}$  are defined for  $\theta = 2$  by

$$\alpha_1 := \sigma_1, \quad \alpha_2 := \sigma_2, \quad \beta := \sigma_3,$$

and for  $\theta = 3$  by

$$\alpha_j := \begin{pmatrix} 0 & \sigma_j \\ \sigma_j & 0 \end{pmatrix} \text{ for } j = 1, 2, 3 \text{ and } \beta := \begin{pmatrix} I_2 & 0 \\ 0 & -I_2 \end{pmatrix},$$

where  $I_2$  is the 2 × 2-identity matrix. We often make use of the abbreviations

$$\alpha \cdot \nabla := \sum_{j=1}^{\theta} \alpha_j \partial_j \quad and \quad \alpha \cdot x := \sum_{j=1}^{\theta} \alpha_j x_j, \quad x = (x_1, \dots, x_{\theta}) \in \mathbb{C}^{\theta}.$$

The Dirac matrices fulfil the useful relations

$$\alpha_j \alpha_l + \alpha_l \alpha_j = 2\delta_{jl}$$
 and  $\alpha_j \beta + \beta \alpha_j = 0 \quad \forall j, l \in \{1, \dots, \theta\},$  (3.1)

where  $\delta_{jl}$  denotes the Kronecker delta. Using the self-adjointness of the Dirac matrices and (3.1) implies that for  $x \in \mathbb{R}^{\theta}$  the matrix  $\alpha \cdot x$  is self-adjoint and  $(\alpha \cdot x)^2 = |x|^2 I_N$ .

Now, we are in the position to define the free Dirac operator.

**Definition 3.2.** Let  $m \in \mathbb{R}$ . Then, the free Dirac operator H is defined by

$$H = -i(\alpha \cdot \nabla) + m\beta, \qquad \text{dom}\, H = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N).$$

In the next proposition we summarize important properties of the free Dirac operator.

**Proposition 3.3.** The free Dirac operator H is self-adjoint and the following is true:

- (i)  $\sigma(H) = \sigma_{\text{ess}}(H) = (-\infty, -|m|] \cup [|m|, \infty).$
- (ii) For  $z \in \rho(H) = \mathbb{C} \setminus (-\infty, -|m|] \cup [|m|, \infty)$  the resolvent  $R(z) := (H z)^{-1}$  can be expressed by

$$R(z)u(x) = \int_{\mathbb{R}^{\theta}} G_z(x-y)u(y) \, dy, \qquad u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N), \ x \in \mathbb{R}^{\theta}, \tag{3.2}$$

where  $G_z$  is given for  $\theta = 2$  and  $x \in \mathbb{R}^2 \setminus \{0\}$  by

$$G_{z}(x) = \frac{\sqrt{z^{2} - m^{2}}}{2\pi} K_{1} \left( -i\sqrt{z^{2} - m^{2}}|x| \right) \frac{\alpha \cdot x}{|x|} + \frac{1}{2\pi} K_{0} \left( -i\sqrt{z^{2} - m^{2}}|x| \right) (m\beta + zI_{2})$$
(3.3)

and for  $\theta = 3$  and  $x \in \mathbb{R}^3 \setminus \{0\}$  by

$$G_z(x) = \left(zI_4 + m\beta + i\left(1 - i\sqrt{z^2 - m^2}|x|\right)\frac{\alpha \cdot x}{|x|^2}\right)\frac{e^{i\sqrt{z^2 - m^2}|x|}}{4\pi|x|}.$$
 (3.4)

The expressions  $K_0$  and  $K_1$  denote the modified Bessel functions of the second kind of order zero and one, respectively.

(iii) R(z) acts as a bounded operator from  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $H^{r+1}(\mathbb{R}^{\theta}; \mathbb{C}^N)$  for all  $r \in \mathbb{R}$ .

*Proof.* All statements besides (ii) can be proven in a straight forward manner by using the Fourier transform; see for instance [18, Section 2] for  $\theta = 2$  and [71, Theorem 1.1] for  $\theta = 3$ . The formulas for  $G_z$  can be found in [13, eq. (3.2)], [16, eq. (1.19)] or [71, eq. (1.263)].

Next, we provide helpful estimates for the integral kernel  $G_z$ .

**Proposition 3.4.** Let  $z \in \rho(H)$  and  $G_z$  be given by (3.3)–(3.4). Then, there exist  $C_{G,1} = C_{G,1}(m, z) > 0$  and  $C_{G,2} = C_{G,2}(m, z) > 0$  such that for all  $x \in \mathbb{R}^{\theta} \setminus \{0\}$  and  $j, k \in \{1, \ldots, \theta\}$ 

$$|G_{z}(x)| \leq C_{G,1}|x|^{1-\theta}e^{-C_{G,2}|x|},$$
  
$$|\partial_{j}G_{z}(x)| \leq C_{G,1}|x|^{-\theta}e^{-C_{G,2}|x|},$$
  
$$|\partial_{k}\partial_{j}G_{z}(x)| \leq C_{G,1}|x|^{-1-\theta}e^{-C_{G,2}|x|}.$$

*Proof.* We start by noticing that  $G_z = (-i(\alpha \cdot \nabla) + m\beta + zI_N)g_z$ , where

$$g_z(x) = \begin{cases} \frac{1}{2\pi} K_0(-i\sqrt{z^2 - m^2}|x|), & \theta = 2, \\ \frac{1}{4\pi} \frac{e^{i\sqrt{z^2 - m^2}|x|}}{|x|}, & \theta = 3, \end{cases}$$

for  $x \in \mathbb{R}^{\theta} \setminus \{0\}$ . This follows from a direct calculation and from using for the modified Bessel functions the rule  $K'_0 = -K_1$ ; see [55, §10.29 (i)]. Next, we introduce for  $l \in \mathbb{N}_0$ the set of functions

$$P_{l} := \left\{ p \in C(\mathbb{R}^{\theta} \setminus \{0\}) : p(x) = \sum_{j=1}^{m} a_{j} x^{\gamma_{j}} |x|^{-k_{j}} \text{ with } m \in \mathbb{N}, a_{j} \in \mathbb{C}, \\ k_{j} \in \mathbb{N}_{0} \text{ and } \gamma_{j} = (\gamma_{j,1}, \dots, \gamma_{j,\theta}) \in \mathbb{N}_{0}^{\theta} \\ \text{ such that } -l \leq \gamma_{j,1} + \dots + \gamma_{j,\theta} - k_{j} \leq 0 \right\}.$$

Note that for  $p \in P_l$  exists a C > 0 such that

$$|p(x)| \le C(1+|x|^{-l}) \qquad \forall x \in \mathbb{R}^{\theta} \setminus \{0\}.$$
(3.5)

Now, let  $\lambda \in \mathbb{N}_0^{\theta}$  be a multi-index with  $\lambda_1 + \cdots + \lambda_{\theta} = n, n \in \mathbb{N}$ . The chain rule, the product rule and induction show that if  $\theta = 2$ , then there exist functions  $p_{\lambda,l} \in P_l$ ,  $l \in \{0, \ldots, n-1\}$ , such that for  $x \in \mathbb{R}^2 \setminus \{0\}$ 

$$\partial^{\lambda}g_{z}(x) := \partial_{\lambda_{1}} \cdots \partial_{\lambda_{n}}g_{z}(x) = \sum_{l=0}^{n-1} K_{0}^{(n-l)}(-i\sqrt{z^{2}-m^{2}}|x|)p_{\lambda,l}(x), \qquad (3.6)$$

and if  $\theta = 3$ , then there exists a function  $p_{\lambda} \in P_{n+1}$  such that for  $x \in \mathbb{R}^3 \setminus \{0\}$ 

$$\partial^{\lambda}g_{z}(x) = p_{\lambda}(x)e^{i\sqrt{z^{2}-m^{2}|x|}}.$$
(3.7)

By the well-known rules for the derivatives and the asymptotic expansions of the modified Bessel functions from [55, §10.25 (ii), §10.27, §10.29 and §10.30 (i)], there exists an R > 0 such that for all  $x \in \mathbb{R}^2 \setminus \{0\}$  with  $|x| \leq R$  one has

$$\left| K_0^{(k)} \left( -i\sqrt{z^2 - m^2} |x| \right) \right| \le \begin{cases} C |\log|x||, & k = 0, \\ C|x|^{-k}, & k \in \mathbb{N}, \end{cases}$$

and for |x| > R

$$\left|K_{0}^{(k)}\left(-i\sqrt{z^{2}-m^{2}}|x|\right)\right| \leq Ce^{-\operatorname{Im}\sqrt{z^{2}-m^{2}}|x|}, \quad k \in \mathbb{N}_{0}.$$

In particular, there exists a C > 0 such that for all  $x \in \mathbb{R}^2 \setminus \{0\}$ 

$$\left| K_0^{(k)} \left( -i\sqrt{z^2 - m^2} |x| \right) \right| \le \begin{cases} C(1 + |x|^{-1})e^{-\operatorname{Im}\sqrt{z^2 - m^2}|x|}, & k = 0, \\ C(1 + |x|^{-k})e^{-\operatorname{Im}\sqrt{z^2 - m^2}|x|}, & k \in \mathbb{N}. \end{cases}$$
(3.8)

Inserting this and (3.5) into (3.6) and (3.7) yields

$$\left|\partial^{\lambda}g_{z}(x)\right| \leq C(1+|x|^{-n+(2-\theta)})e^{-\operatorname{Im}\sqrt{z^{2}-m^{2}}|x|} \qquad \forall x \in \mathbb{R}^{\theta} \setminus \{0\}.$$

Moreover, using this result and (3.8) for k = 0 yields

$$|G_z(x)| \le C(1+|x|^{1-\theta})e^{-\operatorname{Im}\sqrt{z^2-m^2}|x|},$$
  
$$|\partial_j G_z(x)| \le C(1+|x|^{-\theta})e^{-\operatorname{Im}\sqrt{z^2-m^2}|x|},$$
  
$$|\partial_k \partial_j G_z(x)| \le C(1+|x|^{-1-\theta})e^{-\operatorname{Im}\sqrt{z^2-m^2}|x|},$$

for all  $x \in \mathbb{R}^{\theta} \setminus \{0\}$  and  $j, k \in \{1, \ldots, \theta\}$ , where C = C(m, z) > 0 is a constant which only depends on m and z. Thus, the estimates of the proposition are valid if one chooses  $C_{G,2} \in (0, \operatorname{Im} \sqrt{z^2 - m^2})$  and

$$C_{G,1} = \max_{x \in \mathbb{R}^{\theta} \setminus \{0\}, l \in \{1-\theta, -\theta, -1-\theta\}} C \frac{1+|x|^l}{|x|^l} e^{-(\operatorname{Im} \sqrt{m^2 - z^2} - C_{G,2})|x|} < \infty.$$

Having discussed the free Dirac operator and its resolvent, we define potential and boundary integral operators with the help of  $G_z$ . We start with the potential operator.

**Definition 3.5.** Let  $z \in \rho(H)$ ,  $G_z$  be given by (3.3)–(3.4) and  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1. Then, we introduce the potential operator  $\Phi_z : L^2(\Sigma; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  by

$$\Phi_z \varphi(x) := \int_{\Sigma} G_z(x - y_{\Sigma}) \varphi(y_{\Sigma}) \, d\sigma(y_{\Sigma}), \qquad \varphi \in L^2(\Sigma; \mathbb{C}^N), \ x \in \mathbb{R}^{\theta}$$

which is well-defined and bounded by Proposition 3.4 and [4, Lemma 2.1].

Next, we study the mapping properties of this operator in detail. Note that these properties are well known for the case that  $\Sigma$  is compact; see e.g. [12, Proposition 4.2] or [16, Theorem 4.3].

**Proposition 3.6.** Let  $z \in \rho(H) = \mathbb{C} \setminus ((-\infty, -|m|] \cup [|m|, \infty))$  and let  $\Phi_z$  be given by Definition 3.5. Then, the following is true:

(i) For any  $r \in [0, \frac{1}{2}]$  the operator  $\Phi_z$  gives rise to a bounded operator

$$\Phi_z: H^r(\Sigma; \mathbb{C}^N) \to H^{r+1/2}(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N) = H^{r+1/2}(\Omega_+; \mathbb{C}^N) \oplus H^{r+1/2}(\Omega_-; \mathbb{C}^N).$$

- (ii) For  $\varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$  one has  $(-i(\alpha \cdot \nabla) + m\beta zI_N)(\Phi_z \varphi)_{\pm} = 0.$
- (iii) The adjoint  $\Phi_z^* : L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to L^2(\Sigma; \mathbb{C}^N)$  of  $\Phi_z$  acts on  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  as

$$\Phi_z^* u(x_{\Sigma}) = \int_{\mathbb{R}^{\theta}} G_{\overline{z}}(x_{\Sigma} - y) u(y) \, dy = \boldsymbol{t}_{\Sigma} R(\overline{z}) u(x_{\Sigma}), \qquad x_{\Sigma} \in \Sigma, \qquad (3.9)$$

and  $\Phi_z^*$  gives rise to a bounded operator  $\Phi_z^*: L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N).$ 

Before we prove the assertion we shortly introduce auxiliary operators and make some remarks. Although we are not using boundary triples in the proof explicitly, it is heavily inspired by [16, Theorem 4.3], where boundary triple techniques were used to show the assertion in the case that  $\Sigma$  is compact. To use such techniques, it is helpful to introduce the bounded operators  $\Gamma_0, \Gamma_1 : H^1(\mathbb{R}^\theta \setminus \Sigma; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)$ which are defined by  $\Gamma_0 := i(\alpha \cdot \nu)(\mathbf{t}_{\Sigma}^+ - \mathbf{t}_{\Sigma}^-)$  and  $\Gamma_1 := \frac{1}{2}(\mathbf{t}_{\Sigma}^+ + \mathbf{t}_{\Sigma}^-)$ , where  $\nu$  is the unit outward normal vector field of  $\Sigma$  described in Definition 2.1, and  $\mathbf{t}_{\Sigma}^+$  and  $\mathbf{t}_{\Sigma}^-$  are the trace operators from Remark 2.4. Moreover, let

$$Xu := (-i(\alpha \cdot \nabla) + m\beta)u_+ \oplus (-i(\alpha \cdot \nabla) + m\beta)u_-, \quad \text{dom} \ X := H^1(\mathbb{R}^\theta \setminus \Sigma; \mathbb{C}^N).$$
(3.10)

Readers familiar with boundary triples may notice that  $\{L^2(\Sigma; \mathbb{C}^N), \Gamma_0, \Gamma_1\}$  constitutes a quasi boundary triple for  $\overline{X}$ ; cf. [16, Section 4.2]. By Corollary 2.5 the just introduced operators fulfil for all  $u, v \in H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$  the equation

$$(Xu,v)_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)} - (u,Xv)_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)} = (\Gamma_1 u,\Gamma_0 v)_{L^2(\Sigma;\mathbb{C}^N)} - (\Gamma_0 u,\Gamma_1 v)_{L^2(\Sigma;\mathbb{C}^N)}.$$
 (3.11)

This shows that  $X \upharpoonright \ker \Gamma_0$  is symmetric. Moreover,  $H \subset X \upharpoonright \ker \Gamma_0$ . Thus, as H is self-adjoint,  $H = X \upharpoonright \ker \Gamma_0$ .

Proof of Proposition 3.6. First, Fubini's theorem shows that the representation of  $\Phi_z^*$  in (3.9) is valid. Hence, the mapping properties of  $\mathbf{t}_{\Sigma}$  and R(z) prove assertion (iii).

To verify item (i), we note that by (iii) and antiduality  $\Phi_z$  has the bounded extension

$$\widetilde{\Phi}_z := (\Phi_z^*)' : H^{-1/2}(\Sigma; \mathbb{C}^N) \to L^2(\mathbb{R}^\theta; \mathbb{C}^N) = H^0(\Omega_+; \mathbb{C}^N) \oplus H^0(\Omega_-; \mathbb{C}^N).$$
(3.12)

Next, we show the statement for  $r = \frac{1}{2}$ . If we manage to do that, then the claim for  $r \in [0, \frac{1}{2})$  follows from (3.12) and interpolation.

To prove the claim for  $r = \frac{1}{2}$  we note that with  $X \upharpoonright \ker \Gamma_0 = H$  one can show for  $z \in \rho(H)$  the direct sum decomposition

$$\operatorname{dom} X = \operatorname{dom} H + \operatorname{ker}(X - z) = \operatorname{ker} \Gamma_0 + \operatorname{ker}(X - z),$$

which allows us to define the auxiliary operator

$$\widehat{\Phi}_z := (\Gamma_0 \upharpoonright \ker(X - z))^{-1}.$$
(3.13)

Note that the properties of the trace operator in Proposition 2.3 imply the equality  $\operatorname{ran} \Gamma_0 = H^{1/2}(\Sigma; \mathbb{C}^N)$  and we also have dom  $X = H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$ . Thus,  $\widehat{\Phi}_z$  is a linear operator from  $H^{1/2}(\Sigma; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$ . Next, we show that  $\widehat{\Phi}_z$  is a restriction of  $\Phi_z$ . To see this, we observe for  $v \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ ,  $\varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$ , and  $u = R(\overline{z})v = (H - \overline{z})^{-1}v \in \operatorname{dom} H = \ker \Gamma_0$  with the help of (3.11) that

$$\begin{aligned} (\widehat{\Phi}_{z}\varphi,v)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} &= \left(\widehat{\Phi}_{z}\varphi,(H-\overline{z})u\right)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &= \left(\widehat{\Phi}_{z}\varphi,Hu\right)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} - (z\widehat{\Phi}_{z}\varphi,u)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &= \left(\widehat{\Phi}_{z}\varphi,Xu\right)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} - (X\widehat{\Phi}_{z}\varphi,u)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &= -(\Gamma_{1}\widehat{\Phi}_{z}\varphi,\Gamma_{0}u)_{L^{2}(\Sigma;\mathbb{C}^{N})} + (\Gamma_{0}\widehat{\Phi}_{z}\varphi,\Gamma_{1}u)_{L^{2}(\Sigma;\mathbb{C}^{N})} \\ &= (\varphi,\Gamma_{1}R(\overline{z})v)_{L^{2}(\Sigma;\mathbb{C}^{N})} = (\varphi,(\Phi_{z})^{*}v)_{L^{2}(\Sigma;\mathbb{C}^{N})} \\ &= (\Phi_{z}\varphi,v)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}. \end{aligned}$$

Since this is true for all  $v \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , we conclude  $\widehat{\Phi}_z \varphi = \Phi_z \varphi$ ; hence  $\widehat{\Phi}_z$  is the restriction of  $\Phi_z$  to  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . In particular,  $\Phi_z \varphi \in \ker(X - z)$  by (3.13), which yields item (ii). Eventually, we show that this restriction of  $\Phi_z$  is bounded from  $H^{1/2}(\Sigma; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$ . To see this, we prove that  $\widehat{\Phi}_z$  is closed with respect to these spaces. But this follows from the  $L^2$ -boundedness of  $\Phi_z$  and the fact that  $H^{1/2}(\Sigma; \mathbb{C}^N)$  and  $H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$  are continuously embedded in  $L^2(\Sigma; \mathbb{C}^N)$  and  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , respectively. Thus, the closed graph theorem shows that

$$\widehat{\Phi}_z = \Phi_z \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N) : H^{1/2}(\Sigma; \mathbb{C}^N) \to H^1(\mathbb{R}^\theta \backslash \Sigma; \mathbb{C}^N) = H^1(\Omega_+; \mathbb{C}^N) \oplus H^1(\Omega_-; \mathbb{C}^N)$$

is bounded, which finishes the proof.

Finally, we define a boundary integral operator associated with the free Dirac operator.

**Definition 3.7.** Let  $z \in \rho(H)$  and  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1. Then, we define operator  $\mathcal{C}_z : L^2(\Sigma; \mathbb{C}^N) \to L^2(\Sigma; \mathbb{C}^N)$  as the unique bounded extension of

$$H^{1/2}(\Sigma; \mathbb{C}^N) \ni f \mapsto \frac{1}{2}(\boldsymbol{t}_{\Sigma}^+ + \boldsymbol{t}_{\Sigma}^-)\Phi_z f.$$

We prove in Proposition 3.8 that  $C_z$  is well-defined. We remark that this operator could also be introduced as a strongly integral operator on  $\Sigma$  with the kernel  $G_z$ ; cf. for instance [4, Lemma 3.3] or [16, eq. (4.5)]. However, in our case this abstract definition is more convenient. In particular, we do not have to deal with the existence of principal value integrals and corresponding difficulties.

Next, we summarize important properties of  $C_z$ . Similar as in the case of  $\Phi_z$ , these properties are well known in various settings; see [4, Lemma 3.3] and [16, eq. (4.10) and Theorem 4.3] if  $\Sigma$  is compact, and [19, Lemma 2.1 and Corollary 2.1] if  $\Sigma$  is a plane with a compact perturbation.

**Proposition 3.8.** Let  $z \in \rho(H) = \mathbb{C} \setminus ((-\infty, -|m|] \cup [|m|, \infty))$  and let  $C_z$  be given by Definition 3.7. Then,  $C_z$  is well-defined and the following is true:

- (i) For any  $r \in [-\frac{1}{2}, \frac{1}{2}]$  the map  $\mathcal{C}_z$  has a bounded extension or restriction (depending on r)  $\mathcal{C}_z : H^r(\Sigma; \mathbb{C}^N) \to H^r(\Sigma; \mathbb{C}^N)$ .
- (ii) For any  $r \in (0, \frac{1}{2}]$  and  $\varphi \in H^r(\Sigma; \mathbb{C}^N)$  one has

$$\mathcal{C}_z \varphi = \pm \frac{i}{2} (\alpha \cdot \nu) \varphi + \boldsymbol{t}_{\Sigma}^{\pm} \Phi_z \varphi.$$

*Proof.* (i) First, it follows from Proposition 3.6 (i) and Definition 3.7 that  $C_z$  is a bounded operator in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Next, we show that the antidual  $\mathcal{C}'_{\overline{z}}$  of  $\mathcal{C}_{\overline{z}}$ , which is a bounded map in  $H^{-1/2}(\Sigma; \mathbb{C}^N)$ , is an extension of  $\mathcal{C}_z$ . To see this, let  $\varphi, \psi \in H^{1/2}(\Sigma; \mathbb{C}^N)$ . We use (3.11), Proposition 3.6 (ii), (3.13), and the definition of  $\mathcal{C}_z$  to obtain

$$0 = (X\widehat{\Phi}_{z}\varphi, \widehat{\Phi}_{\overline{z}}\psi)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} - (\widehat{\Phi}_{z}\varphi, X\widehat{\Phi}_{\overline{z}}\psi)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}$$
  
$$= (\mathcal{C}_{z}\varphi, \psi)_{L^{2}(\Sigma;\mathbb{C}^{N})} - (\varphi, \mathcal{C}_{\overline{z}}\psi)_{L^{2}(\Sigma;\mathbb{C}^{N})}$$
  
$$= \langle \mathcal{C}_{z}\varphi, \psi \rangle_{H^{-1/2}(\Sigma;\mathbb{C}^{N}) \times H^{1/2}(\Sigma;\mathbb{C}^{N})} - \langle \varphi, \mathcal{C}_{\overline{z}}\psi \rangle_{H^{-1/2}(\Sigma;\mathbb{C}^{N}) \times H^{1/2}(\Sigma;\mathbb{C}^{N})}$$
  
$$= \langle (\mathcal{C}_{z} - \mathcal{C}_{\overline{z}}')\varphi, \psi \rangle_{H^{-1/2}(\Sigma;\mathbb{C}^{N}) \times H^{1/2}(\Sigma;\mathbb{C}^{N})},$$

where  $\langle \cdot, \cdot \rangle_{H^{-1/2}(\Sigma;\mathbb{C}^N) \times H^{1/2}(\Sigma;\mathbb{C}^N)}$  denotes the sesquilinear duality product, which is antilinear in the second argument, on  $H^{-1/2}(\Sigma;\mathbb{C}^N) \times H^{1/2}(\Sigma;\mathbb{C}^N)$ . Hence,  $\mathcal{C}'_{\overline{z}}$  is an extension of  $\mathcal{C}_z$  which is bounded in  $H^{-1/2}(\Sigma;\mathbb{C}^N)$ . By interpolation, we conclude that  $\mathcal{C}_z$  gives rise to a bounded map in  $H^r(\Sigma;\mathbb{C}^N)$  for any  $r \in [-\frac{1}{2}, \frac{1}{2}]$ , which also implies that  $\mathcal{C}_z$  can be extended to a bounded operator in  $L^2(\Sigma;\mathbb{C}^N)$ . This extension is unique since  $H^{1/2}(\Sigma;\mathbb{C}^N)$  is dense in  $L^2(\Sigma;\mathbb{C}^N)$ , see Proposition 2.2 (ii), and hence  $\mathcal{C}_z: L^2(\Sigma;\mathbb{C}^N) \to L^2(\Sigma;\mathbb{C}^N)$  is well-defined. (ii) First, for  $\varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$  Definition 3.7 and the relation (3.13) imply

$$\begin{aligned} \mathcal{C}_{z}\varphi &= \frac{1}{2}\boldsymbol{t}_{\Sigma}^{+}\left(\Phi_{z}\varphi\right)_{+} + \frac{1}{2}\boldsymbol{t}_{\Sigma}^{-}\left(\Phi_{z}\varphi\right)_{-} \\ &= \mp \frac{1}{2}\left(\boldsymbol{t}_{\Sigma}^{+}\left(\widehat{\Phi}_{z}\varphi\right)_{+} - \boldsymbol{t}_{\Sigma}^{-}\left(\widehat{\Phi}_{z}\varphi\right)_{-}\right) + \boldsymbol{t}_{\Sigma}^{\pm}(\Phi_{z}\varphi)_{\pm} \\ &= \pm \frac{i}{2}(\alpha \cdot \nu)\Gamma_{0}\widehat{\Phi}_{z}\varphi + \boldsymbol{t}_{\Sigma}^{\pm}\left(\Phi_{z}\varphi\right)_{\pm} \\ &= \pm \frac{i}{2}(\alpha \cdot \nu)\varphi + \boldsymbol{t}_{\Sigma}^{\pm}(\Phi_{z}\varphi)_{\pm}, \end{aligned}$$
(3.14)

which is the claimed identity for  $\varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$ . If  $r \in (0, \frac{1}{2})$  and  $\varphi \in H^r(\Sigma; \mathbb{C}^N)$ , then (ii) follows from (3.14) by continuity and density; see Proposition 2.2 (ii).

#### 3.2 Dirac operators with regular potentials

In this brief section we introduce Dirac operators with regular potentials and prove a corresponding resolvent formula.

**Definition 3.9.** Let H be the free Dirac operator given by Definition 3.2 and the operator  $P: L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  be bounded and self-adjoint. Then, we define the self-adjoint operator  $H_P$  by

$$H_P := H + P,$$
 dom  $H_P = \operatorname{dom} H = H^1(\mathbb{R}^\theta; \mathbb{C}^N) \subset L^2(\mathbb{R}^\theta; \mathbb{C}^N).$ 

Next, we provide a Birman-Schwinger principle and a resolvent formula for  $H_P$ . Such principles and formulas are well known in various situations in literature; see [32, 39, 45, 46]. However, for completeness we provide a short proof. We do this by proving a Birman-Schwinger principle and a resolvent formula in a general framework in Lemma 3.10. Afterwards, we apply this lemma to the operator  $H_P$  in Proposition 3.11.

**Lemma 3.10.** Let  $\mathcal{G}$  and  $\mathcal{H}$  be Hilbert spaces,  $\mathcal{T}$  be an unbounded self-adjoint operator in  $\mathcal{H}$ ,  $\mathcal{P} = \mathcal{P}_L \mathcal{P}_R$ , where  $\mathcal{P}_L : \mathcal{G} \to \mathcal{H}$  and  $\mathcal{P}_R : \mathcal{H} \to \mathcal{G}$  are bounded operators such that  $\mathcal{P} = \mathcal{P}_L \mathcal{P}_R$  is self-adjoint,  $z \in \rho(\mathcal{T})$  and  $\mathcal{R}(z) := (\mathcal{T} - z)^{-1}$ . Then,  $\mathcal{T} + \mathcal{P}$ is self-adjoint and the following is true:

- (i)  $z \in \sigma_p(\mathcal{T} + \mathcal{P}) \iff -1 \in \sigma_p(\mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L).$
- (ii) If  $-1 \in \rho(\mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)$ , then  $z \in \rho(\mathcal{T} + \mathcal{P})$  and

$$(\mathcal{T} + \mathcal{P} - z)^{-1} = \mathcal{R}(z) - \mathcal{R}(z)\mathcal{P}_L(I + \mathcal{P}_R\mathcal{R}(z)\mathcal{P}_L)^{-1}\mathcal{P}_R\mathcal{R}(z).$$

Proof. The operator  $\mathcal{T} + \mathcal{P}$  is self-adjoint since  $\mathcal{P}$  is bounded and self-adjoint. Next, we prove (i). Let  $z \in \sigma_p(\mathcal{T} + \mathcal{P})$ . Then, there exists a nonzero  $u \in \text{dom }\mathcal{T}$  such that  $(\mathcal{T} - z)u + \mathcal{P}u = 0$ . Applying  $\mathcal{R}(z)$  and using  $\mathcal{P} = \mathcal{P}_L \mathcal{P}_R$  gives us the equation  $u + \mathcal{R}(z)\mathcal{P}_L\mathcal{P}_R u = 0$ . This implies in particular  $f := \mathcal{P}_R u \neq 0$ . Applying  $\mathcal{P}_R$  yields  $f + \mathcal{P}_R \mathcal{R}(z)\mathcal{P}_L f = 0$ , i.e.  $-1 \in \sigma_p(\mathcal{P}_R \mathcal{R}(z)\mathcal{P}_L)$ . Now, let  $-1 \in \sigma_p(\mathcal{P}_R \mathcal{R}(z)\mathcal{P}_L)$ . Then, there exists a nonzero  $f \in \mathcal{G}$  such that  $f + \mathcal{P}_R \mathcal{R}(z)\mathcal{P}_L f = 0$ . We define  $u := \mathcal{R}(z)\mathcal{P}_L f \neq 0$ . Then,

$$(\mathcal{T} - z)u + \mathcal{P}u = \mathcal{P}_L f + \mathcal{P}_L \mathcal{P}_R \mathcal{R}(z)\mathcal{P}_L f = 0,$$

implying  $z \in \sigma_p(\mathcal{T} + \mathcal{P})$ . To prove (ii) we assume  $-1 \in \rho(\mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)$ . Then,

$$\begin{split} (\mathcal{T} + \mathcal{P} - z) \big( \mathcal{R}(z) - \mathcal{R}(z) \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \big) \\ &= ((\mathcal{T} - z) + \mathcal{P}_L \mathcal{P}_R) \big( \mathcal{R}(z) - \mathcal{R}(z) \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \big) \\ &= I - \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) + \mathcal{P}_L \mathcal{P}_R \mathcal{R}(z) \\ &- \mathcal{P}_L \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \\ &= I - \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \\ &+ \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L) (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \\ &- \mathcal{P}_L \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L (I + \mathcal{P}_R \mathcal{R}(z) \mathcal{P}_L)^{-1} \mathcal{P}_R \mathcal{R}(z) \\ &= I. \end{split}$$

Hence,  $\mathcal{R}(z) - \mathcal{R}(z)\mathcal{P}_L(I + \mathcal{P}_R\mathcal{R}(z)\mathcal{P}_L)^{-1}\mathcal{P}_R\mathcal{R}(z)$  is a right inverse of  $\mathcal{T} + \mathcal{P} - z$ . One shows in the same way that  $\mathcal{R}(z) - \mathcal{R}(z)\mathcal{P}_L(I + \mathcal{P}_R\mathcal{R}(z)\mathcal{P}_L)^{-1}\mathcal{P}_R\mathcal{R}(z)$  is a left inverse of  $\mathcal{T} + \mathcal{P} - z$ . Thus, the resolvent formula for  $\mathcal{T} + \mathcal{P}$  is valid.  $\Box$ 

As a consequence of Lemma 3.10, we get the following result for the operator  $H_P$  from Definition 3.9.

**Proposition 3.11.** Let  $\mathcal{G}$  be a Hilbert space,  $P = P_L P_R$ , where  $P_L : \mathcal{G} \to L^2(\mathbb{R}^\theta; \mathbb{C}^N)$ and  $P_R : L^2(\mathbb{R}^\theta; \mathbb{C}^N) \to \mathcal{G}$  are bounded operators such that  $P = P_L P_R$  is self-adjoint,  $z \in \rho(H)$  and  $R(z) = (H - z)^{-1}$  be the resolvent of the free Dirac operator. Then,  $H_P = H + P$  is self-adjoint and the following statements hold:

(i)  $z \in \sigma_p(H_P) \iff -1 \in \sigma_p(P_R R(z) P_L).$ 

(ii) If 
$$-1 \in \rho(P_R R(z) P_L)$$
, then  $z \in \rho(H_P)$  and  
 $(H_P - z)^{-1} = R(z) - R(z) P_L (I + P_R R(z) P_L)^{-1} P_R R(z).$ 

## 3.3 Dirac operators with $\delta$ -shell potentials

Next, we define Dirac operators with  $\delta$ -shell potentials supported on a special  $C^2$ surface  $\Sigma$ . Such operators are formally given by  $-i(\alpha \cdot \nabla) + m\beta + \widetilde{V}\delta_{\Sigma}$ , where  $\widetilde{V} = \widetilde{V}^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N)$  is the interaction matrix.

**Definition 3.12.** Let  $m \in \mathbb{R}$ ,  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1 and  $\widetilde{V} = \widetilde{V}^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N)$ . Then, we define the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  through

$$\operatorname{dom} H_{\widetilde{V}\delta_{\Sigma}} := \left\{ u \in H^{1}(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^{N}) : i(\alpha \cdot \nu)(\boldsymbol{t}_{\Sigma}^{+} - \boldsymbol{t}_{\Sigma}^{-})u + \frac{\widetilde{V}}{2}(\boldsymbol{t}_{\Sigma}^{+} + \boldsymbol{t}_{\Sigma}^{-})u = 0 \right\}$$
$$H_{\widetilde{V}\delta_{\Sigma}}u := (-i(\alpha \cdot \nabla) + m\beta)u_{+} \oplus (-i(\alpha \cdot \nabla) + m\beta)u_{-}.$$

Using the notations introduced below Proposition 3.6 we can write the operator  $H_{\widetilde{V}\delta_{\Sigma}}$ as  $X \upharpoonright \ker(\Gamma_0 + \widetilde{V}\Gamma_1)$ , where  $\Gamma_0 = i(\alpha \cdot \nu)(\boldsymbol{t}_{\Sigma}^+ - \boldsymbol{t}_{\Sigma}^-)$  and  $\Gamma_1 = \frac{1}{2}(\boldsymbol{t}_{\Sigma}^+ - \boldsymbol{t}_{\Sigma}^-)$ .

The self-adjointness and spectral properties of  $H_{\widetilde{V}\delta_{\Sigma}}$  have been investigated in numerous papers in various situations; see for instance [4, 5, 8, 9, 13, 16, 18, 19, 24, 28, 56, 59, 60] Interaction matrices having the form  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$ , where  $\widetilde{\eta}$  and  $\widetilde{\tau}$  are constants in  $\mathbb{R}$  or sufficiently smooth real-valued functions on  $\Sigma$ , are the most prevalent in literature. They are used to model electrostatic and Lorentz scalar interactions. In this case an explicit condition for self-adjointness is known; cf. Proposition 3.15. However, in our general setting self-adjointness is not guaranteed. We start our analysis by showing that  $H_{\widetilde{V}\delta_{\Sigma}}$  is a densely defined symmetric operator in Lemma 3.13. Afterwards, we show in Proposition 3.14 the strong connection between  $H_{\widetilde{V}\delta_{\Sigma}}$  and  $I + C_z \widetilde{V}$ , where  $C_z$  is the operator introduced in Definition 3.7.

**Lemma 3.13.** Let  $\widetilde{V} = \widetilde{V}^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N)$ . Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is symmetric and densely defined.

Proof. By [54, Corollary 3.5] the set  $C_0^{\infty}(\overline{\Omega_+}; \mathbb{C}^N) \oplus C_0^{\infty}(\overline{\Omega_-}; \mathbb{C}^N) \subset \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  is dense in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Consequently, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is densely defined. Furthermore, since  $H_{\widetilde{V}\delta_{\Sigma}} = X \upharpoonright \ker(\Gamma_0 + \widetilde{V}\Gamma_1)$ , we obtain with (3.11)

$$\begin{split} (H_{\widetilde{V}\delta_{\Sigma}}u,v)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} &- (u,H_{\widetilde{V}\delta_{\Sigma}}v)_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} = (\Gamma_{1}u,\Gamma_{0}v)_{L^{2}(\Sigma;\mathbb{C}^{N})} - (\Gamma_{0}u,\Gamma_{1}v)_{L^{2}(\Sigma;\mathbb{C}^{N})} \\ &= -(\Gamma_{1}u,\widetilde{V}\Gamma_{1}v)_{L^{2}(\Sigma;\mathbb{C}^{N})} + (\widetilde{V}\Gamma_{1}u,\Gamma_{1}v)_{L^{2}(\Sigma;\mathbb{C}^{N})} \\ &= 0 \qquad \forall u,v \in \operatorname{dom} H_{\widetilde{V}\delta_{\Sigma}}. \end{split}$$

Thus,  $H_{\widetilde{V}\delta_{\Sigma}}$  is symmetric.

 $\Box$ 

**Proposition 3.14.** Let  $\widetilde{V} = \widetilde{V}^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N)$ ,  $z \in \rho(H)$ ,  $R(z) = (H - z)^{-1}$  be the resolvent of the free Dirac operator,  $\Phi_z$  be defined as in Definition 3.5 and  $\mathcal{C}_z$  be defined as in Definition 3.7. Then,  $H_{\widetilde{V}\delta_{\Sigma}}$  is closed and  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$ , if and only if  $I + \mathcal{C}_z \widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Moreover, in this case the resolvent formula

$$(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} = R(z) - \Phi_z \widetilde{V} (I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^*$$
(3.15)

applies.

Proof. We start by showing that if  $H_{\widetilde{V}\delta_{\Sigma}}$  is closed and  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$ , then  $I + \mathcal{C}_{z}\widetilde{V}$ is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ . By Proposition 2.2 (iii) and Proposition 3.8 (i)  $\widetilde{V}$  and  $\mathcal{C}_{z}$  act as bounded operators in  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ , respectively. Hence, it suffices to show that  $I + \widetilde{V}\mathcal{C}_{z}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ ; then Proposition 2.29 shows that  $I + \mathcal{C}_{z}\widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ . Moreover, since  $I + \widetilde{V}\mathcal{C}_{z}$  is a bounded operator in  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ , we only have to prove that  $I + \widetilde{V}\mathcal{C}_{z}$  is bijective. Let  $\psi \in H^{1/2}(\Sigma; \mathbb{C}^{N})$  such that  $(I + \widetilde{V}\mathcal{C}_{z})\psi = 0$ . We set  $u = \Phi_{z}\psi$ . Then, Proposition 3.6 (i) implies  $u \in H^{1}(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^{N})$ . Furthermore, Proposition 3.6 (ii) yields  $(-i(\alpha \cdot \nabla) + m\beta - zI_{N})u_{\pm} = 0$  and Proposition 3.8 (ii) gives us

$$\Gamma_0 u = \psi \quad \text{and} \quad \Gamma_1 u = \mathcal{C}_z \psi.$$
 (3.16)

Thus, (3.16) leads to

$$\Gamma_0 u + \tilde{V} \Gamma_1 u = \psi + \tilde{V} \mathcal{C}_z \psi = 0$$

and hence  $u \in \ker(H_{\widetilde{V}\delta_{\Sigma}} - z)$ . We have  $\ker(H_{\widetilde{V}\delta_{\Sigma}} - z) = \{0\}$  since  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$ . This implies u = 0 and therefore (3.16) shows  $\psi = 0$ . Now, we turn to the surjectivity. Let  $\varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$ . Then, according to Proposition 2.3 (i) there exist functions  $w_{\pm} \in H^1(\Omega_{\pm}; \mathbb{C}^N)$  such that  $\mathbf{t}_{\Sigma}^{\pm} w_{\pm} = \frac{\mp i(\alpha \cdot \nu)}{2} \varphi \in H^{1/2}(\Sigma; \mathbb{C}^N)$ . Next, we define  $w := w_{+} \oplus w_{-} \in H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$  and see

$$\Gamma_0 w = \varphi$$
 as well as  $\Gamma_1 w = 0.$  (3.17)

Moreover, let

$$v := (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1}(X - z)w,$$

where X is the operator introduced in (3.10). By definition,  $v \in \text{dom } H_{\widetilde{V}\delta_{\Sigma}} \subset \text{dom } X$ and  $(X - z)(w - v)_{\pm} = 0$ , and thus due to (3.13) and the text below there exists a  $\psi \in H^{1/2}(\Sigma; \mathbb{C}^N)$  such that  $\Phi_z \psi = w - v$ . Hence, we use the relations (3.16) (for  $u = \Phi_z \psi$ ), (3.17) and  $v \in \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  to obtain

$$(I + \widetilde{V}\mathcal{C}_z)\psi = \Gamma_0 \Phi_z \psi + \widetilde{V}\Gamma_1 \Phi_z \psi$$
  
=  $\Gamma_0(w - v) + \widetilde{V}\Gamma_1(w - v)$   
=  $\Gamma_0 w + \widetilde{V}\Gamma_1 w = \varphi.$ 

This proves the surjectivity.

Next, let us prove the reverse direction. We assume that  $I + C_z \widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Then, by Proposition 2.29  $I + \widetilde{V}\mathcal{C}_z$  is also continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . We start by showing that  $\ker(H_{\widetilde{V}\delta_{\Sigma}} - z) = \{0\}$  in this case. To do so, we assume that  $u \in \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  and  $(H_{\widetilde{V}\delta_{\Sigma}} - z)u = 0$ . Then, according to (3.13) and the text below there exists a  $\psi \in H^{1/2}(\Sigma; \mathbb{C}^N)$  such that  $u = \Phi_z \psi$ . Since usatisfies the boundary conditions, the equation  $\Gamma_0 u + \widetilde{V} \Gamma_1 u = 0$  is valid which implies by Proposition 3.8 (ii)

$$\Gamma_0 \Phi_z \psi + V \Gamma_1 \Phi_z \psi = (I + V \mathcal{C}_z) \psi = 0.$$

Hence,  $\psi = 0$  and in turn also u = 0, which shows  $\ker(H_{\widetilde{V}\delta_{\Sigma}} - z) = \{0\}$ . Finally, we show that the expression  $R(z) - \Phi_z \widetilde{V}(I + C_z \widetilde{V})^{-1} \Phi_{\overline{z}}^*$  is a right inverse of  $H_{\widetilde{V}\delta_{\Sigma}} - z$ . This implies ran  $(H_{\widetilde{V}\delta_{\Sigma}} - z) = L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and (3.15). Moreover, as the right-hand side of (3.15) is bounded in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  by Proposition 3.3 (iii) and Proposition 3.6, this also implies that  $H_{\widetilde{V}\delta_{\Sigma}}$  is closed and  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$ .

We start by choosing  $v \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and setting

$$u := R(z)v - \Phi_z \widetilde{V}(I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v.$$

Using dom  $H = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , Proposition 3.6 (iii) and Proposition 3.8 (ii) (applied to  $\varphi = \widetilde{V}(I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v)$  yields

$$\begin{split} V\Gamma_1 u + \Gamma_0 u \\ &= \widetilde{V} \Phi_{\overline{z}}^* v - \widetilde{V} \mathcal{C}_z \widetilde{V} (I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v - i(\alpha \cdot \nu) (-i(\alpha \cdot \nu)) \widetilde{V} (I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v \\ &= \widetilde{V} \Phi_{\overline{z}}^* v - \widetilde{V} (I + \mathcal{C}_z \widetilde{V}) (I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v \\ &= 0. \end{split}$$

Hence,  $u \in \text{dom} H_{\widetilde{V}\delta_{\Sigma}}$ . We get with  $(-i(\alpha \cdot \nabla) + m\beta - zI_N)R(z)v = v$  and Proposition 3.6 (ii) that

$$\begin{split} \left( (H_{\widetilde{V}\delta_{\Sigma}} - z)u \right)_{\pm} &= (-i(\alpha \cdot \nabla) + m\beta - zI_N)u_{\pm} \\ &= (-i(\alpha \cdot \nabla) + m\beta - zI_N)(R(z)v)_{\pm} \\ &- (-i(\alpha \cdot \nabla) + m\beta - zI_N) \left( \Phi_z \widetilde{V}(I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^* v \right)_{\pm} \\ &= v_{\pm}, \end{split}$$

which concludes the proof.

The previous proposition can be used to show the self-adjointness of  $H_{\tilde{V}\delta_{\Sigma}}$ . For instance if  $I + \mathcal{C}_z \widetilde{V}$  and  $I + \mathcal{C}_{\overline{z}} \widetilde{V}$  are continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ , then it follows directly by Lemma 3.13 and Proposition 3.14 that  $H_{\tilde{V}\delta_{\Sigma}}$  is self-adjoint.

For the rest of this section we focus on interaction matrices of the form  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$ with  $\widetilde{\eta}, \widetilde{\tau} \in C_b^1(\Sigma; \mathbb{R})$ . Properties of such Dirac operators which are important with respect to this thesis are given in the following proposition.

**Proposition 3.15.** Let 
$$\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$$
 with  $\widetilde{\eta}, \widetilde{\tau} \in C_b^1(\Sigma; \mathbb{R})$  and  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2$ . If  

$$\inf_{\tau_{\Sigma} \in \Sigma} |\widetilde{d}(x_{\Sigma}) - 4| > 0,$$

then  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint. Moreover, in this case the following is true:

(i) If  $\inf_{x_{\Sigma}\in\Sigma} |\tilde{d}(x_{\Sigma})| > 0$ , then  $UH_{\tilde{V}\delta_{\Sigma}}U = H_{-4(\tilde{V}/\tilde{d})\delta_{\Sigma}}$ , where U is the self-adjoint unitary operator

$$U: L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N), \qquad Uu := u_+ \oplus (-u_-).$$

(ii) If  $\tilde{d} = -4$ , then  $H_{\tilde{V}\delta_{\Sigma}}$  induces confinement, i.e.  $H_{\tilde{V}\delta_{\Sigma}} = H_{\tilde{V}}^+ \oplus H_{\tilde{V}}^-$ , where  $H_{\tilde{V}}^{\pm}$ are operators acting in  $L^2(\Omega_{\pm}; \mathbb{C}^N)$  given by

$$H_{\widetilde{V}}^{\pm}u_{\pm} = -i(\alpha \cdot \nabla)u_{\pm} + m\beta u_{\pm},$$
  
$$\operatorname{dom} H_{\widetilde{V}}^{\pm} = \left\{ u_{\pm} \in H^{1}(\Omega_{\pm}; \mathbb{C}^{N}) : (2I_{N} \mp i(\alpha \cdot \nu)\widetilde{V})\boldsymbol{t}_{\Sigma}^{\pm}u_{\pm} = 0 \right\} \subset L^{2}(\Omega_{\pm}; \mathbb{C}^{N}).$$

(iii) The operator  $I + C_z \widetilde{V}$  is continuously invertible in  $H^r(\Sigma; \mathbb{C}^N)$  for all  $r \in [0, \frac{1}{2}]$ and  $z \in \mathbb{C} \setminus \mathbb{R}$ .

Proof. The self-adjointness of  $H_{\widetilde{V}\delta_{\Sigma}}$  follows from [60, Section 6]. Item (i) and (ii) are well known and have been shown in numerous settings, see for instance [24, Section 4], [37, Theorem 2.3 (d)] or [49] for (i), and [5, Theorem 5.5], [16, Section 5.2] or [60, Example 12] for (ii). Although the proofs do not change in our case, we provide them for completeness. Let us start with (i). After checking the definitions of U and  $H_{\widetilde{V}\delta_{\Sigma}}$ one sees that we only have to prove for  $u \in H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$ 

$$Uu \in \ker \left(\Gamma_0 + \widetilde{V}\Gamma_1\right) \iff u \in \ker \left(\Gamma_0 + \frac{-4\widetilde{V}}{\widetilde{d}}\Gamma_1\right),$$

where  $\Gamma_1$  and  $\Gamma_0$  are the operators introduced above (3.10). Using (3.1), the identity  $\widetilde{V}(\alpha \cdot \nu)\widetilde{V} = (\alpha \cdot \nu)\widetilde{d}$  and the definition of U yields

$$\begin{split} \Gamma_{0}Uu + \widetilde{V}\Gamma_{1}Uu &= 0 \\ \Longleftrightarrow \qquad i(\alpha \cdot \nu)(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} - \boldsymbol{t}_{\Sigma}^{-}(-\boldsymbol{u}_{-})) + \widetilde{V}\frac{1}{2}(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} + \boldsymbol{t}_{\Sigma}^{-}(-\boldsymbol{u}_{-})) &= 0 \\ \Leftrightarrow \qquad -(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} + \boldsymbol{t}_{\Sigma}^{-}\boldsymbol{u}_{-}) + i(\alpha \cdot \nu)\widetilde{V}\frac{1}{2}(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} - \boldsymbol{t}_{\Sigma}^{-}\boldsymbol{u}_{-}) &= 0 \\ \Leftrightarrow \qquad -4\widetilde{V}\frac{1}{2}(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} + \boldsymbol{t}_{\Sigma}^{-}\boldsymbol{u}_{-}) + \widetilde{d}i(\alpha \cdot \nu)(\boldsymbol{t}_{\Sigma}^{+}\boldsymbol{u}_{+} - \boldsymbol{t}_{\Sigma}^{-}\boldsymbol{u}_{-}) &= 0 \\ \Leftrightarrow \qquad -4\widetilde{V}\Gamma_{1} + \widetilde{d}\Gamma_{0}\boldsymbol{u} &= 0 \\ \Leftrightarrow \qquad \Gamma_{0}\boldsymbol{u} + \frac{-4\widetilde{V}}{\widetilde{d}}\Gamma_{1}\boldsymbol{u} &= 0, \end{split}$$

which proves (i).

Next, let us consider (ii). In this setting  $\tilde{d} = -4$ . Moreover, it suffices to show that the domains of  $H_{\tilde{V}\delta_{\Sigma}}$  and  $H_{\tilde{V}}^+ \oplus H_{\tilde{V}}^-$  coincide. Let us start by assuming that  $u \in \text{dom } H_{\tilde{V}\delta_{\Sigma}}$ . It is easy to see that the transmission condition  $\Gamma_0 u + \tilde{V}\Gamma_1 u = 0$  can be rewritten in the form

$$(2I_N - i(\alpha \cdot \nu)\widetilde{V})\boldsymbol{t}_{\Sigma}^+\boldsymbol{u}_+ = (2I_N + i(\alpha \cdot \nu)\widetilde{V})\boldsymbol{t}_{\Sigma}^-\boldsymbol{u}_-.$$
(3.18)

Multiplying this equation with  $(2I_N \mp i(\alpha \cdot \nu)\widetilde{V})$  and using  $((\alpha \cdot \nu)\widetilde{V})^2 = \widetilde{d}I_N = -4I_N$  yields

$$4(2I_N \mp i(\alpha \cdot \nu)\widetilde{V})\boldsymbol{t}_{\Sigma}^{\pm}\boldsymbol{u}_{\pm} = 0.$$

Hence,  $u_{\pm} \in \text{dom} H_{\widetilde{V}}^{\pm}$  and also  $u = u_{+} \oplus u_{-} \in \text{dom} H_{\widetilde{V}}^{+} \oplus H_{\widetilde{V}}^{-}$ . If  $u \in \text{dom} H_{\widetilde{V}}^{+} \oplus H_{\widetilde{V}}^{-}$ , then (3.18) is fulfilled since both sides of the equation are zero. Thus, the equation  $\Gamma_{0}u + \widetilde{V}\Gamma_{1}u = 0$  is also valid, and therefore  $u \in \text{dom} H_{\widetilde{V}\delta_{\Sigma}}^{-}$ .

It remains to prove (iii). For  $r = \frac{1}{2}$  the assertion follows from Proposition 3.14 since  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint. Next, let us consider the case  $r \in [0, 1/2)$ . From the proof of Proposition 3.8 we know that  $(\mathcal{C}_{\overline{z}} \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N))'$ , where ' is used to denote the antidual operator, is a continuous extension of  $\mathcal{C}_z$  to  $H^{-1/2}(\Sigma; \mathbb{C}^N)$ . Moreover, using the symmetry of  $\widetilde{V}$  and the fact that  $\widetilde{V}$  induces a bounded multiplication operator in  $H^{1/2}(\Sigma; \mathbb{C}^N)$  shows that  $\widetilde{V}$  can also be extended to a bounded multiplication operator in  $H^{-1/2}(\Sigma; \mathbb{C}^N)$ . Therefore,

$$\left((I + \widetilde{V}\mathcal{C}_{\overline{z}}) \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N)\right)' = I + (\mathcal{C}_{\overline{z}} \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N))'(\widetilde{V} \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N))' \quad (3.19)$$

is a continuous extension of  $I + C_z \widetilde{V}$  to  $H^{-1/2}(\Sigma; \mathbb{C}^N)$ . Since  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint, we can apply Proposition 3.14 again and obtain that  $I + C_{\overline{z}}\widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Thus, according to Proposition 2.29 the operator  $I + \widetilde{V}C_{\overline{z}}$  is also continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Therefore, the operator in (3.19) has the bounded inverse  $((I + \widetilde{V}C_{\overline{z}})^{-1} \upharpoonright H^{1/2}(\Sigma; \mathbb{C}^N))'$ . Hence, one can use interpolation to show the assertion for  $r \in [0, \frac{1}{2})$ ; cf. Section 2.1 (xiii) and Proposition 2.2 (i).

# 4 Norm resolvent convergence of Dirac operators with general strongly localized potentials

In this chapter, which is based on [14, Section 3], we find in Theorem 4.15 conditions for the norm resolvent convergence of Dirac operators with strongly localized potentials. Moreover, we show in Corollary 4.16 that these conditions are fulfilled if the potential satisfies a certain smallness condition.

Throughout this chapter we assume that  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , is a special  $C^2$ -surface as in Definition 2.1 and  $\varepsilon_{tub} \in (0, \infty)$  is chosen as in Proposition 2.12. Moreover, we assume that

$$q \in L^{\infty}((-1,1);\mathbb{R})$$
 with  $\int_{-1}^{1} q(t) dt = 1$  (4.1)

and

$$V = V^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N), \tag{4.2}$$

which we call the *interaction matrix*. For  $\varepsilon \in (0, \varepsilon_{tub})$  we define

$$V_{\varepsilon}(x) := \begin{cases} \frac{1}{\varepsilon} V(x_{\Sigma}) q\left(\frac{t}{\varepsilon}\right), & x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}, \\ 0, & x \notin \Omega_{\varepsilon}, \end{cases}$$
(4.3)

where  $\iota(x_{\Sigma}, t) = x_{\Sigma} + t\nu(x_{\Sigma})$  for  $(x_{\Sigma}, t) \in \Sigma \times \mathbb{R}$ ; cf. Definition 2.7. By Proposition 2.12 the strongly localized potential  $V_{\varepsilon}$  is well-defined. Furthermore, the properties of q and V imply  $V_{\varepsilon} = V_{\varepsilon}^* \in L^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})$ . In this chapter we study the norm resolvent convergence properties of the self-adjoint operators  $H_{V_{\varepsilon}} = H + V_{\varepsilon}$ ,  $\varepsilon \in (0, \varepsilon_{\text{tub}})$ , which are explicitly given by

$$H_{V_{\varepsilon}} = -i(\alpha \cdot \nabla) + m\beta + V_{\varepsilon}, \qquad \text{dom} \, H_{V_{\varepsilon}} = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N). \tag{4.4}$$

We do this by finding a suitable resolvent formula in terms of Bochner-integral operators, denoted by  $A_{\varepsilon}(z), B_{\varepsilon}(z)$  and  $C_{\varepsilon}(z)$ , in Section 4.1. Afterwards we introduce a shift operator which allows us to prove convergence properties of  $A_{\varepsilon}(z), B_{\varepsilon}(z)$  and  $C_{\varepsilon}(z)$  in Section 4.3. In Section 4.4 we study the limits of these operators and connect them to the resolvent of a Dirac operator with a  $\delta$ -shell potential supported on  $\Sigma$  and a rescaled interaction matrix  $\widetilde{V} = \widetilde{V}^* \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . Lastly, we find in Section 4.5 a condition for the norm resolvent convergence of  $H_{V_{\varepsilon}}$  as  $\varepsilon \to 0$ .

# 4.1 A resolvent formula for Dirac operators with strongly localized potentials

In this section we find a suitable resolvent formula for  $H_{V_{\varepsilon}}$ . In order to be able to apply Proposition 3.11, we need an appropriate factorisation for the potential  $V_{\varepsilon}$ . We recall that  $\Omega_{\varepsilon}$  was defined in Definition 2.7 and introduce the mappings

$$\mathcal{I}_{\varepsilon} : L^{2}((-\varepsilon,\varepsilon); L^{2}(\Sigma; \mathbb{C}^{N})) \to L^{2}(\Omega_{\varepsilon}; \mathbb{C}^{N}), \quad \mathcal{I}_{\varepsilon}f(\iota(x_{\Sigma},t)) := f(t)(x_{\Sigma}), \\
\mathcal{I}_{\varepsilon}^{-1} : L^{2}(\Omega_{\varepsilon}; \mathbb{C}^{N}) \to L^{2}((-\varepsilon,\varepsilon); L^{2}(\Sigma; \mathbb{C}^{N})), \quad \mathcal{I}_{\varepsilon}^{-1}u(t)(x_{\Sigma}) := u(\iota(x_{\Sigma},t)),$$
(4.5)

and

$$\mathcal{S}_{\varepsilon} : \mathcal{B}^{0}(\Sigma) \to L^{2}((-\varepsilon,\varepsilon); L^{2}(\Sigma; \mathbb{C}^{N})), \qquad \mathcal{S}_{\varepsilon}g(t) := \frac{1}{\sqrt{\varepsilon}}g\left(\frac{t}{\varepsilon}\right),$$

$$\mathcal{S}_{\varepsilon}^{-1} : L^{2}((-\varepsilon,\varepsilon); L^{2}(\Sigma; \mathbb{C}^{N})) \to \mathcal{B}^{0}(\Sigma), \quad \mathcal{S}_{\varepsilon}^{-1}g(t) := \sqrt{\varepsilon}g(\varepsilon t),$$

$$(4.6)$$

where  $\mathcal{B}^0(\Sigma) = L^2((-1,1); L^2(\Sigma; \mathbb{C}^N))$ ; cf. Section 2.1 (xi). According to Proposition 2.12 and Proposition 2.18 (iii) for any  $\varepsilon \in (0, \varepsilon_{\text{tub}})$  these mappings are welldefined, bounded, invertible, and their inverses have the claimed form; cf. [7, equations (3.6) and (3.7)]. Moreover, we set  $u_{\varepsilon} := \frac{\chi_{\Omega_{\varepsilon}}}{\sqrt{\varepsilon}}$ , where  $\chi_{\Omega_{\varepsilon}}$  is the characteristic function for  $\Omega_{\varepsilon}$ , and define the operators

$$U_{\varepsilon}: L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\Omega_{\varepsilon}; \mathbb{C}^{N}) \quad \text{and} \quad U_{\varepsilon}^{*}: L^{2}(\Omega_{\varepsilon}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})$$
(4.7)

acting on  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and  $v \in L^2(\Omega_{\varepsilon}; \mathbb{C}^N)$  as

$$U_{\varepsilon}u = (u_{\varepsilon}u) \upharpoonright \Omega_{\varepsilon} \quad \text{and} \quad U_{\varepsilon}^*v = \begin{cases} u_{\varepsilon}v & \text{in } \Omega_{\varepsilon}, \\ 0 & \text{in } \mathbb{R}^{\theta} \setminus \Omega_{\varepsilon}. \end{cases}$$

Using these newly introduced operators shows that when viewed as a multiplication operator  $V_{\varepsilon}$  can be factorised in the following way:

$$V_{\varepsilon} = U_{\varepsilon}^* \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} V q \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}.$$

This factorisation, defining for  $z \in \rho(H)$  and  $\varepsilon \in (0, \varepsilon_{tub})$  the bounded operators

$$A_{\varepsilon}(z) := R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon} : \mathcal{B}^{0}(\Sigma) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}),$$
  

$$B_{\varepsilon}(z) := \mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon} : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma),$$
  

$$C_{\varepsilon}(z) := \mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z) : L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to \mathcal{B}^{0}(\Sigma),$$
  
(4.8)

where we used  $R(z) = (H - z)^{-1}$ , and applying Proposition 3.11 for  $P_L = U_{\varepsilon}^* \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} V q$  $P_R = \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}$  and  $P = P_L P_R = V_{\varepsilon}$  gives us:

**Proposition 4.1.** Let q, V and  $V_{\varepsilon}$ ,  $\varepsilon \in (0, \varepsilon_{tub})$ , be as in (4.1)–(4.3),  $z \in \rho(H)$  and  $R(z) = (H - z)^{-1}$ , where H is the free Dirac operator introduced in Definition 3.2. Then,  $H_{V_{\varepsilon}}$  is self-adjoint and the following holds:

(i)  $z \in \sigma_p(H_{V_{\varepsilon}}) \iff -1 \in \sigma_p(B_{\varepsilon}(z)Vq).$ 

(ii) If  $-1 \in \rho(B_{\varepsilon}(z)Vq)$ , then  $z \in \rho(H_{V_{\varepsilon}})$  and the resolvent formula

$$(H_{V_{\varepsilon}}-z)^{-1} = R(z) - A_{\varepsilon}(z)Vq(I + B_{\varepsilon}(z)Vq)^{-1}C_{\varepsilon}(z)$$

holds.

Having established a resolvent formula for  $H_{V_{\varepsilon}}$  in terms of the operators  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$ and  $C_{\varepsilon}(z)$ , we find in Proposition 4.2 integral representations of these operators, which are more convenient than (4.8).

**Proposition 4.2.** Let  $z \in \rho(H)$ ,  $G_z$  be the integral kernel of  $R(z) = (H - z)^{-1}$ given by (3.3)–(3.4) and W be the Weingarten map defined in Definition 2.11. For  $\varepsilon \in (0, \varepsilon_{tub})$  the operators  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$ , and  $C_{\varepsilon}(z)$  defined by (4.8) have the integral representations

$$\begin{aligned} A_{\varepsilon}(z)f(x) &= \int_{-1}^{1} \int_{\Sigma} G_{z}(x - y_{\Sigma} - \varepsilon s\nu(y_{\Sigma}))f(s)(y_{\Sigma}) \\ &\cdot \det(I - \varepsilon sW(y_{\Sigma})) \, d\sigma(y_{\Sigma}) \, ds, \end{aligned} \\ B_{\varepsilon}(z)f(t)(x_{\Sigma}) &= \int_{-1}^{1} \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - y_{\Sigma} - \varepsilon s\nu(y_{\Sigma}))f(s)(y_{\Sigma}) \\ &\cdot \det(I - \varepsilon sW(y_{\Sigma})) \, d\sigma(y_{\Sigma}) \, ds, \end{aligned}$$
$$C_{\varepsilon}(z)u(t)(x_{\Sigma}) &= \int_{\mathbb{R}^{\theta}} G_{z}(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - y)u(y) \, dy, \end{aligned}$$

for  $f \in \mathcal{B}^0(\Sigma)$ ,  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , a.e.  $x \in \mathbb{R}^{\theta}$ , a.e.  $t \in (-1, 1)$  and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ .

*Proof.* First, we prove the claim for  $C_{\varepsilon}(z)$ . Using (3.2), (4.5) and (4.6) gives us for  $v \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , a.e.  $t \in (-1, 1)$ , and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ 

$$C_{\varepsilon}(z)v(t)(x_{\Sigma}) = \left(\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z)v\right)(t)(x_{\Sigma})$$
  
=  $\sqrt{\varepsilon}(U_{\varepsilon}R(z)v)(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}))$   
=  $\int_{\mathbb{R}^{\theta}}G_{z}(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - y)v(y)\,dy$ 

Next, to show the claim for  $A_{\varepsilon}(z)$  we choose  $f \in \mathcal{B}^0(\Sigma)$ . Applying (3.2), Proposi-

tion 2.12 (iii), (4.5) and (4.6) yields

$$\begin{split} A_{\varepsilon}(z)f(x) &= (R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}f)(x) \\ &= \int_{\mathbb{R}^{\theta}} G_{z}(x-y)\left(U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}f\right)(y)\,dy \\ &= \int_{\Omega_{\varepsilon}}^{\varepsilon} G_{z}(x-y)u_{\varepsilon}(y)\left(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}f\right)(y)\,dy \\ &= \int_{-\varepsilon}^{\varepsilon} \int_{\Sigma} G_{z}(x-(y_{\Sigma}+s\nu(y_{\Sigma})))\frac{1}{\sqrt{\varepsilon}}\left(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}f\right)(y_{\Sigma}+s\nu(y_{\Sigma})) \\ &\cdot \det(I-sW(y_{\Sigma}))\,d\sigma(y_{\Sigma})\,ds \\ &= \int_{-\varepsilon}^{\varepsilon} \int_{\Sigma} G_{z}(x-y_{\Sigma}-s\nu(y_{\Sigma}))\frac{1}{\sqrt{\varepsilon}}(\mathcal{S}_{\varepsilon}f)(s)(y_{\Sigma})\det(I-sW(y_{\Sigma}))\,d\sigma(y_{\Sigma})\,ds \\ &= \int_{-\varepsilon}^{\varepsilon} \int_{\Sigma} G_{z}(x-y_{\Sigma}-s\nu(y_{\Sigma}))\frac{1}{\varepsilon}f\left(\frac{s}{\varepsilon}\right)(y_{\Sigma})\det(I-sW(y_{\Sigma}))\,d\sigma(y_{\Sigma})\,ds \\ &= \int_{-1}^{1} \int_{\Sigma} G_{z}(x-y_{\Sigma}-\varepsilon s\nu(y_{\Sigma}))f(s)(y_{\Sigma})\det(I-\varepsilon sW(y_{\Sigma}))\,d\sigma(y_{\Sigma})\,ds \end{split}$$

for a.e.  $x \in \mathbb{R}^{\theta}$ , which is the claimed identity. The representation for  $B_{\varepsilon}(z)$  follows by combining the last two calculations.

### 4.2 The shift operator

We introduce and study a shift operator which turns out to be useful in the convergence analysis of the maps  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$  and  $C_{\varepsilon}(z)$  in (4.8). For this, we first provide a lemma which later allows us to construct a suitable extension of the normal vector field  $\nu$  on  $\Sigma$ . We mention that this section can be found in a similar form in [14, Section 3.2].

**Lemma 4.3.** Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1,  $\varepsilon_{\text{tub}} > 0$  be as in Proposition 2.12 and  $o \in C_b^1(\Sigma; \mathbb{R}^{n \times k})$  for  $n, k \in \mathbb{N}$ . Then o has an extension  $\widehat{o} \in C_b^1(\mathbb{R}^{\theta}; \mathbb{R}^{n \times k})$  such that  $\operatorname{supp} \widehat{o} \subset \Omega_{\varepsilon_{\text{tub}}}$  and  $\widehat{o}(x_{\Sigma} + t\nu(x_{\Sigma})) = o(x_{\Sigma})$  for  $(x_{\Sigma}, t) \in \Sigma \times (-\frac{\varepsilon_{\text{tub}}}{2}, \frac{\varepsilon_{\text{tub}}}{2})$ .

*Proof.* Before we start, recall the maps  $\iota$ ,  $\iota_l$ ,  $\varkappa_l$ ,  $\nu$  and  $\nu_l$  introduced in Definition 2.7. Next, let us choose a function  $\varpi \in C^1(\mathbb{R})$  such that the support of  $\varpi$  is contained in  $(-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}}), 0 \leq \varpi \leq 1$  and  $\varpi = 1$  on  $(-\frac{\varepsilon_{\text{tub}}}{2}, \frac{\varepsilon_{\text{tub}}}{2})$ . Then, we define  $\hat{o}$  through

$$\widehat{o}(x) = \begin{cases} o(x_{\Sigma})\varpi(t), & x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon_{\text{tub}}} \text{ with } (x_{\Sigma}, t) \in \Sigma \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}}), \\ 0, & x \notin \Omega_{\varepsilon_{\text{tub}}}. \end{cases}$$
(4.10)

By Proposition 2.12  $\hat{o}$  is well-defined. Moreover, it is obvious that  $\hat{o}$  has all the claimed properties besides the  $C^1$ -smoothness and the boundedness of the derivatives. Hence, it remains to show  $\hat{o} \in C_b^1(\mathbb{R}^{\theta}; \mathbb{R}^{n \times k})$ . Since  $\hat{o}$  is supported in  $\Omega_{\varepsilon_{\text{tub}}}$ , it suffices consider  $\hat{o}$  only on  $\Omega_{\varepsilon_{\text{tub}}}$ . To do so, we fix an  $x = x_{\Sigma} + t\nu(x_{\Sigma}) = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon_{\text{tub}}}$  with  $(x_{\Sigma}, t) \in \Sigma \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$ . According to Definition 2.1 there exists an  $l \in \{1, \ldots, p\}$ such that  $x_{\Sigma} \in W_l$ . Hence, we can find an  $x' \in \mathbb{R}^{\theta-1}$  such that  $x_{\Sigma} = \varkappa_l(x')$  and therefore  $x = \iota_l(x', t)$ . Moreover, since  $W_l$  is open, we can choose a  $\delta_0 > 0$  such that  $B(x_{\Sigma}, \delta_0) \subset W_l$ . Thus, the Lipschitz continuity of  $\varkappa_l$ , cf. Proposition 2.9, guarantees the existence of a  $\delta_1 > 0$  such that

$$\varkappa_l(B(x',\delta_1)) \subset B(x_{\Sigma},\delta_0) \cap \Sigma_l \subset W_l \cap \Sigma_l \subset \Sigma,$$

implying  $\iota_l(B(x', \delta_1) \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})) \subset \Omega_{\varepsilon_{\text{tub}}}$  and

$$(\widehat{o} \circ \iota_l)(y', s) = o(\varkappa_l(y'))\varpi(s), \qquad \forall (y', s) \in B(x', \delta_1) \times (-\varepsilon_{\rm tub}, \varepsilon_{\rm tub}). \tag{4.11}$$

Equation (4.11),  $o \in C_b^1(\Sigma; \mathbb{R}^{n \times k})$  and  $\varpi \in C^1(\mathbb{R}; \mathbb{R})$  show that the function

 $\widehat{o} \circ \iota_l \upharpoonright B(x', \delta_1) \times (-\varepsilon_{\mathrm{tub}}, \varepsilon_{\mathrm{tub}})$ 

is  $C^1$ -smooth. The Jacobian of  $\iota_l$  at the point  $(y', s) \in B(x', \delta_1) \times (-\varepsilon_{tub}, \varepsilon_{tub})$  is given by

$$(D\iota_l)(y',s) = \left( (I_{\theta} - sW(\varkappa_l(y'))T_l(y') \quad \nu_l(y') \right)$$
  
with  $T_l(y') = \left( (\partial_1 \varkappa_l)(y') \quad \dots \quad (\partial_{\theta-1} \varkappa_l)(y') \right),$  (4.12)

where  $W(\varkappa_l(y'))$ , which is the Weingarten map introduced in Definition 2.11, is applied column-wise to  $T_l(y')$ . Using the coordinate representation  $L_l(y')$  of  $W(\varkappa_l(y'))$  with respect to the basis  $\{\partial_j \varkappa_l(x') : j = 1, \ldots, \theta - 1\}$  we obtain

$$(D\iota_l)(y',s) = (T_l(y')(I_{\theta-1} - sL_l(y')) \quad \nu_l(y'))$$

and

$$((D\iota_l)(y',s))^T (D\iota_l)(y',s) = \begin{pmatrix} (I_{\theta-1} - s(L_l(y'))^T)(T_l(y'))^T \\ (\nu_l(y'))^T \end{pmatrix} \\ \cdot (T_l(y')(I_{\theta-1} - sL_l(y')) & \nu_l(y')) \\ = \begin{pmatrix} (I_{\theta-1} - s(L_l(y'))^T)(T_l(y'))^T T_l(y')(I_{\theta-1} - sL_l(y')) & 0 \\ 0 & 1 \end{pmatrix}.$$

In particular,  $\det((D\iota_l)(y',s))^2 = \det(I_{\theta-1} - sL_l(y'))^2 \det((T_l(y'))^T T_l(y'))$ . Moreover, note that  $(T_l(y'))^T T_l(y') = M_l(y') = I_{\theta-1} + \nabla \zeta_l(x') \nabla \zeta_l(y')^T$  with  $M_l$  as in the proof of Proposition 2.12, and hence  $\det((T_l(y'))^T T_l(y')) = 1 + |\nabla \zeta_l(y')|^2$ . If we combine this with Proposition 2.12 (ii), then we obtain

$$|\det(D\iota_{l})(y',s)| = \sqrt{1 + |\nabla\zeta_{l}(y')|^{2}} \det(I_{\theta-1} - sL_{l}(y'))$$
  
=  $\sqrt{1 + |\nabla\zeta_{l}(y')|^{2}} \det(I - sW(\varkappa_{l}(y')))$   
>  $\frac{1}{2}.$ 

In particular,  $(D\iota_l)(y', s)$  is invertible and thus the inverse function theorem shows that  $\iota_l$  is locally around (x', t) a diffeomorphism and therefore  $\hat{o} = (\hat{o} \circ \iota_l) \circ \iota_l^{-1}$ is  $C^1$ -smooth around the point  $x = \iota_l(x', t)$ . This shows that  $\hat{o}$  is  $C^1$ -smooth in  $\Omega_{\varepsilon_{\text{tub}}}$ . Furthermore, since  $|\det(D\iota_l)(y', s)| > \frac{1}{2}$  and as  $(D\iota_l)(y's)$  can be bounded by a constant which only depends on  $\zeta_l$  and  $\varepsilon_{\text{tub}}$ ,  $((D\iota_l)(y', s))^{-1}$  can also be bounded by a constant which depends only on  $\zeta_l$  and  $\varepsilon_{\text{tub}}$ . Hence, by the chain rule and  $o \in C_b^1(\Sigma; \mathbb{C}^{n \times k})$  the first order derivatives of  $\hat{o}$  are also bounded, which implies  $\hat{o} \in C_b^1(\mathbb{R}^\theta; \mathbb{C}^{n \times k})$ .

Lemma 4.3 shows that the unit normal vector field  $\nu$  of  $\Sigma$  has an extension in  $C_b^1(\mathbb{R}^{\theta};\mathbb{R}^{\theta})$ . We fix such an extension and denote it also by  $\nu$ . Next, we define for  $\delta \in \mathbb{R}$  the *shift operator* 

$$\tau_{\delta} : L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})$$
  
$$\tau_{\delta}u(x) := u\left(x + \delta\nu(x)\right), \quad x \in \mathbb{R}^{\theta}.$$
(4.13)

In the upcoming proposition we study basic properties of  $\tau_{\delta}$ .

**Proposition 4.4.** Let  $D\nu$  be the Jacobian matrix (of the extension) of  $\nu$ ,  $r \in [0, 1]$ and  $\delta_0 \in (0, \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta}\times\theta)}^{-1})$ . Then, the operators  $\tau_{\delta}, \ \delta \in [-\delta_0, \delta_0]$ , are uniformly bounded in  $H^r(\mathbb{R}^{\theta};\mathbb{C}^N)$  and for  $r' \in [0, r]$  the inequality

$$\|\tau_{\delta} - I\|_{H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to H^{r'}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C|\delta|^{r-r'}$$

$$(4.14)$$

holds for all  $\delta \in [-\delta_0, \delta_0]$ , where C > 0 is independent of  $\delta$ .

*Proof.* Fix  $\delta \in [-\delta_0, \delta_0]$  and observe first that  $I_{\theta} + \delta D\nu(x)$  is invertible for all  $x \in \mathbb{R}^{\theta}$ and the norm of the inverse is bounded by  $(1 - |\delta_0| \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta \times \theta})})^{-1}$ . The same bound holds for the modulus of the eigenvalues of  $(I_{\theta} + \delta D\nu(x))^{-1}$  and hence we conclude

$$\left|\det\left((I_{\theta} + \delta D\nu(x))^{-1}\right)\right| \leq \frac{1}{\left(1 - |\delta_0| \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta\times\theta})}\right)^{\theta}}, \quad x \in \mathbb{R}^{\theta}.$$
(4.15)

We start by showing the uniform boundedness of  $\tau_{\delta}$  for r = 0. Let  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Then, a change of variables and (4.15) lead to

$$\int_{\mathbb{R}^{\theta}} |\tau_{\delta} u(x)|^{2} dx = \int_{\mathbb{R}^{\theta}} |u(x + \delta\nu(x))|^{2} dx$$

$$= \int_{\mathbb{R}^{\theta}} |u(x + \delta\nu(x))|^{2} \frac{|\det(I_{\theta} + \delta D\nu(x))|}{|\det(I_{\theta} + \delta D\nu(x))|} dx$$

$$= \int_{\mathbb{R}^{\theta}} |u(x + \delta\nu(x))|^{2} |\det(I_{\theta} + \delta D\nu(x))| |\det((I_{\theta} + \delta D\nu(x))^{-1})| dx$$

$$\leq \frac{1}{(1 - |\delta_{0}| ||D\nu||_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta \times \theta})})^{\theta}} \int_{\mathbb{R}^{\theta}} |u(x)|^{2} dx,$$
(4.16)

and it follows that the operators  $\tau_{\delta}, \delta \in [-\delta_0, \delta_0]$ , are uniformly bounded in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . To see the uniform boundedness of the operators  $\tau_{\delta}$  in  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , let  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$ and compute in a similar way as above

$$\int_{\mathbb{R}^{\theta}} |D(\tau_{\delta}u)(x)|^{2} dx = \int_{\mathbb{R}^{\theta}} |(Du)(x + \delta\nu(x))(I_{\theta} + \delta D\nu(x))|^{2} dx$$

$$\leq \frac{\left(1 + \delta_{0} \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta\times\theta})}\right)^{2}}{\left(1 - |\delta_{0}|\|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta\times\theta})}\right)^{\theta}} \int_{\mathbb{R}^{\theta}} |Du(x)|^{2} dx.$$

$$(4.17)$$

By density this estimate remains valid for  $u \in H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Therefore, the uniform boundedness of the operators  $\tau_{\delta}$  in  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  follows from (4.16) and (4.17). Eventually, using interpolation one concludes that  $\tau_{\delta}$  is uniformly bounded in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ for any  $r \in [0, 1]$ .

It remains to prove (4.14). Since we have already shown that  $\tau_{\delta}$  is uniformly bounded in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , the claim in (4.14) holds for  $r = r' \in [0, 1]$ . Next, we show (4.14) for r' = 0 and r = 1. With the main theorem of calculus and the chain rule we find for  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$ 

$$\begin{split} \int_{\mathbb{R}^{\theta}} |\tau_{\delta} u(x) - u(x)|^2 \, dx &= \int_{\mathbb{R}^{\theta}} \left| \int_0^{\delta} \frac{d}{dt} u(x + t\nu(x)) dt \right|^2 dx \\ &= \int_{\mathbb{R}^{\theta}} \left| \int_0^{\delta} Du(x + t\nu(x))\nu(x) dt \right|^2 dx \\ &\leq \int_{\mathbb{R}^{\theta}} \left( \int_0^{\delta} |(\tau_t Du)(x)|^2 \, dt \right) \left( \int_0^{\delta} |\nu(x)|^2 \, dt \right) dx \\ &\leq |\delta| \|\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta})}^2 \int_0^{\delta} \|\tau_t Du\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^{N\times\theta})}^2 \, dt \\ &\leq C |\delta| \int_0^{\delta} \|Du\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N)}^2, \end{split}$$

where  $\tau_t Du$  is understood column-wise. By density this estimate is also valid for  $u \in H^1(\mathbb{R}^\theta; \mathbb{C}^N)$  and hence  $\|\tau_\delta - I\|_{H^1(\mathbb{R}^\theta; \mathbb{C}^N) \to L^2(\mathbb{R}^\theta; \mathbb{C}^N)} \leq C|\delta|$ . It remains to prove the claim in the case  $0 \leq r' < r \leq 1$  with  $(r', r) \neq (0, 1)$ . We set  $\mu = r - r' \in (0, 1)$  and  $v = \frac{r'}{1 - (r - r')} \in [0, 1]$ . Then,

$$r' = (1 - \mu)v + \mu 0$$
 and  $r = (1 - \mu)v + \mu 1$ 

and consequently [54, Theorem B.7] implies

$$H^{r'}(\mathbb{R}^{\theta};\mathbb{C}^{N}) = \left[H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}), H^{0}(\mathbb{R}^{\theta};\mathbb{C}^{N})\right]_{\mu} = \left[H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}), L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})\right]_{\mu}$$

and

$$H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N})=\left[H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}),H^{1}(\mathbb{R}^{\theta};\mathbb{C}^{N})\right]_{\mu}$$

Applying (xiii) from Section 2.1 yields

$$\begin{split} |I - \tau_{\delta}\|_{H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to H^{r'}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq C \|I - \tau_{\delta}\|_{[H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}),H^{1}(\mathbb{R}^{\theta};\mathbb{C}^{N})]_{\mu} \to [H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}),L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})]_{\mu}} \\ &\leq C \|I - \tau_{\delta}\|_{H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to H^{\upsilon}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{1-\mu} \|I - \tau_{\delta}\|_{H^{1}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{\mu} \\ &= C |\delta|^{r-r'}, \end{split}$$

which is exactly (4.14). This finishes the proof of this proposition.

We will also need a variant of the shift operator  $\tau_{\delta}$  that acts on functions defined on  $\Omega_{\pm}$ . Since  $\Sigma = \partial \Omega_{\pm}$  fulfils Definition 2.1, we can make use of Stein's extension operator  $E : L^2(\Omega_{\pm}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  given by Proposition 2.6. We then define the shift operator for functions on  $\Omega_{\pm}$  by

$$\tau_{\delta}^{\Omega_{\pm}} := (\tau_{\delta} E(\cdot))_{\pm} : L^2(\Omega_{\pm}; \mathbb{C}^N) \to L^2(\Omega_{\pm}; \mathbb{C}^N).$$
(4.18)

 $\Box$ 

The following properties of  $\tau_{\delta}^{\Omega_{\pm}}$  immediately follow from the properties of E and Proposition 4.4.

**Corollary 4.5.** Let  $D\nu$  be the Jacobian matrix (of the extension) of  $\nu$ ,  $r \in [0, 1]$ and  $\delta_0 \in (0, \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta\times\theta})}^{-1})$ . Then, the operators  $\tau_{\delta}^{\Omega_{\pm}}$ ,  $\delta \in [-\delta_0, \delta_0]$ , are uniformly bounded in  $H^r(\Omega_{\pm};\mathbb{C}^N)$  and for  $r' \in [0, r]$  the inequality

$$\|\tau_{\delta}^{\Omega_{\pm}} - I\|_{H^{r}(\Omega_{\pm};\mathbb{C}^{N}) \to H^{r'}(\Omega_{\pm};\mathbb{C}^{N})} \le C|\delta|^{r-r'}$$

holds for all  $\delta \in [-\delta_0, \delta_0]$ , where C > 0 is independent of  $\delta$ .

Finally, we show that the map  $t \mapsto \tau_{t\delta} u$  has a useful continuity property.

**Proposition 4.6.** Let  $D\nu$  be the Jacobian matrix (of the extension) of  $\nu$ ,  $r \in [0, 1]$ ,  $\delta_0 \in (0, \|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta \times \theta})}^{-1})$ ,  $\delta \in [-\delta_0, \delta_0]$ ,  $u \in H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and  $v \in H^r(\Omega_{\pm}; \mathbb{C}^N)$ . Then, the functions

$$f_u: (-1,1) \to H^r(\mathbb{R}^\theta; \mathbb{C}^N), \qquad t \mapsto \tau_{t\delta} u,$$

and

$$f_v^{\pm}: (-1,1) \to H^r(\Omega_{\pm}; \mathbb{C}^N), \qquad t \mapsto \tau_{t\delta}^{\Omega_{\pm}} v,$$

are continuous.

*Proof.* First, consider  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and let  $t_n, t \in (-1, 1)$  such that  $t_n \to t$  as  $n \to \infty$ . Then, with dominated convergence one gets

$$\lim_{n \to \infty} f_u(t_n) = \lim_{n \to \infty} u((\cdot) + \delta t_n \nu) = u((\cdot) + \delta t \nu) = f_u(t) \quad \text{in } H^1(\mathbb{R}^{\theta}; \mathbb{C}^N).$$

Since  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  is continuously embedded in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , the assertion follows for  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . If  $u \in H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , then there exists a sequence  $(u_n)_{n \in \mathbb{N}}$  in  $C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$  such that  $u_n \to u$  in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$  as  $n \to \infty$ . Applying Proposition 4.4 yields

$$\|f_{u}(t) - f_{u_{n}}(t)\|_{H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N})} = \|\tau_{\delta t}(u - u_{n})\|_{H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \le C\|u - u_{n}\|_{H^{r}(\mathbb{R}^{\theta};\mathbb{C}^{N})}$$

for all  $n \in \mathbb{N}$  and  $t \in (-1, 1)$ . Hence,  $f_{u_n}(t) \to f_u(t)$  uniformly with respect to  $t \in (-1, 1)$  in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$  as  $n \to \infty$ . Thus,  $f_u$  is also continuous.

It remains to verify the claim for  $f_v^{\pm}$ . Let  $t_n, t \in (-1, 1)$  such that  $t_n \to t$  for  $n \to \infty$ . Using the properties of Stein's extension operator E and the above observations, we get that  $f_{Ev}(t_n) \to f_{Ev}(t)$  in  $H^r(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Moreover, the boundedness of the restriction mapping gives us that  $f_v^{\pm}(t_n) = (\tau_{\delta t_n} E v)_{\pm} = (f_{Ev}(t_n))_{\pm}$  converges to  $(f_{Ev}(t))_{\pm} = f_v^{\pm}(t)$  in  $H^r(\Omega_{\pm}; \mathbb{C}^N)$ . This shows the continuity of  $f_v^{\pm}$ .

# **4.3 Convergence of** $A_{\varepsilon}(z)$ , $B_{\varepsilon}(z)$ and $C_{\varepsilon}(z)$

This section is devoted to the convergence analysis of the operators  $A_{\varepsilon}(z), B_{\varepsilon}(z)$ and  $C_{\varepsilon}(z)$  introduced in (4.8) for  $\varepsilon \to 0$ , and is based on [14, Section 3.3]. First, in Proposition 4.8 we study the convergence of  $C_{\varepsilon}(z)$ . Then, a duality argument allows us to investigate the convergence of  $A_{\varepsilon}(z)$  in Proposition 4.9. Eventually, in Proposition 4.10 we consider the convergence of  $B_{\varepsilon}(z)$ .

We choose the number  $\varepsilon_{ABC} > 0$  such that

$$\varepsilon_{ABC} \le \min\left\{\frac{\varepsilon_{\text{tub}}}{4}, \frac{1}{2\|D\nu\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{R}^{\theta\times\theta})}}\right\},$$
(4.19)

where  $\varepsilon_{\text{tub}}$  is specified in Proposition 2.12. Let W be the Weingarten map associated with  $\Sigma$  introduced in Definition 2.11. In our analysis, the multiplication operator  $M_{\varepsilon}: \mathcal{B}^0(\Sigma) \to \mathcal{B}^0(\Sigma)$  acting as

$$M_{\varepsilon}f(t) = \det\left(I - t\varepsilon W\right)f(t) \qquad \text{for a.e. } t \in (-1, 1) \tag{4.20}$$

turns out to be useful. In the following lemma, which is an immediate consequence of Proposition 2.12 (ii) and Proposition 2.19, some relevant properties of  $M_{\varepsilon}$  are stated.

**Lemma 4.7.** For any  $\varepsilon \in (0, \varepsilon_{ABC})$  the operator  $M_{\varepsilon}$  is bounded, invertible,

$$\|M_{\varepsilon}\|_{0\to 0} \le (1+\varepsilon C) \quad and \quad \|M_{\varepsilon} - I\|_{0\to 0} \le \varepsilon C.$$

To formulate the result concerning the convergence of  $C_{\varepsilon}(z)$ , we recall that the embedding  $\mathfrak{J}$  is defined by (2.10) and introduce the operator

$$C_0(z) := \mathfrak{J}\Phi^*_{\overline{z}} : L^2(\mathbb{R}^\theta; \mathbb{C}^N) \to \mathcal{B}^0(\Sigma).$$
(4.21)

In fact, the properties of  $\mathfrak{J}$  and  $\Phi_{\overline{z}}^*$ , see (2.10) and Proposition 3.6 (iii), imply that  $C_0(z)$  gives also rise to a bounded operator from  $L^2(\mathbb{R}^\theta; \mathbb{C}^N)$  to  $\mathcal{B}^{1/2}(\Sigma)$ .

**Proposition 4.8.** Let  $z \in \rho(H)$ ,  $R(z) = (H-z)^{-1}$ ,  $\tau_{(\cdot)}$  be the shift operator in (4.13) and  $\varepsilon \in (0, \varepsilon_{ABC})$  with  $\varepsilon_{ABC}$  satisfying (4.19). Then, for any  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  the relation

$$C_{\varepsilon}(z)u(t) = \mathbf{t}_{\Sigma}\tau_{\varepsilon t}R(z)u \qquad for \ a.e. \ t \in (-1,1)$$

$$(4.22)$$

holds in  $L^2(\Sigma; \mathbb{C}^N)$  and ran  $C_{\varepsilon}(z) \subset \mathcal{B}^{1/2}(\Sigma)$ . Moreover, the operators  $C_{\varepsilon}(z)$  are uniformly bounded from  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $\mathcal{B}^{1/2}(\Sigma)$  and for any  $r \in (0, \frac{1}{2})$  one has

$$\|C_{\varepsilon}(z) - C_0(z)\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N) \to 0} \le C\varepsilon^{1/2-r}.$$
(4.23)

Proof. First, we show (4.22) for  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . By density and continuity, this implies (4.22) for all  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Recall that  $R(z) : H^s(\mathbb{R}^{\theta}; \mathbb{C}^N) \to H^{s+1}(\mathbb{R}^{\theta}; \mathbb{C}^N)$ is bounded for  $s \in \mathbb{R}$ ; see Proposition 3.3 (iii). Hence, by the Sobolev embedding theorem R(z)u is continuous for  $u \in C_0^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and the same is true for  $\tau_{\varepsilon t} R(z)u$ . Furthermore, as  $\tau_{\varepsilon t} R(z)u \in H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  we conclude with Proposition 2.3 (ii) and Proposition 4.2 for  $t \in (-1, 1)$  and  $x_{\Sigma} \in \Sigma$  that

$$\boldsymbol{t}_{\Sigma}\tau_{\varepsilon t}R(z)u(x_{\Sigma}) = \tau_{\varepsilon t}R(z)u(x_{\Sigma}) = \int_{\mathbb{R}^{\theta}} G_{z}(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - y)u(y)dy = C_{\varepsilon}(z)u(t)(x_{\Sigma}).$$

Hence, (4.22) is true.

Next, we show the inclusion ran  $C_{\varepsilon}(z) \subset \mathcal{B}^{1/2}(\Sigma)$ . Assume that  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Then, by the boundedness of the trace operator  $\mathbf{t}_{\Sigma} : H^1(\mathbb{R}^{\theta}; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)$ , see Proposition 2.3 (ii), and Proposition 4.6 it follows that the function  $\mathbf{t}_{\Sigma}\tau_{(\cdot)\varepsilon}R(z)u$  is continuous as a mapping from (-1, 1) to  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . In particular,  $\mathbf{t}_{\Sigma}\tau_{(\cdot)\varepsilon}R(z)u$  is measurable as a mapping from (-1, 1) to  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Using again the boundedness of the trace operator  $\mathbf{t}_{\Sigma} : H^1(\mathbb{R}^{\theta}; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)$  and the uniform boundedness of the shift operator in  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , see Proposition 2.3 (ii) and Proposition 4.4, respectively, we conclude

$$\int_{-1}^{1} \|\boldsymbol{t}_{\Sigma}\tau_{\varepsilon t}R(z)u\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})}^{2} dt \leq \int_{-1}^{1} C \|R(z)u\|_{H^{1}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2} dt \leq C \|u\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2}$$

and therefore  $t_{\Sigma}\tau_{(\cdot)\varepsilon}R(z)u \in \mathcal{B}^{1/2}(\Sigma)$ . Moreover, this also shows that  $C_{\varepsilon}(z)$  is uniformly bounded from  $L^2(\mathbb{R}^{\theta};\mathbb{C}^N)$  to  $\mathcal{B}^{1/2}(\Sigma)$ .

Finally, with Proposition 2.3 (ii), Proposition 4.4 and  $C_0(z) = \Im \Phi_{\overline{z}}^* = \Im t_{\Sigma} R(z)$ , see Proposition 3.6 (iii), we have for  $r \in (0, \frac{1}{2})$  and  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ 

$$\begin{aligned} \|C_{\varepsilon}(z)u - C_{0}(z)u\|_{0}^{2} &= \|\boldsymbol{t}_{\Sigma}\tau_{\varepsilon(\cdot)}R(z)u - \mathfrak{J}\boldsymbol{t}_{\Sigma}R(z)u\|_{0}^{2} \\ &= \int_{-1}^{1} \|\boldsymbol{t}_{\Sigma}(\tau_{\varepsilon t} - I)R(z)u\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} dt \\ &\leq C\int_{-1}^{1} \|(\tau_{\varepsilon t} - I)R(z)u\|_{H^{r+1/2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2} dt \\ &\leq C\int_{-1}^{1} |\varepsilon t|^{1-2r} \|R(z)u\|_{H^{1}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2} dt \\ &\leq C\int_{-1}^{1} \varepsilon^{1-2r} \|u\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2} dt \\ &\leq C\varepsilon^{1-2r} \|u\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}^{2}, \end{aligned}$$

which leads to (4.23). Therefore, all claims are shown.

Using the convergence of  $C_{\varepsilon}(z)$ , it is not difficult to show the convergence of  $A_{\varepsilon}(z)$ . We define the natural candidate for the limit operator by

$$A_0(z) := \Phi_z \mathfrak{J}^* : \mathcal{B}^0(\Sigma) \to L^2(\mathbb{R}^\theta; \mathbb{C}^N).$$
(4.24)

**Proposition 4.9.** Let  $z \in \rho(H)$  and  $\varepsilon \in (0, \varepsilon_{ABC})$  with  $\varepsilon_{ABC}$  satisfying (4.19). Then, for any  $r \in (0, \frac{1}{2})$  one has

$$\|A_{\varepsilon}(z) - A_0(z)\|_{0 \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)} \le C\varepsilon^{1/2 - \varepsilon}$$

and, in particular, the operators  $A_{\varepsilon}(z) : \mathcal{B}^{0}(\Sigma) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})$  are uniformly bounded. *Proof.* Let  $\mathcal{I}_{\varepsilon}$ ,  $S_{\varepsilon}$ , and  $M_{\varepsilon}$  be the operators given by (4.5), (4.6), and (4.20), respectively. One verifies by a direct calculation using Proposition 2.12 (iii), (4.5), and (4.6) that  $(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon})^{*} = M_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}$ . Using this relation we conclude from (4.8) that

$$(A_{\varepsilon}(z)M_{\varepsilon}^{-1})^{*} = M_{\varepsilon}^{-1}(A_{\varepsilon}(z))^{*} = M_{\varepsilon}^{-1}(R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon})^{*}$$

$$= M_{\varepsilon}^{-1}M_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(\overline{z}) = \mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(\overline{z}) = C_{\varepsilon}(\overline{z}).$$

$$(4.25)$$

Moreover,  $(A_0(z))^* = (\Phi_z \mathfrak{J}^*)^* = \mathfrak{J} \Phi_z^* = C_0(\overline{z})$ . Hence, Lemma 4.7 and Proposition 4.8 yield

$$\begin{split} \left\| A_{\varepsilon}(z) - A_{0}(z) \right\|_{0 \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})} \\ &= \left\| A_{\varepsilon}(z) M_{\varepsilon}^{-1}(M_{\varepsilon} - I) + A_{\varepsilon}(z) M_{\varepsilon}^{-1} - A_{0}(z) \right\|_{0 \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})} \\ &\leq C \varepsilon \| C_{\varepsilon}(\overline{z}) \|_{L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to 0} + \| C_{\varepsilon}(\overline{z}) - C_{0}(\overline{z}) \|_{L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to 0} \\ &\leq C \varepsilon^{1/2 - r}, \end{split}$$

which is the claimed estimate.

Next, we study the convergence of the operators  $B_{\varepsilon}(z)$ . We define the limit operator  $B_0(z) : \mathcal{B}^0(\Sigma) \to \mathcal{B}^0(\Sigma)$  which acts on  $f \in \mathcal{B}^0(\Sigma)$  evaluated for a.e.  $t \in (-1, 1)$  as

$$B_0(z)f(t) := \frac{i}{2}(\alpha \cdot \nu) \int_{-1}^1 \operatorname{sign}(t-s)f(s) \, ds + \mathcal{C}_z \int_{-1}^1 f(s) \, ds, \qquad (4.26)$$

where  $C_z : L^2(\Sigma; \mathbb{C}^N) \to L^2(\Sigma; \mathbb{C}^N)$  is the operator defined in Definition 3.7. Using the mapping properties of  $C_z$  from Proposition 3.8 (i) it follows that  $B_0(z)$  can also be regarded as a bounded operator in  $\mathcal{B}^r(\Sigma)$  for any  $r \in [-\frac{1}{2}, \frac{1}{2}]$ . In the following proposition we show that  $B_{\varepsilon}(z)$  converges to  $B_0(z)$ . The proof of this result is more complicated than the proofs of Proposition 4.8 and Proposition 4.9, and therefore some of the more technical calculations are shifted to Appendix B.

**Proposition 4.10.** Let  $z \in \rho(H)$  and  $\varepsilon \in (0, \varepsilon_{ABC})$  with  $\varepsilon_{ABC}$  satisfying (4.19). Then, the operators  $B_{\varepsilon}(z)$  are uniformly bounded in  $\mathcal{B}^{0}(\Sigma)$  and for any  $r \in (0, \frac{1}{2})$  one has

$$\|B_{\varepsilon}(z) - B_0(z)\|_{1/2 \to 0} \le C\varepsilon^{1/2-r}.$$

*Proof.* The proof is split into several steps. Let  $\Phi_z$  be as in Definition 3.5 and let  $\tau_{(.)}^{\Omega_{\pm}}$  be defined by (4.18). We introduce the auxiliary operators

$$\widetilde{B}_{\varepsilon}(z) := B_{\varepsilon}(z) M_{\varepsilon}^{-1} : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma), \qquad (4.27)$$

which are, due to the properties of  $B_{\varepsilon}(z)$  and  $M_{\varepsilon}$  in (4.8) and (4.20), bounded and act on  $f \in \mathcal{B}^0(\Sigma)$  for a.e.  $t \in (-1, 1)$  and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$  as

$$\widetilde{B}_{\varepsilon}(z)f(t)(x_{\Sigma}) = \int_{-1}^{1} \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - y_{\Sigma} - \varepsilon s\nu(y_{\Sigma}))f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, ds.$$
(4.28)

Moreover, we define

$$\overline{B}_{\varepsilon}(z): \mathcal{B}^{1/2}(\Sigma) \to \mathcal{B}^{1/2}(\Sigma)$$

acting on  $f \in \mathcal{B}^{1/2}(\Sigma)$  for a.e.  $t \in (-1, 1)$  as

$$\overline{B}_{\varepsilon}(z)f(t) = \int_{-1}^{t} \boldsymbol{t}_{\Sigma}^{-} \tau_{\varepsilon(t-s)}^{\Omega_{-}} (\Phi_{z}f(s))_{-} ds + \int_{t}^{1} \boldsymbol{t}_{\Sigma}^{+} \tau_{\varepsilon(t-s)}^{\Omega_{+}} (\Phi_{z}f(s))_{+} ds.$$
(4.29)

First, in Step 1 we show that  $\overline{B}_{\varepsilon}(z)$  is bounded and converges to  $B_0(z)$ . Then, in Step 2 we verify an alternative representation of  $\overline{B}_{\varepsilon}(z)$ . In Step 3 we use Appendix B to compare  $\overline{B}_{\varepsilon}(z)$  and  $\widetilde{B}_{\varepsilon}(z)$ , and show that  $\widetilde{B}_{\varepsilon}(z)$  is uniformly bounded in  $\varepsilon$ . In Step 4 we combine the results from Step 1 to Step 3 to conclude the claims of this proposition.
Step 1. First, we note that, due to Definition 2.13 and Proposition 4.6, the function  $(-1,1)^2 \ni (t,s) \mapsto \Theta(\mp(t-s))\tau_{\varepsilon(t-s)}^{\Omega_{\pm}} \in \mathcal{L}(H^1(\Omega_{\pm};\mathbb{C}^N), H^1(\Omega_{\pm};\mathbb{C}^N))$  is measurable, where  $\Theta$  is the Heaviside function. Hence, it follows from the text below Definition 2.13 that the integrands in (4.29) are measurable with respect to  $(t,s) \in (-1,1)^2$ . Moreover, by the mapping properties of  $\mathbf{t}_{\Sigma}^{\pm}$ ,  $\Phi_z$  and  $\tau_{(\cdot)}^{\Omega_{\pm}}$  in Proposition 2.3, Proposition 3.6 and Corollary 4.5 respectively, the integrands are bounded by  $C \|f(s)\|_{H^{1/2}(\Sigma;\mathbb{C}^N)}$  for  $(t,s) \in (-1,1)^2$ . In particular, we conclude that for  $f \in \mathcal{B}^{1/2}(\Sigma)$ 

$$\int_{-1}^{1} \int_{-1}^{1} \left\| \Theta(t-s) \boldsymbol{t}_{\Sigma}^{-} \tau_{\varepsilon(t-s)}^{\Omega_{-}} (\Phi_{z} f(s))_{-} + \Theta(s-t) \boldsymbol{t}_{\Sigma}^{+} \tau_{\varepsilon(t-s)}^{\Omega_{+}} (\Phi_{z} f(s))_{+} \right\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})}^{2} dt ds$$

is finite. Thus, Fubini's theorem for Bochner integrals, see Proposition 2.15, yields the integrability of the integrands in (4.29) with respect to  $s \in (-1, 1)$  for a.e.  $t \in (-1, 1)$  and the measurability of  $t \mapsto \overline{B}_{\varepsilon}(z)f(t)$ . Furthermore, the bound for the integrands also implies  $\|\overline{B}_{\varepsilon}(z)f\|_{1/2} \leq C \|f\|_{1/2}$  for  $f \in \mathcal{B}^{1/2}(\Sigma)$ . Hence,  $\overline{B}_{\varepsilon}(z)$  is well-defined and uniformly bounded in  $\mathcal{B}^{1/2}(\Sigma)$ . We claim that

$$\|\overline{B}_{\varepsilon}(z) - B_0(z)\|_{1/2 \to 0} \le C\varepsilon^{1/2-r}.$$
(4.30)

To see this we remark that with Proposition 3.8 (ii) we have the pointwise representation

$$B_0(z)f(t) = \int_{-1}^t \boldsymbol{t}_{\Sigma}^- (\Phi_z f(s))_- \, ds + \int_t^1 \boldsymbol{t}_{\Sigma}^+ (\Phi_z f(s))_+ \, ds$$

for a.e.  $t \in (-1,1)$  and  $f \in \mathcal{B}^{1/2}(\Sigma)$ . Thus,  $r \in (0,\frac{1}{2})$  and direct estimates show

$$\begin{split} \left\|\overline{B}_{\varepsilon}(z)f - B_{0}(z)f\right\|_{0}^{2} &= \int_{-1}^{1} \left\|\int_{-1}^{t} \mathbf{t}_{\Sigma}^{-} (\tau_{\varepsilon(t-s)}^{\Omega_{-}} - I)(\Phi_{z}f(s))_{-} ds \\ &+ \int_{t}^{1} \mathbf{t}_{\Sigma}^{+} (\tau_{\varepsilon(t-s)}^{\Omega_{+}} - I)(\Phi_{z}f(s))_{+} ds \right\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} dt \\ &\leq \int_{-1}^{1} \left(\int_{-1}^{t} \left\|\mathbf{t}_{\Sigma}^{-} (\tau_{\varepsilon(t-s)}^{\Omega_{-}} - I)(\Phi_{z}f(s))_{-}\right\|_{H^{r}(\Sigma;\mathbb{C}^{N})} ds \\ &+ \int_{t}^{1} \left\|\mathbf{t}_{\Sigma}^{+} (\tau_{\varepsilon(t-s)}^{\Omega_{+}} - I)(\Phi_{z}f(s))_{+}\right\|_{H^{r}(\Sigma;\mathbb{C}^{N})} ds \right)^{2} dt. \end{split}$$
(4.31)

Employing Proposition 2.3, Proposition 3.6 (i) and Corollary 4.5 gives us for all  $s,t\in(-1,1)$ 

$$\left\|\boldsymbol{t}_{\boldsymbol{\Sigma}}^{\pm} \big(\tau_{\boldsymbol{\varepsilon}(t-s)}^{\Omega_{\pm}} - I\big) (\Phi_z f(s))_{\pm}\right\|_{H^r(\boldsymbol{\Sigma}; \mathbb{C}^N)} \leq C \boldsymbol{\varepsilon}^{1/2-r} \|f(s)\|_{H^{1/2}(\boldsymbol{\Sigma}; \mathbb{C}^N)}.$$

Plugging this into (4.31) yields

$$\begin{split} \left\|\overline{B}_{\varepsilon}(z)f - B_{0}(z)f\right\|_{0}^{2} \\ &\leq C\varepsilon^{2(1/2-r)} \int_{-1}^{1} \left(\int_{-1}^{t} \|f(s)\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})} \, ds + \int_{t}^{1} \|f(s)\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})} \, ds\right)^{2} dt \\ &\leq C\varepsilon^{2(1/2-r)} \int_{-1}^{1} \|f(s)\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})}^{2} \, ds = C\varepsilon^{2(1/2-r)} \|f\|_{1/2}^{2}, \end{split}$$

which implies (4.30).

Step 2. We show that the operator  $\overline{B}_{\varepsilon}(z)$  in (4.29) has the alternative representation

$$\overline{B}_{\varepsilon}(z)f(t)(x_{\Sigma}) = \int_{-1}^{1} \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) - y_{\Sigma})f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, ds \qquad (4.32)$$

for  $f \in \mathcal{B}^{1/2}(\Sigma)$ , a.e.  $t \in (-1,1)$  and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ . Let  $f \in \mathcal{B}^{1/2}(\Sigma)$  and  $t, s \in (-1,1)$  be fixed such that t > s. Note that the choice of  $\varepsilon_{\text{tub}}$  in Proposition 2.12 implies  $\varepsilon_{ABC} \leq \frac{\varepsilon_{\text{tub}}}{2} < \frac{\varepsilon_t}{2}$ ; cf. also Proposition 2.9. Hence, by Corollary 2.10 we have  $x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) \in \Omega_{-}$  for all  $x_{\Sigma} \in \Sigma$ . Moreover, we conclude from the representation of  $\Phi_z$  given in Definition 3.5 and the form of the integral kernel  $G_z$ , see (3.3)–(3.4), that  $\Phi_z f(s)$  is continuous away from  $\Sigma$ . Thus, we have for  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ 

$$\begin{aligned} \boldsymbol{t}_{\Sigma}^{-} \boldsymbol{\tau}_{\varepsilon(t-s)}^{\Omega_{-}}(\Phi_{z}f(s))_{-}(x_{\Sigma}) &= (\Phi_{z}f(s))(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma})) \\ &= \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) - y_{\Sigma})f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}). \end{aligned}$$

Analogously, for t < s and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ 

$$\begin{aligned} \boldsymbol{t}_{\Sigma}^{+} \boldsymbol{\tau}_{\varepsilon(t-s)}^{\Omega_{+}}(\Phi_{z}f(s))_{+}(x_{\Sigma}) &= (\Phi_{z}f(s))(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma})) \\ &= \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) - y_{\Sigma})f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}). \end{aligned}$$

Combining the previous two equations yields

$$\int_{-1}^{t} \boldsymbol{t}_{\Sigma}^{-} \tau_{\varepsilon(t-s)}^{\Omega_{-}} (\Phi_{z} f(s))_{-} (x_{\Sigma}) \, ds + \int_{t}^{1} \boldsymbol{t}_{\Sigma}^{+} \tau_{\varepsilon(t-s)}^{\Omega_{+}} (\Phi_{z} f(s))_{+} (x_{\Sigma}) \, ds$$

$$= \int_{-1}^{1} \int_{\Sigma} G_{z} (x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) - y_{\Sigma}) f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, ds.$$

$$(4.33)$$

Moreover, as the integrands on the right-hand side in (4.29) are Bochner integrable (cf. *Step 1*), Proposition 2.18 (iii) shows that the pointwise evaluation of the Bochner

integrals in the definition of  $\overline{B}_{\varepsilon}(z)$  in (4.29) coincides with (4.33), i.e.

$$\overline{B}_{\varepsilon}(z)f(t)(x_{\Sigma}) = \left(\int_{-1}^{t} \boldsymbol{t}_{\Sigma}^{-}\tau_{\varepsilon(t-s)}^{\Omega_{-}} \left(\Phi_{z}f(s)\right)_{-} ds + \int_{t}^{1} \boldsymbol{t}_{\Sigma}^{+}\tau_{\varepsilon(t-s)}^{\Omega_{+}} \left(\Phi_{z}f(s)\right)_{+} ds\right)(x_{\Sigma})$$
$$= \int_{-1}^{t} \boldsymbol{t}_{\Sigma}^{-}\tau_{\varepsilon(t-s)}^{\Omega_{-}} \left(\Phi_{z}f(s)\right)_{-}(x_{\Sigma}) ds + \int_{t}^{1} \boldsymbol{t}_{\Sigma}^{+}\tau_{\varepsilon(t-s)}^{\Omega_{+}} \left(\Phi_{z}f(s)\right)_{+}(x_{\Sigma}) ds.$$

This is exactly the claimed formula in (4.32).

Step 3. By the results in Appendix B the map  $\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)$  admits an extension to a bounded operator from  $\mathcal{B}^0(\Sigma)$  to  $\mathcal{B}^{1/2}(\Sigma)$  and

$$\left\|\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\right\|_{0 \to 1/2} \le C\varepsilon^{1/2} (1 + |\log(\varepsilon)|)^{1/2}.$$
(4.34)

Moreover, we claim that  $\widetilde{B}_{\varepsilon}(z)$  is uniformly bounded in  $\mathcal{B}^0(\Sigma)$ . To see this, observe first that

$$\begin{split} \|\widetilde{B}_{\varepsilon}(z)\|_{1/2 \to 1/2} &\leq \|\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\|_{1/2 \to 1/2} + \|\overline{B}_{\varepsilon}(z)\|_{1/2 \to 1/2} \\ &\leq \|\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\|_{0 \to 1/2} + \|\overline{B}_{\varepsilon}(z)\|_{1/2 \to 1/2}. \end{split}$$

Therefore, the estimate (4.34) and the uniform boundedness of  $\overline{B}_{\varepsilon}(z)$  in  $\mathcal{B}^{1/2}(\Sigma)$ shown in *Step 1* imply that  $\widetilde{B}_{\varepsilon}(z)$  is also uniformly bounded in  $\mathcal{B}^{1/2}(\Sigma)$ . The same is true for  $\widetilde{B}_{\varepsilon}(\overline{z})$  and hence also the antidual

$$(\widetilde{B}_{\varepsilon}(\overline{z}) \upharpoonright \mathcal{B}^{1/2}(\Sigma))' : \mathcal{B}^{-1/2}(\Sigma) \to \mathcal{B}^{-1/2}(\Sigma)$$

is uniformly bounded. Here, we used that  $(\mathcal{B}^{1/2}(\Sigma))'$  can be identified with  $\mathcal{B}^{-1/2}(\Sigma)$ ; see Proposition 2.21 (ii). We claim that  $(\widetilde{B}_{\varepsilon}(\overline{z}) \upharpoonright \mathcal{B}^{1/2}(\Sigma))'$  is an extension of  $\widetilde{B}_{\varepsilon}(z)$ , that is,

$$\widetilde{B}_{\varepsilon}(z)f = (\widetilde{B}_{\varepsilon}(\overline{z}) \upharpoonright \mathcal{B}^{1/2}(\Sigma))'f, \qquad f \in \mathcal{B}^{0}(\Sigma).$$
(4.35)

In fact, the identities  $\widetilde{B}_{\varepsilon}(\overline{z}) = B_{\varepsilon}(\overline{z})M_{\varepsilon}^{-1}$ , (4.8), and  $(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon})^* = M_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}$ , yield for the adjoint of  $\widetilde{B}_{\varepsilon}(\overline{z})$  in  $\mathcal{B}^0(\Sigma)$ 

$$(\widetilde{B}_{\varepsilon}(\overline{z}))^{*} = \left(\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(\overline{z})U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}M_{\varepsilon}^{-1}\right)^{*} = \mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}M_{\varepsilon}^{-1} = \widetilde{B}_{\varepsilon}(z)$$

$$(4.36)$$

and hence Proposition 2.21 (ii) implies for  $f \in \mathcal{B}^0(\Sigma)$  and  $g \in \mathcal{B}^{1/2}(\Sigma)$ 

$$\begin{split} \left\langle (\widetilde{B}_{\varepsilon}(\overline{z}) \upharpoonright \mathcal{B}^{1/2}(\Sigma))' f, g \right\rangle_{\mathcal{B}^{-1/2}(\Sigma) \times \mathcal{B}^{1/2}(\Sigma)} &= \langle f, \widetilde{B}_{\varepsilon}(\overline{z})g \rangle_{\mathcal{B}^{-1/2}(\Sigma) \times \mathcal{B}^{1/2}(\Sigma)} \\ &= (f, \widetilde{B}_{\varepsilon}(\overline{z})g)_{\mathcal{B}^{0}(\Sigma)} \\ &= (\widetilde{B}_{\varepsilon}(z)f, g)_{\mathcal{B}^{0}(\Sigma)} \\ &= \langle \widetilde{B}_{\varepsilon}(z)f, g \rangle_{\mathcal{B}^{-1/2}(\Sigma) \times \mathcal{B}^{1/2}(\Sigma)}, \end{split}$$

where  $\langle \cdot, \cdot \rangle_{\mathcal{B}^{-1/2}(\Sigma) \times \mathcal{B}^{1/2}(\Sigma)}$  denotes the sesquilinear duality product; cf. Section 2.1 (iii). This implies (4.35) and since  $(\widetilde{B}_{\varepsilon}(\overline{z}))'$  and  $\widetilde{B}_{\varepsilon}(z)$  are both uniformly bounded in  $\mathcal{B}^{-1/2}(\Sigma)$  and  $\mathcal{B}^{1/2}(\Sigma)$ , respectively, an interpolation argument, see Proposition 2.21 (i) and Section 2.1 (xiii), leads to the uniform boundedness of  $\widetilde{B}_{\varepsilon}(z)$  in  $\mathcal{B}^{0}(\Sigma)$ .

Step 4. Using the results from Step 1 to Step 3 we now complete the proof of Proposition 4.10. Since  $B_{\varepsilon}(z) = \widetilde{B}_{\varepsilon}(z)M_{\varepsilon}$ , Lemma 4.7 and the uniform boundedness of  $\widetilde{B}_{\varepsilon}(z)$  shown in Step 3 imply the uniform boundedness of  $B_{\varepsilon}(z)$  in  $\mathcal{B}^{0}(\Sigma)$ , proving the first claim of this proposition. Moreover, by applying Lemma 4.7, (4.30), (4.34) and the uniform boundedness of  $\widetilde{B}_{\varepsilon}(z)$  in the space  $\mathcal{B}^{0}(\Sigma)$  we obtain

$$\begin{split} \|B_{\varepsilon}(z) - B_{0}(z)\|_{1/2 \to 0} &= \|\overline{B}_{\varepsilon}(z)M_{\varepsilon} - B_{0}(z)\|_{1/2 \to 0} \\ &\leq \|\overline{B}_{\varepsilon}(z)(M_{\varepsilon} - I)\|_{1/2 \to 0} + \|\overline{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\|_{1/2 \to 0} + \|\overline{B}_{\varepsilon}(z) - B_{0}(z)\|_{1/2 \to 0} \\ &\leq \|\overline{B}_{\varepsilon}(z)(M_{\varepsilon} - I)\|_{0 \to 0} + \|\overline{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\|_{0 \to 1/2} + \|\overline{B}_{\varepsilon}(z) - B_{0}(z)\|_{1/2 \to 0} \\ &\leq C\left(\varepsilon + \varepsilon^{1/2}(1 + |\log(\varepsilon)|)^{1/2} + \varepsilon^{1/2 - r}\right) \\ &< C\varepsilon^{1/2 - r}. \end{split}$$

This is the claimed norm estimate and finishes the proof of this proposition.  $\Box$ 

# 4.4 Properties of the limit operators $A_0(z)$ , $B_0(z)$ and $C_0(z)$

After discussing the convergence properties of  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$  and  $C_{\varepsilon}(z)$ , we discuss in this section the limit operators  $A_0(z)$ ,  $B_0(z)$  and  $C_0(z)$ . In particular, we give in Proposition 4.14 conditions under which

$$(H-z)^{-1} - A_0(z)Vq(I+B_0(z)Vq)^{-1}C_0(z)$$

is the resolvent of  $H_{\tilde{V}\delta_{\Sigma}}$ , where  $\tilde{V} = \tilde{V}(V)$  is a rescaled interaction matrix. This section contains parts of [14, Section 4] and [15].

To study the expression  $(H - z)^{-1} - A_0(z)Vq(I + B_0(z)Vq)^{-1}C_0(z)$  it is essential to investigate the operator  $I + B_0(z)Vq$  and its inverse. This requires some technical preparations and we first introduce the operator

$$T: \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma),$$
$$Tf(t) := \frac{i}{2} \int_{-1}^{1} \operatorname{sign}(t-s) f(s) \, ds$$

and the function

$$Q(t) := -\frac{1}{2} + \int_{-1}^{t} q(s)ds, \quad t \in [-1, 1],$$
(4.37)

where  $q \in L^{\infty}((-1, 1); \mathbb{R})$  is the function introduced in (4.1). The function Q satisfies Q' = q,  $Q(-1) = -\frac{1}{2}$ , and since  $\int_{-1}^{1} q(s) ds = 1$ , also  $Q(1) = -Q(-1) = \frac{1}{2}$ . Moreover, for  $r \in [0, \frac{1}{2}]$  the map T gives rise to a bounded operator in  $\mathcal{B}^{r}(\Sigma)$  and

$$B_0(z) = T(\alpha \cdot \nu) + \mathfrak{J}\mathcal{C}_z \mathfrak{J}^* \tag{4.38}$$

with  $\mathfrak{J}$  from (2.10).

**Lemma 4.11.** Let q and V be as in (4.1) and (4.2), let  $r \in [0, \frac{1}{2}]$ , and assume  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . Then, the following is true:

- (i)  $I + T(\alpha \cdot \nu)Vq$  is boundedly invertible in  $\mathcal{B}^r(\Sigma)$  and its inverse is given by (4.40).
- (ii) If  $f \in \operatorname{ran} \mathfrak{J}$ , *i.e.* f is independent of  $t \in (-1, 1)$ , then the equation

$$(I + T(\alpha \cdot \nu)Vq)^{-1}f(t) = \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp\left(-i(\alpha \cdot \nu)VQ(t)\right)f(t)$$

holds for a.e.  $t \in (-1, 1)$ .

*Proof.* (i) We show that the operator O defined in (4.40) below is the inverse of  $I + T(\alpha \cdot \nu)Vq$ . We start by fixing  $r \in [0, \frac{1}{2}]$  and defining the operators

$$\Xi: \mathcal{B}^{r}(\Sigma) \to H^{r}(\Sigma; \mathbb{C}^{N}),$$
  

$$\Xi f:= \frac{1}{2} \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} i(\alpha \cdot \nu) V \int_{-1}^{1} \exp\left(i(\alpha \cdot \nu)V(Q(s) - \frac{1}{2})\right) q(s) f(s) \, ds,$$
(4.39)

and

$$O: \mathcal{B}^{r}(\Sigma) \to \mathcal{B}^{r}(\Sigma),$$
  

$$Of(t) := f(t) + \exp(-i(\alpha \cdot \nu)VQ(t)) \Xi f$$
  

$$-i(\alpha \cdot \nu)V \int_{-1}^{t} \exp(i(\alpha \cdot \nu)V(Q(s) - Q(t)))q(s)f(s) \, ds.$$
(4.40)

We argue that  $\Xi$  and O are bounded and that  $O = (I + T(\alpha \cdot \nu)Vq)^{-1}$ . First, we verify that  $\Xi$  is well-defined and bounded. Let  $f \in \mathcal{B}^r(\Sigma)$ . Then, the integrand in (4.39) is measurable as a function from (-1, 1) to  $H^r(\Sigma; \mathbb{C}^N)$  since the function  $\left(\exp\left(i(\alpha \cdot \nu)V\left(Q(\cdot) - \frac{1}{2}\right)\right)q(\cdot)f(\cdot),\psi\right)_{H^r(\Sigma;\mathbb{C}^N)}$  is measurable for all  $\psi \in H^r(\Sigma;\mathbb{C}^N)$ ; see Definition 2.13. In fact, the latter function is the pointwise limit of the sequence of measurable functions

$$t \mapsto \sum_{k=0}^{n} \frac{\left(\left(i(\alpha \cdot \nu)V\left(Q(t) - \frac{1}{2}\right)\right)^{k}q(t)f(t),\psi\right)_{H^{r}(\Sigma;\mathbb{C}^{N})}}{k!}$$
$$= \sum_{k=0}^{n} \frac{\left(Q(t) - \frac{1}{2}\right)^{k}q(t)\left((i(\alpha \cdot \nu)V)^{k}f(t),\psi\right)_{H^{r}(\Sigma;\mathbb{C}^{N})}}{k!}$$

Moreover, as  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}$ ,  $\alpha \cdot \nu, V \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  it follows from Proposition 2.2 (iii) that

$$\begin{split} \|\Xi f\|_{H^r(\Sigma;\mathbb{C}^N)} &= \frac{1}{2} \left\| \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} i(\alpha \cdot \nu)V \int_{-1}^1 \exp\left(i(\alpha \cdot \nu)V\left(Q(s) - \frac{1}{2}\right)\right) q(s)f(s) \, ds \right\|_{H^r(\Sigma;\mathbb{C}^N)} \\ &\leq C \int_{-1}^1 \left\| \exp\left(i(\alpha \cdot \nu)V\left(Q(s) - \frac{1}{2}\right)\right) q(s)f(s) \right\|_{H^r(\Sigma;\mathbb{C}^N)} \, ds \end{split}$$

and  $\alpha \cdot \nu, V \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  also implies  $\exp(i(\alpha \cdot \nu)V(Q(s) - \frac{1}{2})) \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  via the power series of the exponential function. Using  $q \in L^{\infty}((-1, 1); \mathbb{R})$  we conclude

$$\begin{split} \|\Xi f\|_{H^{r}(\Sigma;\mathbb{C}^{N})} &\leq C \int_{-1}^{1} \left\| \exp\left( (\alpha \cdot \nu) V(Q(s) - \frac{1}{2}) \right) \right\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N \times N})} \|q\|_{L^{\infty}((-1,1))} \|f(s)\|_{H^{r}(\Sigma;\mathbb{C}^{N})} \, ds \\ &\leq C \int_{-1}^{1} \|f(s)\|_{H^{r}(\Sigma;\mathbb{C}^{N})} \, ds \\ &\leq C \|f\|_{r}. \end{split}$$

This shows that  $\Xi$  is well-defined and bounded. Analogously, one can check that O is well-defined and bounded. Hence, in order to show (i) it suffices to prove

$$(I + T(\alpha \cdot \nu)Vq)Of = O(I + T(\alpha \cdot \nu)Vq)f = f$$
(4.41)

for all  $f \in \mathcal{B}^0(\Sigma)$ . By Proposition 2.18 (iii) this is true, if for  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$  the relation

$$(I + T(\alpha \cdot \nu)Vq)Of(\cdot)(x_{\Sigma}) = O(I + T(\alpha \cdot \nu)Vq)f(\cdot)(x_{\Sigma}) = f(\cdot)(x_{\Sigma})$$

holds a.e. on (-1, 1). For  $f \in \mathcal{B}^0(\Sigma)$  we get from Proposition 2.18 (iii) that the function  $(t, x_{\Sigma}) \mapsto f(t)(x_{\Sigma})$  is in  $L^2((-1, 1) \times \Sigma; \mathbb{C}^N)$  and thus  $f(\cdot)(x_{\Sigma}) \in L^2((-1, 1); \mathbb{C}^N)$ for  $\sigma$ -a.e.  $x_{\Sigma}$  in  $\Sigma$ . We fix such an  $x_{\Sigma} \in \Sigma$  and define  $\varphi := f(\cdot)(x_{\Sigma}) \in L^2((-1, 1); \mathbb{C}^N)$ and  $A := (\alpha \cdot \nu(x_{\Sigma}))V(x_{\Sigma})$ . Then, we have for a.e.  $t \in (-1, 1)$ 

$$(I + T(\alpha \cdot \nu)Vq)Of(t)(x_{\Sigma})$$

$$= \varphi(t) + \exp(-iAQ(t))\Xi f(x_{\Sigma}) - iA \int_{-1}^{t} \exp(iA(Q(s) - Q(t)))q(s)\varphi(s) ds$$

$$+ \frac{i}{2} \int_{-1}^{1} \operatorname{sign}(t - s)Aq(s) \Big(\varphi(s) + \exp(-iAQ(s))\Xi f(x_{\Sigma}) - iA \int_{-1}^{s} \exp(iA(Q(r) - Q(s)))q(r)\varphi(r) dr\Big) ds.$$

$$(4.42)$$

With a direct calculation we find that

$$\begin{aligned} \frac{i}{2} \int_{-1}^{1} \operatorname{sign}(t-s) A \exp(-iAQ(s))q(s) \Xi f(x_{\Sigma}) \, ds \\ &= \frac{1}{2} \left( \int_{-1}^{t} iA \exp(-iAQ(s))q(s) \, ds - \int_{t}^{1} iA \exp(-iAQ(s))q(s) \, ds \right) \Xi f(x_{\Sigma}) \\ &= -\exp(-iAQ(t))\Xi f(x_{\Sigma}) + \frac{1}{2} \left( \exp\left(-i\frac{A}{2}\right) + \exp\left(i\frac{A}{2}\right) \right) \Xi f(x_{\Sigma}) \\ &= -\exp(-iAQ(t))\Xi f(x_{\Sigma}) + \cos\left(\frac{A}{2}\right) \Xi f(x_{\Sigma}). \end{aligned}$$

Furthermore, integration by parts gives us

$$\begin{split} &-\frac{i}{2} \int_{-1}^{1} \operatorname{sign}(t-s) Aq(s) iA \int_{-1}^{s} \exp\left(iA(Q(r)-Q(s))\right) q(r)\varphi(r) \, dr \, ds \\ &= \frac{i}{2} A \int_{-1}^{t} \frac{d}{ds} \left(\exp(-iAQ(s))\right) \int_{-1}^{s} \exp(iAQ(r)) q(r)\varphi(r) \, dr \, ds \\ &- \frac{i}{2} A \int_{t}^{1} \frac{d}{ds} \left(\exp(-iAQ(s))\right) \int_{-1}^{s} \exp(iAQ(r)) q(r)\varphi(r) \, dr \, ds \\ &= iA \exp(-iAQ(t)) \int_{-1}^{t} \exp(iAQ(r)) q(r)\varphi(r) \, dr \\ &- \frac{i}{2} A \int_{-1}^{1} \exp(iA(Q(r)-\frac{1}{2})) q(r)\varphi(r) \, dr - \frac{i}{2} A \int_{-1}^{1} \operatorname{sign}(t-s) q(s)\varphi(s) \, ds \\ &= iA \exp(-iAQ(t)) \int_{-1}^{t} \exp(iAQ(r)) q(r)\varphi(r) \, dr \\ &- \frac{i}{2} A \int_{-1}^{1} \operatorname{sign}(t-s) q(s)\varphi(s) \, ds - \cos\left(\frac{A}{2}\right) \Xi f(x_{\Sigma}). \end{split}$$

A combination of the last two calculations with (4.42) yields

$$(I + T(\alpha \cdot \nu)Vq)Of(t)(x_{\Sigma}) = \varphi(t).$$

One verifies in a very similar way that O is also the left inverse of  $I + T(\alpha \cdot \nu)Vq$ . Consequently, (4.41) is true.

(ii) Let  $f \in \operatorname{ran} \mathfrak{J}$ , that is, f is independent of  $t \in (-1, 1)$ . Instead of inserting f in (4.40) we find it more convenient and easier to verify this claim directly by showing

$$(I + T(\alpha \cdot \nu)Vq)\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp(-i(\alpha \cdot \nu)VQ)f = f.$$

Similarly as above it suffices to prove

$$\left(I + T(\alpha \cdot \nu)Vq\right)\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp\left(-i(\alpha \cdot \nu)VQ\right)f(\cdot)(x_{\Sigma}) = f(\cdot)(x_{\Sigma})$$

for  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$  a.e. on (-1, 1). Thus, we again fix  $x_{\Sigma} \in \Sigma$  and use the same abbreviations as in the proof of (i). Since here f is constant with respect to t,  $\varphi = f(t)(x_{\Sigma})$  is also independent of t. We then compute

$$(I + T(\alpha \cdot \nu)Vq)\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp(-i(\alpha \cdot \nu)VQ)f(t)(x_{\Sigma})$$

$$= \cos\left(\frac{A}{2}\right)^{-1}\exp(-iAQ(t))\varphi + \frac{i}{2}\int_{-1}^{1}\operatorname{sign}(t-s)Aq(s)\cos\left(\frac{A}{2}\right)^{-1}\exp(-iAQ(s))\varphi \,ds$$

$$= \cos\left(\frac{A}{2}\right)^{-1}\exp(-iAQ(t))\varphi - \frac{\cos\left(\frac{A}{2}\right)^{-1}}{2}\int_{-1}^{1}\operatorname{sign}(t-s)\frac{d}{ds}\exp(-iAQ(s))\,ds\varphi$$

$$= \cos\left(\frac{A}{2}\right)^{-1}\exp(-iAQ(t))\varphi$$

$$- \frac{\cos\left(\frac{A}{2}\right)^{-1}}{2}\left(2\exp(-iAQ(t)) - \exp(i\frac{A}{2}) - \exp(-i\frac{A}{2})\right)\varphi$$

$$= \varphi = f(t)(x_{\Sigma})$$

for a.e.  $t \in (-1, 1)$ , which shows (ii).

In the next lemma we study relations connecting the interaction matrix V from (4.2) and the matrix  $\widetilde{V} = VS$  with

$$S := \operatorname{sinc}\left(\frac{(\alpha \cdot \nu)V}{2}\right) \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}$$
(4.43)

if  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . We call S the scaling matrix.

**Lemma 4.12.** Let  $z \in \rho(H)$ , q and V be as in (4.1) and (4.2), respectively, assume  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  and set  $\widetilde{V} = VS$ , where S is the scaling matrix from (4.43). Then, the following is true:

(i)  $S, \widetilde{V} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  and, in particular, the multiplication by  $\widetilde{V}$  gives rise to a bounded operator in  $H^r(\Sigma; \mathbb{C}^N)$  for  $r \in [0, 1]$ .

(ii) 
$$\mathfrak{J}^*q\cos\left(\frac{(\alpha\cdot\nu)V}{2}\right)^{-1}\exp(-i(\alpha\cdot\nu)VQ)\mathfrak{J}=S.$$

(iii) 
$$(I + B_0(z)Vq)(I + T(\alpha \cdot \nu)Vq)^{-1}\mathfrak{J} = \mathfrak{J}(I + \mathcal{C}_z\widetilde{V}).$$

Proof. (i) From  $\alpha \cdot \nu, V \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  we conclude, using the power series of sinc and Proposition 2.2 (iv),  $\operatorname{sinc}\left(\frac{(\alpha \cdot \nu)V}{2}\right) \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . Furthermore, the assumption  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  and Proposition 2.2 (iv) imply  $S \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ and  $\widetilde{V} = VS \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . Since the multiplication by any  $B \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ gives rise to a bounded operator in  $H^r(\Sigma; \mathbb{C}^N)$  for  $r \in [0, 1]$ , the same is true for  $\widetilde{V}$ ; cf. Proposition 2.2 (iii).

(ii) Recall that  $\mathfrak{J}$  is defined by (2.10) and that its adjoint acts as

$$\mathfrak{J}^*f = \int_{-1}^1 f(t) \, dt, \qquad f \in \mathcal{B}^0(\Sigma).$$

As in the proof of the previous lemma we use the abbreviation  $A = (\alpha \cdot \nu)V$ . Then, with Proposition 2.18 (iii) we get for  $\psi \in L^2(\Sigma; \mathbb{C}^N)$ 

$$\mathfrak{J}^* q \cos\left(\frac{A}{2}\right)^{-1} \exp(-iAQ) \mathfrak{J} \psi = \int_{-1}^1 \cos\left(\frac{A}{2}\right)^{-1} \exp(-iAQ(s)) q(s)(\mathfrak{J} \psi)(s) \, ds$$
$$= \cos\left(\frac{A}{2}\right)^{-1} \int_{-1/2}^{1/2} \exp(-iAr) \, dr \psi$$
$$= \cos\left(\frac{A}{2}\right)^{-1} \int_0^{1/2} 2\cos(Ar) \, dr \psi.$$

Using the power series of sinc one verifies  $\int_0^{1/2} 2\cos(Ar) dr = \operatorname{sinc}\left(\frac{A}{2}\right)$  and hence

$$\mathfrak{J}^*q\cos\left(\frac{A}{2}\right)^{-1}\exp(-iAQ)\mathfrak{J}\psi = \operatorname{sinc}\left(\frac{A}{2}\right)\cos\left(\frac{A}{2}\right)^{-1}\psi = S\psi$$

for  $\psi \in L^2(\Sigma; \mathbb{C}^N)$ . This shows (ii).

(iii) For  $\psi \in L^2(\Sigma; \mathbb{C}^N)$  item (ii) from the current lemma, (4.38) and Lemma 4.11 (ii) imply

$$(I + B_0(z)Vq)(I + T(\alpha \cdot \nu)Vq)^{-1}\mathfrak{J}\psi$$
  
=  $\mathfrak{J}\psi + \mathfrak{J}\mathcal{C}_z\mathfrak{J}^*Vq(I + T(\alpha \cdot \nu)Vq)^{-1}\mathfrak{J}\psi$   
=  $\mathfrak{J}\psi + \mathfrak{J}\mathcal{C}_z\mathfrak{J}^*Vq\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp(-i(\alpha \cdot \nu)VQ)\mathfrak{J}\psi$   
=  $\mathfrak{J}\psi + \mathfrak{J}\mathcal{C}_zVS\psi$   
=  $\mathfrak{J}(I + \mathcal{C}_z\widetilde{V})\psi.$ 

**Proposition 4.13.** Let  $z \in \rho(H)$ ,  $r \in [0, \frac{1}{2}]$ , q and V be as in (4.1) and (4.2), assume  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  and set  $\tilde{V} = VS$ . Moreover, assume that  $I + \mathcal{C}_z \tilde{V}$  is continuously invertible in  $H^r(\Sigma; \mathbb{C}^N)$ . Then, the operator  $I + B_0(z)Vq$  is continuously invertible in the space  $\mathcal{B}^r(\Sigma)$  and

$$(I + T(\alpha \cdot \nu)Vq)^{-1}\mathfrak{J}(I + \mathcal{C}_{z}\widetilde{V})^{-1} = (I + B_{0}(z)Vq)^{-1}\mathfrak{J}.$$
(4.44)

Proof. The operator  $B_0(z)$  is according to the text below (4.26) bounded in  $\mathcal{B}^r(\Sigma)$ . Moreover, as  $V \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  and  $q \in L^{\infty}((-1, 1))$ , V as well as q induce by Proposition 2.2 (iii) and (2.9) also bounded operators in  $\mathcal{B}^r(\Sigma)$ . Hence,  $I + B_0(z)Vq$  is bounded in  $\mathcal{B}^r(\Sigma)$  and therefore it suffices to show that  $I + B_0(z)Vq$  is bijective in  $\mathcal{B}^r(\Sigma)$ .

Let us start with the injectivity. To do so, we use the representation of  $B_0(z)$  given by (4.38) and assume

$$(I + B_0(z)Vq)f = (I + T(\alpha \cdot \nu)Vq)f + \Im \mathcal{C}_z V \Im^* qf = 0$$

for a  $f \in \mathcal{B}^r(\Sigma)$ . Applying the operator  $(I + T(\alpha \cdot \nu)Vq)^{-1}$  yields

$$f + (I + T(\alpha \cdot \nu)Vq)^{-1} \Im \mathcal{C}_z V \Im^* q f = 0.$$

Using Lemma 4.11 (ii) gives us

$$f + \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \exp\left(-i(\alpha \cdot \nu)VQ\right) \Im \mathcal{C}_z V \Im^* q f = 0.$$
(4.45)

By applying  $\mathfrak{I}^*q$  and Lemma 4.12 (ii) we obtain

$$\mathfrak{I}^* q f + \mathfrak{I}^* q \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \exp\left(-i(\alpha \cdot \nu)VQ\right) \mathfrak{I}\mathcal{C}_z V \mathfrak{I}^* q f$$

$$= (I + S\mathcal{C}_z V) \mathfrak{I}^* q f = 0.$$
(4.46)

Since  $I + \mathcal{C}_z \widetilde{V}$  is continuously invertible in  $H^r(\Sigma; \mathbb{C}^N)$  and  $\widetilde{V} = VS$ ,  $I + S\mathcal{C}_z V$ is by Proposition 2.29 also continuously invertible in  $H^r(\Sigma; \mathbb{C}^N)$ . Hence, applying  $(I + S\mathcal{C}_z V)^{-1}$  to (4.46) yields  $\mathfrak{I}^*qf = 0$  and thus (4.45) shows f = 0.

Next, we show the surjectivity. Let  $g \in \mathcal{B}^r(\Sigma)$ . We set  $f_g = (I + T(\alpha \cdot \nu)Vq)^{-1}(g + \Im\psi)$ , where

$$\psi = -(I + \mathcal{C}_z \widetilde{V})^{-1} \mathcal{C}_z \mathfrak{I}^* V q (I + T(\alpha \cdot \nu) V q)^{-1} g$$

Applying  $(I + B_0(z)Vq) = I + (T(\alpha \cdot \nu) + \Im C_z \Im^*)Vq$  to  $f_g$  and Lemma 4.12 (iii) yield

$$(I + B_0(z)Vq)f_g = g + \Im \mathcal{C}_z \Im^* Vq(I + T(\alpha \cdot \nu)Vq)^{-1}g + (I + B_0(z)Vq)(I + T(\alpha \cdot \nu)Vq)^{-1}\Im \psi = g + \Im \mathcal{C}_z \Im^* Vq(I + T(\alpha \cdot \nu)Vq)^{-1}g + \Im(I + \mathcal{C}_z \widetilde{V})\psi = g,$$

which shows that  $I + B_0(z)Vq$  is continuously invertible in  $\mathcal{B}^r(\Sigma)$ . Applying the operators  $(I + \mathcal{C}_z \widetilde{V})^{-1}$  and  $(I + B_0(z)Vq)^{-1}$  from the left and from the right to Lemma 4.12 (iii), respectively, yields (4.44).

Having provided all these preliminary results, we are well-equipped to prove the main result of this section, which is a resolvent formula for  $H_{\tilde{V}\delta_{\Sigma}}$  in terms of the operators  $A_0(z)$ ,  $B_0(z)$  and  $C_0(z)$ .

**Proposition 4.14.** Let  $z \in \rho(H)$ ,  $R(z) = (H-z)^{-1}$  be the resolvent of the free Dirac operator, q and V be as in (4.1) and (4.2), assume  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ and set  $\widetilde{V} = VS$ . Moreover, assume that  $I + \mathcal{C}_z \widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Then,  $I + B_0(z)Vq$  is continuously invertible in  $\mathcal{B}^{1/2}(\Sigma)$ ,  $H_{\widetilde{V}\delta_{\Sigma}}$  is closed,  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$  and the formula

$$(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} = R(z) - A_0(z)Vq(I + B_0(z)Vq)^{-1}C_0(z)$$

holds.

*Proof.* Since  $I + C_z \widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ , Proposition 3.14 implies that  $H_{\widetilde{V}\delta_{\Sigma}}$  is closed,  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$  and

$$(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} = R(z) - \Phi_z \widetilde{V} (I + \mathcal{C}_z \widetilde{V})^{-1} \Phi_{\overline{z}}^*.$$
(4.47)

Proposition 4.13 and the assumption  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$  show that the operator  $I + B_0(z)Vq$  is continuously invertible in  $\mathcal{B}^{1/2}(\Sigma)$ . Hence, by (4.21), (4.44) and Lemma 4.11 (ii) we get for  $v \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ 

$$(I + B_0(z)Vq)^{-1}C_0(z)v$$

$$= (I + T(\alpha \cdot \nu)Vq)^{-1}\mathfrak{J}(I + \mathcal{C}_z \widetilde{V})^{-1}\Phi_{\overline{z}}^*v$$

$$= \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\exp(-i(\alpha \cdot \nu)VQ)\mathfrak{J}(I + \mathcal{C}_z \widetilde{V})^{-1}\Phi_{\overline{z}}^*v.$$
(4.48)

With  $A_0(z) = \Phi_z \mathfrak{J}^*$ , Lemma 4.12 (ii), and  $VS = \widetilde{V}$  we conclude

$$\begin{aligned} A_0(z)Vq(I+B_0(z)Vq)^{-1}C_0(z)v \\ &= \Phi_z V\mathfrak{J}^*q\cos\left(\frac{(\alpha\cdot\nu)V}{2}\right)^{-1}\exp(-i(\alpha\cdot\nu)VQ)\mathfrak{J}(I+\mathcal{C}_z\widetilde{V})^{-1}\Phi_{\overline{z}}^*v \\ &= \Phi_z VS(I+\mathcal{C}_z\widetilde{V})^{-1}\Phi_{\overline{z}}^*v \\ &= \Phi_z\widetilde{V}(I+\mathcal{C}_z\widetilde{V})^{-1}\Phi_{\overline{z}}^*v. \end{aligned}$$

Inserting this observation into (4.47) yields the claimed resolvent formula.

### 4.5 Convergence conditions for Dirac operators with general strongly localized potentials

Now, we are in a position to state the first main result of this thesis, which provides sufficient conditions for the norm resolvent convergence of  $H_{V_{\varepsilon}}$ .

**Theorem 4.15.** Let q and V be as in (4.1) and (4.2),  $\varepsilon_{ABC} > 0$  as in (4.19),  $V_{\varepsilon}$  be defined by (4.3) and assume that for some  $z \in \rho(H)$  the following conditions are fulfilled:

- (i) There exists an  $\varepsilon_{\text{conv}} \in (0, \varepsilon_{ABC}]$  such that the inverse  $(I + B_{\varepsilon}(z)Vq)^{-1}$  exists for  $\varepsilon \in (0, \varepsilon_{\text{conv}})$  and is uniformly bounded in  $\mathcal{B}^0(\Sigma)$ .
- (ii)  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N}).$
- (iii) The operator  $I + \mathcal{C}_z \widetilde{V}$  ( $\widetilde{V} = VS$  with S from (4.43)) is bijective in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ .

Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint,  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}}) \cap \rho(H_{V_{\varepsilon}})$  for all  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ and for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}$$
(4.49)

for all  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ . In particular,  $H_{V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* Throughout this proof we assume  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ . By Proposition 4.14 we have that  $I + B_0(z)Vq$  is continuously invertible in  $B^{1/2}(\Sigma)$ ,  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$  and

$$(H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1}=(H-z)^{-1}-A_0(z)Vq(I+B_0(z)Vq)^{-1}C_0(z).$$

The assumptions and Proposition 4.1 (ii) guarantee  $z \in \rho(H_{V_{\varepsilon}})$  as well as

$$(H_{V_{\varepsilon}}-z)^{-1} = (H-z)^{-1} - A_{\varepsilon}(z)Vq(I+B_{\varepsilon}(z)Vq)^{-1}C_{\varepsilon}(z)$$

Subtracting the above two equations yields

$$(H_{V_{\varepsilon}}-z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1} = -A_{\varepsilon}(z)Vq(I+B_{\varepsilon}(z)Vq)^{-1}C_{\varepsilon}(z) + A_{0}(z)Vq(I+B_{0}(z)Vq)^{-1}C_{0}(z) = -A_{\varepsilon}(z)Vq(I+B_{\varepsilon}(z)Vq)^{-1}(C_{\varepsilon}(z)-C_{0}(z))$$

$$(4.50) - A_{\varepsilon}(z)Vq((I+B_{\varepsilon}(z)Vq)^{-1} - (I+B_{0}(z)Vq)^{-1})C_{0}(z) - (A_{\varepsilon}(z)-A_{0}(z))Vq(I+B_{0}(z)Vq)^{-1}C_{0}(z).$$

In the following we use the uniform boundedness of  $A_{\varepsilon}(z) : \mathcal{B}^{0}(\Sigma) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})$ and  $(I + B_{\varepsilon}(z)Vq)^{-1} : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma)$ ; cf. Proposition 4.9 and assumption (i). Employing this and Proposition 4.8 we see that

$$\begin{aligned} \left\| A_{\varepsilon}(z) Vq(I + B_{\varepsilon}(z) Vq)^{-1} (C_{\varepsilon}(z) - C_{0}(z)) \right\|_{L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})} \\ &\leq C \| C_{\varepsilon}(z) - C_{0}(z) \|_{L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to 0} \\ &\leq C \varepsilon^{1/2 - r}. \end{aligned}$$

$$(4.51)$$

Since  $C_0(z) : L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to \mathcal{B}^{1/2}(\Sigma)$  and  $(I + B_0(z)Vq)^{-1} : \mathcal{B}^{1/2}(\Sigma) \to \mathcal{B}^{1/2}(\Sigma)$  are bounded, see (4.21) and Proposition 4.14, we get from Proposition 4.10

$$\begin{aligned} \left\| A_{\varepsilon}(z) Vq \left( (I + B_{\varepsilon}(z) Vq)^{-1} - (I + B_{0}(z) Vq)^{-1} \right) C_{0}(z) \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq C \left\| (I + B_{\varepsilon}(z) Vq)^{-1} - (I + B_{0}(z) Vq)^{-1} \right\|_{1/2 \to 0} \\ &\leq C \left\| (I + B_{\varepsilon}(z) Vq)^{-1} \right\|_{0 \to 0} \\ &\quad \cdot \left\| (B_{\varepsilon}(z) - B_{0}(z)) Vq \right\|_{1/2 \to 0} \left\| (I + B_{0}(z) Vq)^{-1} \right\|_{1/2 \to 1/2} \\ &\leq C \varepsilon^{1/2 - r}. \end{aligned}$$

$$(4.52)$$

Eventually, in a similar way as in (4.51), we find with Proposition 4.9 that

$$\begin{aligned} \left\| (A_{\varepsilon}(z) - A_0(z)) Vq(I + B_0(z) Vq)^{-1} C_0(z) \right\|_{L^2(\mathbb{R}^{\theta}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)} \\ &\leq C \|A_{\varepsilon}(z) - A_0(z)\|_{0 \to L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)} \\ &\leq C \varepsilon^{1/2 - r}. \end{aligned}$$

$$(4.53)$$

Combining (4.51)-(4.53) with (4.50) shows (4.49).

It remains to prove the self-adjointness of  $H_{\widetilde{V}\delta_{\Sigma}}$ . Let us first consider the case  $z \in \mathbb{R}$ . In this case  $(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1}$  is a bounded self-adjoint operator with the dense range dom  $H_{\widetilde{V}\delta_{\Sigma}}$ ; cf. Lemma 3.13. Thus,

$$H_{\widetilde{V}\delta_{\Sigma}} = ((H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1})^{-1} + z$$

is also self-adjoint. Now, let  $z \in \mathbb{C} \setminus \mathbb{R}$ . Since  $(H_{V_{\varepsilon}} - z)^{-1}$  converges in the the operator norm to  $(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1}$ , the adjoint resolvent  $(H_{V_{\varepsilon}} - \overline{z})^{-1}$  converges also in the operator norm to  $((H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1})^*$ . Furthermore, ran  $(H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} = \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  is dense in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  by Lemma 3.13. Hence, it follows from [62, Theorem VIII.22] that there exists a self-adjoint operator  $\widetilde{H}$  such that  $z \in \rho(\widetilde{H})$  and  $(\widetilde{H} - z)^{-1} = (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1}$ . This implies  $\widetilde{H} = H_{\widetilde{V}\delta_{\Sigma}}$  and therefore  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}})$ .

To make the conditions of Theorem 4.15 more tangible, we show in the upcoming corollary that these conditions are satisfied if  $\|V\|_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$  is sufficiently small.

**Corollary 4.16.** Let q and V be as in (4.1) and (4.2),  $\varepsilon_{ABC} > 0$  as in (4.19),  $V_{\varepsilon}$  be defined by (4.3) and  $z \in \rho(H)$ . If  $||V||_{W_{\infty}^{1}(\Sigma; \mathbb{C}^{N \times N})}$  is sufficiently small, then the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint,  $z \in \rho(H_{\widetilde{V}\delta_{\Sigma}}) \cap \rho(H_{V_{\varepsilon}})$  for all  $\varepsilon \in (0, \varepsilon_{ABC})$ , and for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2 - r}$$

for all  $\varepsilon \in (0, \varepsilon_{ABC})$ . In particular,  $H_{V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* Let us shortly check that the conditions of Theorem 4.15 are fulfilled. We start with (i). By Proposition 2.2 (iii), Proposition 4.10 and the comments below (2.9) we have

$$\sup_{\varepsilon \in (0,\varepsilon_{ABC})} \|B_{\varepsilon}(z)Vq\|_{0 \to 0} \le C \|q\|_{L^{\infty}((-1,1))} \|V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N \times N})} \sup_{\varepsilon \in (0,\varepsilon_{ABC})} \|B_{\varepsilon}(z)\|_{0 \to 0} < 1$$

if  $||V||_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$  is sufficiently small, i.e. (i) of Theorem 4.15 is fulfilled if we set  $\varepsilon_{\text{conv}} = \varepsilon_{ABC}$  and  $||V||_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$  is sufficiently small. For condition (ii) we use the power series of cos and cosh to estimate

$$\left\|\cos\left(\frac{(\alpha\cdot\nu)V}{2}\right) - I_N\right\|_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})} \le \cosh\left(\frac{\|(\alpha\cdot\nu)V\|_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}}{2}\right) - 1$$

In particular, if  $\|V\|_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$  is small enough, then  $\cos\left(\frac{(\alpha\cdot\nu)V}{2}\right)$  is invertible in  $W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})$  and

$$\left\|\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1}\right\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N \times N})} \leq \frac{1}{2 - \cosh\left(\frac{\|(\alpha \cdot \nu)V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N \times N})}}{2}\right)}$$

Now, let us turn to (iii). Using  $\tilde{V} = VS$ , (4.43), the power series of sinc and the estimate from above we obtain

$$\begin{split} \|\widetilde{V}\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} \\ &\leq \|V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} 2 \frac{\sinh\left(\frac{\|(\alpha\cdot\nu)V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})}}{2}\right)}{\|(\alpha\cdot\nu)V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} \left(2 - \cosh\left(\frac{\|(\alpha\cdot\nu)V\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})}}{2}\right)\right)}. \end{split}$$

Hence,

$$\|\mathcal{C}_{z}\widetilde{V}\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})\to H^{1/2}(\Sigma;\mathbb{C}^{N})} \leq C\|\mathcal{C}_{z}\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})\to H^{1/2}(\Sigma;\mathbb{C}^{N})}\|\widetilde{V}\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$$

is smaller than one if  $\|V\|_{W^1_{\infty}(\Sigma;\mathbb{C}^{N\times N})}$  is sufficiently small. In particular, in this case  $I + \mathcal{C}_z \widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma;\mathbb{C}^N)$ , i.e. (iii) of Theorem 4.15 is also fulfilled.

### 5 An explicit convergence condition for Dirac operators with strongly localized electrostatic and Lorentz scalar potentials

In Theorem 4.15 we provided conditions, which imply the norm resolvent convergence of  $H_{V_{\varepsilon}}$  as  $\varepsilon \to 0$ , with  $H_{V_{\varepsilon}}$  as in (4.4). Unfortunately, the conditions in Theorem 4.15 and Corollary 4.16 are not explicit and it may be hard to verify them. The aim of this chapter is to show that under the assumptions

$$q \in L^{\infty}((-1,1);[0,\infty))$$
 such that  $\int_{-1}^{1} q(t)dt = 1$  (5.1)

and

$$V = \eta I_N + \tau \beta \quad \text{with} \quad \eta, \tau \in C_b^1(\Sigma; \mathbb{R}),$$
(5.2)

one can simplify the three conditions in Theorem 4.15 to the explicit and simple condition

$$\sup_{x_{\Sigma} \in \Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}, \qquad d = \eta^2 - \tau^2.$$
(5.3)

We focus on interaction matrices of the form  $V = \eta I_N + \tau \beta$ , which model electrostatic and Lorentz scalar interactions, as they are the most prevalent interaction matrices in literature; cf. Section 3.3. In this situation we get with the help of (3.1) the identity  $((\alpha \cdot \nu)V)^2 = dI_N$ . Hence, the power series expansions of cos and sinc give us

$$\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right) = \cos\left(\frac{\sqrt{d}}{2}\right)I_N \quad \text{and} \quad \operatorname{sinc}\left(\frac{(\alpha \cdot \nu)V}{2}\right) = \operatorname{sinc}\left(\frac{\sqrt{d}}{2}\right)I_N. \tag{5.4}$$

This and (5.3) immediately imply  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ , i.e. (ii) in Theorem 4.15 is fulfilled. Moreover, in this case the scaling matrix, see (4.43), is given by

$$S = \operatorname{sinc}\left(\frac{(\alpha \cdot \nu)V}{2}\right) \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} = \frac{\operatorname{sinc}\left(\frac{\sqrt{d}}{2}\right)}{\cos\left(\frac{\sqrt{d}}{2}\right)} I_N = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right) I_N, \tag{5.5}$$

where tanc is the function from item (xx) of Section 2.1, and therefore

$$\widetilde{V} = VS = \widetilde{\eta}I_N + \widetilde{\tau}\beta \quad \text{with} \quad (\widetilde{\eta}, \widetilde{\tau}) = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)(\eta, \tau)$$

Thus,

$$\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = 4 \tan\left(\frac{\sqrt{d}}{2}\right)^2 \tag{5.6}$$

and by (5.3)

$$\inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma}) - 4| > 0$$

Now, Proposition 3.15 shows that  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and that  $I + \mathcal{C}_{z}\widetilde{V}$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$  for  $z \in \mathbb{C} \setminus \mathbb{R}$ . In particular, (ii) of Theorem 4.15 is also fulfilled for  $z \in \mathbb{C} \setminus \mathbb{R}$ . Thus, in order to prove the norm resolvent convergence of  $H_{V_{\varepsilon}}$ one only has to guarantee that (i) of Theorem 4.15 is satisfied for  $z \in \mathbb{C} \setminus \mathbb{R}$ , i.e. it is suffices to show that the inverse of  $I + B_{\varepsilon}(z)Vq$  is uniformly bounded in  $\mathcal{B}^0(\Sigma)$ .

We proceed in the current chapter as follows: First, we show in Section 5.1 the uniform boundedness of  $(I+B_{\varepsilon}(z)Vq)^{-1}$  in  $\mathcal{B}^{0}(\Sigma)$  if  $\Sigma$  is a rotated  $C_{b}^{2}$ -graph. Afterwards, we use in the proof of the main theorem, Theorem 5.20, a partition of unity to reduce the general case of a special  $C^{2}$ -surface to the case of a rotated  $C_{b}^{2}$ -graph. However, in this case the proof of the norm resolvent convergence follows immediately from the comments above and Section 5.1. Before we continue, let us mention that this chapter is based on [15].

# 5.1 Analysis of $I + B_{\varepsilon}(z)Vq$ for rotated $C_b^2$ -graphs

In this section we show that if  $\Sigma$  is a rotated  $C_b^2$ -graph and if  $\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}$ ,  $d = \eta^2 - \tau^2$ , then (i) of Theorem 4.15 is fulfilled, i.e.  $(I + B_{\varepsilon}(z)Vq)^{-1}$  is uniformly bounded in  $\mathcal{B}^0(\Sigma)$ . To prove this result a very careful and deep analysis of  $I + B_{\varepsilon}(z)Vq$ is necessary. We do this by meticulously studying  $I + B_{\varepsilon}(z)Vq$  in the case where  $\Sigma$ is a hyperplane and  $\eta$  and  $\tau$  are constant in Section 5.1.1. Then, we use a parameter dependent partition of unity in Section 5.1.2 to transfer the results to the case where  $\Sigma$  is a rotated  $C_b^2$ -graph and  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ .

Recall the bounded operators  $\widetilde{B}_{\varepsilon}(z) : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma)$  and  $\overline{B}_{\varepsilon}(z) : \mathcal{B}^{1/2}(\Sigma) \to \mathcal{B}^{1/2}(\Sigma)$ from (4.27) and (4.29), respectively. By (B.1) the difference of  $\widetilde{B}_{\varepsilon}(z)$  and  $\overline{B}_{\varepsilon}(z)$  can be extended to a bounded operator mapping from  $\mathcal{B}^{0}(\Sigma)$  to  $\mathcal{B}^{1/2}(\Sigma)$ . In particular, this extension acts also as a bounded operator in  $\mathcal{B}^{0}(\Sigma)$ . Thus,

$$\overline{B}_{\varepsilon}(z) = \widetilde{B}_{\varepsilon}(z) + (\overline{B}_{\varepsilon}(z) - \widetilde{B}_{\varepsilon}(z))$$

can also be extended to a bounded operator in  $\mathcal{B}^0(\Sigma)$ . We denote this extension also by  $\overline{B}_{\varepsilon}(z)$  and according to (4.28), (B.3), (B.15) and (B.16) it has the representation

$$\overline{B}_{\varepsilon}(z)f(t)(x_{\Sigma}) = \int_{-1}^{1} \int_{\Sigma} G_{z}(x_{\Sigma} + \varepsilon(t-s)\nu(x_{\Sigma}) - y_{\Sigma})f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, ds \qquad (5.7)$$

for  $f \in \mathcal{B}^0(\Sigma)$ , a.e.  $t \in (-1, 1)$  and  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ . According to Lemma 4.7, (4.27) and (B.1) we have

$$\begin{aligned} \|\overline{B}_{\varepsilon}(z) - B_{\varepsilon}(z)\|_{0 \to 0} &\leq \|\overline{B}_{\varepsilon}(z) - \widetilde{B}_{\varepsilon}(z)\|_{0 \to 0} + \|\widetilde{B}_{\varepsilon}(z) - B_{\varepsilon}(z)\|_{0 \to 0} \\ &\leq \|\overline{B}_{\varepsilon}(z) - \widetilde{B}_{\varepsilon}(z)\|_{0 \to 1/2} + \|B_{\varepsilon}(z)M_{\varepsilon}^{-1}(I - M_{\varepsilon})\|_{0 \to 0} \\ &\leq C\varepsilon^{1/2}(1 + |\log(\varepsilon)|)^{1/2} \quad \forall \varepsilon \in (0, \varepsilon_{ABC}) \end{aligned}$$
(5.8)

with  $\varepsilon_{ABC}$  defined by (4.19). This leads us to studying  $\overline{B}_{\varepsilon}(z)$  instead of  $B_{\varepsilon}(z)$ . Next, we transfer this operator, which acts in  $\mathcal{B}^0(\Sigma)$ , to  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . To do so, let  $\zeta \in C_b^2(\mathbb{R}^{\theta}; \mathbb{R}), \, \kappa \in \mathrm{SO}(\theta)$  and

$$\Sigma = \Sigma_{\zeta,\kappa} := \{ \kappa(x', \zeta(x')) : x' \in \mathbb{R}^{\theta - 1} \},\$$

where we used the convention from Section 2.1 (xvi) that  $(x', \zeta(x'))$  is an abbreviation for  $(x'^T, \zeta(x'))^T$ . Then, we introduce the isomorphism

$$\iota_{\zeta,\kappa}: L^2(\Sigma_{\zeta,\kappa}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N), \quad (\iota_{\zeta,\kappa}f)(x') := f\big(\kappa(x', \zeta(x'))\big).$$
(5.9)

Through transforming integrals on  $\Sigma_{\zeta,\kappa}$  to integrals on  $\mathbb{R}^{\theta-1}$  we get the following norms for  $\iota_{\zeta,\kappa}$  and its inverse:

$$\|\iota_{\zeta,\kappa}\|_{L^{2}(\Sigma_{\zeta,\kappa};\mathbb{C}^{N})\to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} = \sup_{x'\in\mathbb{R}^{\theta-1}} (1+|\nabla\zeta(x')|^{2})^{-1/4}, \|\iota_{\zeta,\kappa}^{-1}\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to L^{2}(\Sigma_{\zeta,\kappa};\mathbb{C}^{N})} = \sup_{x'\in\mathbb{R}^{\theta-1}} (1+|\nabla\zeta(x')|^{2})^{1/4}.$$
(5.10)

Note that the definition of  $H^r(\Sigma_{\zeta,\kappa}; \mathbb{C}^N)$ ,  $r \in [0,2]$ , see (2.2), implies that  $\iota_{\zeta,\kappa}$  also acts as an isomorphic operator from  $H^r(\Sigma_{\zeta,\kappa}; \mathbb{C}^N)$  to  $H^r(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  for  $r \in [0,2]$ . Recall that in this case  $\iota_{\zeta,\kappa}$  can also be viewed as a bounded operator from  $\mathcal{B}^r(\Sigma_{\zeta,\kappa})$ to  $\mathcal{B}^r(\mathbb{R}^{\theta-1})$  which has the same norm as the operator acting from  $H^r(\Sigma_{\zeta,\kappa}; \mathbb{C}^N)$  to  $H^r(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ ; cf. (2.9).

In the upcoming lines we often use  $\Sigma_{\zeta,\kappa}$  in the upper index of various already introduced objects which depended on  $\Sigma$ . In this way we emphasize that the object with the upper index depends on  $\zeta$  and  $\kappa$ .

We introduce for  $\varepsilon \in (0, \varepsilon_{ABC}^{\Sigma_{\zeta,\kappa}})$  the operators

$$D_{\varepsilon}^{\zeta,\kappa}(z) := \iota_{\zeta,\kappa} \overline{B}_{\varepsilon}^{\Sigma_{\zeta,\kappa}}(z) \iota_{\zeta,\kappa}^{-1} : \mathcal{B}^{0}(\mathbb{R}^{\theta-1}) \to \mathcal{B}^{0}(\mathbb{R}^{\theta-1}),$$
  
$$D_{0}^{\zeta,\kappa}(z) := \iota_{\zeta,\kappa} B_{0}^{\Sigma_{\zeta,\kappa}}(z) \iota_{\zeta,\kappa}^{-1} : \mathcal{B}^{0}(\mathbb{R}^{\theta-1}) \to \mathcal{B}^{0}(\mathbb{R}^{\theta-1}).$$
(5.11)

The results from Proposition 4.10 and (5.8) imply that  $D_{\varepsilon}^{\zeta,\kappa}(z)$  is uniformly bounded with respect to  $\varepsilon \in (0, \varepsilon_{ABC}^{\Sigma_{\zeta,\kappa}})$  and for  $r \in (0, 1/2)$  the inequality

$$\|D_0^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z)\|_{1/2 \to 0} = \|\iota_{\zeta,\kappa}(B_0^{\Sigma_{\zeta,\kappa}}(z) - \overline{B}_{\varepsilon}^{\Sigma_{\zeta,\kappa}}(z))\iota_{\zeta,\kappa}^{-1}\|_{1/2 \to 0} \le C\varepsilon^{1/2-r}$$
(5.12)

holds for all  $\varepsilon \in (0, \varepsilon_{ABC}^{\Sigma_{\zeta,\kappa}})$ . In particular,  $D_{\varepsilon}^{\zeta,\kappa}(z)f$  converges as  $\varepsilon \to 0$  to  $D_0^{\zeta,\kappa}(z)f$  in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  for  $f \in \mathcal{B}^{1/2}(\mathbb{R}^{\theta-1})$ . Furthermore,  $\mathcal{B}^{1/2}(\mathbb{R}^{\theta-1})$  is by Proposition 2.21 (iii) a dense subset of  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Combining these considerations with the uniform boundedness of  $D_{\varepsilon}^{\zeta,\kappa}(z)$  in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  shows that for all  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$ 

$$D_{\varepsilon}^{\zeta,\kappa}(z)f \xrightarrow{\varepsilon \to 0} D_0^{\zeta,\kappa}(z)f \quad \text{in } \mathcal{B}^0(\mathbb{R}^{\theta-1}).$$
 (5.13)

Using (5.7) and (5.9), and setting

$$\varkappa_{\zeta,\kappa} = \kappa(\cdot,\zeta(\cdot)) \quad \text{and} \quad \nu_{\zeta,\kappa} = \nu^{\Sigma_{\zeta,\kappa}} \circ \varkappa_{\zeta,\kappa} = \frac{\kappa(-\nabla\zeta,1)}{\sqrt{1+|\nabla\zeta|^2}}$$
(5.14)

yields for  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$  and a.e.  $(t, x') \in (-1, 1) \times \mathbb{R}^{\theta-1}$ 

$$D_{\varepsilon}^{\zeta,\kappa}(z)f(t)(x') = \int_{-1}^{1} \int_{\mathbb{R}^{\theta-1}} G_z \Big(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \varepsilon(t-s)\nu_{\zeta,\kappa}(x')\Big) \\ \cdot \sqrt{1 + |\nabla\zeta(y')|^2} f(s)(y') \, dy' \, ds.$$
(5.15)

Another useful representation is given by

$$D_{\varepsilon}^{\zeta,\kappa}(z)f(t) = \int_{-1}^{1} d_{\varepsilon(t-s)}^{\zeta,\kappa}(z)f(s)\,ds, \quad f \in \mathcal{B}^{0}(\mathbb{R}^{\theta-1}), t \in (-1,1),$$
(5.16)

with the operator

$$d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z): L^{2}(\Sigma; \mathbb{C}^{N}) \to L^{2}(\Sigma; \mathbb{C}^{N}),$$
  
$$d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z)g(x') = \int_{\mathbb{R}^{\theta-1}} G_{z} \big(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x')\big) \\ \cdot \sqrt{1 + |\nabla\zeta(y')|^{2}}g(y') \, dy',$$
(5.17)

for  $\widetilde{\varepsilon} \in (-2\varepsilon_{ABC}^{\Sigma_{\zeta,\kappa}}, 2\varepsilon_{ABC}^{\Sigma_{\zeta,\kappa}}) \setminus \{0\}$ . For the interaction strengths  $\eta^{\Sigma_{\zeta,\kappa}}, \tau^{\Sigma_{\zeta,\kappa}} \in C_b^1(\Sigma_{\zeta,\kappa}; \mathbb{R})$  we also define the matrix-valued function

$$Q_{\eta,\tau}^{\zeta,\kappa} := V^{\Sigma_{\zeta,\kappa}} \circ \varkappa_{\zeta,\kappa} = \eta^{\Sigma_{\zeta,\kappa}} \circ \varkappa_{\zeta,\kappa} I_N + \tau^{\Sigma_{\zeta,\kappa}} \circ \varkappa_{\zeta,\kappa} \beta.$$
(5.18)

There holds  $Q_{\eta,\tau}^{\zeta,\kappa} = \iota_{\zeta,\kappa} V^{\Sigma_{\zeta,\kappa}} \iota_{\zeta,\kappa}^{-1}$  in the sense of operators in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ .

#### 5.1.1 Hyperplanes and constant interaction strengths

In this section we assume that  $\Sigma = \Sigma_{y_0,\kappa}$ , for a  $y_0 \in \mathbb{R}$  and a  $\kappa \in \mathrm{SO}(\theta)$ , i.e.  $\Sigma$  is an affine  $(\theta - 1)$ -dimensional hyperplane in  $\mathbb{R}^{\theta}$ . We also assume that the interaction strengths are constant and given by  $\eta, \tau \in \mathbb{R}$ , i.e.  $\eta^{\Sigma_{y_0,\kappa}} \equiv \eta \in \mathbb{R}$  and  $\tau^{\Sigma_{y_0,\kappa}} \equiv \tau \in \mathbb{R}$ . This implies that  $Q_{\eta,\tau}^{y_0,\kappa}$  is equal to the constant matrix

$$Q_{\eta,\tau} := \eta I_N + \tau \beta \tag{5.19}$$

in this case. The main goal of this section is to show that for every compact set  $S\subset \mathbb{R}^2$  satisfying

$$\max_{(\eta,\tau)\in S} \eta^2 - \tau^2 < \frac{\pi^2}{4},$$

there exists an  $\varepsilon_{\text{pl},2} = \varepsilon_{\text{pl},2}(S)$  such that  $\|(I + D_{\varepsilon}^{y_0,\kappa}(z)Q_{\eta,\tau})^{-1}\|_{0\to 0}$  is uniformly bounded with respect to  $(\varepsilon, y_0, (\eta, \tau), \kappa) \in (0, \varepsilon_{\text{pl},2}) \times \mathbb{R} \times S \times \text{SO}(\theta)$ ; cf. Corollary 5.9. This result plays a major role when we prove the uniform boundedness of the operators  $(I + B_{\varepsilon}(z)Vq)^{-1}$  in the case that  $\Sigma$  is a rotated  $C_b^2$ -graph in Section 5.1.2.

We proceed in this section as follows: First, we use the Fourier transform to turn  $D_{\varepsilon}^{y_0,\kappa}(z)$  into a decomposable operator with frequency dependent fiber operators; see Lemma 5.1–Definition 5.3. Then, up to Lemma 5.7, we find and analyse suitable approximations for the fiber operators for high and low frequencies. Finally, we use these results to prove the main statements (Proposition 5.8 and Corollary 5.9) of this section.

Before we start, let us fix some notations. In the current setting the normal vector  $\nu_{\zeta,\kappa}$  is constant and given by  $\kappa e_{\theta}$ , where  $e_{\theta}$  is the  $\theta$ -th Euclidean unit vector. Moreover, the map  $\iota^{\Sigma_{y_0,\kappa}}$ , see Definition 2.7, is a bijective isomorphism and the Weingarten map  $W^{\Sigma_{y_0,\kappa}}$  is zero in this case. Hence, by revisiting Proposition 2.12 we can set  $\varepsilon_{tub}^{\Sigma_{y_0,\kappa}}$  to  $\infty$ . Thus, (4.19) and the constancy of the unit-vector (and its constant extension to  $\mathbb{R}^{\theta}$ ) lets us also set  $\varepsilon_{ABC}^{\Sigma_{y_0,\kappa}}$  to infinity. According to (5.14) and (5.15)  $D_{\varepsilon}^{y_0,\kappa}(z)$  has for  $\varepsilon \in (0,\infty)$  and  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$  the representation

$$D_{\varepsilon}^{y_0,\kappa}(z)f(t)(x') = \int_{-1}^{1} \int_{\mathbb{R}^{\theta-1}} G_z\big(\kappa(x'-y',\varepsilon(t-s))\big)f(s)(y')\,dy'\,ds$$
(5.20)

for a.e.  $(t, x') \in (-1, 1) \times \mathbb{R}^{\theta-1}$ , which shows that  $D_{\varepsilon}^{y_0,\kappa}(z)$  is independent of  $y_0 \in \mathbb{R}$ . Furthermore, (5.13) implies that also  $D_0^{y_0,\kappa}(z)$  is independent of  $y_0$ . Thus, w.l.o.g. we can set  $y_0 = 0$ . We define the matrices

$$\widetilde{\alpha}_j := \alpha \cdot \kappa e_j, \ j \in \{1, \dots, \theta\}, \quad \text{and} \quad \widetilde{\alpha} \cdot \xi := \sum_{j=1}^{\sigma} \widetilde{\alpha}_j \xi_j, \ \xi \in \mathbb{R}^{\theta}, \tag{5.21}$$

as well as

$$\widetilde{\alpha}' \cdot \xi' = \sum_{j=1}^{\theta-1} \widetilde{\alpha}_j \xi'_j, \qquad \xi' \in \mathbb{R}^{\theta-1},$$
(5.22)

for convenience. Similarly to the  $\alpha$ -matrices from Definition 3.1, the  $\tilde{\alpha}$ -matrices are self-adjoint, unitary and fulfil the relations

$$\widetilde{\alpha_j}\widetilde{\alpha_l} + \widetilde{\alpha_l}\widetilde{\alpha_j} = 2\delta_{jl} \quad \text{and} \quad \widetilde{\alpha_j}\beta + \beta\widetilde{\alpha_j} = 0 \quad \forall j, l \in \{1, \dots, \theta\}.$$
(5.23)

Using these rules one easily concludes

$$(\widetilde{\alpha} \cdot \xi)^2 = |\xi|^2 I_N$$
 and  $(\widetilde{\alpha}' \cdot \xi')^2 = |\xi'|^2 I_N$   $\forall \xi = (\xi', \xi_\theta) \in \mathbb{R}^{\theta}.$  (5.24)

We start by calculating the Fourier transform of the function  $G_z(\kappa(\cdot, \tilde{\varepsilon}))$  for fixed  $\tilde{\varepsilon} \neq 0$ ; cf. [60, eqs. (44)–(45)] for similar considerations.

**Lemma 5.1.** Let  $z \in \rho(H)$ ,  $G_z$  be the integral kernel of  $(H - z)^{-1}$  given by (3.3)–(3.4),  $\tilde{\varepsilon} \neq 0$  and  $\mathcal{F}$  be the Fourier transform in  $\mathbb{R}^{\theta-1}$  from Section 2.1 (xvii). Then,

$$\mathcal{F}G_z(\kappa(\cdot,\widetilde{\varepsilon})) = \left(\frac{\widetilde{\alpha}'\cdot(\cdot) + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\cdot|^2}} + \widetilde{\alpha}_{\theta}\operatorname{sign}(\widetilde{\varepsilon})\right) \frac{ie^{|\widetilde{\varepsilon}|i\sqrt{z^2 - m^2 - |\cdot|^2}}}{2\sqrt{(2\pi)^{\theta - 1}}}.$$

*Proof.* Let  $\mathcal{F}$ ,  $\mathcal{F}_1$ ,  $\mathcal{F}_2$  and  $\mathcal{F}_{1,2}$  be as defined in Section 2.1 (xvii). We start by considering  $\mathcal{F}_1G_z(\kappa(\cdot))$ . Since  $G_z(\kappa(\cdot)) \in L^1(\mathbb{R}^\theta; \mathbb{C}^{N \times N}) \subset \mathcal{S}'(\mathbb{R}^\theta; \mathbb{C}^{N \times N})$ , see Proposition 3.4, the expression  $\mathcal{F}_1G_z(\kappa(\cdot))$  is well-defined in  $\mathcal{S}'(\mathbb{R}^\theta; \mathbb{C}^{N \times N})$ . Moreover,  $\mathcal{F}_1G_z(\kappa(\cdot)) = \mathcal{F}_2^{-1}\mathcal{F}_{1,2}G_z(\kappa(\cdot))$ . Thus, we calculate  $\mathcal{F}_{1,2}G_z(\kappa(\cdot))$  next. The function  $G_z$  satisfies the equation

$$(-i(\alpha \cdot \nabla) + m\beta - zI_N)G_z = \delta I_N,$$

with  $\delta$  denoting the  $\delta$ -distribution supported in {0}. Hence, the standard rules for the Fourier transform, see [63, Chapter IX], show

$$(\alpha \cdot (\cdot) + m\beta - zI_N)\mathcal{F}_{1,2}G_z = \frac{1}{\sqrt{(2\pi)^{\theta}}}I_N \text{ in } \mathcal{S}'(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N}).$$

Furthermore,  $G_z \in L^1(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})$  implies  $\mathcal{F}_{1,2}G_z \in C_0(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})$ ; see [63, Theorem IX.7]. Thus, using the properties of  $\alpha_j, j \in \{1, \ldots, \theta\}$ , and  $\beta$  yields

$$\mathcal{F}_{1,2}G_z(\xi) = \frac{1}{\sqrt{(2\pi)^{\theta}}} (\alpha \cdot \xi + m\beta - zI_N)^{-1} = \frac{\alpha \cdot \xi + m\beta + zI_N}{(|\xi|^2 + m^2 - z^2)\sqrt{(2\pi)^{\theta}}} \quad \forall \xi \in \mathbb{R}^{\theta}.$$

Consequently,

$$(\mathcal{F}_{1,2}G_z)(\kappa\xi) = \frac{\alpha \cdot (\kappa\xi) + m\beta + zI_N}{(|\kappa\xi|^2 + m^2 - z^2)\sqrt{(2\pi)^{\theta}}} \quad \forall \xi \in \mathbb{R}^{\theta}.$$

Additionally,  $\kappa \in SO(\theta)$  gives us

$$\mathcal{F}_{1,2}G_z(\kappa(\cdot))(\xi) = (\mathcal{F}_{1,2}G_z)(\kappa\xi) = \frac{\widetilde{\alpha}\cdot\xi + m\beta + zI_N}{(|\xi|^2 + m^2 - z^2)\sqrt{(2\pi)^{\theta}}} \quad \forall \xi \in \mathbb{R}^{\theta}.$$
 (5.25)

Next, we determine  $\mathcal{F}_2^{-1}\mathcal{F}_{1,2}G_z$ . We claim that for a.e.  $(\xi', x_\theta) \in \mathbb{R}^{\theta}$  the equation

$$\mathcal{F}_{2}^{-1}\mathcal{F}_{1,2}G_{z}(\kappa(\cdot))(\xi', x_{\theta}) = \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_{N}}{\sqrt{z^{2} - m^{2} - |\xi'|^{2}}} + \widetilde{\alpha}_{\theta}\operatorname{sign}(x_{\theta})\right)\frac{ie^{|x_{\theta}|i\sqrt{z^{2} - m^{2} - |\xi'|^{2}}}{2\sqrt{(2\pi)^{\theta - 1}}}$$
(5.26)

holds. We verify (5.26) by applying  $\mathcal{F}_2$  and comparing the result with (5.25). As the right-hand side of (5.26) decays exponentially for  $|x_{\theta}| \to \infty$ , we can use the

$$\frac{1}{\sqrt{2\pi}} \int_{\mathbb{R}} \left( \frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(x_{\theta}) \right)$$
$$\cdot \frac{ie^{i(-x_{\theta}\xi_{\theta} + |x_{\theta}|\sqrt{z^2 - m^2 - |\xi'|^2})}}{2\sqrt{(2\pi)^{\theta - 1}}} dx_{\theta} = \frac{\widetilde{\alpha} \cdot \xi + m\beta + zI_N}{(|\xi|^2 + m^2 - z^2)\sqrt{(2\pi)^{\theta}}},$$

which verifies (5.26). Therefore,  $\mathcal{F}_1G_z(\kappa(\cdot)) = \mathcal{F}_2^{-1}\mathcal{F}_{1,2}G_z(\kappa(\cdot))$  can be represented by the function

$$\mathbb{R}^{\theta} \ni (\xi', x_{\theta}) \mapsto \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(x_{\theta})\right) \frac{ie^{|x_{\theta}|i\sqrt{z^2 - m^2 - |\xi'|^2}}}{2\sqrt{(2\pi)^{\theta - 1}}}$$

Moreover, since  $G_z(\kappa(\cdot, \tilde{\varepsilon})) \in L^1(\mathbb{R}^{\theta-1}; \mathbb{C}^{N \times N})$  for  $\tilde{\varepsilon} \neq 0$ , which follows from Proposition 3.4, this shows that for  $\tilde{\varepsilon} \neq 0$  and  $\xi' \in \mathbb{R}^{\theta-1}$  the equation

$$\mathcal{F}G_{z}(\kappa(\cdot,\widetilde{\varepsilon}))(\xi') = \frac{1}{\sqrt{(2\pi)^{\theta-1}}} \int_{\mathbb{R}^{\theta-1}} G_{z}(\kappa(x',\widetilde{\varepsilon})) e^{-i\langle x',\xi'\rangle} \, dx'$$
$$= \mathcal{F}_{1}G_{z}(\kappa(\cdot))(\xi',\widetilde{\varepsilon})$$
$$= \left(\frac{\widetilde{\alpha}'\cdot\xi'+m\beta+zI_{N}}{\sqrt{z^{2}-m^{2}-|\xi'|^{2}}}+\widetilde{\alpha}_{\theta}\mathrm{sign}(\widetilde{\varepsilon})\right) \frac{ie^{|\widetilde{\varepsilon}|i\sqrt{z^{2}-m^{2}-|\xi'|^{2}}}}{2\sqrt{(2\pi)^{\theta-1}}}$$

holds.

**Proposition 5.2.** Let  $z \in \rho(H)$ ,  $\varepsilon > 0$  and  $\mathcal{F}$  be the Fourier transform in  $\mathbb{R}^{\theta-1}$  from Section 2.1 (xvii). Then, for  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$ 

$$\mathcal{F}D^{0,\kappa}_{\varepsilon}(z)\mathcal{F}^{-1}f(t)(\xi') = \int_{-1}^{1} \left( \frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(t-s) \right) \\ \cdot \frac{ie^{|\varepsilon(t-s)|i\sqrt{z^2 - m^2 - |\xi'|^2}}}{2} f(s)(\xi') \, ds$$
(5.27)

and

$$\mathcal{F}D_0^{0,\kappa}(z)\mathcal{F}^{-1}f(t)(\xi') = \int_{-1}^1 \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_\theta \operatorname{sign}(t-s)\right) \\ \cdot \frac{if(s)(\xi')}{2} \, ds$$
(5.28)

for a.e  $(t,\xi') \in (-1,1) \times \mathbb{R}^{\theta-1}$ .

*Proof.* We start with the case  $\varepsilon > 0$ . Using (5.20) shows

$$D^{0,\kappa}_{\varepsilon}(z)f(t) = \int_{-1}^{1} G_z(\kappa(\cdot,\varepsilon(t-s))) * f(s) \, ds$$

for a.e.  $t \in (-1, 1)$  and all  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Thus, Lemma 5.1 and [63, Theorem IX.4] prove the statement for  $\varepsilon > 0$ . It remains to consider the operator  $D_0^{0,\kappa}(z)$ . We start by defining

$$\widetilde{D}_0^{0,\kappa}(z) : \mathcal{B}^0(\mathbb{R}^{\theta-1}) \to \mathcal{B}^0(\mathbb{R}^{\theta-1}),$$
  
$$\widetilde{D}_0^{0,\kappa}(z)f(t)(\xi') := \int_{-1}^1 \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_\theta \operatorname{sign}(t-s)\right) \frac{if(s)(\xi')}{2} \, ds.$$

Next, let  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$ . From (5.27) and the dominated convergence theorem, see Proposition 2.16, one obtains that  $\mathcal{F}D^{0,\kappa}_{\varepsilon}(z)\mathcal{F}^{-1}f$  converges for  $\varepsilon \to 0$  to  $\widetilde{D}^{0,\kappa}_0(z)f$ in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Thus, the boundedness of  $\mathcal{F}$  and  $\mathcal{F}^{-1}$  in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  (and therefore also in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ , cf. (2.9)), implies that  $D^{0,\kappa}_{\varepsilon}(z)f$  converges to  $\mathcal{F}^{-1}\widetilde{D}^{0,\kappa}_0(z)\mathcal{F}f$  in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Moreover, by (5.13)  $D^{0,\kappa}_{\varepsilon}(z)f$  converges to  $D^{0,\kappa}_0(z)f$  in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Hence,  $D^{0,\kappa}_0(z)f = \mathcal{F}^{-1}\widetilde{D}^{0,\kappa}_0(z)\mathcal{F}f$  which proves (5.28).

The structure of  $\mathcal{F}D^{0,\kappa}_{\varepsilon}(z)\mathcal{F}^{-1}$  and  $\mathcal{F}D^{0,\kappa}_{0}(z)\mathcal{F}^{-1}$  inspires us to change our viewpoint. Namely, instead of viewing these operators in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$  we consider them as operators in the isometrically isomorphic space  $L^{2}(\mathbb{R}^{\theta-1}; L^{2}((-1,1); \mathbb{C}^{N}))$ . In the context of direct integrals the notation  $\int_{\mathbb{R}^{\theta-1}}^{\oplus} L^{2}((-1,1); \mathbb{C}^{N}) d\xi'$  for this space is also common. Considered as operators acting in this space  $\mathcal{F}D^{0,\kappa}_{\varepsilon}(z)\mathcal{F}^{-1}$  and  $\mathcal{F}D^{0,\kappa}_{0}(z)\mathcal{F}^{-1}$  are decomposable operators in  $L^{2}(\mathbb{R}^{\theta-1}; L^{2}((-1,1); \mathbb{C}^{N}))$  with the following fibers.

**Definition 5.3.** Let  $\varepsilon > 0$ ,  $\xi' \in \mathbb{R}^{\theta-1}$  and  $z \in \rho(H)$ . We define

$$\mathfrak{D}_{\varepsilon,\xi'}(z) : L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N),$$
$$\mathfrak{D}_{\varepsilon,\xi'}(z)f(t) := \int_{-1}^1 \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(t-s)\right) \cdot \frac{ie^{|\varepsilon(t-s)|i\sqrt{z^2 - m^2 - |\xi'|^2}}}{2}f(s) \, ds$$

and

$$\mathfrak{D}_{0,\xi'}(z) : L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N),\\ \mathfrak{D}_{0,\xi'}(z)f(t) := \int_{-1}^1 \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_\theta \operatorname{sign}(t-s)\right) \frac{if(s)}{2} \, ds.$$

**Remark 5.4.** These operators still depend on the rotation matrix  $\kappa \in SO(\theta)$  since the  $\tilde{\alpha}$ 's depend on  $\kappa$ . As we use these operators only as auxiliary operators in this section, we omit  $\kappa$ . Next, we explain the ideas mentioned above Definition 5.3 in a more rigorous way. Using

$$\begin{aligned} \mathcal{B}^{0}(\mathbb{R}^{\theta-1}) &= L^{2}((-1,1); L^{2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N})) \\ &\simeq L^{2}((-1,1) \times \mathbb{R}^{\theta-1}; \mathbb{C}^{N}) \simeq L^{2}(\mathbb{R}^{\theta-1}; L^{2}((-1,1); \mathbb{C}^{N})), \end{aligned}$$

see Proposition 2.18 (iii), allows us to define the isometric isomorphism

$$\begin{aligned} \mathbf{i} : \mathcal{B}^0(\mathbb{R}^{\theta-1}) &\to L^2(\mathbb{R}^{\theta-1}; L^2((-1,1); \mathbb{C}^N)), \\ \mathbf{i} f(\xi')(t) := \mathcal{F} f(t)(\xi') \quad \text{for a.e. } (\xi',t) \in \mathbb{R}^{\theta-1} \times (-1,1). \end{aligned}$$

Thus, by Proposition 5.2 and Definition 5.3 we obtain for  $\varepsilon \geq 0$ 

$$\mathfrak{i} D^{0,\kappa}_{\varepsilon}(z) \mathfrak{i}^{-1} : L^2(\mathbb{R}^{\theta-1}; L^2((-1,1); \mathbb{C}^N)) \to L^2(\mathbb{R}^{\theta-1}; L^2((-1,1); \mathbb{C}^N)),$$
  
$$\mathfrak{i} D^{0,\kappa}_{\varepsilon}(z) \mathfrak{i}^{-1} f(\xi') = \mathfrak{D}_{\varepsilon,\xi'}(z) f(\xi'),$$
(5.29)

i.e.  $\mathfrak{i}D^{0,\kappa}_{\varepsilon}(z)\mathfrak{i}^{-1}$  is a decomposable operator in the sense of (2.12) which is induced by the operator-valued function  $\zeta' \mapsto \mathfrak{D}_{\varepsilon,\xi'}(z)$ . Hence, Proposition 2.19 lets us transfer results regarding  $\mathfrak{D}_{\varepsilon,\xi'}(z)$  to  $D^{0,\kappa}_{\varepsilon}(z)$  and vice versa.

Next, we study the operator  $\mathfrak{D}_{\varepsilon,\xi'}(z)$  in detail. For this purpose, we introduce the auxiliary operator

$$\mathfrak{H}_{\rho,w'}: L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N),$$

$$(\mathfrak{H}_{\rho,w'}f)(t) := \int_{-1}^1 \left(\widetilde{\alpha}' \cdot w' + i\widetilde{\alpha}_{\theta} \mathrm{sign}(t-s)\right) \frac{e^{-\rho|t-s|}}{2} f(s) \, ds,$$
(5.30)

for  $\rho \in [0,\infty)$  and  $w' \in \mathbb{R}^{\theta-1}$  with |w'| = 1. It is easy to check that  $\mathfrak{H}_{\rho,w'}$  is a self-adjoint Hilbert-Schmidt operator.

**Lemma 5.5.** Let  $\varepsilon > 0$ ,  $\xi' \in \mathbb{R}^{\theta-1} \setminus \{0\}$  and  $z \in \rho(H)$ . Then, there exists a constant  $C_{\mathrm{pl},1} > 0$  which only depends on m and z such that

$$\begin{aligned} \|\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{D}_{0,\xi'}(z)\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} &\leq C_{\mathrm{pl},1}\varepsilon(1+|\xi'|), \\ \|\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} &\leq \frac{C_{\mathrm{pl},1}}{1+|\xi'|}. \end{aligned}$$

*Proof.* In this proof C > 0 denotes a constant which may change in-between lines, but only depends on m and z. We start by estimating the kernel of  $\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{D}_{0,\xi'}(z)$ . We can bound this kernel by

$$C\left|1 - e^{\varepsilon|t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}}\right| \le C\varepsilon|t-s|\sqrt{|z^2 - m^2 - |\xi'|^2|}$$
$$\le C\varepsilon(1+|\xi'|).$$

Hence, there exists a constant  $C_{pl,1} > 0$  such that

$$\|\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{D}_{0,\xi'}(z)\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \le C_{\mathrm{pl},1}\varepsilon(1+|\xi'|).$$

Next, we estimate the kernel of  $\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}$  by

$$\frac{1}{2} \left| \left( e^{-\varepsilon |t-s||\xi'|} - e^{\varepsilon |t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}} \right) \left( \widetilde{\alpha}' \cdot \frac{\xi'}{|\xi'|} + i \operatorname{sign}(t-s) \widetilde{\alpha}_{\theta} \right) \right| \\
+ \frac{1}{2} \left| e^{\varepsilon |t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}} \left( \widetilde{\alpha}' \cdot \frac{\xi'}{|\xi'|} \right) \left( 1 - \frac{i|\xi'|}{\sqrt{z^2 - m^2 - |\xi'|^2}} \right) \right| \quad (5.31) \\
+ \frac{1}{2} \left| e^{\varepsilon |t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}} \frac{m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} \right|.$$

The first term in (5.31) can be estimated for  $\varepsilon |t-s| \leq 1$  by

$$C\left|e^{-\varepsilon|t-s||\xi'|} - e^{\varepsilon|t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}}\right| \le C\varepsilon|t-s|\left||\xi'| + i\sqrt{z^2 - m^2 - |\xi'|^2}\right|.$$

$$\le C\frac{|z^2 - m^2|}{\left||\xi'| - i\sqrt{z^2 - m^2 - |\xi'|^2}\right|}$$

$$\le \frac{C}{1 + |\xi'|},$$
(5.32)

where we used  $z \in \rho(H) = \mathbb{C} \setminus ((-\infty, -|m|] \cup [|m|, \infty))$  as well as  $\operatorname{Im} \sqrt{w} > 0$  for  $w \in \mathbb{C} \setminus [0, \infty)$ . For  $\varepsilon |t - s| > 1$ , we get

$$C\left|e^{-\varepsilon|t-s||\xi'|} - e^{\varepsilon|t-s|i\sqrt{z^2 - m^2 - |\xi'|^2}}\right| \le C\left(e^{-|\xi'|} + e^{-\operatorname{Im}\sqrt{z^2 - m^2 - |\xi'|^2}}\right) \le \frac{C}{1 + |\xi'|}.$$
 (5.33)

Similarly as we estimated the first term in the case  $\varepsilon |t - s| \le 1$ , the second term in (5.31) can be bounded by

$$\begin{split} C \bigg| 1 - \frac{i|\xi'|}{\sqrt{z^2 - m^2 - |\xi'|^2}} \bigg| &= C \frac{|z^2 - m^2|}{\left|\sqrt{z^2 - m^2 - |\xi'|^2} + i|\xi'|\right| \left|\sqrt{z^2 - m^2 - |\xi'|^2}\right|} \\ &\leq \frac{C}{(1 + |\xi'|)^2} \leq \frac{C}{1 + |\xi'|}. \end{split}$$

One also sees that the third term in (5.31) is smaller than  $\frac{C}{1+|\xi'|}$  for a sufficiently large C. Summing up, we have that the kernel of  $\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}$  can be bounded by  $\frac{C}{1+|\xi'|}$  and therefore if  $C_{\mathrm{pl},1}$  is chosen sufficiently large, then

$$\left\|\mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}\right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \le \frac{C_{\mathrm{pl},1}}{1+|\xi'|}.$$

**Lemma 5.6.** Let  $\rho \geq 0$ ,  $w' \in \mathbb{R}^{\theta-1}$  with |w'| = 1 and q be as in (5.1). Then,  $\sigma(\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q}) \subset [-\frac{2}{\pi},\frac{2}{\pi}].$ 

*Proof.* To shorten notation, we set  $\widetilde{\alpha}_{\pm} := \widetilde{\alpha}' \cdot w' \pm i \widetilde{\alpha}_{\theta}$ . Then,

$$\mathfrak{H}_{\rho,w'}f(t) = \frac{1}{2}\int_{t}^{1} e^{-\rho|t-s|}\widetilde{\alpha}_{-}f(s)\,ds + \frac{1}{2}\int_{-1}^{t} e^{-\rho|t-s|}\widetilde{\alpha}_{+}f(s)\,ds$$

for  $f \in L^2((-1,1); \mathbb{C}^N)$ . Using the rules for the  $\tilde{\alpha}$  matrices from (5.23) yields the orthogonality relation ran  $\tilde{\alpha}_+ \perp \operatorname{ran} \tilde{\alpha}_-$ . Hence, as  $q \ge 0$  a.e. on (-1,1) by (5.1), we have for  $f \in L^2((-1,1); \mathbb{C}^N)$ 

$$\begin{aligned} \left\|\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q}f\right\|_{L^{2}((-1,1);\mathbb{C}^{N})}^{2} \\ &= \frac{1}{4}\int_{-1}^{1}q(t)\left|\int_{t}^{1}e^{-\rho|t-s|}\widetilde{\alpha}_{-}\sqrt{q(s)}f(s)\,ds\right|^{2}dt \\ &+ \frac{1}{4}\int_{-1}^{1}q(t)\left|\int_{-1}^{t}e^{-\rho|t-s|}\widetilde{\alpha}_{+}\sqrt{q(s)}f(s)\,ds\right|^{2}dt. \end{aligned}$$
(5.34)

We start by estimating the first term on the right-hand side. We define the function  $\mathscr{Q}(t) = -1 + \int_{-1}^{t} q(s) \, ds, t \in [-1, 1]$ . Then,  $\mathscr{Q}' = q, \, \mathscr{Q}(-1) = -1$  and  $\mathscr{Q}(1) = 0$  since  $\int_{-1}^{1} q(s) \, ds = 1$ . Applying the Cauchy-Schwarz inequality and Fubini's theorem gives us

$$\begin{split} \frac{1}{4} \int_{-1}^{1} q(t) \left| \int_{t}^{1} e^{-\rho|t-s|} \widetilde{\alpha}_{-} \sqrt{q(s)} f(s) \, ds \right|^{2} dt \\ &= \frac{1}{4} \int_{-1}^{1} q(t) \left| \int_{t}^{1} \frac{\sqrt{\cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right)}}{\sqrt{\cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right)}} e^{-\rho|t-s|} \widetilde{\alpha}_{-} \sqrt{q(s)} f(s) \, ds \right|^{2} dt \\ &\leq \frac{1}{4} \int_{-1}^{1} q(t) \left( \int_{t}^{1} \cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right) q(s) \, ds \right) \left( \int_{t}^{1} \frac{1}{\cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right)} |\widetilde{\alpha}_{-} f(s)|^{2} \, ds \right) dt \\ &= \frac{1}{2\pi} \int_{-1}^{1} -\sin\left(\frac{\pi}{2}\mathcal{Q}(t)\right) q(t) \left( \int_{t}^{1} \frac{1}{\cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right)} |\widetilde{\alpha}_{-} f(s)|^{2} \, ds \right) dt \\ &= \frac{1}{2\pi} \int_{-1}^{1} \left( \int_{-1}^{s} -\sin\left(\frac{\pi}{2}\mathcal{Q}(t)\right) q(t) \, dt \right) \frac{1}{\cos\left(\frac{\pi}{2}\mathcal{Q}(s)\right)} |\widetilde{\alpha}_{-} f(s)|^{2} \, ds \\ &= \frac{1}{\pi^{2}} \int_{-1}^{1} |\widetilde{\alpha}_{-} f(s)|^{2} \, ds. \end{split}$$

The same trick with  $\mathscr{Q} + 1$  instead of  $\mathscr{Q}$  yields that the second term of the righthand side of equation (5.34) can be estimated by  $\frac{1}{\pi^2} \int_{-1}^{1} |\widetilde{\alpha}_+ f(s)|^2 ds$ . These estimates,  $\operatorname{ran} \widetilde{\alpha}_+ \perp \operatorname{ran} \widetilde{\alpha}_-$  and (5.24) imply

$$\begin{split} \left\| \sqrt{q} \mathfrak{H}_{\rho,w'} \sqrt{q} f \right\|_{L^2((-1,1);\mathbb{C}^N)}^2 &\leq \frac{1}{\pi^2} \int_{-1}^1 |\widetilde{\alpha}_- f(s)|^2 + |\widetilde{\alpha}_+ f(s)|^2 \, ds \\ &= \frac{1}{\pi^2} \int_{-1}^1 |(\widetilde{\alpha}_- + \widetilde{\alpha}_+) f(s)|^2 \, ds \\ &= \frac{1}{\pi^2} \int_{-1}^1 |2(\widetilde{\alpha}' \cdot w') f(s)|^2 \, ds \\ &= \frac{4}{\pi^2} \|f\|_{L^2((-1,1);\mathbb{C}^N)}^2. \end{split}$$

Since  $\mathfrak{H}_{\rho,w'}$  is self-adjoint in  $L^2((-1,1);\mathbb{C}^N)$ , we obtain  $\sigma(\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q}) \subset [-\frac{2}{\pi},\frac{2}{\pi}]$ .  $\Box$ 

Having studied the spectrum of  $\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q}$ , we employ this knowledge to study the bounded invertibility of  $I + \mathfrak{H}_{\rho,w'}Q_{\eta,\tau}q$ . Recall that  $Q_{\eta,\tau} = \eta I_N + \tau\beta$  for  $\eta, \tau \in \mathbb{R}$ .

**Lemma 5.7.** Let  $\rho \geq 0$ ,  $w' \in \mathbb{R}^{\theta-1}$  with |w'| = 1,  $\eta, \tau \in \mathbb{R}$ ,  $Q_{\eta,\tau} = \eta I_N + \tau \beta$ ,  $d = \eta^2 - \tau^2$ ,

$$c(d) := \begin{cases} \sqrt{d\frac{2}{\pi}}, & d \ge 0, \\ 0, & d < 0, \end{cases}$$

and q be as in (5.1). If  $d < \frac{\pi^2}{4}$ , then c(d) < 1,  $I + \mathfrak{H}_{\rho,w'}Q_{\eta,\tau}q$  is continuously invertible in  $L^2((-1,1);\mathbb{C}^N)$  and the norm of the inverse is bounded by the constant

$$C_{\rm pl,2} = C_{\rm pl,2}(\eta,\tau) := 4 \|q\|_{L^{\infty}((-1,1))} (|\eta| + |\tau|) \frac{1 + (|\eta| + |\tau|)\frac{2}{\pi}}{(1 - c(d))\pi} + 1.$$
(5.35)

*Proof.* The identities  $\mathfrak{H}_{\rho,w'}Q_{\eta,\tau} = Q_{\eta,-\tau}\mathfrak{H}_{\rho,w'}$  and  $Q_{\eta,-\tau}Q_{\eta,\tau} = dI_N$ , which follow from (5.23), give us

$$I - d(\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q})^2 = (I + \sqrt{q}\mathfrak{H}_{\rho,w'}Q_{\eta,\tau}\sqrt{q})(I - \sqrt{q}\mathfrak{H}_{\rho,w'}Q_{\eta,\tau}\sqrt{q}).$$

If  $d < \frac{\pi^2}{4}$ , then Lemma 5.6 implies  $1 \in \rho(d(\sqrt{q}\mathfrak{H}_{\rho,w'}\sqrt{q})^2)$  and therefore the operator  $I + \sqrt{q}\mathfrak{H}_{\rho,w'}Q_{\eta,\tau}\sqrt{q}$  is also continuously invertible in  $L^2((-1,1);\mathbb{C}^N)$  and

$$\begin{split} \left\| (I + \sqrt{q} \mathfrak{H}_{\rho,w'} Q_{\eta,\tau} \sqrt{q})^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &= \left\| (I - d(\sqrt{q} \mathfrak{H}_{\rho,w'} \sqrt{q})^2)^{-1} (I - \sqrt{q} \mathfrak{H}_{\rho,w'} Q_{\eta,\tau} \sqrt{q}) \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &\leq \frac{1 + (|\eta| + |\tau|)^2_{\pi}}{1 - c(d)}. \end{split}$$

Moreover, applying Proposition 2.29 shows that  $I + \mathfrak{H}_{\rho,w'}Q_{\eta,\tau}q$  is also continuously invertible in  $L^2((-1,1);\mathbb{C}^N)$  and

$$\begin{split} \left\| (I + \mathfrak{H}_{\rho,w'}Q_{\eta,\tau}q)^{-1} \right\|_{L^{2}((-1,1);\mathbb{C}^{N}) \to L^{2}((-1,1);\mathbb{C}^{N})} \\ &= \left\| \mathfrak{H}_{\rho,w'}Q_{\eta,\tau}\sqrt{q}(I + \sqrt{q}\mathfrak{H}_{\rho,w'}Q_{\eta,\tau}\sqrt{q})^{-1}\sqrt{q} - I \right\|_{L^{2}((-1,1);\mathbb{C}^{N}) \to L^{2}((-1,1);\mathbb{C}^{N})} \\ &\leq \|q\|_{L^{\infty}((-1,1))}(|\eta| + |\tau|) \|\mathfrak{H}_{\rho,w'}\|_{L^{2}((-1,1);\mathbb{C}^{N}) \to L^{2}((-1,1);\mathbb{C}^{N})} \frac{1 + (|\eta| + |\tau|)\frac{2}{\pi}}{1 - c(d)} + 1 \\ &\leq 4 \|q\|_{L^{\infty}((-1,1))}(|\eta| + |\tau|) \frac{1 + (|\eta| + |\tau|)\frac{2}{\pi}}{(1 - c(d))\pi} + 1, \end{split}$$

where we used Lemma 5.6 (for  $q = \frac{1}{2}$ ) to estimate  $\|\mathfrak{H}_{\rho,w'}\|_{L^2((-1,1);\mathbb{C}^N)\to L^2((-1,1);\mathbb{C}^N)}$  by  $\frac{4}{\pi}$ .

In the last part of Section 5.1.1 we use our findings to prove a norm estimate for the operator  $(I + D_{\varepsilon}^{0,\kappa}(z)Q_{\eta,\tau}q)^{-1}$ .

**Proposition 5.8.** Let  $\kappa \in SO(\theta)$ ,  $\eta, \tau \in \mathbb{R}$  fulfil  $d = \eta^2 - \tau^2 < \frac{\pi^2}{4}$ ,  $Q_{\eta,\tau} = \eta I_N + \tau \beta$ , q be as in (5.1) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, let

$$C_{\text{pl},3} = C_{\text{pl},3}(\eta,\tau,\kappa) := 2 \max\{C_{\text{pl},2}, \left\| (I + D_0^{0,\kappa}(z)Q_{\eta,\tau}q)^{-1} \right\|_{0\to 0} \}$$
(5.36)

with  $C_{\rm pl,2} = C_{\rm pl,2}(\eta, \tau)$  from Lemma 5.7 and

$$\varepsilon_{\rm pl,1} = \varepsilon_{\rm pl,1}(\eta,\tau,\kappa) := (C_{\rm pl,3}C_{\rm pl,1}(|\eta| + |\tau|) \|q\|_{L^{\infty}((-1,1))})^{-2}$$
(5.37)

with  $C_{\text{pl},1}$  from Lemma 5.5. Then,

$$\sup_{\varepsilon \in (0,\varepsilon_{\mathrm{pl},1})} \left\| (I + D^{0,\kappa}_{\varepsilon}(z)Q_{\eta,\tau}q)^{-1} \right\|_{0\to 0} \le C_{\mathrm{pl},3} < \infty.$$

Proof. We start by arguing that  $C_{\mathrm{pl},3} < \infty$ . The assumption  $d < \frac{\pi^2}{4}$  guarantees  $C_{\mathrm{pl},2} < \infty$ . Moreover, applying Proposition 3.15 (iii) and Proposition 4.13 (for  $V = Q_{\eta,\tau} = \eta I_2 + \tau \beta = \text{const.}$  and r = 0) shows that  $I + B_0^{\Sigma y_0,\kappa}(z)Q_{\eta,\tau}q$  is continuously invertible in  $\mathcal{B}^0(\Sigma_{y_0,\kappa})$ . Let us shortly explain why Proposition 4.13 is indeed applicable. We have to show  $\cos\left(\frac{(\alpha \cdot \nu)Q_{\eta,\tau}}{2}\right)^{-1} \in W^1_{\infty}(\Sigma_{y_0,\kappa}; \mathbb{C}^{N \times N})$  and that  $I + \mathcal{C}_z^{\Sigma y_0,\kappa}\widetilde{Q_{\eta,\tau}}$  is continuously invertible in  $L^2(\Sigma_{y_0,\kappa}; \mathbb{C}^N)$ , where

$$\widetilde{Q_{\eta,\tau}} = Q_{\eta,\tau} \operatorname{sinc}\left(\frac{(\alpha \cdot \nu)Q_{\eta,\tau}}{2}\right) \cos\left(\frac{(\alpha \cdot \nu)Q_{\eta,\tau}}{2}\right)^{-1}.$$

In the same way as in (5.4)–(5.6) we get  $\cos\left(\frac{(\alpha \cdot \nu)Q_{\eta,\tau}}{2}\right) = \cos\left(\frac{\sqrt{d}}{2}\right) = \text{const.}$  and  $\widetilde{Q_{\eta,\tau}} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$  with  $(\widetilde{\eta},\widetilde{\tau}) = \tan\left(\frac{\sqrt{d}}{2}\right)(\eta,\tau)$ . Hence,  $d < \frac{\pi^2}{4}$  implies  $\cos\left(\frac{\sqrt{d}}{2}\right) \neq 0$  and therefore  $\cos\left(\frac{(\alpha \cdot \nu)Q_{\eta,\tau}}{2}\right)^{-1} \in W^1_{\infty}(\Sigma_{y_0,\kappa}; \mathbb{C}^{N \times N})$ . Furthermore, we have

$$\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = 4 \tan\left(\frac{\sqrt{d}}{2}\right)^2 < 4$$

and thus Proposition 3.15 (iii) implies that  $I + C_z^{\Sigma_{y_0,\kappa}} \widetilde{Q_{\eta,\tau}}$  is continuously invertible in  $L^2(\Sigma_{y_0,\kappa}; \mathbb{C}^N)$ . Hence, the assumptions of Proposition 4.13 are satisfied and its application is justified. By (5.11)  $B_0^{\Sigma_{0,\kappa}}(z)$  is related to  $D_0^{0,\kappa}(z)$  via an isometric isomorphism and therefore  $I + D_0^{0,\kappa}(z)Q_{\eta,\tau}q$  is continuously invertible in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ , which proves  $C_{\mathrm{pl},3} < \infty$ .

According to Lemma 5.5 we have

$$\begin{split} \|\mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q\|_{L^2((-1,1);\mathbb{C}^N)\to L^2((-1,1);\mathbb{C}^N)} \\ &\leq C_{\mathrm{pl},1}\|q\|_{L^{\infty}((-1,1))}\frac{|\eta| + |\tau|}{1 + |\xi'|} = \frac{\varepsilon_{\mathrm{pl},1}^{-1/2}}{C_{\mathrm{pl},3}(1 + |\xi'|)} \end{split}$$

for  $\xi' \in \mathbb{R}^{\theta-1} \setminus \{0\}$  and all  $\varepsilon > 0$ . Hence, if we set  $R := \varepsilon_{\text{pl},1}^{-1/2} - 1$ , the choices of  $C_{\text{pl},3}$ ,  $\varepsilon_{\text{pl},1}$  and R yield

$$\begin{split} \left\| (I + \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ & \cdot \left\| \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ & \leq C_{\mathrm{pl},2} \frac{\varepsilon_{\mathrm{pl},1}^{-1/2}}{C_{\mathrm{pl},3}(1+R)} \leq \frac{C_{\mathrm{pl},3}}{2} \cdot \frac{\varepsilon_{\mathrm{pl},1}^{-1/2}}{C_{\mathrm{pl},3}\varepsilon_{\mathrm{pl},1}^{-1/2}} = \frac{1}{2} \end{split}$$

for  $0 \neq |\xi'| \geq R$  and  $\varepsilon > 0$ . In particular, Proposition 2.28 shows that

$$I + \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q = I + \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q + \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q - \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q$$

is continuously invertible in  $L^2((-1,1);\mathbb{C}^N)$  and that the corresponding norm estimate

$$\begin{aligned} \left\| (I + \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &\leq \frac{\left\| (I + \mathfrak{H}_{|\xi'|\varepsilon,\xi'/|\xi'|}Q_{\eta,\tau}q)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)}}{1 - \frac{1}{2}} \\ &\leq 2\frac{C_{\mathrm{pl},3}}{2} = C_{\mathrm{pl},3} \end{aligned}$$
(5.38)

is valid for  $0 \neq |\xi'| \geq R$  and  $\varepsilon > 0$ .

Having found an estimate for  $0 \neq |\xi'| \geq R$ , we aim to find a similar estimate for  $0 \neq |\xi'| < R$ . Again, according to Lemma 5.5 we have

$$\begin{split} \|\mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q - \mathfrak{D}_{0,\xi'}(z)Q_{\eta,\tau}q\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &\leq C_{\mathrm{pl},1}\|q\|_{L^{\infty}((-1,1))}(|\eta| + |\tau|)\varepsilon(1 + |\xi'|) = \varepsilon \frac{\varepsilon_{\mathrm{pl},1}^{-1/2}(1 + |\xi'|)}{C_{\mathrm{pl},3}} \end{split}$$

for  $\xi' \in \mathbb{R}^{\theta-1} \setminus \{0\}$  and  $\varepsilon > 0$ . Moreover, Proposition 2.19, (5.29) and (5.36) imply

ess 
$$\sup_{\xi' \in \mathbb{R}^{\theta-1}} \left\| (I + \mathfrak{D}_{0,\xi'}(z)Q_{\eta,\tau}q)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)}$$
  
=  $\left\| (I + D_0^{0,\kappa}(z)Q_{\eta,\tau}q)^{-1} \right\|_{0\to 0} \leq \frac{C_{\mathrm{pl},3}}{2}.$ 

Hence, as  $1 + |\xi'| < 1 + R = \varepsilon_{\text{pl},1}^{-1/2}$ , we estimate similarly as in the first part of the proof for  $\varepsilon \in (0, \varepsilon_{\text{pl},1})$ 

$$\begin{aligned} & \operatorname{ess\ sup}_{|\xi'| < R} \left\| \left( I + \mathfrak{D}_{\varepsilon,\xi'}(z) Q_{\eta,\tau} q \right)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &= \operatorname{ess\ sup}_{|\xi'| < R} \left\| \left[ I + \left( I + \mathfrak{D}_{0,\xi'}(z) Q_{\eta,\tau} q \right)^{-1} \left( \mathfrak{D}_{\varepsilon,\xi'}(z) - \mathfrak{D}_{0,\xi'}(z) \right) Q_{\eta,\tau} q \right]^{-1} \right. \\ & \left. \cdot \left( I + \mathfrak{D}_{0,\xi'}(z) Q_{\eta,\tau} q \right)^{-1} \right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ & \left. < \frac{1}{1 - \frac{C_{\mathrm{pl},3}}{2} \cdot \frac{\varepsilon \varepsilon_{\mathrm{pl},1}^{-1/2}(1+R)}{C_{\mathrm{pl},3}}} \cdot \frac{C_{\mathrm{pl},3}}{2} = \frac{1}{1 - \frac{\varepsilon \varepsilon_{\mathrm{pl},1}^{-1}}{2}} \cdot \frac{C_{\mathrm{pl},3}}{2} < C_{\mathrm{pl},3}. \end{aligned}$$
(5.39)

Combining (5.38) and (5.39), and applying Proposition 2.19 gives us

$$\begin{split} \left\| (I + D_{\varepsilon}^{0,\kappa}(z)Q_{\eta,\tau}q)^{-1} \right\|_{0\to 0} \\ &= \max \Big\{ \operatorname{ess\,sup}_{|\xi'| \ge R} \big\| (I + \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q)^{-1} \big\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)}, \\ & \operatorname{ess\,sup}_{|\xi'| < R} \big\| (I + \mathfrak{D}_{\varepsilon,\xi'}(z)Q_{\eta,\tau}q)^{-1} \big\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \Big\} \\ &\le C_{\mathrm{pl},3} \quad \forall \varepsilon \in (0,\varepsilon_{\mathrm{pl},1}). \end{split}$$

**Corollary 5.9.** Let  $z \in \mathbb{C} \setminus \mathbb{R}$ , q be as in (5.1),  $S \subset \mathbb{R}^2$  be a compact set and  $\max_{(\eta,\tau)\in S} \eta^2 - \tau^2 < \frac{\pi^2}{4}$ . Then, there exists an  $\varepsilon_{\text{pl},2} = \varepsilon_{\text{pl},2}(S) > 0$  such that

$$\sup_{(\varepsilon,y_0,(\eta,\tau),\kappa)\in(0,\varepsilon_{\mathrm{pl},2})\times\mathbb{R}\times S\times\mathrm{SO}(\theta)}\left\|\left(I+D_{\varepsilon}^{y_0,\kappa}(z)Q_{\eta,\tau}q\right)^{-1}\right\|_{0\to0}<\infty$$

*Proof.* Since  $D_{\varepsilon}^{y_0,\kappa}(z) = D_{\varepsilon}^{0,\kappa}(z)$ , see the text below (5.20), the assertion follows directly from Proposition 5.8 if we can show

$$\sup_{((\eta,\tau),\kappa)\in S\times\mathrm{SO}(\theta)} C_{\mathrm{pl},3}(\eta,\tau,\kappa) < \infty \quad \text{and} \quad \inf_{((\eta,\tau),\kappa)\in S\times\mathrm{SO}(\theta)} \varepsilon_{\mathrm{pl},1}(\eta,\tau,\kappa) > 0, \quad (5.40)$$

with  $C_{\text{pl},3}$  and  $\varepsilon_{\text{pl},1}$  as in Proposition 5.8. Note also that as S is bounded, the first inequality in (5.40) and (5.37) imply the second inequality in (5.40). Moreover, the assumption  $\max_{(\eta,\tau)\in S} \eta^2 - \tau^2 < \frac{\pi^2}{4}$  implies  $\max_{(\eta,\tau)\in S} C_{\text{pl},2}(\eta,\tau) < \infty$ , where  $C_{\text{pl},2}$  is defined by (5.35). Hence, it follows from (5.36) that (5.40) is valid if

$$\sup_{((\eta,\tau),\kappa)\in S\times\mathrm{SO}(\theta)} \left\| (I+D_0^{0,\kappa}(z)Q_{\eta,\tau}q)^{-1} \right\|_{0\to 0} < \infty.$$
(5.41)

By the representation of  $D_0^{0,\kappa}(z)$  in Proposition 5.2 and (5.21),  $D_0^{0,\kappa}(z)Q_{\eta,\tau}q$  depends with respect to the operator norm in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  continuously on  $\eta$ ,  $\tau$  and  $\kappa$ . Thus, as  $S \times SO(\theta)$  is compact, (5.41) is indeed true.

#### 5.1.2 General rotated $C_b^2$ -graphs

After treating the case of affine hyperplanes, we turn to the case where  $\Sigma$  is a rotated  $C_b^2$ -graph. To do so, we fix in this section

$$\zeta \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R}) \quad \text{and} \quad \kappa \in \mathrm{SO}(\mathbb{R}^{\theta}), \tag{5.42}$$

and we assume

$$\Sigma = \Sigma_{\zeta,\kappa} = \{\kappa(x',\zeta(x')) : x' \in \mathbb{R}^{\theta-1}\}.$$
(5.43)

Before we resume, let us mention that in the current section we use the upper index notation introduced above (5.11), cf. (5.14) and (5.18), only for objects which do not correspond to the fixed rotated graph  $\Sigma = \Sigma_{\zeta,\kappa}$ ; i.e. we write  $\overline{B}_{\varepsilon}(z)$ ,  $\nu$ ,  $\eta$ , etc. instead of  $\overline{B}_{\varepsilon}^{\Sigma_{\zeta,\kappa}}(z)$ ,  $\nu^{\Sigma_{\zeta,\kappa}}$ ,  $\eta^{\Sigma_{\zeta,\kappa}}$ , etc., respectively.

Recall from (5.14) that  $\varkappa_{\zeta,\kappa}(x') = \kappa(x',\zeta(x'))$  and  $\nu_{\zeta,\kappa}(x') = \nu(\varkappa_{\zeta,\kappa}(x'))$  for  $x' \in \mathbb{R}^{\theta-1}$ . According to Proposition 2.9 (i) (in the current case we have p = 1,  $\varkappa_1 = \varkappa_{\zeta,\kappa}$ ,  $\Sigma_1 = \Sigma = \Sigma_{\zeta,\kappa}$ ), there exists a  $C_{\iota,1} > 0$  such that for all  $\tilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC})$  and  $x', y' \in \mathbb{R}^{\theta-1}$ 

$$C_{\iota,1}^{-1}(|x'-y'|+|\widetilde{\varepsilon}|) \le |\varkappa_{\zeta,\kappa}(x')-\varkappa_{\zeta,\kappa}(y')+\widetilde{\varepsilon}\nu_{\zeta,\kappa}(x')| \le C_{\iota,1}(|x'-y'|+|\widetilde{\varepsilon}|) \quad (5.44)$$

with  $\varepsilon_{ABC} > 0$  from (4.19). Furthermore, combining the estimates from Proposition 3.4 for  $G_z$ ,  $z \in \rho(H)$ , with (5.44) gives us for all  $x', y' \in \mathbb{R}^{\theta-1}$ ,  $j \in \{1, \ldots, \theta\}$  and  $\tilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC})$  the inequalities

$$|G_{z}(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'))| \leq C_{G,1}C_{\iota,1}^{\theta-1}(|x'-y'|+|\widetilde{\varepsilon}|)^{1-\theta}e^{-\frac{C_{G,2}}{C_{\iota,1}}|x'-y'|},$$
  
$$|\partial_{j}G_{z}(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'))| \leq C_{G,1}C_{\iota,1}^{\theta}(|x'-y'|+|\widetilde{\varepsilon}|)^{-\theta}e^{-\frac{C_{G,2}}{C_{\iota,1}}|x'-y'|}.$$
  
(5.45)

We are going to prove the uniform boundedness of  $(I + B_{\varepsilon}(z)Vq)^{-1}$  in  $\mathcal{B}^{0}(\Sigma)$  with respect to  $\varepsilon \in (0, \varepsilon_{\text{conv}})$  for a suitable  $\varepsilon_{\text{conv}} \in (0, \varepsilon_{ABC}]$ . According to (5.8), (5.10) and (5.11) this is equivalent to proving the uniform boundedness of  $(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1}$ in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$ .

We start by analysing  $D_{\varepsilon}^{\zeta,\kappa}(z)$  locally. To proceed, we need to introduce further notations. For  $x'_0 \in \mathbb{R}^{\theta-1}$  we define

$$\zeta_{x'_0}(x') := \zeta(x'_0) + \langle \nabla \zeta(x'_0), x' - x'_0 \rangle, \qquad x' \in \mathbb{R}^{\theta - 1}.$$
(5.46)

Moreover, we define the localization parameter  $a_{\varepsilon} := \varepsilon^{1/6}$  for  $\varepsilon \in (0, \varepsilon_{ABC})$ . Next, we introduce a family of auxiliary operators. For this, we choose a  $C^{\infty}$ -function  $\omega$  with  $0 \le \omega \le 1, \omega = 1$  on  $\mathbb{R}^{\theta-1} \setminus B(0, 1)$  and  $\omega = 0$  on B(0, 1/2). We use this function to cut out the singular part of the kernel of  $D_{\varepsilon}^{\zeta,\kappa}(z)$ ; cf. (5.15). More precisely, by analogy

with (5.16) and (5.17), we define for  $\varepsilon \in (0, \varepsilon_{ABC})$  and  $\tilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$  the operators

$$e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z) : L^{2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N}),$$

$$e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z)g(x') := \int_{\mathbb{R}^{\theta-1}} G_{z}(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \tilde{\varepsilon}\nu_{\zeta,\kappa}(x'))\omega\left(\frac{x'-y'}{a_{\varepsilon}}\right)$$

$$\cdot \sqrt{1 + |\nabla\zeta(y')|^{2}}g(y') \, dy',$$
(5.47)

and

$$E_{\varepsilon}(z): \mathcal{B}^{0}(\mathbb{R}^{\theta-1}) \to \mathcal{B}^{0}(\mathbb{R}^{\theta-1}), \qquad E_{\varepsilon}(z)f(t) := \int_{-1}^{1} e_{\varepsilon(t-s)}^{a_{\varepsilon}}(z)f(s) \, ds. \tag{5.48}$$

First, we prove preliminary results for  $e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z)$  and  $d_{\tilde{\varepsilon}}^{\zeta,\kappa}(z)$ . Afterwards, we transfer these results to the operators  $E_{\varepsilon}(z)$  and  $D_{\varepsilon}^{\zeta,\kappa}(z)$  in Proposition 5.14.

**Lemma 5.10.** Let  $z \in \rho(H)$ ,  $\tilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$ ,  $\varepsilon \in (0, \varepsilon_{ABC})$  as well as  $a_{\varepsilon} = \varepsilon^{1/6}$ . Then, the operator  $e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z)$  acts as a bounded operator from  $L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and

$$\|e_{\widetilde{\varepsilon}}^{a_{\varepsilon}}(z)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \leq C\frac{1+|\log(\varepsilon)|}{a_{\varepsilon}},$$

where C > 0 does not depend on  $\tilde{\varepsilon}$  and  $\varepsilon$ . Moreover, for  $f \in L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  the mapping  $(-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\} \ni \tilde{\varepsilon} \mapsto e^{a_{\varepsilon}}_{\tilde{\varepsilon}}(z) f \in H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  is continuous.

*Proof.* We aim to prove the assertion by applying Lemma C.1. To do so, it is necessary to find suitable estimates for the kernel of  $e^{a_{\varepsilon}}_{\widetilde{\varepsilon}}(z)$  which is for  $x', y' \in \mathbb{R}^{\theta-1}$  given by

$$k(x',y') := G_z(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'))\omega(\frac{x'-y'}{a_\varepsilon})\sqrt{1 + |\nabla\zeta(y')|^2}.$$

We notice as  $G_z \in C^{\infty}(\mathbb{R}^{\theta} \setminus \{0\}; \mathbb{C}^{N \times N}), \zeta \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R})$  and  $\omega \in C_b^{\infty}(\mathbb{R}^{\theta-1}; \mathbb{R})$ , and as  $\omega$  cuts out the singularity of  $G_z$ , we have  $k \in C_b^1(\mathbb{R}^{\theta-1} \times \mathbb{R}^{\theta-1}; \mathbb{C}^{N \times N})$ . Furthermore, using (5.45),  $0 \leq \omega \leq 1$ , supp  $\omega \subset \mathbb{R}^{\theta-1} \setminus B(0, 1/2)$  and  $\zeta \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R})$  immediately gives us for  $x' \neq y' \in \mathbb{R}^{\theta-1}$ 

$$\begin{aligned} |k(x',y')| &\leq C\chi_{\mathbb{R}^{\theta-1}\setminus B(0,1/2)}(\frac{x'-y'}{a_{\varepsilon}})(|x'-y'|+|\widetilde{\varepsilon}|)^{1-\theta}e^{-c|x'-y'|} \\ &\leq C\chi_{\mathbb{R}^{\theta-1}\setminus B(0,1/2)}(\frac{x'-y'}{a_{\varepsilon}})|x'-y'|^{1-\theta}e^{-c|x'-y'|}, \end{aligned}$$

where  $c = \frac{C_{G,2}}{C_{\iota,1}}$  with  $C_{\iota,1} > 0$  from (5.44) and  $C_{G,2}$  from Proposition 3.4. Next, we estimate the derivatives of k. The *l*-th derivative,  $l \in \{1, \ldots, \theta - 1\}$ , with respect to

x' is given by

$$\frac{d}{dx_l'}k(x',y') = \sum_{j=1}^{\theta} \left( (\partial_j G_z)(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x')) \right) \\ \cdot \frac{d}{dx_l'}(\varkappa_{\zeta,\kappa}(x')[j] + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x')[j])\omega(\frac{x'-y'}{a_{\varepsilon}})\sqrt{1 + |\nabla\zeta(y')|^2} \right) \\ + G_z(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'))\frac{1}{a_{\varepsilon}}(\partial_l\omega)(\frac{x'-y'}{a_{\varepsilon}})\sqrt{1 + |\nabla\zeta(y')|^2}$$

for  $x' \neq y' \in \mathbb{R}^{\theta-1}$ , where v[j] denotes the *j*-th component of a vector v. Applying (5.45), and the properties of  $\zeta$  and  $\omega$  again lets us estimate

$$\begin{aligned} \left| \frac{d}{dx_{l}'} k(x', y') \right| &\leq C \chi_{\mathbb{R}^{\theta-1} \setminus B(0, 1/2)} \left( \frac{x' - y'}{a_{\varepsilon}} \right) \left( (|x' - y'| + |\widetilde{\varepsilon}|)^{-\theta} e^{-c|x' - y'|} \right) \\ &+ \frac{1}{a_{\varepsilon}} (|x' - y'| + |\widetilde{\varepsilon}|)^{1-\theta} e^{-c|x' - y'|} \right) \\ &\leq C \chi_{\mathbb{R}^{\theta-1} \setminus B(0, 1/2)} \left( \frac{x' - y'}{a_{\varepsilon}} \right) \left( |x' - y'|^{-\theta} + \frac{1}{a_{\varepsilon}} |x' - y'|^{1-\theta} \right) e^{-c|x' - y'|}. \end{aligned}$$

Thus, if we set  $\widetilde{k}(z') := C\chi_{\mathbb{R}^{\theta-1}\setminus B(0,1/2)}(\frac{z'}{a_{\varepsilon}})\Big(|z'|^{-\theta} + \frac{1}{a_{\varepsilon}}|z'|^{1-\theta}\Big)e^{-c|z'|}$  for  $z' \in \mathbb{R}^{\theta-1}\setminus\{0\}$  we get

$$|k(x',y')|, \sum_{l=1}^{\theta-1} \left| \frac{d}{dx'_l} k(x',y') \right| \le \widetilde{k}(x'-y') \quad \forall x' \neq y' \in \mathbb{R}^{\theta-1}.$$
(5.49)

Hence,  $e^{a_{\varepsilon}}_{\widetilde{\varepsilon}}(z)$  acts by Lemma C.1 as a bounded operator from  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  and

$$\|e_{\widetilde{\varepsilon}}^{a_{\varepsilon}}(z)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \leq C\|k\|_{L^{1}(\mathbb{R}^{\theta-1})}.$$

Now, the norm estimate in the assertion follows from

$$\begin{split} \|\widetilde{k}\|_{L^{1}(\mathbb{R}^{\theta-1})} &= C \int_{\mathbb{R}^{\theta-1}} \chi_{\mathbb{R}^{\theta-1} \setminus B(0,1/2)}(\frac{z'}{a_{\varepsilon}}) \left( |z'|^{-\theta} + \frac{1}{a_{\varepsilon}} |z'|^{1-\theta} \right) e^{-c|z'|} dz' \\ &\leq C \int_{a_{\varepsilon}/2}^{\infty} \left( r^{-\theta} + \frac{1}{a_{\varepsilon}} r^{1-\theta} \right) e^{-cr} r^{\theta-2} dr \\ &\leq C \left( \frac{1}{a_{\varepsilon}} + \frac{1 + |\log(a_{\varepsilon})|}{a_{\varepsilon}} \right) \\ &\leq C \frac{1 + |\log(a_{\varepsilon})|}{a_{\varepsilon}} \leq C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}}. \end{split}$$

Finally, we prove the continuity. For this, let  $\tilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$  and  $(\tilde{\varepsilon}_n)_{n \in \mathbb{N}}$  be a sequence such that  $\tilde{\varepsilon}_n \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$  for all  $n \in \mathbb{N}$  and  $\tilde{\varepsilon}_n \to \tilde{\varepsilon}$ 

as  $n \to \infty$ . Using the dominated convergence theorem and (5.49) shows that for  $f \in L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N) \ e_{\tilde{\varepsilon}_n}^{a_{\varepsilon}}(z) f$  and  $\partial_l e_{\tilde{\varepsilon}_n}^{a_{\varepsilon}}(z) f$ ,  $l \in \{1, \ldots, \theta - 1\}$ , converge pointwise to  $e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z) f$  and  $\partial_l e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z) f$ ,  $l \in \{1, \ldots, \theta - 1\}$ , respectively. Furthermore, the estimate from (5.49) shows that the functions  $|e_{\tilde{\varepsilon}_n}^{a_{\varepsilon}}(z)f|$  and  $|\partial_l e_{\tilde{\varepsilon}_n}^{a_{\varepsilon}}(z)f|$ ,  $l \in \{1, \ldots, \theta - 1\}$ , are independently of  $n \in \mathbb{N}$  pointwise bounded by the function  $|f| * \tilde{k}$  which is by Young's inequality square integrable as  $f \in L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and  $\tilde{k} \in L^1(\mathbb{R}^{\theta-1})$ . Hence, applying the dominated converge theorem again shows that  $e_{\tilde{\varepsilon}_n}^{a_{\varepsilon}}(z)f$  converges to  $e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z)f$  in  $H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ .

**Lemma 5.11.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta_{x'_0}$  be as in (5.46),  $z \in \rho(H)$ ,  $\psi \in C^1_b(\mathbb{R}^{\theta-1})$  and  $\widetilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$ . Then,

$$\left\| \left[ d_{\widetilde{\varepsilon}}^{\zeta_{x_0'},\kappa}(z),\psi\right] \right\|_{L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)\to H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)} \le C \|\psi\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})}(1+|\log|\widetilde{\varepsilon}||),$$

where C > 0 does not depend on  $\tilde{\varepsilon}$  and  $x'_0 \in \mathbb{R}^{\theta-1}$ . Moreover, for  $f \in L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ the mapping  $(-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\} \ni \tilde{\varepsilon} \mapsto [d_{\tilde{\varepsilon}}^{\zeta_{x'_0},\kappa}(z), \psi] f \in H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  is continuous.

*Proof.* We prove this result in the same vein as the previous lemma, i.e. we estimate the kernel of  $[d_{\tilde{\varepsilon}}^{\zeta_{x'_0},\kappa}(z),\psi]$  and its partial derivatives, and apply Lemma C.1. The kernel of  $[d_{\tilde{\varepsilon}}^{\zeta_{x'_0},\kappa}(z),\psi]$  is given by

$$k(x',y') := G_z(\varkappa_{\zeta_{x'_0},\kappa}(x') - \varkappa_{\zeta_{x'_0},\kappa}(y') + \tilde{\varepsilon}\nu_{\zeta_{x'_0},\kappa}(x'))\sqrt{1 + |\nabla\zeta_{x'_0}(y')|^2(\psi(y') - \psi(x'))}$$

for  $x', y' \in \mathbb{R}^{\theta-1}$ . The representations

$$\varkappa_{\zeta_{x'_{0}},\kappa}(x') = \kappa(x',\zeta(x'_{0}) + \langle \nabla\zeta(x'_{0}), x' - x'_{0} \rangle), 
\nu_{\zeta_{x'_{0}},\kappa}(x') = \frac{\kappa(-\nabla\zeta(x'_{0}),1)}{\sqrt{1 + |\nabla\zeta(x'_{0})|^{2}}} = \nu_{\zeta,\kappa}(x'_{0}), 
\nabla\zeta_{x'_{0}}(x') = \nabla\zeta(x'_{0}),$$
(5.50)

for  $x' \in \mathbb{R}^{\theta-1}$  show that k can be simplified to

$$k(x',y') := G_z(\kappa(x'-y', \langle \nabla\zeta(x'_0), x'-y'\rangle) + \tilde{\epsilon}\nu_{\zeta,\kappa}(x'_0))\sqrt{1 + |\nabla\zeta(x'_0)|^2}(\psi(y') - \psi(x')).$$

Moreover, with (5.50) and  $\kappa \in SO(\theta)$  one gets

$$\begin{aligned} |\kappa(x'-y',\langle\nabla\zeta(x'_0),x'-y'\rangle) + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'_0)|^2 \\ &= |x'-y'|^2 + \langle\nabla\zeta(x'_0),x'-y'\rangle^2 + \widetilde{\varepsilon}^2 \\ &\leq |x'-y'|^2(1+\|\nabla\zeta\|^2_{L^{\infty}(\mathbb{R}^{\theta-1};\mathbb{R}^{\theta-1})}) + \widetilde{\varepsilon}^2. \end{aligned}$$

In particular, we can choose a  $C_{\text{lin}} > 0$  which does not depend on  $x'_0$  and  $\tilde{\varepsilon}$  such that

$$(C_{\text{lin}})^{-1}(|x'-y'|+|\widetilde{\varepsilon}|) \leq |\kappa(x'-y',\langle\nabla\zeta(x'_0),x'-y'\rangle) + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'_0)| \\ \leq C_{\text{lin}}(|x'-y'|+|\widetilde{\varepsilon}|).$$
(5.51)

Then, Proposition 3.4, (5.51),  $\psi \in C_b^1(\mathbb{R}^{\theta-1})$  and  $\zeta \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  yield

$$\begin{aligned} |k(x',y')| &\leq C(|x'-y'|+|\widetilde{\varepsilon}|)^{1-\theta} e^{-c'|x'-y'|} \|\psi\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})} |x'-y'| \\ &\leq C \|\psi\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})} (|x'-y'|+|\widetilde{\varepsilon}|)^{2-\theta} e^{-c'|x'-y'|} \quad \forall x',y' \in \mathbb{R}^{\theta-1}, \end{aligned}$$

where  $c' = \frac{C_{G,2}}{C_{\text{lin}}}$  with  $C_{G,2}$  from Proposition 3.4. The *l*-th derivative,  $l \in \{1, \ldots, \theta-1\}$ , with respect to  $x' \in \mathbb{R}^{\theta-1}$  of k is given by

$$\begin{aligned} \frac{d}{dx_l'}k(x',y') &= \Big(\sum_{j=1}^{\theta} (\partial_j G_z)(\kappa(x'-y',\langle \nabla \zeta(x_0'), x'-y'\rangle) + \tilde{\varepsilon}\nu_{\zeta,\kappa}(x_0')) \\ &\cdot \big(\kappa(e_l',\partial_l \zeta(x_0'))\big)[j](\psi(y')-\psi(x')) \\ &- G_z(\kappa(x'-y',\langle \nabla \zeta(x_0'), x'-y'\rangle) + \tilde{\varepsilon}\nu_{\zeta,\kappa}(x_0'))(\partial_l \psi)(x')\big) \\ &\cdot \sqrt{1+|\nabla \zeta(x_0')|^2}, \end{aligned}$$

where  $e'_l$  denotes the *l*-th Euclidean unit vector in  $\mathbb{R}^{\theta-1}$  and  $(\kappa(e'_l, \partial_l \zeta(x'_0)))[j]$  denotes the *j*-th entry of the vector  $\kappa(e'_l, \partial_l \zeta(x'_0))$ . Using Proposition 3.4, (5.51),  $\psi \in C^1_b(\mathbb{R}^{\theta-1})$  and  $\zeta \in C^2_b(\mathbb{R}^{\theta-1}; \mathbb{R})$  again gives us

$$\begin{aligned} \left| \frac{d}{dx_{l}'} k(x', y') \right| &\leq C \|\psi\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})} \Big( (|x' - y'| + |\widetilde{\varepsilon}|)^{-\theta} e^{-c'|x' - y'|} |x' - y'| \\ &+ (|x' - y'| + |\widetilde{\varepsilon}|)^{1-\theta} e^{-c'|x' - y'|} \Big) \\ &\leq C \|\psi\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})} (|x' - y'| + |\widetilde{\varepsilon}|)^{1-\theta} e^{-c'|x' - y'|} \end{aligned}$$

for all  $x', y' \in \mathbb{R}^{\theta-1}$ , where C > 0 can be chosen independently of  $x'_0$  and  $\tilde{\varepsilon}$ . Setting

$$\widetilde{k}(z') = C \|\psi\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})} \big( (|z'| + |\widetilde{\varepsilon}|)^{2-\theta} + (|z'| + |\widetilde{\varepsilon}|)^{1-\theta} \big) e^{-c'|z'}$$

for  $z' \in \mathbb{R}^{\theta-1}$  leads to

$$|k(x',y')|, \sum_{l=1}^{\theta-1} \left| \frac{d}{dx'_l} k(x',y') \right| \le \widetilde{k}(x'-y') \quad \forall x',y' \in \mathbb{R}^{\theta-1}.$$

Now, Lemma C.1 shows

$$\begin{split} \left\| \left[ d_{\widehat{\varepsilon}}^{\zeta_{x_0'},\kappa}(z),\psi\right] \right\|_{L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N) \to H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)} &\leq C \int_{\mathbb{R}^{\theta-1}} \widetilde{k}(z') \, dz' \\ &\leq C \|\psi\|_{W_{\infty}^1(\mathbb{R}^{\theta-1})} \int_0^{\infty} \left( (r+|\widetilde{\varepsilon}|)^{2-\theta} + (r+|\widetilde{\varepsilon}|)^{1-\theta} \right) e^{-c'r} r^{\theta-2} \, dr \\ &\leq C \|\psi\|_{W_{\infty}^1(\mathbb{R}^{\theta-1})} \int_0^{\infty} \left( 1 + (r+|\widetilde{\varepsilon}|)^{-1} \right) e^{-c'r} \, dr \\ &\leq C \|\psi\|_{W_{\infty}^1(\mathbb{R}^{\theta-1})} (1+|\log|\widetilde{\varepsilon}||). \end{split}$$

The assertion regarding the continuity can be proven in a similar way as in the previous lemma.  $\hfill \Box$ 

**Lemma 5.12.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta$  and  $\kappa$  be as in (5.42),  $\zeta_{x'_0}$  be as in (5.46),  $z \in \rho(H)$ ,  $a_{\varepsilon} = \varepsilon^{1/6}$  and  $\widetilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$ . Then, there exists an  $\varepsilon_{\text{gr},1} = \varepsilon_{\text{gr},1}(\zeta)$  in the interval  $(0, \varepsilon_{ABC}]$  such that for all  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  the inequality

$$\left\|\chi_{B(x'_{0},3a_{\varepsilon})}\left(d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z)-d_{\widetilde{\varepsilon}}^{\zeta_{x'_{0}},\kappa}(z)\right)\chi_{B(x'_{0},3a_{\varepsilon})}\right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \leq Ca_{\varepsilon}(1+|\log|\widetilde{\varepsilon}||)$$

holds, where C > 0 does not depend on  $\varepsilon$ ,  $\tilde{\varepsilon}$  and  $x'_0$ .

*Proof.* We prove this statement by estimating the kernel of the operator

$$\chi_{B(x'_{0},a_{\varepsilon})} \left( d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) - d_{\widetilde{\varepsilon}}^{\zeta_{x'_{0}},\kappa}(z) \right) \chi_{B(x'_{0},a_{\varepsilon})}$$

and applying Lemma C.1. The mentioned kernel is given by

$$k(x',y') := \chi_{B(x'_0,3a_{\varepsilon})}(x') \left( G_z(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'))\sqrt{1 + |\nabla\zeta(y')|^2} - G_z(\kappa(x'-y',\langle\nabla\zeta(x'_0),x'-y'\rangle) + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'_0))\sqrt{1 + |\nabla\zeta(x'_0)|^2} \right) \cdot \chi_{B(x'_0,3a_{\varepsilon})}(y')$$

for  $x' \neq y' \in \mathbb{R}^{\theta-1}$ . If  $x' \notin B(x'_0, 3a_{\varepsilon})$  or  $y' \notin B(x'_0, 3a_{\varepsilon})$ , then k(x', y') = 0. Thus, we assume from now on  $x', y' \in B(x'_0, 3a_{\varepsilon})$ . Using Lemma 2.8, Proposition 3.4 and  $\zeta \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R})$  gives us

$$|k(x',y')| \leq \sqrt{\theta} \sup_{v \in [0,1], j \in \{1,\dots\theta\}} |\partial_j G_z(w_v) \sqrt{1 + |\nabla \zeta(y')|^2} ||w_1 - w_0| + |G_z(w_0) (\sqrt{1 + |\nabla \zeta(y')|^2} - \sqrt{1 + |\nabla \zeta(x'_0)|^2})|$$
(5.52)  
$$\leq C \Big( \sup_{v \in [0,1]} |w_v|^{-\theta} |w_1 - w_0| + |w_0|^{1-\theta} |x'_0 - y'| \Big)$$

with

$$w_{\upsilon} = \upsilon \big( \varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon} \nu_{\zeta,\kappa}(x') \big) + (1 - \upsilon) \big( \kappa(x' - y', \langle \nabla \zeta(x'_0), x' - y' \rangle) + \widetilde{\varepsilon} \nu_{\zeta,\kappa}(x'_0) \big)$$

for  $v \in [0, 1]$ . We remark that we were able to apply Lemma 2.8 since  $w_v \neq 0$  for all  $v \in [0, 1]$ , which turns out be true in the course of the proof; cf. (5.54). Next, let us estimate

$$\begin{aligned} |w_{1} - w_{0}| &= |\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x') - \kappa(x' - y', \langle\nabla\zeta(x'_{0}), x' - y'\rangle) - \widetilde{\varepsilon}\nu_{\zeta,\kappa}(x'_{0})| \\ &= |\kappa(x' - y', \zeta(x') - \zeta(y')) \\ &- \kappa(x' - y', \langle\nabla\zeta(x'_{0}), x' - y'\rangle) + \widetilde{\varepsilon}(\nu_{\zeta,\kappa}(x') - \nu_{\zeta,\kappa}(x'_{0}))| \\ &\leq |\kappa(0, \zeta(x') - \zeta(y') - \langle\nabla\zeta(x'_{0}), x' - y'\rangle)| + |\widetilde{\varepsilon}(\nu_{\zeta,\kappa}(x') - \nu_{\zeta,\kappa}(x'_{0}))|. \end{aligned}$$

As  $\zeta \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  and  $\kappa \in SO(\theta)$ , there exists a  $C_{\zeta} > 0$  such that

$$|\nu_{\zeta,\kappa}(x') - \nu_{\zeta,\kappa}(x'_0)| = \left|\frac{\kappa(-\nabla\zeta(x'),1)}{\sqrt{1+|\nabla\zeta(x')|^2}} - \frac{\kappa(-\nabla\zeta(x'_0),1)}{\sqrt{1+|\nabla\zeta(x'_0)|^2}}\right| \le C_{\zeta}|x'-x'_0| \le 3C_{\zeta}a_{\varepsilon}$$

and

$$\begin{aligned} |\kappa(0,\zeta(x') - \zeta(y') - \langle \nabla\zeta(x'_0), x' - y' \rangle)| \\ &= |\zeta(x') - \zeta(y') - \langle \nabla\zeta(x'_0), x' - y' \rangle| \\ &= \left| \int_0^1 \langle \nabla\zeta(y' + t(x' - y')) - \nabla\zeta(x'_0), x' - y' \rangle \, dt \right| \\ &\leq C_{\zeta} \left| \int_0^1 |t(x' - x'_0) + (1 - t)(y' - x'_0)| |x' - y'| \, dt \right| \\ &\leq 3C_{\zeta} a_{\varepsilon} |x' - y'|, \end{aligned}$$

where we used  $x', y' \in B(x'_0, 3a_{\varepsilon})$ . Hence, if  $\varepsilon_{\text{gr},1} = \varepsilon_{\text{gr},1}(\zeta) > 0$  is chosen sufficiently small, then for all  $a_{\varepsilon} \in (0, \varepsilon_{\text{gr},1}^{1/6})$  the inequality

$$|w_1 - w_0| \le 3C_{\zeta} a_{\varepsilon}(|x' - y'| + |\tilde{\varepsilon}|) \le \frac{1}{2C_{\iota,1}}(|x' - y'| + |\tilde{\varepsilon}|)$$
(5.53)

holds with  $C_{\iota,1} > 0$  from (5.44). Therefore, we can use (5.44) to estimate  $|w_v|$ ,  $v \in [0, 1]$ , from below by

$$|w_{v}| = |vw_{1} + (1 - v)w_{0}| = |w_{1} + (1 - v)(w_{0} - w_{1})| \ge |w_{1}| - |w_{1} - w_{0}|$$
  
$$\ge \frac{1}{C_{\iota,1}}(|x' - y'| + |\widetilde{\varepsilon}|) - \frac{1}{2C_{\iota,1}}(|x' - y'| + |\widetilde{\varepsilon}|) = \frac{1}{2C_{\iota,1}}(|x' - y'| + |\widetilde{\varepsilon}|).$$
(5.54)

Thus, by plugging (5.53) and (5.54) into (5.52), the kernel k can be estimated for  $a_{\varepsilon} = \varepsilon^{1/6}$  with  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  by

$$|k(x',y')| \leq \begin{cases} 0, & x' \notin B(x'_0, 3a_{\varepsilon}) \text{ or } y' \notin B(x'_0, 3a_{\varepsilon}), \\ Ca_{\varepsilon}(|x'-y'|+|\widetilde{\varepsilon}|)^{1-\theta}, & x' \in B(x'_0, 3a_{\varepsilon}) \text{ and } y' \in B(x'_0, 3a_{\varepsilon}). \end{cases}$$
Applying Lemma C.1 yields

$$\begin{aligned} \left\| \chi_{B(x'_{0},3a_{\varepsilon})} \left( d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) - d_{\widetilde{\varepsilon}}^{\zeta_{x'_{0}},\kappa}(z) \right) \chi_{B(x'_{0},3a_{\varepsilon})} \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \\ &\leq Ca_{\varepsilon} \int_{B(0,6a_{\varepsilon})}^{6a_{\varepsilon}} (r+|\widetilde{\varepsilon}|)^{1-\theta} dz' \\ &\leq Ca_{\varepsilon} \int_{0}^{6a_{\varepsilon}} (r+|\widetilde{\varepsilon}|)^{1-\theta} r^{\theta-2} dr \\ &\leq Ca_{\varepsilon} \int_{0}^{6a_{\varepsilon}} (r+|\widetilde{\varepsilon}|)^{-1} dr \\ &\leq Ca_{\varepsilon} (1+|\log|\widetilde{\varepsilon}||). \end{aligned}$$

**Corollary 5.13.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta$  and  $\kappa$  be as in (5.42),  $\zeta_{x'_0}$  be as in (5.46),  $\Sigma$  be as in (5.43),  $\eta, \tau \in C^1_b(\Sigma; \mathbb{R})$ ,  $Q^{\zeta,\kappa}_{\eta,\tau}$  be as in (5.18),  $z \in \rho(H)$ ,  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  with  $\varepsilon_{\text{gr},1}$  chosen as in the previous lemma,  $a_{\varepsilon} = \varepsilon^{1/6}$  and  $\widetilde{\varepsilon} \in (-2\varepsilon_{ABC}, 2\varepsilon_{ABC}) \setminus \{0\}$ . Then,

$$\begin{aligned} \left\| \chi_{B(x'_{0},3a_{\varepsilon})} \left( d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa} \right. \\ \left. - d_{\widetilde{\varepsilon}}^{\zeta_{x'_{0}},\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa}(x'_{0}) \right) \chi_{B(x'_{0},3a_{\varepsilon})} \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \leq Ca_{\varepsilon}(1 + |\log|\widetilde{\varepsilon}||), \end{aligned}$$

where C > 0 does not depend on  $\varepsilon$ ,  $\widetilde{\varepsilon}$  and  $x'_0$ .

*Proof.* The previous lemma and  $Q_{\eta,\tau}^{\zeta,\kappa} \in C_b^1(\mathbb{R}^{\theta-1};\mathbb{C}^{N\times N})$  yield

$$\begin{aligned} \left\| \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} \left( d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa} - d_{\widetilde{\varepsilon}}^{\varsigma_{u_{0}}^{\prime},\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa}(x_{0}^{\prime}) \right) \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \\ &\leq \left\| \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) \left( Q_{\eta,\tau}^{\zeta,\kappa} - Q_{\eta,\tau}^{\zeta,\kappa}(x_{0}^{\prime}) \right) \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \\ &+ \left\| \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} \left( d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) - d_{\widetilde{\varepsilon}}^{\varsigma_{u_{0}}^{\prime},\kappa}(z) \right) \chi_{B(x_{0}^{\prime},3a_{\varepsilon})} Q_{\eta,\tau}^{\zeta,\kappa}(x_{0}^{\prime}) \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \\ &\leq C \left( \left\| d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z) \right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} a_{\varepsilon} + a_{\varepsilon}(1 + \left| \log \left| \widetilde{\varepsilon} \right| \right|) \right), \end{aligned}$$

where C > 0 does not depend on  $\varepsilon$ ,  $\tilde{\varepsilon}$  and  $x'_0$ . Moreover, (5.17), (5.44), (5.45) and  $\zeta \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  let us estimate the kernel k of  $d_{\tilde{\varepsilon}}^{\zeta,\kappa}(z)$  by

$$|k(x',y')| \le C(|x'-y'|+|\widetilde{\varepsilon}|)^{1-\theta} e^{-c|x'-y'|} \quad \forall x',y' \in \mathbb{R}^{\theta-1},$$

where  $c = \frac{C_{G,2}}{C_{\iota,1}} > 0$  with  $C_{\iota,1}$  from (5.44) and  $C_{G,2}$  from Proposition 3.4, and therefore Lemma C.1 implies

$$\begin{split} \|d_{\widetilde{\varepsilon}}^{\zeta,\kappa}(z)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} &\leq C \int_{\mathbb{R}^{\theta-1}} (|z'|+|\widetilde{\varepsilon}|)^{1-\theta} e^{-c|z'|} \, dz' \\ &\leq C \int_{0}^{\infty} (r+|\widetilde{\varepsilon}|)^{1-\theta} e^{-cr} r^{\theta-2} \, dr \\ &\leq C(1+|\log|\widetilde{\varepsilon}||), \end{split}$$

which yields the assertion.

 $\Box$ 

Now, we can transfer the results for  $d_{\tilde{\varepsilon}}^{\zeta,\kappa}(z)$  and  $e_{\tilde{\varepsilon}}^{a_{\varepsilon}}(z)$  from Lemma 5.10, Lemma 5.11 and Corollary 5.13 to the operators  $D_{\varepsilon}^{\zeta,\kappa}(z)$  and  $E_{\varepsilon}(z)$  via their representations in (5.16) and (5.48), respectively. We obtain the following statement.

**Proposition 5.14.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta$  and  $\kappa$  be as in (5.42),  $\Sigma$  be as in (5.43),  $\zeta_{x'_0}$  be as in (5.46),  $\eta, \tau \in C^1_b(\Sigma; \mathbb{R})$ ,  $Q^{\zeta,\kappa}_{\eta,\tau}$  be as in (5.18),  $z \in \rho(H)$ ,  $\psi \in C^1_b(\mathbb{R}^{\theta-1})$ ,  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  with  $\varepsilon_{\text{gr},1}$  chosen as in Lemma 5.12 and  $a_{\varepsilon} = \varepsilon^{1/6}$ . Then, the operators  $E_{\varepsilon}(z)$  and  $[D^{\zeta,\kappa}_{\varepsilon}(z), \psi]$  act as bounded operators from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^1(\mathbb{R}^{\theta-1})$  and

$$\begin{split} \|E_{\varepsilon}(z)\|_{0\to 1} &\leq C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}}, \\ \|\left[D_{\varepsilon}^{\zeta_{x'_{0}},\kappa}(z),\psi\right]\right\|_{0\to 1} &\leq C \|\psi\|_{W^{1}_{\infty}(\mathbb{R}^{\theta-1})}(1 + |\log(\varepsilon)|), \\ \|\chi_{B(x'_{0},3a_{\varepsilon})}\left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{x'_{0}},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x'_{0})\right)\chi_{B(x'_{0},3a_{\varepsilon})}\|_{0\to 0} &\leq Ca_{\varepsilon}(1 + |\log(\varepsilon)|), \\ where C > 0 \text{ does not depend on } x'_{0} \text{ and } \varepsilon. \end{split}$$

*Proof.* First, we consider  $[D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z),\psi]$ . We start by showing that  $[D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z),\psi]$  is well-defined as an operator from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^1(\mathbb{R}^{\theta-1})$ . Let  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$  and set  $g = [D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z),\psi]f$ . It follows from (5.16) that g has the representation

$$g(t) = \int_{-1}^{1} [d_{\varepsilon(t-s)}^{\zeta_{x'_0},\kappa}(z),\psi]f(s)\,ds, \qquad t \in (-1,1).$$

Since  $[d_{\varepsilon(t-s)}^{\zeta_{x'_0},\kappa}(z),\psi]$  has the continuity property from Lemma 5.11 and  $f \in \mathcal{B}^0(\mathbb{R}^{\theta-1})$ , the text below Definition 2.13 implies that the function

$$(-1,1) \times (-1,1) \ni (t,s) \mapsto [d_{\varepsilon(t-s)}^{\zeta_{x'_0},\kappa}(z),\psi]f(s) \in H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$$

is measurable. According to Lemma 5.11 we have

$$\int_{-1}^{1} \left( \int_{-1}^{1} \| [d_{\varepsilon(t-s)}^{\zeta_{x_{0}'},\kappa}(z),\psi]f(s)\|_{H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} ds \right)^{2} dt 
\leq C \|\psi\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})}^{2} \int_{-1}^{1} \left( \int_{-1}^{1} (1+|\log|\varepsilon(t-s)||)\|f(s)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} ds \right)^{2} dt.$$
(5.55)

This expression can be estimated with the Cauchy-Schwarz inequality and Fubini's theorem by

$$C\|\psi\|_{W^{1}_{\infty}(\mathbb{R}^{\theta-1})}^{2} \int_{-1}^{1} \int_{-1}^{1} (1+|\log|\varepsilon(t-s)||) ds$$
  

$$\cdot \int_{-1}^{1} (1+|\log|\varepsilon(t-s)||) \|f(s)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})}^{2} ds dt$$
  

$$\leq C\|\psi\|_{W^{1}_{\infty}(\mathbb{R}^{\theta-1})}^{2} \left(\int_{-2}^{2} (1+|\log|\varepsilon s||) ds\right)^{2} \int_{-1}^{1} \|f(s)\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})}^{2} ds$$
  

$$\leq C(\|\psi\|_{W^{1}_{\infty}(\mathbb{R}^{\theta-1})}(1+|\log(\varepsilon)|) \|f\|_{0})^{2}.$$
(5.56)

In particular, applying the Cauchy-Schwarz inequality again gives us

$$\begin{split} \int_{-1}^{1} \int_{-1}^{1} \left\| \left[ d_{\varepsilon(t-s)}^{\zeta_{x_{0}'},\kappa}(z),\psi\right] f(s) \right\|_{H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} ds \, dt \\ & \leq \sqrt{2} \sqrt{\int_{-1}^{1} \left( \int_{-1}^{1} \left\| \left[ d_{\varepsilon(t-s)}^{\zeta_{x_{0}'},\kappa}(z),\psi\right] f(s) \right\|_{H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} ds \right)^{2} dt} < \infty. \end{split}$$

Thus, (2.8) and Fubini's theorem, see Proposition 2.15, shows that  $g = [D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z),\psi]f$  is well-defined and measurable as a function from (-1,1) to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ . Moreover, (5.55) and (5.56) also give us the norm estimate as

$$\begin{split} \|g\|_{1}^{2} &= \int_{-1}^{1} \|g(t)\|_{H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})}^{2} dt \\ &\leq \int_{-1}^{1} \left( \int_{-1}^{1} \|[d_{\varepsilon(t-s)}^{\zeta_{x_{0}'},\kappa}(z),\psi]f(s)\|_{H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \, ds \right)^{2} dt \\ &\leq C \Big( \|\psi\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})} (1+|\log(\varepsilon)|) \|f\|_{0} \Big)^{2}. \end{split}$$

The proof for  $E_{\varepsilon}(z)$  can be done in exactly the same way. Moreover,

$$\chi_{B(x_0',3a_{\varepsilon})} \left( D_{\varepsilon}^{\zeta,\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{x_0'},\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa}(x_0') \right) \chi_{B(x_0',3a_{\varepsilon})}$$

is a well-defined operator in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . So in this case it is sufficient to show the norm estimate, which can be proven in the same way as the norm estimate for  $[D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z),\psi]$ .

As the last part of our local analysis we state a result concerning the inverse of  $I + D_{\varepsilon}^{\zeta_{x'_0},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x'_0)q$  for  $x'_0 \in \mathbb{R}^{\theta-1}$ . This is an important result since these operators play an essential role when constructing the inverse of  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$ .

**Proposition 5.15.** Let  $\zeta$  and  $\kappa$  be as in (5.42),  $\Sigma$  be as in (5.43),  $\zeta_{x'_0}$  be as in (5.46),  $\eta, \tau \in C^1_b(\Sigma; \mathbb{R}), d = \eta^2 - \tau^2$  satisfy

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4},$$

q be as in (5.1),  $Q_{\eta,\tau}^{\zeta,\kappa}$  be as in (5.18) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\varepsilon_{\mathrm{gr},2} > 0$ such that the operators  $(I + D_{\varepsilon}^{\zeta_{x_0'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x_0')q)^{-1}$  are uniformly bounded in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ with respect to  $\varepsilon \in (0, \varepsilon_{\mathrm{gr},2})$  and  $x_0' \in \mathbb{R}^{\theta-1}$ . *Proof.* Note that

$$Q_{\eta,\tau}^{\zeta,\kappa}(x_0') = \eta(\varkappa_{\zeta,\kappa}(x_0'))I_N + \tau(\varkappa_{\zeta,\kappa}(x_0'))\beta = Q_{\eta(\varkappa_{\zeta,\kappa}(x_0')),\tau(\varkappa_{\zeta,\kappa}(x_0'))}$$

for  $x'_0 \in \mathbb{R}^{\theta-1}$ ; cf. (5.18) and (5.19) and the text below (5.43). Moreover, for every  $x'_0 \in \mathbb{R}^{\theta-1}$  the set  $\Sigma_{\zeta_{x'_0},\kappa}$  is an affine hyperplane in  $\mathbb{R}^{\theta}$  and therefore there exists a  $\widetilde{y}_0(x'_0) \in \mathbb{R}$  and a  $\widetilde{\kappa}(x'_0) \in \mathrm{SO}(\theta)$  such that

$$\Sigma_{\zeta_{x'_0},\kappa} = \{\kappa(x',\zeta_{x'_0}(x')) : x' \in \mathbb{R}^{\theta-1}\}$$
  
=  $\widetilde{\kappa}(x'_0) (\mathbb{R}^{\theta-1} \times \{\widetilde{y}_0(x'_0)\}) = \Sigma_{\widetilde{y}_0(x'_0),\widetilde{\kappa}(x'_0)}.$  (5.57)

Hence, we get from (5.9), (5.10) and (5.11)

$$\begin{split} \left\| (I + D_{\varepsilon}^{\zeta_{x_0'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x_0')q)^{-1} \right\|_{0\to 0} \\ &= \left\| (I + \overline{B}_{\varepsilon}^{\Sigma_{\zeta_{x_0'},\kappa}}(z)Q_{\eta(\varkappa_{\zeta,\kappa}(x_0')),\tau(\varkappa_{\zeta,\kappa}(x_0'))}q)^{-1} \right\|_{0\to 0} \\ &= \left\| (I + \overline{B}_{\varepsilon}^{\Sigma_{\widetilde{y_0}(x_0'),\widetilde{\kappa}(x_0')}}(z)Q_{\eta(\varkappa_{\zeta,\kappa}(x_0')),\tau(\varkappa_{\zeta,\kappa}(x_0'))}q)^{-1} \right\|_{0\to 0} \\ &= \left\| (I + D_{\varepsilon}^{\widetilde{y_0}(x_0'),\widetilde{\kappa}(x_0')}(z)Q_{\eta(\varkappa_{\zeta,\kappa}(x_0')),\tau(\varkappa_{\zeta,\kappa}(x_0'))}q)^{-1} \right\|_{0\to 0}. \end{split}$$

Now, the result follows from applying Corollary 5.9 (for  $S = \operatorname{ran}(\eta, \tau)$ ) if one chooses  $\varepsilon_{\operatorname{gr},2} = \varepsilon_{\operatorname{pl},2}(\overline{\operatorname{ran}(\eta, \tau)}) > 0$ , where  $\varepsilon_{\operatorname{pl},2}$  was introduced in Corollary 5.9.

Inspired by the local principle in [60, Proposition 5], see also [59, 61], we construct partitions of unity which allow us to globalize the established local results. We start by choosing a partition of unity  $(\phi_{n'})_{n' \in \mathbb{Z}^{\theta-1}}$  for  $\mathbb{R}^{\theta-1}$  with uniformly bounded derivatives which satisfies  $\operatorname{supp} \phi_{n'} \subset B(n', 1)$  for  $n' \in \mathbb{Z}^{\theta-1}$ . Moreover, let  $(\vartheta_{n'})_{n' \in \mathbb{Z}^{\theta-1}}$  be a sequence of functions with uniformly bounded derivatives which fulfils  $0 \leq \vartheta_{n'} \leq 1$ ,  $\vartheta_{n'} = 1$  on B(n', 2) and  $\operatorname{supp} \vartheta_{n'} \subset B(n', 3)$  for  $n' \in \mathbb{Z}^{\theta-1}$ . According to Proposition A.2 such sequences exist. By defining for  $a \in (0, (\varepsilon_{ABC})^{1/6})$  and  $n' \in \mathbb{Z}^{\theta-1}$  the functions  $\phi_{n'}^a(\cdot) = \phi_{n'}(\cdot/a)$  and  $\vartheta_{n'}^a(\cdot) = \vartheta_{n'}(\cdot/a)$  we obtain similar sequences; in particular  $(\phi_{n'}^a)_{n' \in \mathbb{Z}^{\theta-1}}$  is a partition of unity for  $\mathbb{R}^{\theta-1}$  with scaled supports. Furthermore, there exists a C > 0 which does not depend on a such that

$$\sup_{n' \in \mathbb{Z}^{\theta-1}} \max\{ \|\phi_{n'}^a\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})}, \|\vartheta_{n'}^a\|_{W^1_{\infty}(\mathbb{R}^{\theta-1})} \} < \frac{C}{a};$$
(5.58)

cf. Corollary A.3. Before we use these essential observations in the proof of Proposition 5.17 to construct the right inverse of  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$ , we state a helpful preliminary lemma.

**Lemma 5.16.** Let  $z \in \rho(H)$ ,  $\varepsilon \in (0, \varepsilon_{ABC})$  and  $a_{\varepsilon} = \varepsilon^{1/6}$ . Then,

$$(1 - \vartheta_{n'}^{a_{\varepsilon}})E_{\varepsilon}(z)\phi_{n'}^{a_{\varepsilon}} = (1 - \vartheta_{n'}^{a_{\varepsilon}})D_{\varepsilon}^{\zeta,\kappa}(z)\phi_{n'}^{a_{\varepsilon}}.$$

*Proof.* We prove this lemma by showing that the difference of the integral kernels of  $(1 - \vartheta_{n'}^{a_{\varepsilon}})E_{\varepsilon}(z)\phi_{n'}^{a_{\varepsilon}}$  and  $(1 - \vartheta_{n'}^{a_{\varepsilon}})D_{\varepsilon}^{\zeta,\kappa}(z)\phi_{n'}^{a_{\varepsilon}}$  is zero. According to (5.15), (5.47) and (5.48) this difference is given by

$$(1 - \vartheta_{n'}^{a_{\varepsilon}}(x')) \left( \omega(\frac{x'-y'}{a_{\varepsilon}}) - 1 \right) \phi_{n'}^{a_{\varepsilon}}(y') \cdot G_{z}(\varkappa_{\zeta,\kappa}(x') - \varkappa_{\zeta,\kappa}(y') + \varepsilon(t-s)\nu_{\zeta,\kappa}(x')) \sqrt{1 + |\nabla\zeta(y')|^{2}}$$
(5.59)

for all  $x', y' \in \mathbb{R}^{\theta-1}$  and  $t, s \in (-1, 1)$ . If  $y' \notin B(a_{\varepsilon}n', a_{\varepsilon})$  for an  $n' \in \mathbb{R}^{\theta-1}$ , then  $\frac{y'}{a_{\varepsilon}} \notin B(n', 1) \supset \operatorname{supp} \phi_{n'}$  and therefore

$$\phi_{n'}^{a_{\varepsilon}}(y') = \phi_{n'}(\frac{y'}{a_{\varepsilon}}) = 0.$$

Furthermore, if  $x' \in B(a_{\varepsilon}n', 2a_{\varepsilon})$ , then  $\frac{x'}{a_{\varepsilon}} \in B(n', 2)$  and hence as  $\vartheta_{n'} = 1$  on B(n', 2) we have

$$1 - \vartheta_{n'}^{a_{\varepsilon}}(x') = 1 - \vartheta_{n'}(\frac{x'}{a_{\varepsilon}}) = 0.$$

These two observations show that if  $y' \notin B(a_{\varepsilon}n', a_{\varepsilon})$  or  $x' \in B(a_{\varepsilon}n', 2a_{\varepsilon})$ , then (5.59) vanishes. Thus, it remains to consider the case  $y' \in B(a_{\varepsilon}n', a_{\varepsilon})$  and  $x' \notin B(a_{\varepsilon}n', 2a_{\varepsilon})$ . However, this implies  $|x' - y'| > a_{\varepsilon}$ . In this case we use  $\omega = 1$  on  $\mathbb{R}^{\theta-1} \setminus B(0, 1)$ , see the text above (5.47), to obtain

$$\omega(\frac{x'-y'}{a_{\varepsilon}}) - 1 = 0.$$

This shows that (5.59) vanishes for all  $x', y' \in \mathbb{R}^{\theta-1}$  and  $t, s \in (-1, 1)$ .

**Proposition 5.17.** Let  $\zeta$  and  $\kappa$  be as in (5.42),  $\Sigma$  be as in (5.43),  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ ,  $d = \eta^2 - \tau^2$  satisfy

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4},\tag{5.60}$$

q be as in (5.1),  $Q_{\eta,\tau}^{\zeta,\kappa}$  be as in (5.18) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\varepsilon_{\mathrm{gr},3}$  in the interval  $(0, \varepsilon_{ABC}]$ , with  $\varepsilon_{ABC} > 0$  chosen according to (4.19), such that the operator  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$  has a right inverse which is uniformly bounded in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  with respect to  $\varepsilon \in (0, \varepsilon_{\mathrm{gr},3})$ .

*Proof.* The proof is split into four steps. In Step 1 we define a first approximation for the right inverse of  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$  denoted by  $R_{\varepsilon}$ . Moreover, in this step we also show that  $R_{\varepsilon}$  is uniformly bounded in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$  with respect to  $\varepsilon$ . Then, in Step 2 we calculate  $(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon}$ . Afterwards, we find in Step 3 that the product  $(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon}$  equals  $I + K_{\varepsilon} + L_{\varepsilon}$ , where  $K_{\varepsilon}$  and  $L_{\varepsilon}$  fulfil the inequalities

$$\|K_{\varepsilon}\|_{0\to 1} \le C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}^2} \quad \text{and} \quad \|L_{\varepsilon}\|_{0\to 0} \le C a_{\varepsilon} (1 + |\log(\varepsilon)|).$$
(5.61)

Based on these observations we use Proposition 2.30 in *Step 4* to prove the assertion. *Step 1.* We define for  $\varepsilon \in (0, \min\{\varepsilon_{ABC}, \varepsilon_{gr,2}\})$  with  $\varepsilon_{ABC}$  and  $\varepsilon_{gr,2}$  chosen as in (4.19) and Proposition 5.15, respectively,  $a_{\varepsilon} = \varepsilon^{1/6}$  and  $n' \in \mathbb{Z}^{\theta-1}$ 

$$R_{n',\varepsilon} := (I + D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q)^{-1}$$

with  $\zeta_{a_{\varepsilon}n'} = \zeta_{x'_0}$  as in (5.46) for  $x'_0 = a_{\varepsilon}n'$ . These operators are uniformly bounded in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  with respect to  $n' \in \mathbb{Z}^{\theta-1}$  and  $\varepsilon \in (0, \min\{\varepsilon_{ABC}, \varepsilon_{gr,2}\})$  according to Proposition 5.15. Therefore, Proposition C.3 (see also (v) in Section 2.1) shows that

$$R_{\varepsilon}: \mathcal{B}^{0}(\mathbb{R}^{\theta-1}) \to \mathcal{B}^{0}(\mathbb{R}^{\theta-1}),$$
$$R_{\varepsilon}:=\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} \phi_{n'}^{a_{\varepsilon}} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}},$$

is well-defined and uniformly bounded by

$$\|R_{\varepsilon}\|_{0\to 0} \le 11^{\theta-1} \sup_{n' \in \mathbb{Z}^{\theta-1}} \left\| (I + D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q)^{-1} \right\|_{0\to 0} \le C,$$
(5.62)

where C > 0 does not depend on  $\varepsilon$ . Furthermore, since  $\vartheta_{n'}^{a_{\varepsilon}} \phi_{n'}^{a_{\varepsilon}} = \phi_{n'}^{a_{\varepsilon}}$  by construction, we have  $R_{\varepsilon} = \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \phi_{n'}^{a_{\varepsilon}} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}}$ .

Step 2. Applying  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$  to  $R_{\varepsilon}$  yields

$$(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon} = \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} (I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}}$$
$$= \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}}$$
$$+ \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} (1 - \vartheta_{n'}^{a_{\varepsilon}})D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}}.$$

Moreover, using Lemma 5.16 gives us

$$\begin{split} (I+D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon} &= \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}(I+D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}} \\ &+ \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} (1-\vartheta_{n'}^{a_{\varepsilon}})E_{\varepsilon}(z)Q_{\eta,\tau}^{\zeta,\kappa}q\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}} \\ &= \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}(I+D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q-E_{\varepsilon}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)\phi_{n'}^{a_{\varepsilon}}R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}} \\ &+ E_{\varepsilon}(z)Q_{n,\tau}^{\zeta,\kappa}qR_{\varepsilon}. \end{split}$$

Writing  $D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q\phi_{n'}^{a_{\varepsilon}}$  as

$$D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q\phi_{n'}^{a_{\varepsilon}} + \left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')\right)q\phi_{n'}^{a_{\varepsilon}} \\ = \phi_{n'}^{a_{\varepsilon}}D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q + \left[D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z),\phi_{n'}^{a_{\varepsilon}}\right]Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q \\ + \left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')\right)q\phi_{n'}^{a_{\varepsilon}}$$

and introducing the operators  $L_{n',\varepsilon} := \left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{a\varepsilon n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')\right)q\phi_{n'}^{a_{\varepsilon}}$  and  $K_{n',\varepsilon} := \left[D_{\varepsilon}^{\zeta_{a\varepsilon n'},\kappa}(z), \phi_{n'}^{a_{\varepsilon}}\right]Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q - E_{\varepsilon}(z)Q_{\eta,\tau}^{\zeta,\kappa}q\phi_{n'}^{a_{\varepsilon}}$  yields

$$(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon}$$

$$= \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}\phi_{n'}^{a_{\varepsilon}}(I + D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q)R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}}$$

$$+ \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}(K_{n',\varepsilon} + L_{n',\varepsilon})R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}} + E_{\varepsilon}(z)Q_{\eta,\tau}^{\zeta,\kappa}qR_{\varepsilon}$$

$$= I + \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}}(K_{n',\varepsilon} + L_{n',\varepsilon})R_{n',\varepsilon}\vartheta_{n'}^{a_{\varepsilon}} + E_{\varepsilon}(z)Q_{\eta,\tau}^{\zeta,\kappa}qR_{\varepsilon},$$
(5.63)

where

$$\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}}\vartheta_{n'}^{a_{\varepsilon}}\phi_{n'}^{a_{\varepsilon}}\vartheta_{n'}^{a_{\varepsilon}} = \sum_{n'\in\mathbb{Z}^{\theta-1}}\vartheta_{n'}^{a_{\varepsilon}}\phi_{n'}^{a_{\varepsilon}}\vartheta_{n'}^{a_{\varepsilon}} = \sum_{n'\in\mathbb{Z}^{\theta-1}}\phi_{n'}^{a_{\varepsilon}} = 1$$

was used.

Step 3. We start this step by setting  $\frac{1}{2}$ 

$$K_{\varepsilon} := \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} K_{n',\varepsilon} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}} + E_{\varepsilon}(z) Q_{\eta,\tau}^{\zeta,\kappa} q R_{\varepsilon},$$
$$L_{\varepsilon} := \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} L_{n',\varepsilon} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}}.$$

Then, (5.63) shows  $(I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)R_{\varepsilon} = I + K_{\varepsilon} + L_{\varepsilon}$ . Since  $R_{\varepsilon}$  and  $D_{\varepsilon}^{\zeta,\kappa}(z)$  are uniformly bounded in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$ , see *Step 1* and the text above (5.12), respectively, this implies that also  $K_{\varepsilon} + L_{\varepsilon}$  is uniformly bounded in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$ . Moreover, Proposition C.3, Proposition 5.14, Proposition 5.15, (5.62) and  $Q_{\eta,\tau}^{\zeta,\kappa} \in C_{b}^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N\times N})$  imply

$$\begin{split} \|K_{\varepsilon}\|_{0\to1} &\leq \Big\|\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} K_{n',\varepsilon} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}} \Big\|_{0\to1} + \|E_{\varepsilon}(z) Q_{\eta,\tau}^{\zeta,\kappa} q R_{\varepsilon}\|_{0\to1} \\ &\leq \frac{C}{a_{\varepsilon}} \sup_{n'\in\mathbb{Z}^{\theta-1}} \|K_{n',\varepsilon} R_{n',\varepsilon}\|_{0\to1} + \|E_{\varepsilon}(z) Q_{\eta,\tau}^{\zeta,\kappa} q R_{\varepsilon}\|_{0\to1} \\ &\leq \frac{C}{a_{\varepsilon}} \sup_{n'\in\mathbb{Z}^{\theta-1}} \|[D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z), \phi_{n'}^{a_{\varepsilon}}] Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')q - E_{\varepsilon}(z) Q_{\eta,\tau}^{\zeta,\kappa} q \phi_{n'}^{a_{\varepsilon}}\|_{0\to1} + \|E_{\varepsilon}(z)\|_{0\to1} \\ &\leq \frac{C}{a_{\varepsilon}} \sup_{n'\in\mathbb{Z}^{\theta-1}} \left( \|[D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z), \phi_{n'}^{a_{\varepsilon}}]\|_{0\to1} + \|E_{\varepsilon}(z)\|_{0\to1} \right) + \|E_{\varepsilon}(z)\|_{0\to1} \\ &\leq C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}} \Big( \sup_{n'\in\mathbb{Z}^{\theta-1}} \|\phi_{n'}^{a_{\varepsilon}}\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})} + \frac{1}{a_{\varepsilon}} + 1 \Big) \\ &\leq C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}} \Big( \frac{2}{a_{\varepsilon}} + 1 \Big) \\ &\leq C \frac{1 + |\log(\varepsilon)|}{a_{\varepsilon}^{2}} \Big( \frac{2}{a_{\varepsilon}} + 1 \Big) \end{split}$$

Similarly, we estimate  $L_{\varepsilon}$  with Proposition C.3, Proposition 5.14 and Proposition 5.15 by

$$\begin{split} \|L_{\varepsilon}\|_{0\to0} &= \left\|\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} L_{n',\varepsilon} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}}\right\|_{0\to0} \\ &= \left\|\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta_{n'}^{a_{\varepsilon}} \chi_{B(a_{\varepsilon}n',3a_{\varepsilon})} L_{n',\varepsilon} R_{n',\varepsilon} \vartheta_{n'}^{a_{\varepsilon}}\right\|_{0\to0} \\ &\leq C \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\|\chi_{B(a_{\varepsilon}n',3a_{\varepsilon})} L_{n',\varepsilon}\right\|_{0\to0} \\ &= C \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\|\chi_{B(a_{\varepsilon}n',3a_{\varepsilon})} \left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')\right)q\phi_{n'}^{a_{\varepsilon}}\right\|_{0\to0} \\ &= C \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\|\chi_{B(a_{\varepsilon}n',3a_{\varepsilon})} \left(D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa} - D_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n')\right)q\chi_{B(a_{\varepsilon}n',3a_{\varepsilon})}\phi_{n'}^{a_{\varepsilon}}\right\|_{0\to0} \\ &\leq Ca_{\varepsilon}(1+|\log(\varepsilon)|). \end{split}$$

This shows that (5.61) is valid and hence completes Step 3.

Step 4. Revisiting the considerations from the beginning of the current chapter shows that for  $V = \eta I_N + \tau \beta$ ,  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  fulfilling (5.60), the following holds:  $\cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N}),$ 

$$\widetilde{V} = V \operatorname{sinc}\left(\frac{(\alpha \cdot \nu)V}{2}\right) \cos\left(\frac{(\alpha \cdot \nu)V}{2}\right)^{-1} = \widetilde{\eta} I_N + \widetilde{\tau}\beta \quad \text{with} \quad (\widetilde{\eta}, \widetilde{\tau}) = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)(\eta, \tau),$$

and  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2$  fulfils  $\inf_{x_{\Sigma} \in \Sigma} |\tilde{d}(x_{\Sigma}) - 4| > 0$ . Thus, by Proposition 3.15 (iii) and Proposition 4.13 the operator  $I + B_0(z)Vq$  is continuously invertible in  $\mathcal{B}^0(\Sigma)$  and  $\mathcal{B}^{1/2}(\Sigma)$ . Hence, (5.9) and (5.11) imply that also  $I + D_0^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$  is continuously invertible in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  and  $\mathcal{B}^{1/2}(\mathbb{R}^{\theta-1})$ . Applying Proposition 2.30 (for the choices  $\mathcal{A} = I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$ ,  $\mathcal{A}_0 = I + D_0^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$ ,  $\mathcal{T} = R_{\varepsilon}$ ,  $\mathcal{K}_1 = 0$ ,  $\mathcal{K}_2 = K_{\varepsilon} + L_{\varepsilon}$ ) shows that if the operator norm of

$$\widetilde{L}_{\varepsilon} := (D_0^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q(I + D_0^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1}(K_{\varepsilon} + L_{\varepsilon})$$
$$(= \mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2)$$

is bounded by  $\frac{1}{2}$ , then  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q$  has a right inverse which is bounded by

$$2 \left\| R_{\varepsilon} - (I + D_0^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1}(K_{\varepsilon} + L_{\varepsilon}) \right\|_{0\to 0}$$

In particular, as  $K_{\varepsilon} + L_{\varepsilon}$  and  $R_{\varepsilon}$  are uniformly bounded in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$ , this would yield the assertion. Using the estimates for  $L_{\varepsilon}$  and  $K_{\varepsilon}$  from *Step 3* as well as (5.12) and the text above it, we obtain for a fixed  $r \in (0, \frac{1}{6})$ 

$$\begin{split} \|\tilde{L}_{\varepsilon}\|_{0\to0} &\leq \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q(I + D_{0}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1}K_{\varepsilon} \right\|_{0\to0} \\ &+ \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q(I + D_{0}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1}L_{\varepsilon} \right\|_{0\to0} \\ &\leq \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q \right\|_{1/2\to0} \\ &\cdot \left\| (I + D_{0}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1} \right\|_{1/2\to1/2} \|K_{\varepsilon}\|_{0\to1/2} \\ &+ \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q \right\|_{0\to0} \\ &\leq \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q \right\|_{1/2\to0} \\ &\cdot \left\| (I + D_{0}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1} \right\|_{1/2\to1/2} \|K_{\varepsilon}\|_{0\to1} \\ &+ \left\| (D_{0}^{\zeta,\kappa}(z) - D_{\varepsilon}^{\zeta,\kappa}(z))Q_{\eta,\tau}^{\zeta,\kappa}q \right\|_{0\to0} \\ &\cdot \left\| (I + D_{0}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}q)^{-1} \right\|_{0\to0} \|L_{\varepsilon}\|_{0\to0} \\ &\leq C \Big( \frac{\varepsilon^{1/2-r}(1 + |\log(\varepsilon)|)}{a_{\varepsilon}^{2}} + a_{\varepsilon}(1 + |\log(\varepsilon)|) \Big) \\ &= C(1 + |\log(\varepsilon)|)(\varepsilon^{1/6-r} + \varepsilon^{1/6}) \leq C(1 + |\log(\varepsilon)|)\varepsilon^{1/6-r}. \end{split}$$

This shows that if we choose  $\varepsilon_{\text{gr},3} > 0$  sufficiently small, then  $\|\widetilde{L}_{\varepsilon}\|_{0\to 0} < \frac{1}{2}$  for all  $\varepsilon \in (0, \varepsilon_{\text{gr},3})$ .

Finally, we are able to state the main result of Section 5.1 in the following proposition. **Proposition 5.18.** Let  $\Sigma$  be a rotated  $C_b^2$ -graph as described in the beginning of Section 5.1.2,  $z \in \mathbb{C} \setminus \mathbb{R}$ , q and  $V = \eta I_N + \tau \beta$  be as in (5.1) and (5.2),  $d = \eta^2 - \tau^2$  fulfil

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}$$

and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\varepsilon_{\text{conv}} \in (0, \varepsilon_{ABC}]$ , with  $\varepsilon_{ABC} > 0$  chosen according to (4.19), such that  $I + B_{\varepsilon}(z)Vq$  has an inverse which is uniformly bounded in  $\mathcal{B}^{0}(\Sigma)$ with respect to  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ .

Proof. From the previous proposition, (5.10), (5.11) and (5.18) we directly get that the operator  $I + \overline{B}_{\varepsilon}(z)Vq$  has a right inverse which is uniformly bounded with respect to  $\varepsilon \in (0, \varepsilon_{\text{gr},3})$ . Using (5.8) shows that then  $I + B_{\varepsilon}(z)Vq$  has a right inverse which is uniformly bounded for  $\varepsilon \in (0, \varepsilon_{\text{conv}})$  if  $\varepsilon_{\text{conv}} > 0$  is chosen small enough. Moreover, since  $z \in \mathbb{C} \setminus \mathbb{R}$ , Proposition 4.1 (i) implies that  $I + B_{\varepsilon}(z)Vq$  has also a left inverse which yields that  $I + B_{\varepsilon}(z)Vq$  is invertible and its inverse is uniformly bounded.  $\Box$ 

## 5.2 Main results

In this section we state and prove the main results of this chapter. After dealing with the cases where  $\Sigma$  is a hyperplane in Section 5.1.1 and a rotated graph of a  $C_b^2$ -function in Section 5.1.2, we return to our general assumption that  $\Sigma \subset \mathbb{R}^{\theta}$  is a special  $C^2$ -surface as in Definition 2.1. We start by providing a useful lemma regarding Dirac operators with  $\delta$ -shell potentials.

**Lemma 5.19.** Let  $O_1, O_2 \subset \mathbb{R}^{\theta}$  be open sets such that their boundaries  $S_1 = \partial O_1$ ,  $S_2 = \partial O_2$  are special  $C^2$ -surfaces as in Definition 2.1,  $\widehat{V}_1 = \widehat{V}_1^* \in W_{\infty}^1(S_1; \mathbb{C}^{N \times N})$ ,  $\widehat{V}_2 = \widehat{V}_2^* \in W_{\infty}^1(S_2; \mathbb{C}^{N \times N})$ ,  $\varphi \in C_b^1(\mathbb{R}^{\theta})$  and  $H_{\widehat{V}_1\delta_{S_1}}$ ,  $H_{\widehat{V}_2\delta_{S_2}}$  be Dirac operators with  $\delta$ -shell potentials as in Definition 3.12. Moreover, assume that there exists an open set  $O \subset \mathbb{R}^{\theta}$  such that  $\operatorname{supp} \varphi \subset O$  and  $O_1 \cap O = O_2 \cap O$ , and  $\widehat{V}_1 = \widehat{V}_2 \sigma$ -a.e on  $S_1 \cap O(=S_2 \cap O)$ . Then,  $H_{\widehat{V}_1\delta_{S_1}}\varphi = H_{\widehat{V}_2\delta_{S_2}}\varphi$ , where  $\varphi$  is viewed as an multiplication operator in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , and for  $u \in \operatorname{dom} H_{\widehat{V}_j\delta_{S_j}}$ ,  $j \in \{1, 2\}$ , one has  $u \in \operatorname{dom} H_{\widehat{V}_j\delta_{S_j}}\varphi$ and

$$\varphi H_{\widehat{V}_j \delta_{S_i}} u = H_{\widehat{V}_j \delta_{S_i}} \varphi u + i(\alpha \cdot \nabla \varphi) u.$$

Proof. In this proof we use the notations  $O_{j,+} = O_j$  and  $O_{j,-} = \mathbb{R}^{\theta} \setminus \overline{O_j}$  for  $j \in \{1, 2\}$ . Moreover,  $\nu_j$  denotes the unit normal vector field on  $S_j$  pointing outwards of  $O_{j,+}$ . Now, let us start by showing  $H_{\widehat{V}_1\delta_{S_1}}\varphi = H_{\widehat{V}_2\delta_{S_2}}\varphi$ . To do so, let  $u \in \text{dom } H_{\widehat{V}_1\delta_{S_1}}\varphi$ . Then,  $\varphi u \in \text{dom } H_{\widehat{V}_1\delta_{S_1}} \subset H^1(\mathbb{R}^{\theta} \setminus S_1)$ . In particular,  $(\varphi u) \upharpoonright O_{1,\pm} \in H^1(O_{1,\pm}; \mathbb{C}^N)$ . Furthermore,  $(\varphi u) \upharpoonright O_{2,\pm} \in H^1(O_{2,\pm}; \mathbb{C}^N)$  since supp  $\varphi \subset O$  and  $O_{1,\pm} \cap O = O_{2,\pm} \cap O$ . Thus, we can apply the trace operator to  $(\varphi u) \upharpoonright O_{2,\pm}$  and obtain

$$\begin{split} & \left(i(\alpha \cdot \nu_2)(\boldsymbol{t}_{S_2}^+ - \boldsymbol{t}_{S_2}^-) + \frac{V_2}{2}(\boldsymbol{t}_{S_2}^+ + \boldsymbol{t}_{S_2}^-)\right)\varphi u \\ & = \begin{cases} \left(i(\alpha \cdot \nu_2)(\boldsymbol{t}_{S_2}^+ - \boldsymbol{t}_{S_2}^-) + \frac{\hat{V}_2}{2}(\boldsymbol{t}_{S_2}^+ + \boldsymbol{t}_{S_2}^-)\right)\varphi u & \text{ on } S_2 \cap O, \\ 0 & \text{ on } S_2 \setminus O, \end{cases} \\ & = \begin{cases} \left(i(\alpha \cdot \nu_1)(\boldsymbol{t}_{S_1}^+ - \boldsymbol{t}_{S_1}^-) + \frac{\hat{V}_1}{2}(\boldsymbol{t}_{S_1}^+ + \boldsymbol{t}_{S_1}^-)\right)\varphi u & \text{ on } S_1 \cap O, \\ 0 & \text{ on } S_1 \setminus O, \end{cases} \\ & = 0, \end{split}$$

where we used  $\varphi u \in \text{dom } H_{\widehat{V}_1 \delta_{S_1}}$ ,  $\text{supp } \varphi \subset O$  and  $S_1 \cap O = S_2 \cap O$ . Hence,  $u\varphi$  fulfils the boundary condition

$$\left(i(\alpha \cdot \nu_2)(\mathbf{t}_{S_2}^+ - \mathbf{t}_{S_2}^-) + \frac{\widehat{V}_2}{2}(\mathbf{t}_{S_2}^+ + \mathbf{t}_{S_2}^-)\right)\varphi u = 0,$$

i.e.  $\varphi u \in \operatorname{dom} H_{\widehat{V}_2 \delta_{S_2}}$ . This shows  $\operatorname{dom} H_{\widehat{V}_1 \delta_{S_1}} \varphi \subset \operatorname{dom} H_{\widehat{V}_2 \delta_{S_2}} \varphi$ . The reverse inclusion can be proven in the same way. Moreover, for  $u \in \operatorname{dom} H_{\widehat{V}_1 \delta_{S_1}} \varphi = \operatorname{dom} H_{\widehat{V}_2 \delta_{S_2}} \varphi$  the equality

$$\begin{aligned} H_{\widehat{V}_{1}\delta_{S_{1}}}\varphi u &= \begin{cases} (-i(\alpha\cdot\nabla)+m\beta)(\varphi u) \upharpoonright O_{1,\pm}\cap O & \text{in } O_{1,\pm}\cap O, \\ 0 & \text{else,} \end{cases} \\ &= \begin{cases} (-i(\alpha\cdot\nabla)+m\beta)(\varphi u) \upharpoonright O_{2,\pm}\cap O & \text{in } O_{2,\pm}\cap O, \\ 0 & \text{else,} \end{cases} \\ &= H_{\widehat{V}_{2}\delta_{S_{2}}}\varphi u \end{aligned}$$

is valid, showing  $H_{\widehat{V}_1\delta_{S_1}}\varphi = H_{\widehat{V}_2\delta_{S_2}}\varphi$ . With similar arguments as above one proves  $\varphi u \in \operatorname{dom} H_{\widehat{V}_j\delta_{S_j}}$ ,  $j \in \{1, 2\}$ , for  $u \in \operatorname{dom} H_{\widehat{V}_j\delta_{S_j}}$ . Moreover, the product rule gives us for  $u \in \operatorname{dom} H_{\widehat{V}_i\delta_{S_i}}$ 

$$\varphi H_{\widehat{V}_j \delta_{S_j}} u = H_{\widehat{V}_j \delta_{S_j}} \varphi u + i(\alpha \cdot \nabla \varphi) u.$$

The upcoming theorem, which is one of the main results of this thesis, shows that for electrostatic and Lorentz scalar interactions, i.e.  $V = \eta I_N + \tau \beta$ ,  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ , the simple condition

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}, \qquad d = \eta^2 - \tau^2, \tag{5.64}$$

guarantees the norm resolvent convergence of  $H_{V_{\varepsilon}}$  for  $\varepsilon \to 0$ .

**Theorem 5.20.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (5.64),  $V_{\varepsilon}$  be defined by (4.3) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, set  $\widetilde{V} = \widetilde{\eta} I_N + \widetilde{\tau} \beta$  with  $(\widetilde{\eta}, \widetilde{\tau}) = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)(\eta, \tau)$  where  $d = \eta^2 - \tau^2$ . Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and there exists an  $\varepsilon_{\operatorname{conv}} > 0$  such that for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2 - r}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ . In particular,  $H_{V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* First, we observe that (5.64),  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  and  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 = 4 \tan\left(\frac{\sqrt{d}}{2}\right)^2$  imply

$$\inf_{x_{\Sigma}\in\Sigma}|\widetilde{d}(x_{\Sigma})-4|>0$$

and thus  $H_{\tilde{V}\delta_{\Sigma}}$  is self-adjoint by Proposition 3.15. Moreover, we note that if  $\Sigma$  is a rotated  $C_b^2$ -graph, then the assertion follows directly from Theorem 4.15, the text in the beginning of Chapter 5 and Proposition 5.18.

In the general setting  $\Sigma \subset \mathbb{R}^{\theta}$  is a special  $C^2$ -surface as in Definition 2.1. The surface  $\Sigma$  is in this case a subset of  $\bigcup_{l=1}^{p} \Sigma_l$  with  $\Sigma_l = \{\kappa_l(x', \zeta_l(x')) : x' \in \mathbb{R}^{\theta-1}\}$  for  $l \in \{1, \ldots, p\}$ . Thus, we prove the general case by reducing it to the case of rotated  $C_b^2$ -graphs. However, V is only defined on  $\Sigma$  and therefore in general only on parts of  $\Sigma_l, l \in \{1, \ldots, p\}$ . Hence, we define suitable extensions of V to  $\Sigma_l$  in the following way: Let  $\hat{V}, \hat{\eta}, \hat{\tau}$  be the  $C_b^1$ -extensions of  $V, \tau$  and  $\eta$  defined by (4.10). Moreover, we set  $V_l := \hat{V} \upharpoonright \Sigma_l, \eta_l := \hat{\eta} \upharpoonright \Sigma_l, \tau_l := \hat{\tau} \upharpoonright \Sigma_l$  and  $d_l := \eta_l^2 - \tau_l^2$  for  $l \in \{1, \ldots, p\}$ . Then, these functions also satisfy

$$\sup_{x_{\Sigma_l}\in\Sigma_l} d_l(x_{\Sigma_l}) < \frac{\pi^2}{4} \qquad \forall l \in \{1,\dots,p\}$$
(5.65)

by construction as  $d = \eta^2 - \tau^2$  satisfies (5.64).

In order to be able to reduce the general case, we choose the  $C^1$ -partition of unity  $\widehat{\varphi}_1, \ldots, \widehat{\varphi}_p \in C_b^1(\mathbb{R}^{\theta})$  for  $\Omega_{\frac{\varepsilon_{\text{tub}}}{2}}$  from Corollary A.5, which fulfils  $\operatorname{supp} \widehat{\varphi}_l \cap \Sigma \subset W_l$ , where  $W_1, \ldots, W_p$  is the open cover of  $\Sigma$  from Definition 2.1. Moreover, let for  $\varepsilon \in (0, \varepsilon_{\text{tub}}) \Omega_{l,\varepsilon}$  be the tubular neighbourhood of  $\Sigma_l$  and  $V_{l,\varepsilon}$  be defined analogously to  $V_{\varepsilon}$  in (4.3). We claim that  $\Omega_{\varepsilon} \cap \operatorname{supp} \widehat{\varphi}_l = \Omega_{\varepsilon,l} \cap \operatorname{supp} \widehat{\varphi}_l$  for all  $\varepsilon \in (0, \varepsilon_{\text{tub}})$  and  $l \in \{1, \ldots, p\}$ . Indeed, if  $x \in \Omega_{\varepsilon} \cap \operatorname{supp} \widehat{\varphi}_l$ , then there exists  $(x_{\Sigma}, t) \in \Sigma \times (-\varepsilon, \varepsilon)$  such that  $x = x_{\Sigma} + t\nu(x_{\Sigma})$ . The equation  $0 \neq \widehat{\varphi}_l(x) = \widehat{\varphi}_l(x_{\Sigma})\varpi(t)$ , see Corollary A.5, implies  $x_{\Sigma} \in \operatorname{supp} \widehat{\varphi}_l \cap \Sigma \subset W_l \cap \Sigma = W_l \cap \Sigma_l$ . Thus,  $x_{\Sigma} \in \Sigma_l$  and  $\nu(x_{\Sigma}) = \nu_l(x_{\Sigma})$ . Consequently,  $x = x_{\Sigma} + t\nu_l(x_{\Sigma}) \in \Omega_{l,\varepsilon} \cap \operatorname{supp} \widehat{\varphi}_l$  and additionally

$$V_{\varepsilon}(x) = V(x_{\Sigma}) \frac{q\left(\frac{t}{\varepsilon}\right)}{\varepsilon} = V_{l}(x_{\Sigma}) \frac{q\left(\frac{t}{\varepsilon}\right)}{\varepsilon} = V_{l,\varepsilon}(x).$$

The reverse inclusion can be shown in exactly the same way. This implies in particular  $V_{l,\varepsilon}\widehat{\varphi}_l = V_{\varepsilon}\widehat{\varphi}_l$  for  $l \in \{1, \ldots, p\}$  and  $\varepsilon \in (0, \varepsilon_{tub})$ . Hence, we get for  $l \in \{1, \ldots, p\}$  and  $\varepsilon \in (0, \varepsilon_{tub})$  the identity

$$H_{V_{l,\varepsilon}}\widehat{\varphi}_{l}u = H_{V_{\varepsilon}}\widehat{\varphi}_{l}u \quad \forall u \in \operatorname{dom} H_{V_{\varepsilon}} = \operatorname{dom} H_{V_{l,\varepsilon}} = H^{1}(\mathbb{R}^{\theta}; \mathbb{C}^{N}),$$

where  $H_{V_{l,\varepsilon}} = H + V_{l,\varepsilon}$  and  $H_{V_{\varepsilon}} = H + V_{\varepsilon}$  with H being the free Dirac operator; cf. (4.4) and Definition 3.2. Additionally to  $\hat{\varphi}_1, \ldots, \hat{\varphi}_p$ , we introduce  $\hat{\varphi}_{p+1} := 1 - \sum_{l=1}^p \hat{\varphi}_l$ . The fact that  $\hat{\varphi}_1, \ldots, \hat{\varphi}_p$  is a partition of unity for  $\Omega_{\frac{\varepsilon_{\text{tub}}}{2}}$  shows that for all  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{2})$  $V_{\varepsilon} \hat{\varphi}_{p+1} = 0$  and thus  $H_{V_{\varepsilon}} \hat{\varphi}_{p+1} u = H \hat{\varphi}_{p+1} u$  for  $u \in H^1(\mathbb{R}^\theta; \mathbb{C}^N)$ . These observations, setting  $H_{V_{p+1,\varepsilon}} = H$  and applying the product rule yields for  $l \in \{1, \ldots, p+1\}$ ,  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{2})$  and  $u \in \text{dom } H_{V_{\varepsilon}} = H^1(\mathbb{R}^\theta; \mathbb{C}^N)$ 

$$\widehat{\varphi}_l H_{V_{\varepsilon}} u = H_{V_{\varepsilon}} \widehat{\varphi}_l u + i(\alpha \cdot \nabla \widehat{\varphi}_l) u = H_{V_{l,\varepsilon}} \widehat{\varphi}_l u + i(\alpha \cdot \nabla \widehat{\varphi}_l).$$
(5.66)

Next, we construct a resolvent for  $H_{V_{\varepsilon}}$  in terms of the operators  $H_{V_{l,\varepsilon}}$ . We use (5.66) to get for  $u \in \text{dom } H_{V_{\varepsilon}}$  and  $z \in \mathbb{C} \setminus \mathbb{R}$ 

$$\left(\sum_{l=1}^{p+1} (H_{V_{l,\varepsilon}} - z)^{-1} \widehat{\varphi}_{l}\right) (H_{V_{\varepsilon}} - z) u$$

$$= \sum_{l=1}^{p+1} (H_{V_{l,\varepsilon}} - z)^{-1} (H_{V_{l,\varepsilon}} - z) \widehat{\varphi}_{l} u + i (H_{V_{l,\varepsilon}} - z)^{-1} (\alpha \cdot \nabla \widehat{\varphi}_{l}) u$$

$$= \sum_{l=1}^{p+1} \widehat{\varphi}_{l} u + i (H_{V_{l,\varepsilon}} - z)^{-1} (\alpha \cdot \nabla \widehat{\varphi}_{l}) u$$

$$= \left(I + \sum_{l=1}^{p+1} i (H_{V_{l,\varepsilon}} - z)^{-1} (\alpha \cdot \nabla \widehat{\varphi}_{l})\right) u.$$
(5.67)

In particular, if  $|\text{Im} z| > \sum_{l=1}^{p+1} \| \alpha \cdot \nabla \widehat{\varphi}_l \|_{L^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})}$ , then

$$I + \sum_{l=1}^{p+1} i(H_{V_{l,\varepsilon}} - z)^{-1} (\alpha \cdot \nabla \widehat{\varphi}_l)$$

is continuously invertible in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and

$$(H_{V_{\varepsilon}} - z)^{-1} = \left(I + \sum_{l=1}^{p+1} i(H_{V_{l,\varepsilon}} - z)^{-1} (\alpha \cdot \nabla \widehat{\varphi}_l)\right)^{-1} \left(\sum_{l=1}^{p+1} (H_{V_{l,\varepsilon}} - z)^{-1} \widehat{\varphi}_l\right)$$
(5.68)

for all  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{2})$ . Next, we find a similar resolvent formula for the resolvent of  $H_{\widetilde{V}\delta_{\Sigma}}$ . Again, it is important to establish relations between  $H_{\widetilde{V}\delta_{\Sigma}}$  and  $H_{\widetilde{V}_l\delta_{\Sigma_l}}$ .

Here,  $\widetilde{V} = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)V$ ,  $\widetilde{V}_{l} = \operatorname{tanc}\left(\frac{\sqrt{d_{l}}}{2}\right)V_{l}$ ,  $l \in \{1, \ldots, p\}$ ,  $H_{\widetilde{V}\delta_{\Sigma}}$  is defined as in Definition 3.12 and  $H_{\widetilde{V}_{l}\delta_{\Sigma_{l}}}$  is defined in the same way as  $H_{\widetilde{V}\delta_{\Sigma}}$  (with  $\Sigma_{l}$  instead of  $\Sigma$  and  $\widetilde{V}_{l}$  instead of V). Furthermore, we set  $\widetilde{V}_{p+1} = 0$  and  $\Sigma_{p+1} = \Sigma$ . In this case  $H_{\widetilde{V}_{p+1}\delta_{\Sigma_{p+1}}}$  coincides with the free Dirac operator H. Note that we already know from the comments at the beginning of the proof that (5.64) and (5.65) imply that  $H_{\widetilde{V}\delta_{\Sigma}}$ and  $H_{\widetilde{V}_{l}\delta_{\Sigma_{l}}}$ ,  $l \in \{1, \ldots, p\}$ , are self-adjoint in  $L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N})$ . Moreover, Lemma 5.19 shows that if  $u \in \operatorname{dom} H_{\widetilde{V}\delta_{\Sigma}}$ , then  $\widehat{\varphi}_{l}u \in \operatorname{dom} H_{\widetilde{V}_{l}\delta_{\Sigma_{l}}}$  and

$$\widehat{\varphi}_l H_{\widetilde{V}\delta_{\Sigma}} u = H_{\widetilde{V}\delta_{\Sigma}} \widehat{\varphi}_l u + i(\alpha \cdot \nabla \widehat{\varphi}_l) u = H_{\widetilde{V}_l \delta_{\Sigma_l}} \widehat{\varphi}_l u + i(\alpha \cdot \nabla \widehat{\varphi}_l) u$$
(5.69)

for all  $l \in \{1, \ldots, p+1\}$ . Using (5.69) one argues with the same steps as in (5.67) that for  $u \in \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  and  $z \in \mathbb{C} \setminus \mathbb{R}$ 

$$\Big(\sum_{l=1}^{p+1} (H_{\widetilde{V}_l\delta_{\Sigma_l}} - z)^{-1}\widehat{\varphi}_l\Big)(H_{\widetilde{V}\delta_{\Sigma}} - z)u = \Big(I + \sum_{l=1}^{p+1} i(H_{\widetilde{V}_l\delta_{\Sigma_l}} - z)^{-1}(\alpha \cdot \nabla\widehat{\varphi}_l)\Big)u.$$

Hence, if  $|\operatorname{Im} z| > \sum_{l=1}^{p+1} \|\alpha \cdot \nabla \widehat{\varphi}_l\|_{L^{\infty}(\mathbb{R}^{\theta};\mathbb{C}^{N\times N})}$ , then  $I + \sum_{l=1}^{p+1} i(H_{\widetilde{V}_l\delta_{\Sigma_l}} - z)^{-1}(\alpha \cdot \nabla \widehat{\varphi}_l)$  is continuously invertible in  $L^2(\mathbb{R}^{\theta};\mathbb{C}^N)$  and

$$(H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1} = \left(I + \sum_{l=1}^{p+1} i(H_{\widetilde{V}_l\delta_{\Sigma_l}}-z)^{-1}(\alpha \cdot \nabla\widehat{\varphi}_l)\right)^{-1} \left(\sum_{l=1}^{p+1} (H_{\widetilde{V}_l\delta_{\Sigma_l}}-z)^{-1}\widehat{\varphi}_l\right).$$
(5.70)

Thus, if  $|\operatorname{Im} z| > \sum_{l=1}^{p+1} \|\alpha \cdot \nabla \widehat{\varphi}_l\|_{L^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})}$ , then the assertion follows from comparing (5.68) and (5.70) with one another, and the case of the rotated  $C_b^2$ -graph. Moreover, we can get rid of the assumption  $|\operatorname{Im} z| > \sum_{l=1}^{p+1} \|\alpha \cdot \nabla \widehat{\varphi}_l\|_{L^{\infty}(\mathbb{R}^{\theta}; \mathbb{C}^{N \times N})}$  by applying the resolvent identity

$$(H_{\widetilde{V}\delta_{\Sigma}} - w)^{-1} - (H_{V_{\varepsilon}} - w)^{-1} = (I + (w - z)(H_{\widetilde{V}\delta_{\Sigma}} - w)^{-1}) \cdot ((H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} - (H_{V_{\varepsilon}} - z)^{-1})(I + (w - z)(H_{V_{\varepsilon}} - w)^{-1}) \quad \forall z, w \in \mathbb{C} \setminus \mathbb{R}.$$

This completes the proof of Theorem 5.20.

In the next theorem we add a magnetic term to V and show that then  $H_{V_{\varepsilon}}$  also converges to a Dirac operator with  $\delta$ -shell potential with electrostatic and Lorentz scalar interactions. However, in the following theorem the rescaling of the interaction strengths is different.

**Theorem 5.21.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta + \pi(\alpha \cdot \nu)$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ and the unit normal vector  $\nu$  on  $\Sigma$ ,  $d = \eta^2 - \tau^2$  satisfy

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4} \quad \text{and} \quad \inf_{x_{\Sigma}\in\Sigma} |d(x_{\Sigma})| > 0,$$

 $V_{\varepsilon}$  be defined by (4.3) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, set

$$(\widetilde{\eta},\widetilde{\tau}) := \frac{-2}{\sqrt{d}\tan\left(\frac{\sqrt{d}}{2}\right)}(\eta,\tau)$$

and  $\widetilde{V} := \widetilde{\eta}I_N + \widetilde{\tau}\beta$ . Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and there exists an  $\varepsilon_{\text{conv}} > 0$  such that for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ . In particular,  $H_{V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* We start by calculating

$$\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = \frac{4(\eta^2 - \tau^2)}{d\tan\left(\frac{\sqrt{d}}{2}\right)^2} = \frac{4}{\tan\left(\frac{\sqrt{d}}{2}\right)^2}.$$
(5.71)

Thus, the assumptions regarding d imply that the  $\inf_{x_{\Sigma} \in \Sigma} |d(x_{\Sigma}) - 4| > 0$  and hence  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint according to Proposition 3.15.

For the remaining proof we use an idea from [24, Section 8]. This idea makes use of the fact stated in Proposition 3.15 (i) that the two Dirac operators  $H_{\tilde{V}\delta_{\Sigma}}$  and  $H_{-4(\tilde{V}/\tilde{d})\delta_{\Sigma}}$  are unitarily equivalent. More precisely,  $H_{\tilde{V}\delta_{\Sigma}} = UH_{-4(\tilde{V}/\tilde{d})\delta_{\Sigma}}U$ , where Uis the self-adjoint unitary multiplication operator in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  which is induced by the function  $w = \chi_{\Omega_+} - \chi_{\Omega_-}$ . If we set  $F := \eta I_N + \tau \beta$  and  $\tilde{F} := \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)F$ , we get in the current setting

$$\widetilde{V} = \frac{-2}{\sqrt{d}\tan\left(\frac{\sqrt{d}}{2}\right)}F$$

and by (5.71)

$$\frac{-4}{\widetilde{d}}\widetilde{V} = \frac{2\tan\left(\frac{\sqrt{d}}{2}\right)}{\sqrt{d}}F = \tan\left(\frac{\sqrt{d}}{2}\right)F = \widetilde{F}.$$

In particular,

$$UH_{\widetilde{F}\delta_{\Sigma}}U = H_{\widetilde{V}\delta_{\Sigma}}.$$

Note also that Theorem 5.20 shows that  $H_{F_{\varepsilon}}$ , where  $F_{\varepsilon}$  is analogously defined as  $V_{\varepsilon}$  in (4.3) with F instead of V, converges to  $H_{\tilde{F}\delta_{\Sigma}}$  in the norm resolvent sense. In this proof we find unitary multiplication operators  $W_{\varepsilon}$  such that

$$W_{\varepsilon}^* H_{F_{\varepsilon}} W_{\varepsilon} = H_{V_{\varepsilon}}$$

and  $W_{\varepsilon} \to U$  for  $\varepsilon \to 0$  in a suitable sense. Furthermore, using the convergence properties of these operators and  $H_{F_{\varepsilon}}$  we show that  $H_{V_{\varepsilon}}$  converges in the norm resolvent

sense to  $H_{\tilde{V}\delta_{\Sigma}}$ . Having explained the ideas, we start the main part of the proof by defining for  $\varepsilon \in (0, \varepsilon_{tub})$  the function

$$w_{\varepsilon}: \mathbb{R}^{\theta} \to \mathbb{C}, \qquad w_{\varepsilon}(x) := \begin{cases} 1, & x \in \Omega_{+} \setminus \Omega_{\varepsilon}, \\ e^{i\pi \int_{-1}^{t/\varepsilon} q(s) \, ds}, & x = x_{\Sigma} + t\nu(x_{\Sigma}) \in \Omega_{\varepsilon}, \\ -1, & x \in \Omega_{-} \setminus \Omega_{\varepsilon}. \end{cases}$$

This function is well-defined according to Proposition 2.12. Moreover,  $\int_{-1}^{1} q(s) ds = 1$ shows that  $w_{\varepsilon}$  is continuous. We define  $W_{\varepsilon}$  to be the unitary multiplication operator in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  induced by  $w_{\varepsilon}$ . Next, we show  $w_{\varepsilon} \in W^1_{\infty}(\mathbb{R}^{\theta})$ . To do so, let us fix  $x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}$  and  $x' \in \mathbb{R}^{\theta-1}$  such that  $\varkappa_l(x') = x_{\Sigma}$  for a  $l \in \{1, \ldots, p\}$ . This implies  $x = \iota_l(x', t)$  with  $\iota_l$  from Definition 2.7. In the proof of Lemma 4.3 we showed that  $\iota_l : \mathbb{R}^{\theta-1} \times \mathbb{R} \to \mathbb{R}^{\theta}$  is locally around (x', t) a diffeomorphism. Moreover, using  $w_{\varepsilon} \circ \iota_l(x', t) = e^{i\pi \int_{-1}^{t/\varepsilon} q(s) ds}$ , the chain rule and formula (4.12) for  $(D\iota_l)(x', t)$  yields

$$\nabla w_{\varepsilon}(x) = \left( ((D\iota_l)(x',t))^{-1} \right)^T \nabla (w_{\varepsilon} \circ \iota_l)(x',t) \\ = \left( * | \nu_l(x') \right) \frac{i\pi q\left(\frac{t}{\varepsilon}\right)}{\varepsilon} e^{i\pi \int_{-1}^{t/\varepsilon} q(s) \, ds} e_{\theta} \\ = \nu(x_{\Sigma}) \frac{i\pi q\left(\frac{t}{\varepsilon}\right)}{\varepsilon} w_{\varepsilon}(x),$$

where  $e_{\theta}$  is the  $\theta$ -th Euclidean unit vector in  $\mathbb{R}^{\theta}$ . Hence,

$$\nabla w_{\varepsilon}(x) = \begin{cases} \nu(x_{\Sigma}) \frac{i\pi q\left(\frac{t}{\varepsilon}\right)}{\varepsilon} w_{\varepsilon}(x), & x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}, \\ 0, & x \notin \Omega_{\varepsilon}, \end{cases}$$

and  $w_{\varepsilon} \in W^1_{\infty}(\mathbb{R}^{\theta})$ . These considerations,  $\frac{1}{w_{\varepsilon}} = \overline{w_{\varepsilon}} \in W^1_{\infty}(\mathbb{R}^{\theta})$ , and the definition of  $V_{\varepsilon}$  in (4.3) show dom  $W^*_{\varepsilon}H_{F_{\varepsilon}}W_{\varepsilon} = \operatorname{dom} H_{V_{\varepsilon}} = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and

$$W_{\varepsilon}^{*}H_{F_{\varepsilon}}W_{\varepsilon} = H_{F_{\varepsilon}} - i\overline{w_{\varepsilon}}(\alpha \cdot \nabla w_{\varepsilon}) = H_{V_{\varepsilon}};$$

cf. [24, Section 8, below the proof of Theorem 2.6]. We note that  $w_{\varepsilon}$  converges pointwise to  $w = \chi_{\Omega_+} - \chi_{\Omega_-}$  and therefore  $W_{\varepsilon}$  converges in the strong sense to the operator U. In addition, for  $\varepsilon \in (0, \varepsilon_{\text{tub}})$  the estimate

$$\|(W^*_{\varepsilon} - U)u\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N)} \le \varepsilon^{1/2} C \|u\|_{H^1(\mathbb{R}^{\theta}\setminus\Sigma;\mathbb{C}^N)} \quad \forall u \in H^1(\mathbb{R}^{\theta}\setminus\Sigma)$$
(5.72)

is also valid. We postpone the verification of this fact to the end of the proof. Moreover,  $(H_{\widetilde{F}\delta_{\Sigma}}-z)^{-1}$  also acts as a bounded operator from  $L^2(\mathbb{R}^{\theta};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta}\setminus\Sigma;\mathbb{C}^N)$ . Indeed, one can show this by using dom  $H_{\widetilde{F}\delta_{\Sigma}} \subset H^1(\mathbb{R}^{\theta}\setminus\Sigma;\mathbb{C}^N)$ , see Definition 3.12, the boundedness of  $(H_{\widetilde{F}\delta_{\Sigma}}-z)^{-1}$  acting as an operator in  $L^2(\mathbb{R}^{\theta};\mathbb{C}^N)$ , the continuous embedding of  $H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N)$  in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and the closed graph theorem. Thus, (5.72) gives us for  $\varepsilon \in (0, \varepsilon_{tub})$ 

$$\left\| (W_{\varepsilon}^* - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \right\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N) \to L^2(\mathbb{R}^{\theta};\mathbb{C}^N)} \le C\varepsilon^{1/2}.$$

This observation, the norm resolvent convergence of  $H_{F_{\varepsilon}}$  (see Theorem 5.20) and the fact that U and  $W_{\varepsilon}$  are unitary operators let us estimate for  $\varepsilon \in (0, \varepsilon_{\text{conv}})$  (with  $\varepsilon_{\text{conv}} > 0$  from Theorem 5.20)

$$\begin{split} \left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &= \| W_{\varepsilon}^{*}(H_{F_{\varepsilon}} - z)^{-1} W_{\varepsilon} - U(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} U \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq \| W_{\varepsilon}^{*}((H_{F_{\varepsilon}} - z)^{-1} - (H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1}) W_{\varepsilon} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} W_{\varepsilon} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| U(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1}(W_{\varepsilon} - U) \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &= \| (H_{F_{\varepsilon}} - z)^{-1} - (H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} (W_{\varepsilon} - U) \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - \overline{z})^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - \overline{z})^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - \overline{z})^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \| (W_{\varepsilon}^{*} - U)(H_{\widetilde{F}\delta_{\Sigma}} - \overline{z})^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq C \varepsilon^{1/2-r} + C \varepsilon^{1/2} \\ &\leq C \varepsilon^{1/2-r}, \end{aligned}$$

where  $r \in (0, \frac{1}{2})$ . Hence, it only remains to prove (5.72). We start by choosing  $u = u_+ \oplus u_-$  with  $u_{\pm} \in C_0^{\infty}(\overline{\Omega_{\pm}}; \mathbb{C}^N)$  and get for  $\varepsilon \in (0, \varepsilon_{\text{tub}})$ 

$$\begin{aligned} \|(W_{\varepsilon}^* - U)u\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N)}^2 &= \int_{\Omega_{\varepsilon}} |(\overline{w_{\varepsilon}(x)} - \chi_{\Omega_+}(x) + \chi_{\Omega_-}(x))u(x)|^2 \, dx \\ &\leq 4 \int_{\Omega_{\varepsilon}} |u(x)|^2 \, dx. \end{aligned}$$

Moreover, by Corollary 2.10 (where we use  $\varepsilon_{tub} < \varepsilon_{\iota}$ ) and Proposition 2.12 we obtain

$$\begin{split} \int_{\Omega_{\varepsilon}} |u(x)|^2 \, dx &\leq C \int_{-\varepsilon}^{\varepsilon} \int_{\Sigma} |u(x_{\Sigma} + t\nu(x_{\Sigma}))|^2 \, d\sigma(x_{\Sigma}) \, dt \\ &= C\varepsilon \Big( \int_{-1}^{0} \int_{\Sigma} |u_+(x_{\Sigma} + t\varepsilon\nu(x_{\Sigma}))|^2 \, d\sigma(x_{\Sigma}) \, dt \\ &+ \int_{0}^{1} \int_{\Sigma} |u_-(x_{\Sigma} + t\varepsilon\nu(x_{\Sigma}))|^2 \, d\sigma(x_{\Sigma}) \, dt \Big). \end{split}$$

Next, we estimate the term  $\int_{-1}^{0} \int_{\Sigma} |u_{+}(x_{\Sigma} + t\varepsilon\nu(x_{\Sigma}))|^{2} d\sigma(x_{\Sigma}) dt$ . Note that the smoothness of u implies that for  $t \in (-1, 0)$  the function

$$\Sigma \ni x_{\Sigma} \mapsto u_+(x_{\Sigma} + t\varepsilon\nu(x_{\Sigma}))$$

coincides with the trace of the function  $\tau_{\varepsilon t}^{\Omega_{+}} u$ , where  $\tau_{\varepsilon t}^{\Omega_{\pm}}$  is the shift operator introduced in (4.18) (for  $\delta = \varepsilon t$ ). Thus,

$$\int_{-1}^{0} \int_{\Sigma} |u_{+}(x_{\Sigma} + t\varepsilon\nu(x_{\Sigma}))|^{2} d\sigma(x_{\Sigma}) dt = \int_{-1}^{0} \int_{\Sigma} |\mathbf{t}_{\Sigma}\tau_{\varepsilon t}^{\Omega_{+}}u(x_{\Sigma})|^{2} d\sigma(x_{\Sigma}) dt.$$

Now, Proposition 2.3 and Corollary 4.5 let us estimate this term in the following way:

$$\int_{-1}^{0} \int_{\Sigma} |\boldsymbol{t}_{\Sigma} \tau_{\varepsilon t}^{\Omega_{+}} u_{+}(x_{\Sigma})|^{2} d\sigma(x_{\Sigma}) dt = \int_{-1}^{0} \|\boldsymbol{t}_{\Sigma} \tau_{\varepsilon t}^{\Omega_{+}} u_{+}\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} dt$$

$$\leq \int_{-1}^{0} \|\boldsymbol{t}_{\Sigma} \tau_{\varepsilon t}^{\Omega_{+}} u_{+}\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})}^{2} dt$$

$$\leq C \int_{-1}^{0} \|\tau_{\varepsilon t}^{\Omega_{+}} u_{+}\|_{H^{1}(\Omega_{+};\mathbb{C}^{N})}^{2} dt$$

$$\leq C \|u_{+}\|_{H^{1}(\Omega_{+};\mathbb{C}^{N})}^{2}.$$

Therefore,

$$\int_{-1}^{0} \int_{\Sigma} |u_{+}(x_{\Sigma} + t\nu(x_{\Sigma}))|^{2} \, d\sigma(x_{\Sigma}) \, dt \leq C ||u_{+}||^{2}_{H^{1}(\Omega_{+};\mathbb{C}^{N})};$$

in the same way one gets

$$\int_{0}^{1} \int_{\Sigma} |u_{-}(x_{\Sigma} + t\nu(x_{\Sigma}))|^{2} \, d\sigma(x_{\Sigma}) \, dt \leq C ||u_{-}||^{2}_{H^{1}(\Omega_{-};\mathbb{C}^{N})}.$$

This implies (5.72) for  $u \in C_0^{\infty}(\overline{\Omega_+}; \mathbb{C}^N) \oplus C_0^{\infty}(\overline{\Omega_-}; \mathbb{C}^N)$ , which is a dense subspace of  $H^1(\mathbb{R}^{\theta} \setminus \Sigma; \mathbb{C}^N) = H^1(\Omega_+; \mathbb{C}^N) \oplus H^1(\Omega_-; \mathbb{C}^N)$ ; see e.g. [54, Chapter 3]. Hence, (5.72) is valid.

An immediate consequence of the two previous theorems is the following corollary.

**Corollary 5.22.** Let q be as in (5.1),  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$  with  $\widetilde{\eta}, \widetilde{\tau} \in C_b^1(\Sigma; \mathbb{R}), \ \widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2$  satisfy either

$$\sup_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| < 4 \quad \text{or} \quad \inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| > 4,$$

and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, let the interaction strengths  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  be given by

$$(\eta,\tau) := \begin{cases} \frac{2 \arctan\left(\frac{\sqrt{d}}{2}\right)}{\sqrt{d}}(\widetilde{\eta},\widetilde{\tau}), & \sup_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| < 4, \\ \frac{-2 \arctan\left(\frac{2}{\sqrt{d}}\right)}{\sqrt{d}}(\widetilde{\eta},\widetilde{\tau}), & \inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| > 4, \end{cases} \\ V := \begin{cases} \eta I_{N} + \tau\beta, & \sup_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| < 4, \\ \eta I_{N} + \tau\beta + \pi(\alpha \cdot \nu), & \inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| > 4, \end{cases} \end{cases}$$

and  $V_{\varepsilon}$  be defined by (4.3). Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and there exists an  $\varepsilon_{\text{conv}} > 0$  such that for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conv}})$ . In particular,  $H_{V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* By definition, we have

$$d = \eta^2 - \tau^2 = \begin{cases} 4 \arctan\left(\frac{\sqrt{\tilde{d}}}{2}\right)^2, & \sup_{x_{\Sigma} \in \Sigma} |\tilde{d}(x_{\Sigma})| < 4, \\ 4 \arctan\left(\frac{2}{\sqrt{\tilde{d}}}\right)^2, & \inf_{x_{\Sigma} \in \Sigma} |\tilde{d}(x_{\Sigma})| > 4. \end{cases}$$

Hence, as  $\arctan(t)^2 < \frac{\pi^2}{16}$  for  $t \in [0, 1)$  and  $t \in i\mathbb{R}$ , the inequality  $\sup_{x_{\Sigma} \in \Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}$  holds in both cases. Thus, if  $|\tilde{d}(x_{\Sigma})| < 4$ , then the assertion follows from Theorem 5.20 and  $\inf_{x_{\Sigma} \in \Sigma} |\tilde{d}(x_{\Sigma})| > 4$ , then the assertion is a consequence of Theorem 5.21.

Corollary 5.22 is particularly interesting in the case that  $\tilde{\eta}, \tilde{\tau} \in \mathbb{R}$ . Then, the conditions of Corollary 5.22 reduce to  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 \neq \pm 4$ . The two excluded cases  $\tilde{d} = -4$ and  $\tilde{d} = 4$  are called the confinement case and the critical case, respectively; cf. the discussion below (1.3). We show in Corollary 7.5 that by an additional scaling of the strongly localized potentials one can also approximate Dirac operators with  $\delta$ -shell potentials in the confinement case and therefore all Dirac operators with  $\delta$ -shell potentials and noncritical electrostatic and Lorentz scalar interaction strengths can be approximated in the norm resolvent sense by Dirac operators with strongly localized potentials. Moreover, we provide a counterexample in Theorem 6.1 in the critical case.

## 6 Counterexamples

In Theorem 5.20 we showed that for  $V = \eta I_N + \tau \beta$ ,  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ ,  $d = \eta^2 - \tau^2$ , the condition

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < \frac{\pi^2}{4}$$

guarantees that  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense for  $\varepsilon \to 0$  to  $H_{\widetilde{V}\delta_{\Sigma}}$  with  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$ , where  $(\widetilde{\eta}, \widetilde{\tau}) = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)(\eta, \tau)$ . In this chapter we show that this condition is optimal by providing counterexamples in the case of constant interaction strengths, i.e.  $\eta, \tau \in \mathbb{R}$ . As already mentioned in the introduction, we have to consider the case  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 = 4$ , which is the so-called critical case, and  $\tilde{d} \neq 4$ , separately. From Proposition 3.15 we know that  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint if  $d \neq 4$ . However, if  $d \neq 4$ , then Proposition 3.15 is not applicable and  $H_{\tilde{V}\delta_{\Sigma}}$  is generally not selfadjoint. Thus, in the latter case it is not meaningful to ask the question whether  $H_{V_{\varepsilon}}$ converges in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$ . However, under certain assumptions regarding  $\Sigma$ , the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is essentially self-adjoint. The first counterexample given in Theorem 6.1 deals with this situation. In particular, we show the following: If  $\Sigma$  is a compact and smooth hypersurface,  $d \geq \frac{\pi^2}{4}$  and  $\tilde{d} = 4$  (i.e.  $d = (2k+1)^2 \frac{\pi^2}{4}$ ,  $k \in \mathbb{N}_0$ , then  $H_{V_{\varepsilon}}$  does not converge in the norm resolvent sense to the closure of  $H_{\widetilde{V}\delta_{\Sigma}}$ . Afterwards, we consider the case where  $\Sigma$  is an affine hyperplane in  $\mathbb{R}^{\theta}$ and find out that in this situation  $\sigma(H_{V_{\varepsilon}}) = \mathbb{R}$  if  $\eta, \tau \in \mathbb{R}$  are chosen such that  $d = \eta^2 - \tau^2 > \frac{\pi^2}{4}$  and  $\varepsilon > 0$  is sufficiently small. Furthermore, by combining this result with known spectral properties of  $H_{\tilde{V}\delta_{\Sigma}}$  we show in Corollary 6.6 that if  $\Sigma$  is an affine hyperplane,  $d \ge \frac{\pi^2}{4}$ ,  $d \ne (2k+1)^2 \pi^2$  for  $k \in \mathbb{N}_0$  and  $\tilde{d} \ne 4$ , then  $H_{V_{\varepsilon}}$  does not converge to  $H_{\tilde{V}\delta_{\Sigma}}$  in the norm resolvent sense. Finally, we transfer this result in Theorem 6.7 to the case where  $\Sigma$  is a special C<sup>2</sup>-surface which contains a flat part. The mentioned counterexamples are particularly interesting since it is known in various situations which are included in the counterexamples that  $H_{V_{\varepsilon}}$  converges in the strong resolvent sense to  $H_{\widetilde{V}\delta_{\Sigma}}$ ; see [18, Theorem 7.2], [24, Theorem 2.6] and [74, Theorem 2.1].

**Theorem 6.1.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a compact  $C^{\infty}$ -smooth hypersurface, q be as in (4.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ ,  $d = \eta^2 - \tau^2 \geq \frac{\pi^2}{4}$  such that  $d \neq (2k+1)^2 \pi^2$  for  $k \in \mathbb{N}_0$  is fulfilled and  $V_{\varepsilon}$  be defined by (4.3). Moreover, let  $\widetilde{V} = \widetilde{\eta} I_N + \widetilde{\tau} \beta$  with  $(\widetilde{\eta}, \widetilde{\tau}) = \operatorname{tanc}(\frac{\sqrt{d}}{2})(\eta, \tau)$  and  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = 4$ . Then,  $H_{\widetilde{V}\delta_{\Sigma}}$  is not self-adjoint but essentially self-adjoint. Furthermore,  $H_{V_{\varepsilon}}$  does not converge in the norm resolvent sense to the closure of  $H_{\widetilde{V}\delta_{\Sigma}}$  for  $\varepsilon \to 0$ . Proof. The claims regarding the (essential) self-adjointness follow from [13, Theorem 4.11] for  $\theta = 2$  and from [19, Theorem 3.1 (ii)] for  $\theta = 3$ . Thus, it only remains to prove the non-convergence statement. By Proposition 2.24 we can w.l.o.g. assume m > 0. If  $\Sigma$  is compact, then supp  $V_{\varepsilon} \subset \Omega_{\varepsilon}$  is compact. Hence, according to [54, Theorem 3.27 (ii)]  $V_{\varepsilon}$  induces a compact operator from  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . Moreover, we know from Proposition 3.3 (iii) that  $(H - z)^{-1}$  is bounded acting as an operator from  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ . In turn, the resolvent difference

$$(H-z)^{-1} - (H_{V_{\varepsilon}} - z)^{-1} = (H_{V_{\varepsilon}} - z)^{-1} V_{\varepsilon} (H-z)^{-1}$$

is compact in  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , which shows  $\sigma_{\text{ess}}(H_{V_{\varepsilon}}) = \sigma_{\text{ess}}(H) = (-\infty, -m] \cup [m, \infty)$ . Equipped with this preliminary observation, we now prove the claim about the nonconvergence by contradiction. We assume that  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense to  $\overline{H_{V_{\delta_{\Sigma}}}}$  for  $\varepsilon \to 0$ . Then, Proposition 2.25 (ii) yields

$$\sigma_{\rm ess}(\overline{H_{\widetilde{V}\delta_{\Sigma}}}) = (-\infty, -m] \cup [m, \infty).$$

Furthermore, [13, Theorem 1.2 and Theorem 1.3] and [20] give us  $-\frac{\tilde{\tau}}{\tilde{\eta}}m \in \sigma_{\text{ess}}(\overline{H_{\tilde{V}\delta_{\Sigma}}})$ . However, in the current case  $\tilde{\eta}^2 - \tilde{\tau}^2 = 4$  and therefore  $-\frac{\tilde{\tau}}{\tilde{\eta}}m \in (-m,m)$  which leads to a contradiction.

Next, we consider the case  $\Sigma = \kappa(\mathbb{R}^{\theta-1} \times \{y_0\}) = \Sigma_{y_0,\kappa}$  with  $\kappa \in \mathrm{SO}(\theta)$  and  $y_0 \in \mathbb{R}$ , i.e.  $\Sigma$  is an affine hyperplane in  $\mathbb{R}^{\theta}$ . Moreover, we assume  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$  such that  $d = \eta^2 - \tau^2 > \frac{\pi^2}{4}$ . We show in Theorem 6.5 that under this set of assumptions  $\sigma(H_{V_{\varepsilon}}) = \mathbb{R}$  for  $\varepsilon > 0$  sufficiently small. In Corollary 6.6 we utilize this knowledge to show that in this situation  $H_{V_{\varepsilon}}$  does not converge in the norm resolvent sense to  $H_{\widetilde{V}\delta_{\Sigma}}$ .

We start by applying the coordinate transformation  $\tilde{x}(x) = \kappa^T x - y_0 e_{\theta}$ , where  $x \in \mathbb{R}^{\theta}$ and  $e_{\theta}$  is the  $\theta$ -th Euclidean unit vector in  $\mathbb{R}^{\theta}$ , to  $H_{V_{\varepsilon}}$ . This transformation turns the operator  $H_{V_{\varepsilon}}$  into the unitarily equivalent Dirac operator

$$\begin{aligned} H_{V_{\varepsilon}}u(\widetilde{x}) &= H_{V_{\varepsilon}}u(\widetilde{x}(x)) \\ &= -i\alpha \cdot \nabla_{x}u(\widetilde{x}(x)) + m\beta u(\widetilde{x}(x)) + V_{\varepsilon}(x)u(\widetilde{x}(x)) \\ &= -i\widetilde{\alpha} \cdot \nabla_{\widetilde{x}}u(\widetilde{x}) + m\beta u(\widetilde{x}) + V_{\varepsilon}(x)u(\widetilde{x}), \quad \widetilde{x} = \widetilde{x}(x) \in \mathbb{R}^{\theta}, \end{aligned}$$

for  $u \in \text{dom } \widetilde{H}_{V_{\varepsilon}} = H_{V_{\varepsilon}} = H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , where  $\widetilde{\alpha} = (\widetilde{\alpha}_1, \dots, \widetilde{\alpha}_{\theta})$  is defined as in (5.21), i.e.  $\widetilde{\alpha} = (\widetilde{\alpha}_1, \dots, \widetilde{\alpha}_{\theta})$  with  $\widetilde{\alpha}_l = \alpha \cdot \kappa e_j, j \in \{1, \dots, \theta\}$ , where  $e_j$  is the *j*-th Euclidean unit vector. Furthermore, for  $\widetilde{x} = (\widetilde{x}', \widetilde{x}_{\theta}) = \widetilde{x}(x) \in \mathbb{R}^{\theta}$  we get

$$x = \kappa \widetilde{x} + y_0 \kappa e_\theta = \kappa(\widetilde{x}', y_0) + \widetilde{x}_\theta \kappa e_\theta$$
, where  $\kappa(\widetilde{x}', y_0) \in \Sigma$  and  $\kappa e_\theta \perp \Sigma$ .

In particular, combining this observation with the definition of  $V_{\varepsilon}$  in (4.3), and introducing  $\tilde{q}$  as the zero extension of q to  $\mathbb{R}$  with  $q \in L^{\infty}((-1,1);\mathbb{R})$  as in (4.1), we obtain

$$V_{\varepsilon}(x) = V \frac{\widetilde{q}(\frac{x_{\theta}}{\varepsilon})}{\varepsilon} \quad \text{for } \widetilde{x} = (\widetilde{x}', \widetilde{x}_{\theta}) = \widetilde{x}(x) \in \mathbb{R}^{\theta}.$$
(6.1)

Hence,

$$\dim H_{V_{\varepsilon}} = H^{1}(\mathbb{R}^{\theta}; \mathbb{C}^{N}), \widetilde{H}_{V_{\varepsilon}}u(\widetilde{x}) = -i\widetilde{\alpha} \cdot \nabla_{\widetilde{x}}u(\widetilde{x}) + m\beta u(\widetilde{x}) + V\frac{\widetilde{q}(\frac{\widetilde{x}_{\theta}}{\varepsilon})}{\varepsilon}u(\widetilde{x}), \quad \widetilde{x} = (\widetilde{x}', \widetilde{x}_{\theta}) \in \mathbb{R}^{\theta}.$$

$$(6.2)$$

Applying the Fourier transform with respect to the first  $\theta - 1$  variables, which is denoted by  $\mathcal{F}_1$  and defined in Section 2.1 (xvii), and identifying  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  with the Bochner space  $L^2(\mathbb{R}^{\theta-1}; L^2(\mathbb{R}; \mathbb{C}^N))$ , see Proposition 2.18 (iii), we get similarly as in [18, eq. (2.3) and the text below] that  $\mathcal{F}_1 \widetilde{H}_{V_{\varepsilon}} \mathcal{F}_1^{-1}$  is unitarily equivalent to the decomposable operator in  $L^2(\mathbb{R}^{\theta-1}; L^2(\mathbb{R}; \mathbb{C}^N))$  which is induced by the mapping

$$\mathbb{R}^{\theta-1} \ni \xi' \mapsto H_{V_{\varepsilon},\xi'}$$

with

$$H_{V_{\varepsilon},\xi'} := \widetilde{\alpha}' \cdot \xi' - i\widetilde{\alpha}_{\theta} \frac{d}{d\widetilde{x}_{\theta}} + m\beta + V \frac{\widetilde{q}(\frac{\cdot}{\varepsilon})}{\varepsilon}, \quad \text{dom} \, H_{V_{\varepsilon},\xi'} := H^1(\mathbb{R}; \mathbb{C}^N), \tag{6.3}$$

for  $\xi' \in \mathbb{R}^{\theta-1}$ ; cf. Proposition 2.20. The notation  $\widetilde{\alpha}' \cdot \xi'$  was introduced in (5.22). By Proposition 2.20 we can analyse the spectrum of  $H_{V_{\varepsilon}}$  by studying the spectrum of  $H_{V_{\varepsilon},\xi'}$  for  $\xi' \in \mathbb{R}^{\theta-1}$ . Next, we introduce the operators

$$H_{0,\xi'} := \widetilde{\alpha}' \cdot \xi' - i\widetilde{\alpha}_{\theta} \frac{d}{d\widetilde{x}_{\theta}} + m\beta, \qquad \operatorname{dom} H_{0,\xi'} := H^{1}(\mathbb{R}; \mathbb{C}^{N}),$$
$$V_{R}^{\varepsilon} : L^{2}(\mathbb{R}; \mathbb{C}^{N}) \to L^{2}((-1,1); \mathbb{C}^{N}), \qquad V_{R}^{\varepsilon} u(\widetilde{x}_{\theta}) := \frac{1}{\sqrt{\varepsilon}} u(\varepsilon \widetilde{x}_{\theta}),$$

and

$$V_L^{\varepsilon}: L^2((-1,1); \mathbb{C}^N) \to L^2(\mathbb{R}; \mathbb{C}^N), \qquad V_L^{\varepsilon} u(\widetilde{x}_{\theta}) := \begin{cases} \frac{1}{\sqrt{\varepsilon}} u\left(\frac{\widetilde{x}_{\theta}}{\varepsilon}\right), & |\widetilde{x}_{\theta}| < \varepsilon, \\ 0, & |\widetilde{x}_{\theta}| \ge \varepsilon. \end{cases}$$

Then,

$$H_{V_{\varepsilon},\xi'} = H_{0,\xi'} + V_L^{\varepsilon} V q V_R^{\varepsilon}.$$

Let us start analysing  $H_{V_{\varepsilon},\xi'}$  by studying the operator  $H_{0,\xi'}$ . Thereby, we proceed in a similar way as in [18, Section 2]. Applying the (one-dimensional) Fourier transform shows that  $H_{0,\xi'}$  is unitarily equivalent to the matrix multiplication operator induced by the matrix-valued function

$$M_{\xi'}: \mathbb{R} \to \mathbb{C}^{N \times N}, \qquad M_{\xi'}(\xi_{\theta}) = \widetilde{\alpha} \cdot (\xi', \xi_{\theta}) + m\beta$$

This implies that the operator  $H_{0,\xi'}$  is self-adjoint. The spectrum of  $H_{0,\xi'}$  is given by the closure of the image of the eigenvalue curves corresponding to  $M_{\xi'}$ ; see [36, Proposition 1]. Using the rules for the Dirac matrices from (5.23) one concludes

$$(M_{\xi'}(\xi_{\theta}))^2 = (|\xi'|^2 + \xi_{\theta}^2 + m^2)I_N$$

for  $\xi = (\xi', \xi_{\theta}) \in \mathbb{R}^{\theta}$ . Moreover, by the structure of the Dirac matrices,  $M_{\xi'}(\xi_{\theta})$ cannot be a multiple of  $I_N$  if  $\xi \neq 0$  or  $m \neq 0$ ; cf. Definition 3.1 and (5.21). Thus, the eigenvalues of  $M_{\xi'}(\xi_{\theta})$  are given by  $\pm \sqrt{|\xi'|^2 + \xi_{\theta}^2 + m^2}$ , and hence the spectrum of  $H_{0,\xi'}$  is given by  $(-\infty, -\sqrt{|\xi'|^2 + m^2}] \cup [\sqrt{|\xi'|^2 + m^2}, \infty)$ . Applying the rules for the Dirac matrices again yields

$$H_{0,\xi'}^2 = \left(-\frac{d^2}{d\tilde{x}_{\theta}^2} + |\xi'|^2 + m^2\right) I_N, \quad \text{dom} \ H_{0,\xi'}^2 = H^2(\mathbb{R}; \mathbb{C}^N).$$

It is well known that

$$(H_{0,\xi'}^2 - z^2)^{-1} f(t) = \int_{-\infty}^{\infty} \frac{i e^{i|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}}}{2\sqrt{z^2 - m^2 - |\xi'|^2}} f(s) \, ds, \quad f \in L^2(\mathbb{R}, \mathbb{C}^N), \, t \in \mathbb{R};$$

see for instance [66, eq. (8.7)]. Note also that we switched from the variable  $\tilde{x}_{\theta}$  to t for convenience. The resolvent representation of  $H^2_{0,\xi'}$  implies

$$(H_{0,\xi'} - z)^{-1} f(t) = (H_{0,\xi'} + z)(H_{0,\xi'}^2 - z^2)^{-1} f(t)$$
  
=  $\int_{-\infty}^{\infty} \left( \frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(t - s) \right) \frac{ie^{i|t - s|\sqrt{z^2 - m^2 - |\xi'|^2}}}{2} f(s) \, ds$ 

for  $f \in L^2(\mathbb{R}; \mathbb{C}^N)$  and  $t \in \mathbb{R}$ . Using Lemma 3.10 (i) one obtains that  $z \in \rho(H_{0,\xi'})$  is in the point spectrum of  $H_{V_{\varepsilon},\xi'}$  if and only if

$$-1 \in \sigma_p(\mathfrak{D}_{\varepsilon,\xi'}(z)Vq) \quad \text{with} \quad \mathfrak{D}_{\varepsilon,\xi'}(z) = V_R^{\varepsilon}(H_{0,\xi'}-z)^{-1}V_L^{\varepsilon}. \tag{6.4}$$

The operator  $\mathfrak{D}_{\varepsilon,\xi'}(z): L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)$  has the explicit representation

$$\mathfrak{D}_{\varepsilon,\xi'}(z)g(t) = \int_{-1}^{1} \left(\frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{\sqrt{z^2 - m^2 - |\xi'|^2}} + \widetilde{\alpha}_{\theta} \operatorname{sign}(t-s)\right) \frac{ie^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}}}{2}g(s) \, ds$$

for  $g \in L^2((-1,1); \mathbb{C}^N)$  and  $t \in (-1,1)$ . Recall that this operator was already derived in a different way in Section 5.1.1; cf. Lemma 5.1, Proposition 5.2 and Definition 5.3. In the following lemma we compare this operator with the operator  $\mathfrak{H}_{0,\xi'/|\xi'|}(z)$  which was defined in (5.30) and which is for  $\xi' \in \mathbb{R}^{\theta-1} \setminus \{0\}, g \in L^2((-1,1); \mathbb{C}^N)$  and  $t \in (-1,1)$  given by

$$\mathfrak{H}_{0,\xi'/|\xi'|}g(t) = \int_{-1}^{1} \left( \widetilde{\alpha}' \cdot \frac{\xi'}{|\xi'|} + i\widetilde{\alpha}_{\theta} \operatorname{sign}(t-s) \right) \frac{1}{2} g(s) \, ds.$$

**Lemma 6.2.** Let  $\varepsilon > 0$ ,  $0 \neq \xi' \in \mathbb{R}^{\theta-1}$  and  $z \in (-|m|, |m|)$ . Then,

$$\begin{split} \|\mathfrak{H}_{0,\xi'/|\xi'|} - \mathfrak{D}_{\varepsilon,\xi'}(z)\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &\leq \sqrt{N} \Big( 4\varepsilon \sqrt{m^2 + |\xi'|^2} + \frac{2|m|}{|\xi'|} + \frac{m^2}{2|\xi'|^2} \Big). \end{split}$$

*Proof.* Similarly as in Lemma 5.5, we prove the result by estimating the difference of the kernels of both operators. This difference is given by

$$\frac{\widetilde{\alpha}' \cdot \xi'}{2|\xi'|} \left( 1 - \frac{i|\xi'|e^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}}}{\sqrt{z^2 - m^2 - |\xi'|^2}} \right) - \frac{i(m\beta + zI_N)e^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}}}{2\sqrt{z^2 - m^2 - |\xi'|^2}} + i\frac{\widetilde{\alpha}_{\theta}}{2}\mathrm{sign}(t-s)(1 - e^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}}).$$
(6.5)

Before we estimate the individual terms of this sum, we mention that for a unitary self-adjoint matrix  $A \in \mathbb{C}^{N \times N}$  the Frobenius-norm is given by

$$|A|^2 = \operatorname{tr}(AA^*) = \operatorname{tr}(I_N) = N.$$

This applies in particular to  $\tilde{\alpha}' \cdot \frac{\xi'}{|\xi'|}$ ,  $\tilde{\alpha}_{\theta}$ ,  $\beta$  and  $I_N$ . Hence, the third term in (6.5) is bounded by

$$\frac{\sqrt{N}}{2} \Big| 1 - e^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}} \Big| \le \varepsilon \sqrt{N} \sqrt{m^2 + |\xi'|^2 - z^2} \le \varepsilon \sqrt{N} \sqrt{m^2 + |\xi'|^2},$$

where we used  $z \in (-|m|, |m|)$  and  $|t - s| \leq 2$ . The second term in (6.5) can be estimated by

$$\frac{\sqrt{N}(|m|+|z|)}{2\sqrt{|\xi'|^2+m^2-z^2}} \le \frac{\sqrt{N}|m|}{|\xi'|}.$$

Finally, we estimate the first term in (6.5) by

$$\begin{split} & \frac{\sqrt{N}}{2} \Big| 1 - e^{i\varepsilon|t-s|\sqrt{z^2 - m^2 - |\xi'|^2}} \Big| + \frac{\sqrt{N} \Big(\sqrt{|\xi'|^2 + m^2 - z^2} - |\xi'|\Big)}{2\sqrt{|\xi'|^2 + m^2 - z^2}} \\ & \leq \varepsilon \sqrt{N} \sqrt{m^2 + |\xi'|^2} + \frac{\sqrt{N}(m^2 - z^2)}{2\sqrt{|\xi'|^2 + m^2 - z^2} (\sqrt{|\xi'|^2 + m^2 - z^2} + |\xi'|)} \\ & \leq \varepsilon \sqrt{N} \sqrt{m^2 + |\xi'|^2} + \frac{\sqrt{N}m^2}{4|\xi'|^2}, \end{split}$$

where we used again  $z \in (-|m|, |m|)$  and  $|t - s| \leq 2$ . Combining these results shows that the kernel of  $\mathfrak{H}_{0,\xi'/|\xi'|}(z) - \mathfrak{D}_{\varepsilon,\xi'}(z)$  can be estimated by the expression  $\sqrt{N}\left(2\varepsilon\sqrt{m^2+|\xi'|^2}+\frac{|m|}{|\xi'|}+\frac{m^2}{4|\xi'|^2}\right)$ . This estimate and the Schur test, see [44, Chapter III, Example 2.4], yield

$$\begin{split} \|\mathfrak{H}_{0,\xi'/|\xi'|} - \mathfrak{D}_{\varepsilon,\xi'}(z)\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} \\ &\leq \int_{-1}^1 \sqrt{N} \Big( 2\varepsilon \sqrt{m^2 + |\xi'|^2} + \frac{|m|}{|\xi'|} + \frac{m^2}{2|\xi'|^2} \Big) \, dt \\ &= \sqrt{N} \Big( 4\varepsilon \sqrt{m^2 + |\xi'|^2} + \frac{2|m|}{|\xi'|} + \frac{m^2}{2|\xi'|^2} \Big). \end{split}$$

**Lemma 6.3.** Let q be as in (4.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ ,  $d = \eta^2 - \tau^2 > 0$ , and  $\xi' \in \mathbb{R}^{\theta-1} \setminus \{0\}$ . Then, the (nonzero) eigenvalues of  $\mathfrak{H}_{0,\xi'/|\xi'|}Vq$  are given by  $\lambda_k = \frac{2\sqrt{d}}{(2k+1)\pi}, k \in \mathbb{Z}$ .

*Proof.* Let  $\lambda \in \mathbb{C} \setminus \{0\}$  and  $g \in L^2((-1,1);\mathbb{C}^N)$  such that  $\mathfrak{H}_{0,\xi'/|\xi'|}Vqg = \lambda g$ , i.e.

$$\int_{-1}^{1} \left( \widetilde{\alpha}' \cdot \frac{\xi'}{|\xi'|} + i \operatorname{sign}(t-s) \widetilde{\alpha}_{\theta} \right) \frac{1}{2} Vq(s)g(s) \, ds = \lambda g(t) \quad \forall t \in (-1,1).$$

Differentiating both sides gives us

$$i\widetilde{\alpha}_{\theta}Vqg = \lambda g'. \tag{6.6}$$

Hence,  $g(t) = \exp\left(\frac{i\tilde{\alpha}_{\theta}}{\lambda}VQ(t)\right)v$  for a  $v \in \mathbb{C}^N \setminus \{0\}$ , where Q is the primitive function of q chosen in such a way that  $Q(1) = -Q(-1) = \frac{1}{2}$ ; cf. (4.37). Note that the equation  $\mathfrak{H}_{0,\xi'/|\xi'|}Vqg = \lambda g$  implies

$$g(1) + g(-1) = \int_{-1}^{1} \frac{\widetilde{\alpha}' \cdot \xi'}{|\xi'|\lambda} Vq(s)g(s) \, ds$$
  
=  $-\frac{\widetilde{\alpha}' \cdot \xi'}{|\xi'|} i\widetilde{\alpha}_{\theta} \int_{-1}^{1} \frac{i\widetilde{\alpha}_{\theta}}{\lambda} Vq(s) \exp\left(\frac{i\widetilde{\alpha}_{\theta}}{\lambda} VQ(t)\right) v \, ds.$  (6.7)

Calculating the integral on the right-hand side leads to

$$g(1) + g(-1) = 2\cos\left(\frac{\widetilde{\alpha}_{\theta}}{2\lambda}V\right)v = 2\frac{\widetilde{\alpha}' \cdot \xi'}{|\xi'|}\widetilde{\alpha}_{\theta}\sin\left(\frac{\widetilde{\alpha}_{\theta}}{2\lambda}V\right)v = \frac{\widetilde{\alpha}' \cdot \xi'}{|\xi'|\lambda}V\operatorname{sinc}\left(\frac{\widetilde{\alpha}_{\theta}}{2\lambda}V\right)v.$$

Noticing  $(\widetilde{\alpha}_{\theta}V)^2 = (\widetilde{\alpha}_{\theta}(\eta I_N + \tau\beta))^2 = dI_N$  yields

$$\cos\left(\frac{\sqrt{d}}{2\lambda}\right)v = \frac{\widetilde{\alpha}' \cdot \xi'}{2|\xi'|\lambda} (\eta I_N + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)v.$$

Thus,  $v \neq 0$  implies

$$0 = \det\left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)I_N - \frac{\tilde{\alpha}'\cdot\xi'}{2|\xi'|\lambda}(\eta I_N + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)\right)$$

Now, using the rules from (5.23) and (5.24) shows that this is equivalent to

$$0 = \det\left(\left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)I_{N} - \frac{\tilde{\alpha}'\cdot\xi'}{2|\xi'|\lambda}(\eta I_{N} + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)\right) \\ \cdot \beta\left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)I_{N} - \frac{\tilde{\alpha}'\cdot\xi'}{2|\xi'|\lambda}(\eta I_{N} + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)\right)\right) \\ = \det\left(\left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)I_{N} - \frac{\tilde{\alpha}'\cdot\xi'}{2|\xi'|\lambda}(\eta I_{N} + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)\right) \\ \cdot \left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)I_{N} + \frac{\tilde{\alpha}'\cdot\xi'}{2|\xi'|\lambda}(\eta I_{N} + \tau\beta)\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)\right)\beta\right) \\ = \det\left(\left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)^{2} - \frac{d}{4\lambda^{2}}\operatorname{sinc}\left(\frac{\sqrt{d}}{2\lambda}\right)^{2}\right)\beta\right) \\ = \left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)^{2} - \sin\left(\frac{\sqrt{d}}{2\lambda}\right)^{2}\right)^{N}\det(\beta) = \left(\cos\left(\frac{\sqrt{d}}{2\lambda}\right)^{2} - \sin\left(\frac{\sqrt{d}}{2\lambda}\right)^{2}\right)^{N}(-1)^{N/2}$$

and therefore  $\lambda = \frac{2\sqrt{d}}{(2k+1)\pi} = \lambda_k$  for  $k \in \mathbb{Z}$ .

Finally, let us shortly argue that every  $\lambda_k$ ,  $k \in \mathbb{Z}$ , is an eigenvalue of  $\mathfrak{H}_{0,\xi'/|\xi'|}Vq$ . Taking the the same steps as before in the reverse direction one can construct a nonzero smooth function  $g \in L^2((-1,1);\mathbb{C}^N)$  which fulfils (6.6) and (6.7) for  $\lambda = \lambda_k$ . Integrating (6.6) over the intervals (-1,t) and (t,1), subtracting these two integrals and applying (6.7) gives us

$$\int_{-1}^{t} i\widetilde{\alpha}_{\theta} Vq(s)g(s)\,ds - \int_{t}^{1} i\widetilde{\alpha}_{\theta} Vq(s)g(s)\,ds = \lambda_{k} \big( (g(t) - g(-1)) - (g(1) - g(t)) \big)$$
$$= 2\lambda_{k}g(t) - \int_{-1}^{1} \frac{\widetilde{\alpha}' \cdot \xi'}{|\xi'|} Vq(s)g(s)\,ds.$$

Hence, dividing by two and rearranging the terms leads to  $\mathfrak{H}_{0,\xi'/|\xi'|}Vqg = \lambda_k g$ ; i.e.  $\lambda_k$  is an eigenvalue of  $\mathfrak{H}_{0,\xi'/|\xi'|}Vq$ .

**Lemma 6.4.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ ,  $d = \eta^2 - \tau^2 > \frac{\pi^2}{4}$ and  $H_{V_{\varepsilon,\xi'}}$ ,  $\xi' \in \mathbb{R}^{\theta-1}$ , be defined by (6.3). Then, there exists an  $\varepsilon_{\text{cex}} > 0$  such that for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and  $z \in (-|m|, |m|)$  exists a  $\xi'_{\varepsilon,z}$  with  $z \in \sigma_p(H_{V_{\varepsilon},\xi'_{\varepsilon,z}})$ .

Proof. Since  $\eta^2 - \tau^2 > \frac{\pi^2}{4} > 0$ , we have  $\operatorname{sign}(\eta) = \operatorname{sign}(\eta + \tau) = \operatorname{sign}(\eta - \tau)$ . Moreover,  $q \ge 0$  a.e. by assumption (5.1). Therefore, Vq can be rewritten as  $\operatorname{sign}(\eta)D^2$  with  $D = \sqrt{q}\operatorname{diag}(\sqrt{|\eta + \tau|}I_{N/2}, \sqrt{|\eta - \tau|}I_{N/2})$ . Hence, the eigenvalues of  $\mathfrak{D}_{\varepsilon,\xi'}(z)Vq$  and  $\operatorname{sign}(\eta)D\mathfrak{D}_{\varepsilon,\xi'}(z)D, \xi' \in \mathbb{R}^{\theta-1}$ , coincide. Now, (6.4) shows that in order to verify the assertion, we have to prove that there exists an  $\varepsilon_{\text{cex}} > 0$  such that for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and  $z \in (-|m|, |m|)$  exists a  $\xi'_{\varepsilon,z}$  with  $1 \in \sigma_p(-\operatorname{sign}(\eta)D\mathfrak{D}_{\varepsilon,\xi'_{\varepsilon,z}}(z)D)$ . Before we proceed with the proof, we introduce two useful functions. Let  $\mathcal{S}(L^2((-1,1);\mathbb{C}^N))$ be the space of all self-adjoint and compact operators on  $L^2((-1,1);\mathbb{C}^N)$  equipped with the operator norm topology. Note that  $\mathfrak{D}_{\varepsilon,\xi'}(z)$  and  $\mathfrak{H}_{0,\xi'/|\xi'|}$  are Hilbert-Schmidt operators, since their kernels are bounded, which implies

$$\operatorname{sign}(\eta) D\mathfrak{D}_{\varepsilon,\xi'}(z) D, \operatorname{sign}(\eta) D\mathfrak{H}_{0,\xi'/|\xi'|} D \in \mathcal{S}(L^2((-1,1);\mathbb{C}^N)).$$

The mapping

$$\lambda_0: \mathcal{S}(L^2((-1,1);\mathbb{C}^N)) \to \mathbb{R}$$

which maps a compact self-adjoint operator to its biggest nonnegative element of the spectrum is continuous; see [33, Lemma 3.10]. Furthermore, we define for a fixed  $\xi'_0 \in \mathbb{R}^{\theta-1} \setminus \{0\}$ , and  $z \in \mathbb{C} \setminus \mathbb{R}$  and  $\varepsilon > 0$  the continuous function

$$\mu_{\varepsilon,z}: (0,\infty) \to \mathcal{S}(L^2((-1,1);\mathbb{C}^N)), \quad r \mapsto -\operatorname{sign}(\eta)D\mathfrak{D}_{\varepsilon,r\xi'_0}(z)D.$$

Our goal is to find  $r \in (0, \infty)$  such that  $\lambda_0 \circ \mu_{\varepsilon,z}(r) = 1$ , as this implies that z is an eigenvalue of the operator  $H_{V_{\varepsilon},r\xi'_0}$ . Since Proposition 2.29 shows that the spectra of  $-\operatorname{sign}(\eta)D\mathfrak{H}_{0,\xi'/|\xi'|}(z)D$  and  $-\mathfrak{H}_{0,\xi'/|\xi'|}(\eta I_N + \tau\beta)q$  coincide, Lemma 6.3 implies  $\lambda_0(-\operatorname{sign}(\eta)D\mathfrak{H}_{0,\xi'/|\xi'|}D) = \frac{2\sqrt{d}}{\pi} > 1$ . Moreover, as  $\lambda_0$  is continuous, there exists a  $\delta > 0$  such that for all  $A \in \mathcal{S}(L^2((-1,1);\mathbb{C}^N))$  with

$$\|A + \operatorname{sign}(\eta) D\mathfrak{H}_{0,\xi'/|\xi'|} D\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} < \delta$$

also  $\lambda_0(A) > 1$ . According to Lemma 6.2 there exist  $\varepsilon_{\text{cex}} > 0$  and  $r_{\text{cex}} > 0$  such that for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and all  $z \in (-|m|, |m|)$ 

$$\left\|\mu_{\varepsilon,z}(r_{\text{cex}}) + \operatorname{sign}(\eta) D\mathfrak{H}_{0,\xi'/|\xi'|} D\right\|_{L^2((-1,1);\mathbb{C}^N) \to L^2((-1,1);\mathbb{C}^N)} < \delta,$$

implying  $\lambda_0 \circ \mu_{\varepsilon,z}(r_{\text{cex}}) > 1$  for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and all  $z \in (-|m|, |m|)$ . Moreover, it is easy to see that for fixed  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and  $z \in (-|m|, |m|)$  the operator  $\mu_{\varepsilon,z}(r)$ converges for  $r \to \infty$  in the operator norm to zero. Therefore, also  $\lambda_0 \circ \mu_{\varepsilon,z}(r)$  converges for  $r \to \infty$  to zero. Let us summarize, for  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and  $z \in (-|m|, |m|)$  the function  $\lambda_0 \circ \mu_{\varepsilon,z}$  is continuous,  $\lambda_0 \circ \mu_{\varepsilon,z}(r_{\text{cex}}) > 1$  and  $\lambda_0 \circ \mu_{\varepsilon,z}(r) \xrightarrow{r \to \infty} 0$ . Hence, there exists an  $r_{\varepsilon,z} \in (r_{\text{cex}}, \infty)$  such that  $\lambda_0 \circ \mu_{\varepsilon,z}(r_{\varepsilon,z}) = 1$  and therefore  $z \in \sigma_p(H_{V_{\varepsilon}, r_{\varepsilon,z}\xi'_0})$ . Setting  $\xi_{\varepsilon,z} := r_{\varepsilon,z}\xi'_0$  concludes the proof.

**Theorem 6.5.** Let  $\Sigma = \kappa(\mathbb{R}^{\theta-1} \times \{y_0\})$ , i.e.  $\Sigma$  is an affine hyperplane, q as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ ,  $d = \eta^2 - \tau^2 > \frac{\pi^2}{4}$  and  $V_{\varepsilon}$  be defined by (4.3). Then, there exists an  $\varepsilon_{\text{cex}} > 0$  such that  $\sigma(H_{V_{\varepsilon}}) = \mathbb{R}$  for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$ .

*Proof.* Let  $\varepsilon_{\text{cex}} > 0$  be chosen as in Lemma 6.4. We claim that in this case for all  $z \in \mathbb{R}$ and  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  exists a  $\xi'_{\varepsilon,z} \in \mathbb{R}^{\theta-1}$  such that  $z \in \sigma(H_{V_{\varepsilon},\xi'_{\varepsilon,z}})$ . If  $z \in (-|m|, |m|)$ , then this follows from Lemma 6.4. Now, let us assume that  $z \in (-\infty, -|m|] \cup [|m|, \infty)$ . The difference

$$H_{V_{\varepsilon},0} - H_{0,0} = V \frac{\widetilde{q}(\frac{\cdot}{\varepsilon})}{\varepsilon}$$

is a compactly supported  $L^{\infty}$ -function and therefore induces by [54, Theorem 3.27 (ii)] a compact operator from  $H^1(\mathbb{R}; \mathbb{C}^N) = \text{dom } H_{0,0}$  to  $L^2(\mathbb{R}; \mathbb{C}^N)$ . Moreover, the graph norm of  $H_{0,0}$  is equivalent to the  $H^1(\mathbb{R}; \mathbb{C}^N)$ -norm and hence V is relatively compact with respect to the operator  $H_{0,0}$ . Thus, [44, Chapter IV, Theorem 5.35] implies

$$(-\infty, -|m|] \cup [|m|, \infty) = \sigma(H_{0,0}) = \sigma_{\mathrm{ess}}(H_{0,0}) = \sigma_{\mathrm{ess}}(H_{V_{\varepsilon},0}) \subset \sigma(H_{V_{\varepsilon},0}).$$

Hence, the claim that for all  $\varepsilon \in (0, \varepsilon_{\text{cex}})$  and all  $z \in \mathbb{R}$  exists a  $\xi'_{\varepsilon,z}$  such that  $z \in \sigma(H_{V_{\varepsilon}, \xi'_{\varepsilon,z}})$  is valid.

We finish the proof by showing that  $z \in \sigma(H_{V_{\varepsilon},\xi'_{\varepsilon,z}})$  for a fixed  $\xi'_{\varepsilon,z}$  implies  $z \in \sigma(H_{V_{\varepsilon}})$ . The assertion follows from Proposition 2.20 and the text above (6.3) if we can show that for all  $\delta > 0$  exists a  $\gamma_{\delta} > 0$  such that  $(z - \delta, z + \delta) \cap \sigma(H_{V_{\varepsilon},\xi'}) \neq \emptyset$  for all  $\xi' \in B(\xi'_{\varepsilon,z}, \gamma_{\delta})$ . We assume that our claim is not true. In this case there exists a  $\delta' > 0$  and a sequence  $(\xi'_n)_{n \in \mathbb{N}}$  such that  $\xi'_n \xrightarrow{n \to \infty} \xi'_{\varepsilon,z}$  and  $(z - \delta', z + \delta') \cap \sigma(H_{V_{\varepsilon},\xi'_n}) = \emptyset$  for all  $n \in \mathbb{N}$ . Note that for  $w \in \mathbb{C} \setminus \mathbb{R}$  holds

$$\begin{split} \left\| (H_{V_{\varepsilon},\xi'_{\varepsilon,z}} - w)^{-1} - (H_{V_{\varepsilon},\xi'_{n}} - w)^{-1} \right\|_{L^{2}(\mathbb{R};\mathbb{C}^{N}) \to L^{2}(\mathbb{R};\mathbb{C}^{N})} \\ &= \left\| (H_{V_{\varepsilon},\xi'_{\varepsilon,z}} - w)^{-1} \widetilde{\alpha}' \cdot (\xi'_{n} - \xi'_{\varepsilon,z}) (H_{V_{\varepsilon},\xi'_{n}} - w)^{-1} \right\|_{L^{2}(\mathbb{R};\mathbb{C}^{N}) \to L^{2}(\mathbb{R};\mathbb{C}^{N})} \\ &\leq \frac{1}{(\operatorname{Im} w)^{2}} |\xi'_{n} - \xi'_{\varepsilon,z}| \xrightarrow{n \to \infty} 0, \end{split}$$

i.e.  $H_{V_{\varepsilon},\xi'_n}$  converges in the norm resolvent sense to  $H_{V_{\varepsilon},\xi'_{\varepsilon,z}}$ . Moreover,

$$(z - \delta', z + \delta') \cap \sigma(H_{V_{\varepsilon}, \xi'_n}) = \emptyset, \quad n \in \mathbb{N},$$

and Proposition 2.25 (i) imply the contradiction  $(z - \delta', z + \delta') \cap \sigma(H_{V_{\varepsilon}, \xi'_{\varepsilon,z}}) = \emptyset$ .  $\Box$ 

Using Theorem 6.5 leads us to the second counterexample concerning the norm resolvent convergence of  $H_{V_{\varepsilon}}$ .

**Corollary 6.6.** Let the assumptions of Theorem 6.5 be satisfied and assume additionally  $d \neq (2k+1)^2 \pi^2$  for  $k \in \mathbb{N}_0$ . Moreover, let  $(\tilde{\eta}, \tilde{\tau}) = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)(\eta, \tau),$  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 \neq 4$  and  $\tilde{V} = \tilde{\eta}I_N + \tilde{\tau}\beta$ . Then,  $H_{V_{\varepsilon}}$  does not converge in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$  for  $\varepsilon \to 0$ .

*Proof.* By Proposition 2.24 it is no restriction to assume m > 0. We prove this corollary by contradiction. Thus, we assume that  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$ . Then, Theorem 6.5 and Proposition 2.25 (i) imply  $\sigma(H_{\tilde{V}\delta_{\Sigma}}) = \mathbb{R}$ .

However, this is not the case according to [18, Theorem 6.2 (c)] ( $\theta = 2$ ), where  $H_{\widetilde{V}\delta_{\Sigma}}$  was defined and studied via direct integral methods, and [19, Theorem 4.1] ( $\theta = 3$ ). Note that these two references only consider  $\Sigma = \{0\} \times \mathbb{R}$  and  $\Sigma = \mathbb{R}^2 \times \{0\}$ , respectively. However, by a suitable rotation and translation one can transform  $H_{\widetilde{V}\delta_{\Sigma}}$  to a Dirac operator with a  $\delta$ -shell potential supported on  $\{0\} \times \mathbb{R}$  or  $\mathbb{R}^2 \times \{0\}$ , respectively, and a different set of Dirac matrices; cf. [60, Proposition 4] and (6.2). The exact form of the Dirac matrices does not influence the spectrum and therefore the results from [18, 19] are also valid for the case  $\Sigma = \kappa(\mathbb{R}^{\theta-1} \times \{y_0\})$ .

Having established a counterexample for the case where  $\Sigma$  is an affine hyperplane, we consider the case where  $\Sigma$  is a special  $C^2$ -surface containing a flat part in the next theorem.

**Theorem 6.7.** Let  $\Sigma \subset \mathbb{R}^{\theta}$  be a special  $C^2$ -surface as in Definition 2.1 and assume additionally that there exist  $\kappa \in SO(\theta)$ ,  $y_0 \in \mathbb{R}$ ,  $\delta > 0$  and  $x'_0 \in \mathbb{R}^{\theta-1}$  such that  $\Sigma \supset \kappa(B(x'_0, \delta) \times \{y_0\})$ , i.e.  $\Sigma$  contains a flat part. Moreover, let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$ ,  $d = \eta^2 - \tau^2 > \frac{\pi^2}{4}$  such that  $d \neq (2k+1)^2 \pi^2$  for  $k \in \mathbb{N}_0$ ,  $V_{\varepsilon}$  be defined by (4.3),  $(\tilde{\eta}, \tilde{\tau}) = tanc(\frac{\sqrt{d}}{2})(\eta, \tau)$ ,  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 \neq 4$  and  $\tilde{V} = \tilde{\eta} I_N + \tilde{\tau}\beta$ . Then,  $H_{V_{\varepsilon}}$  does not converge in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}$ .

Proof. We prove this theorem by contradiction. Thus, we assume that  $H_{V_{\varepsilon}}$  converges in the norm resolvent sense to  $H_{\widetilde{V}\delta_{\Sigma}}$  and proceed as follows: In Step 1 we compare the two operators  $H_{V_{\varepsilon}}$  and  $H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}$ , where  $V_{\varepsilon}^{\Sigma_{y_0,\kappa}}$  is defined as  $V_{\varepsilon}$  in (4.3) with  $\Sigma$ substituted by  $\Sigma_{y_0,\kappa} = \kappa(\mathbb{R}^{\theta-1} \times \{y_0\})$ . Then, in Step 2, we construct a resolvent formula for  $H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}$  in terms of  $(H_{V_{\varepsilon}} - z)^{-1}$ . Finally, in Step 3, we use this formula to show that  $H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma_{y_0,\kappa}}}$  for  $\varepsilon \to 0$ ; this contradicts Corollary 6.6. Step 1. As in (6.1) we can represent  $V_{\varepsilon}^{\Sigma_{y_0,\kappa}}$ ,  $\varepsilon \in (0,\infty)$ , by

$$V_{\varepsilon}^{\Sigma_{y_0,\kappa}}(x) = V \frac{\widetilde{q}\left(\frac{\widetilde{x}_{\theta}}{\varepsilon}\right)}{\varepsilon} \quad \text{for } x \in \mathbb{R}^{\theta} \text{ and } \widetilde{x} = (\widetilde{x}', \widetilde{x}_{\theta}) = \kappa^T x - y_0 e_{\theta}, \tag{6.8}$$

where  $\widetilde{q}$  is the zero extension q to  $\mathbb{R}$  and  $e_{\theta}$  is the  $\theta$ -th Euclidean unit vector. Next, we compare  $V_{\varepsilon}$  and  $V_{\varepsilon}^{\Sigma_{y_0,\kappa}}$ . To guarantee the well-definedness of  $V_{\varepsilon}$  we assume  $\varepsilon \in (0, \varepsilon_{\text{tub}})$  with  $\varepsilon_{\text{tub}} > 0$  as in Proposition 2.12. Let

$$x \in \kappa(B(x'_0, \delta) \times (y_0 - \varepsilon_{\text{tub}}, y_0 + \varepsilon_{\text{tub}})).$$

Then,  $x = \kappa(\tilde{x}', y_0) + \tilde{x}_{\theta}\kappa e_{\theta}$  for a vector  $\tilde{x} = (\tilde{x}', \tilde{x}_{\theta}) \in B(x'_0, \delta) \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$ . In particular, through defining  $x_{\Sigma} := \kappa(\tilde{x}', y_0)$  we get  $x_{\Sigma} \in \kappa(B(x'_0, \delta) \times \{y_0\}) \subset \Sigma$  and that  $\nu(x_{\Sigma}) = \kappa e_{\theta}$  is the corresponding normal vector at this point. Hence, by setting

 $t := \widetilde{x}_{\theta} \in (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$ , we have  $x = x_{\Sigma} + t\nu(x_{\Sigma}) \in \Omega_{\varepsilon_{\text{tub}}}$ , and by (4.3) and (6.8)

$$V_{\varepsilon}(x) = \begin{cases} V \frac{q\left(\frac{\widetilde{x}_{\theta}}{\varepsilon}\right)}{\varepsilon}, & |\widetilde{x}_{\theta}| < \varepsilon, \\ 0, & |\widetilde{x}_{\theta}| \ge \varepsilon, \end{cases} \\ = V \frac{\widetilde{q}\left(\frac{\widetilde{x}_{\theta}}{\varepsilon}\right)}{\varepsilon} = V_{\varepsilon}^{\Sigma_{y_0,\kappa}}(x), \end{cases}$$

implying

$$V_{\varepsilon} \upharpoonright \kappa(B(x'_{0}, \delta) \times (y_{0} - \varepsilon_{\text{tub}}, y_{0} + \varepsilon_{\text{tub}}))$$
  
=  $V_{\varepsilon}^{\Sigma_{y_{0},\kappa}} \upharpoonright \kappa(B(x'_{0}, \delta) \times (y_{0} - \varepsilon_{\text{tub}}, y_{0} + \varepsilon_{\text{tub}})).$  (6.9)

In particular, if  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and  $\operatorname{supp} u \subset \kappa(B(x'_0, \delta) \times (y_0 - \varepsilon_{\operatorname{tub}}, y_0 + \varepsilon_{\operatorname{tub}}))$ , then

$$V_{\varepsilon}u = V_{\varepsilon}^{\Sigma_{y_0,\kappa}}u$$
 and if  $u \in H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ , then also  $H_{V_{\varepsilon}}u = H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}u$ .

Step 2. Based on Step 1 we construct a resolvent for  $H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}$  in terms of  $(H_{V_{\varepsilon}}-z)^{-1}$  in this step. We start by constructing a partition of unity for the strip

$$\kappa \left( \mathbb{R}^{\theta - 1} \times \left( y_0 - \frac{\varepsilon_{\text{tub}}}{4}, y_0 + \frac{\varepsilon_{\text{tub}}}{4} \right) \right).$$

To do so, let  $(\phi_{n'}^a)_{n'\in\mathbb{Z}^{\theta-1}}$  with fixed  $a \in (0, \max\{1, \frac{\delta}{3}\})$  be the partition of unity for  $\mathbb{R}^{\theta-1}$  from Corollary A.3 (for  $n = \theta - 1$ ). Moreover, let  $(\vartheta_{n'}^a)_{n'\in\mathbb{Z}^{\theta-1}}$  be also chosen as in Corollary A.3. These sequences possess the following properties:

$$\operatorname{supp} \phi_{n'}^a \subset B(an', a), \operatorname{supp} \vartheta_{n'}^a \subset B(an', 3a) \text{ and } \phi_{n'}^a \vartheta_{n'}^a = \phi_{n'}^a \text{ for all } n' \in \mathbb{Z}^{\theta - 1}.$$

Moreover, the  $W^1_{\infty}$ -norms of  $\phi^a_{n'}$  and  $\vartheta^a_{n'}$  are uniformly bounded with respect to  $n' \in \mathbb{Z}^{\theta-1}$ . Now, we choose a function  $\varrho \in C_0^{\infty}(\mathbb{R})$  such that  $0 \leq \varrho \leq 1$ ,  $\varrho = 1$  on  $(y_0 - \frac{\varepsilon_{\text{tub}}}{4}, y_0 + \frac{\varepsilon_{\text{tub}}}{4})$  and  $\text{supp } \varrho \subset (y_0 - \frac{\varepsilon_{\text{tub}}}{2}, y_0 + \frac{\varepsilon_{\text{tub}}}{2})$ , and define for  $n' \in \mathbb{Z}^{\theta-1}$ 

$$\psi_{n'}^a := (\phi_{n'}^a \otimes \rho)(\kappa^T(\cdot) - y_0 e_\theta) \quad \text{as well as} \quad \varsigma_{n'}^a := \left(\vartheta_{n'}^a \otimes \varrho(\frac{\cdot}{2})\right)(\kappa^T(\cdot) - y_0 e_\theta)$$

Then,  $(\psi_{n'}^a)_{n' \in \mathbb{Z}^{\theta-1}}$  is a partition of unity for  $\kappa \left(\mathbb{R}^{\theta-1} \times (y_0 - \frac{\varepsilon_{\text{tub}}}{4}, y_0 + \frac{\varepsilon_{\text{tub}}}{4})\right)$ , the inclusions

$$\operatorname{supp} \psi_{a'}^{a} \subset \kappa \left( B(an', a) \times \left( y_0 - \frac{\varepsilon_{\operatorname{tub}}}{2}, y_0 + \frac{\varepsilon_{\operatorname{tub}}}{2} \right) \right)$$
$$\operatorname{supp} \zeta_{a'}^{a} \subset \kappa \left( B(an', 3a) \times \left( y_0 - \varepsilon_{\operatorname{tub}}, y_0 + \varepsilon_{\operatorname{tub}} \right) \right)$$

hold and  $\psi_{n'}^a \varsigma_{n'}^a = \psi_{n'}^a$  for all  $n' \in \mathbb{Z}^{\theta-1}$ . Now, let the unitary translation operator  $T_{n'}$ ,  $n' \in \mathbb{Z}^{\theta-1}$ , which translates the argument of a function parallel to  $\Sigma_{y_0,\kappa}$ , be given by

$$T_{n'}: L^2(\mathbb{R}^\theta; \mathbb{C}^N) \to L^2(\mathbb{R}^\theta; \mathbb{C}^N), \quad T_{n'}u(x) = u(x + \kappa(an' - x'_0, 0)).$$

As  $T_{n'}$  translates the argument of a function parallel to  $\Sigma_{y_0,\kappa}$ , we see with the help of (6.8) that  $V_{\varepsilon}^{\Sigma_{y_0,\kappa}}$  commutes with  $T_{n'}$ . Hence, also  $H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}$  commutes with  $T_{n'}$ . Moreover, for all  $n' \in \mathbb{Z}^{\theta-1}$  and  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  the inclusions

$$\sup T_{n'}\psi_{n'}^{a} u \subset \kappa \left( B(x'_{0},a) \times \left(y_{0} - \frac{\varepsilon_{\text{tub}}}{2}, y_{0} + \frac{\varepsilon_{\text{tub}}}{2}\right) \right), \\ \operatorname{supp} T_{n'}\varsigma_{n'}^{a} u \subset \kappa \left( B(x'_{0},3a) \times \left(y_{0} - \varepsilon_{\text{tub}}, y_{0} + \varepsilon_{\text{tub}}\right) \right),$$

are valid. These considerations and (6.9) show that we have for all  $n' \in \mathbb{Z}^{\theta-1}$  and  $u \in H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ 

$$H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}\varsigma_{n'}^a u = T_{n'}^{-1}H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}T_{n'}\varsigma_{n'}^a u = T_{n'}^{-1}H_{V_{\varepsilon}}T_{n'}\varsigma_{n'}^a u.$$
(6.10)

We define  $\psi := 1 - \sum_{n' \in \mathbb{Z}} \psi_{n'}^a$ . By definition  $\psi = 0$  on  $\kappa \left( \mathbb{R}^{\theta - 1} \times \left( y_0 - \frac{\varepsilon_{\text{tub}}}{4}, y_0 + \frac{\varepsilon_{\text{tub}}}{4} \right) \right)$ . Consequently,  $H \psi u = H_{V_{\varepsilon}^{\Sigma y_0, \kappa}} \psi u$  for  $u \in H^1(\mathbb{R}^{\theta - 1}; \mathbb{C}^N)$  and  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{4})$ .

Next, we introduce for a fixed  $z \in \mathbb{C} \setminus \mathbb{R}$  and for  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{4})$  the two series

$$\mathscr{R}_{\varepsilon,1} := (H-z)^{-1}\psi + \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\text{st.}} \psi_{n'}^{a}T_{n'}^{-1}(H_{V_{\varepsilon}}-z)^{-1}T_{n'}\varsigma_{n'}^{a},$$
$$\mathscr{R}_{\varepsilon,2} := (H-z)^{-1}i(\alpha\cdot\nabla\psi) + \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\text{st.}}\psi_{n'}^{a}T_{n'}^{-1}(H_{V_{\varepsilon}}-z)^{-1}T_{n'}i(\alpha\cdot\nabla\varsigma_{n'}^{a}).$$

We claim that both  $\mathscr{R}_{\varepsilon,1}$  and  $\mathscr{R}_{\varepsilon,2}$  are well-defined acting as operators in  $L^2(\mathbb{R}^{\theta};\mathbb{C}^N)$ . Setting  $\mathcal{A}_{n'} := \psi_{n'}^a T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} \varsigma_{n'}^a$  and using the properties of  $\psi_{n'}$  and  $\vartheta_{n'}$ shows that for every  $n'_0 \in \mathbb{Z}^{\theta-1}$  exist at most  $7^{\theta-1}$  indices  $n' \in \mathbb{Z}^{\theta-1}$  such that  $\mathcal{A}_{n'_0} \mathcal{A}_{n'}^* \neq 0$  and  $\mathcal{A}_{n'_0}^* \mathcal{A}_{n'} \neq 0$ . Hence, Lemma C.2 implies that the family of operators  $(\mathcal{A})_{n' \in \mathbb{Z}^{\theta-1}} = (\psi_{n'}^a T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} \varsigma_{n'}^a)_{n' \in \mathbb{Z}^{\theta-1}}$  is strongly summable and therefore

$$\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}}\psi_{n'}^{a}T_{n'}^{-1}(H_{V_{\varepsilon}}-z)^{-1}T_{n'}\varsigma_{n'}^{a}$$

is well-defined and its norm is bounded by

$$C \sup_{n' \in \mathbb{Z}^{\theta-1}} \left\| \psi_{n'}^{a} T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} \varsigma_{n'}^{a} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}$$

$$\leq C \sup_{n' \in \mathbb{Z}^{\theta-1}} \left\| T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}$$

$$= C \left\| (H_{V_{\varepsilon}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})}.$$

Thus,  $\mathscr{R}_{\varepsilon,1}$  is bounded by

$$C\Big(\big\|(H-z)^{-1}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)}+\big\|(H_{V_\varepsilon}-z)^{-1}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)}\Big).$$

Similar arguments and the fact that the derivatives of  $\zeta_{n'}^a$  are uniformly bounded with respect to  $n' \in \mathbb{Z}^{\theta-1}$  show that  $\mathscr{R}_{\varepsilon,2}$  is well-defined and bounded by the same expression. Moreover, by choosing Im z big enough (independent of  $\varepsilon$ ) we can guarantee  $\|\mathscr{R}_{\varepsilon,2}\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)} < \frac{1}{2}$  and hence  $I + \mathscr{R}_{\varepsilon,2}$  is boundedly invertible. Next, we show that  $(I + \mathscr{R}_{\varepsilon,2})^{-1}\mathscr{R}_{\varepsilon,1}$  is the inverse of  $H^{\Sigma y_0,\kappa}_{V_{\varepsilon}} - z$ . We do this by determining  $(I + \mathscr{R}_{\varepsilon,2})^{-1}\mathscr{R}_{\varepsilon,1}(H_{V_{\varepsilon}^{\Sigma y_0,\kappa}} - z)u$  for  $u \in H^1(\mathbb{R}^\theta; \mathbb{C}^N)$ . The product rule gives us

$$\varsigma^a_{n'}H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}u = H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}}\varsigma^a_{n'}u + i(\alpha\cdot\nabla\varsigma^a_{n'})u \quad \text{ and } \quad \psi Hu = H\psi u + i(\alpha\cdot\nabla\psi)u.$$

Combining this observation with (6.10) yields

$$\begin{split} (I + \mathscr{R}_{\varepsilon,2})^{-1} \mathscr{R}_{\varepsilon,1} (H_{V_{\varepsilon}^{\Sigma y_{0},\kappa}} - z) u \\ &= (I + \mathscr{R}_{\varepsilon,2})^{-1} \Big( (H - z)^{-1} \psi (H_{V_{\varepsilon}^{\Sigma y_{0},\kappa}} - z) u \\ &+ \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} \varsigma_{n'}^{a} (H_{V_{\varepsilon}^{\Sigma y_{0},\kappa}} - z) u \Big) \\ &= (I + \mathscr{R}_{\varepsilon,2})^{-1} \Big( (H - z)^{-1} (H_{V_{\varepsilon}^{\Sigma y_{0},\kappa}} - z) \psi u \\ &+ \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} (H_{V_{\varepsilon}^{\Sigma y_{0},\kappa}} - z) \varsigma_{n'}^{a} u \\ &+ \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} (H - z)^{-1} i (\alpha \cdot \nabla \psi) u \\ &+ \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} i (\alpha \cdot \nabla \varsigma_{n'}^{a}) u \Big) \\ &= (I + \mathscr{R}_{\varepsilon,2})^{-1} \Big( (H - z)^{-1} (H - z) \psi u \\ &+ \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{V_{\varepsilon}} - z)^{-1} T_{n'} T_{n'}^{-1} (H_{V_{\varepsilon}} - z) T_{n'} \varsigma_{n'}^{a} u + \mathscr{R}_{\varepsilon,2} u \Big) \\ &= (I + \mathscr{R}_{\varepsilon,2})^{-1} \Big( \psi u + \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \psi_{n'}^{a} \varsigma_{n'}^{a} u + \mathscr{R}_{\varepsilon,2} u \Big) \\ &= (I + \mathscr{R}_{\varepsilon,2})^{-1} (I + \mathscr{R}_{\varepsilon,2}) u = u. \end{split}$$

Hence,  $(I + \mathscr{R}_{\varepsilon,2})^{-1} \mathscr{R}_{\varepsilon,1}$  is indeed the inverse of  $H_{V_{\varepsilon}}^{\Sigma_{y_0,\kappa}} - z$ .

Step 3. Similarly as we defined  $\mathscr{R}_{\varepsilon,1}$  and  $\mathscr{R}_{\varepsilon,2}$ , we also introduce

$$\mathscr{R}_{0,1} := (H-z)^{-1}\psi + \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\text{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} T_{n'}\varsigma_{n'}^{a},$$
$$\mathscr{R}_{0,2} := (H-z)^{-1} i(\alpha \cdot \nabla \psi) + \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\text{st.}} \psi_{n'}^{a} T_{n'}^{-1} (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} T_{n'} i(\alpha \cdot \nabla \varsigma_{n'}^{a}).$$

Applying Lemma C.2 shows on the one hand that these operators are bounded by

$$C\Big(\big\|(H-z)^{-1}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)} + \sup_{n'\in\mathbb{Z}^{\theta-1}}\big\|T_{n'}^{-1}(H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1}T_{n'}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)}\Big)$$
$$= C\Big(\big\|(H-z)^{-1}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)} + \big\|(H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1}\big\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to L^2(\mathbb{R}^\theta;\mathbb{C}^N)}\Big)$$

and on the other hand that the differences  $\mathscr{R}_{\varepsilon,2} - \mathscr{R}_{0,2}$  and  $\mathscr{R}_{\varepsilon,1} - \mathscr{R}_{0,1}$  are bounded by

$$C \| (H_{V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \stackrel{\varepsilon \to 0}{\longrightarrow} 0$$

Hence,  $(H_{V_{\varepsilon}^{\Sigma_{y_0,\kappa}}} - z)^{-1} = (I + \mathscr{R}_{\varepsilon,2})^{-1} \mathscr{R}_{\varepsilon,1}$  converges to  $(I + \mathscr{R}_{0,2})^{-1} \mathscr{R}_{0,1}$  in the operator norm. It remains to prove  $(I + \mathscr{R}_{0,2})^{-1} \mathscr{R}_{0,1} = (H_{\widetilde{V}\delta_{\Sigma_{y_0,\kappa}}} - z)^{-1}$ . We start by noticing that  $H_{\widetilde{V}\delta_{\Sigma_{y_0,\kappa}}}$  also commutes with  $T_{n'}, n' \in \mathbb{Z}^{\theta-1}$ , since  $T_{n'}$  translates the argument of a function only along  $\Sigma_{y_0,\kappa}$  and  $\widetilde{V}$  is constant. According to Lemma 5.19  $u \in \text{dom } H_{\widetilde{V}\delta_{\Sigma_{y_0,\kappa}}}$  (or equivalently  $T_{n'}u = u((\cdot) + \kappa(an' - x'_0, 0)) \in \text{dom } H_{\widetilde{V}\delta_{\Sigma_{y_0,\kappa}}}$  for a  $n' \in \mathbb{Z}^{\theta-1}$ ) implies  $T_{n'}\varsigma_{n'}^a u = \varsigma_{n'}^a((\cdot) + \kappa(an' - x'_0, 0))u((\cdot) + \kappa(an' - x'_0, 0)) \in \text{dom } H_{\widetilde{V}\delta_{\Sigma}}$  and

$$H_{\widetilde{V}\delta_{\Sigma_{y_{0},\kappa}}}\varsigma_{n'}^{a}u = T_{n'}^{-1}H_{\widetilde{V}\delta_{\Sigma_{y_{0},\kappa}}}T_{n'}\varsigma_{n'}^{a}u = T_{n'}^{-1}H_{\widetilde{V}\delta_{\Sigma}}T_{n'}\varsigma_{n'}^{a}u$$
$$\varsigma_{n'}^{a}H_{\widetilde{V}\delta_{\Sigma_{y_{0},\kappa}}}u = H_{\widetilde{V}\delta_{\Sigma_{y_{0},\kappa}}}\varsigma_{n'}^{a}u + i(\alpha \cdot \nabla\varsigma_{n'}^{a})u.$$

Using these observations one shows  $(H_{\tilde{V}\delta_{\Sigma_{y_0,\kappa}}}-z)^{-1}=(I+\mathscr{R}_{0,2})^{-1}\mathscr{R}_{0,1}$  in the same way as in (6.11).

## 7 Consequences of the approximation results

In this chapter we present various consequences of the approximation results shown in Chapter 4 and Chapter 5. In particular, in Section 7.1, we show a scheme to approximate Dirac operators with  $\delta$ -shell potentials that induce confinement; cf. [65] for the one-dimensional counterpart. Furthermore, in Section 7.2, we transfer results from Dirac operators with  $\delta$ -shell potentials to Dirac operators with strongly localized potentials to study their discrete and essential spectrum.

## 7.1 Approximation of Dirac operators with $\delta$ -shell potentials that induce confinement

In Corollary 5.22 we saw that all Dirac operators with  $\delta$ -shell potentials and interaction matrices of the form  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$ ,  $\widetilde{\eta}, \widetilde{\tau} \in C_b^1(\Sigma; \mathbb{R})$ , with  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2$ , satisfying

$$\sup_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| < 4 \quad \text{or} \quad \inf_{x_{\Sigma}\in\Sigma} |\widetilde{d}(x_{\Sigma})| > 4$$

can be approximated by Dirac operators with strongly localized potentials. However, this excludes the case where  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 = -4$ . According to Proposition 3.15 (ii) this is a particular interesting case as it induces confinement, i.e. the operator  $H_{\tilde{V}\delta_{\Sigma}}$  splits into the orthogonal sum  $H_{\tilde{V}}^+ \oplus H_{\tilde{V}}^-$  with

dom 
$$H_{\widetilde{V}}^{\pm} = \{ u_{\pm} \in H^1(\Omega_{\pm}; \mathbb{C}^N) : (2I_N \mp i(\alpha \cdot \nu)\widetilde{V}) \mathbf{t}_{\Sigma}^{\pm} u_{\pm} = 0 \} \subset L^2(\Omega_{\pm}; \mathbb{C}^N),$$
  
 $H_{\widetilde{V}}^{\pm} u_{\pm} = -i(\alpha \cdot \nabla) u_{\pm} + m\beta u_{\pm},$ 

which means on a physical level that  $\Sigma$  becomes impermeable for a particle. This raises the question whether there is also a way to approximate such Dirac operators by Dirac operators with strongly localized potentials. Inspired by the rescaling formula in (5.6) one would have to choose  $V = \eta I_N + \tau \beta$  with  $d = \eta^2 - \tau^2 = -\infty$  as the interaction matrix in the approximating operators to obtain a Dirac operator with  $\delta$ -shell potential and  $\tilde{d} = \tilde{\eta}^2 - \tilde{\tau}^2 = -4$  in the limit. We rigorously realize this idea by choosing  $\varepsilon$ -dependent interaction strengths  $\eta_{\varepsilon}$  and  $\tau_{\varepsilon}$  such that  $\eta_{\varepsilon}^2 - \tau_{\varepsilon}^2 \xrightarrow{\varepsilon \to 0} -\infty$ . The same approach was used in [65, Chapter 3] when dealing with the one-dimensional case. We begin by explaining the setting. We choose q as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  and assume that  $V_{\varepsilon}$  is defined as in (4.3). Moreover, we set  $d = \eta^2 - \tau^2$ , assume

$$\sup_{x_{\Sigma} \in \Sigma} d(x_{\Sigma}) < 0 \quad \text{and} \quad \text{sign}(\tau) = \text{const.}, \tag{7.1}$$

and fix a continuous function  $f: (0, \varepsilon_{tub}) \to (0, \infty)$ , with  $\varepsilon_{tub} > 0$  from Proposition 2.12, which satisfies the condition

$$\lim_{\varepsilon \to 0} f(\varepsilon) = \infty \quad \text{and} \quad \lim_{\varepsilon \to 0} f(\varepsilon)^4 \varepsilon^{1/2 - r} = 0 \tag{7.2}$$

for an  $r \in (0, \frac{1}{2})$ . We also introduce the  $\varepsilon$ -depend interaction strengths

$$(\eta_{\varepsilon}, \tau_{\varepsilon}) = f(\varepsilon)(\eta, \tau) \text{ and } d_{\varepsilon} = \eta_{\varepsilon}^2 - \tau_{\varepsilon}^2 = f(\varepsilon)^2 d.$$
 (7.3)

Then,  $f(\varepsilon)V = \eta_{\varepsilon}I_N + \tau_{\varepsilon}\beta$ . Now, we are interested in the norm resolvent convergence of the operator  $H_{f(\varepsilon)V_{\varepsilon}}$ . Note that for a fixed  $\varepsilon_0 \in (0, \varepsilon_{\text{tub}})$   $H_{f(\varepsilon_0)V_{\varepsilon}}$  converges by Theorem 5.20 in the norm resolvent sense to  $H_{\widetilde{V}_{\varepsilon_0}\delta_{\Sigma}}$  for  $\varepsilon \to 0$  with

$$\widetilde{V}_{\varepsilon_0} := \operatorname{tanc}\left(\frac{\sqrt{d_{\varepsilon_0}}}{2}\right) f(\varepsilon_0) V.$$

Our goal is to prove that the limit of  $H_{f(\varepsilon)V_{\varepsilon}}$  is the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  with

$$\begin{split} \widetilde{V} &= \lim_{\varepsilon \to 0} \widetilde{V}_{\varepsilon} = \lim_{\varepsilon \to 0} \tan\left(\frac{\sqrt{d_{\varepsilon}}}{2}\right) f(\varepsilon) V \\ &= \lim_{\varepsilon \to 0} \frac{2 \tan\left(\frac{f(\varepsilon)\sqrt{d}}{2}\right)}{f(\varepsilon)\sqrt{d}} f(\varepsilon) V = \frac{2i}{\sqrt{d}} V = \frac{2}{\sqrt{|d|}} V \end{split}$$

Consequently,  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$  with  $(\widetilde{\eta}, \widetilde{\tau}) = \frac{2}{\sqrt{|d|}}(\eta, \tau)$  and therefore  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = -4$ , i.e.  $H_{\widetilde{V}\delta_{\Sigma}}$  induces confinement.

We start by proving several preparatory statements.

**Lemma 7.1.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (7.1),  $V_{\varepsilon}$  be defined by (4.3),  $f(\varepsilon)$  be the function from (7.2) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, let  $\widetilde{V}_{\varepsilon} = \operatorname{tanc}(\frac{\sqrt{d_{\varepsilon}}}{2})f(\varepsilon)V$ ,  $\widetilde{V} = \frac{2}{\sqrt{|d|}}V$  and  $d_{\inf} := \inf_{x_{\Sigma}\in\Sigma}\sqrt{|d(x_{\Sigma})|} > 0$ , where  $d = \eta^2 - \tau^2$  and  $d_{\varepsilon} = f(\varepsilon)^2 d$ . Then,  $H_{\widetilde{V}_{\varepsilon}\delta_{\Sigma}}$  and  $H_{\widetilde{V}\delta_{\Sigma}}$  are self-adjoint and there exists an  $\varepsilon_{\operatorname{conf},1} \in (0, \varepsilon_{\operatorname{tub}})$  such that

$$\left\| (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} - (H_{\widetilde{V}_{\varepsilon}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq Cf(\varepsilon)e^{-f(\varepsilon)d_{\inf}}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf},1})$ .
Proof. The interaction strengths corresponding to the interaction matrix  $\widetilde{V}_{\varepsilon}$  are given by  $(\widetilde{\eta}_{\varepsilon}, \widetilde{\tau}_{\varepsilon}) = \operatorname{tanc}(\frac{\sqrt{d_{\varepsilon}}}{2})(\eta_{\varepsilon}, \tau_{\varepsilon})$ . Using  $(\eta_{\varepsilon}, \tau_{\varepsilon}) = f(\varepsilon)(\eta, \tau)$  and  $d(x_{\Sigma}) < 0, x_{\Sigma} \in \Sigma$ , yields  $\widetilde{d}_{\varepsilon}(x_{\Sigma}) = \widetilde{\eta}_{\varepsilon}(x_{\Sigma})^2 - \widetilde{\tau}_{\varepsilon}(x_{\Sigma})^2$ 

$$\begin{split} \tilde{d}_{\varepsilon}(x_{\Sigma}) &= \tilde{\eta}_{\varepsilon}(x_{\Sigma})^2 - \tilde{\tau}_{\varepsilon}(x_{\Sigma})^2 \\ &= 4 \tan\left(\frac{f(\varepsilon)\sqrt{d(x_{\Sigma})}}{2}\right)^2 \\ &= -4 \tanh\left(f(\varepsilon)\frac{\sqrt{|d(x_{\Sigma})|}}{2}\right)^2 \\ &\leq -4 \tanh\left(f(\varepsilon)\frac{d_{\inf}}{2}\right)^2 \\ &< 0 \end{split}$$

for all  $x_{\Sigma} \in \Sigma$ . Moreover, by construction  $\tilde{d} = -4$ . Thus, Proposition 3.14 implies that both  $H_{\tilde{V}_{\varepsilon}\delta_{\Sigma}}$  and  $H_{\tilde{V}\delta_{\Sigma}}$  are self-adjoint, and that  $I + \mathcal{C}_{z}\tilde{V}_{\varepsilon}$  as well as  $I + \mathcal{C}_{z}\tilde{V}$  are continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Moreover, the resolvent formulas

$$(H_{\widetilde{V}_{\varepsilon}\delta_{\Sigma}}-z)^{-1} = (H-z)^{-1} - \Phi_{z}\widetilde{V}_{\varepsilon}(I+\mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1}\Phi_{z}^{*},$$
  
$$(H_{\widetilde{V}\delta_{\Sigma}}-z)^{-1} = (H-z)^{-1} - \Phi_{z}\widetilde{V}(I+\mathcal{C}_{z}\widetilde{V})^{-1}\Phi_{z}^{*},$$

hold by Proposition 3.14. Thus, the difference of the resolvents is given by

$$-\Phi_{z}\widetilde{V}_{\varepsilon}(I+\mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1}\Phi_{z}^{*}+\Phi_{z}\widetilde{V}(I+\mathcal{C}_{z}\widetilde{V})^{-1}\Phi_{z}^{*}$$
  
$$=\Phi_{z}(\widetilde{V}-\widetilde{V}_{\varepsilon})(I+\mathcal{C}_{z}\widetilde{V})^{-1}\Phi_{z}^{*}+\Phi_{z}\widetilde{V}_{\varepsilon}\big((I+\mathcal{C}_{z}\widetilde{V})^{-1}-(I+\mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1}\big)\Phi_{z}^{*}.$$
(7.4)

Simple calculations show

$$\widetilde{V}_{\varepsilon} = \operatorname{tanc}\left(\frac{\sqrt{d_{\varepsilon}}}{2}\right) f(\varepsilon) V = 2 \frac{\operatorname{tan}\left(\frac{\sqrt{d_{\varepsilon}}}{2}\right)}{\sqrt{d_{\varepsilon}}} f(\varepsilon) V$$
$$= 2 \frac{\operatorname{tanh}\left(f(\varepsilon)\frac{\sqrt{|d|}}{2}\right)}{f(\varepsilon)\sqrt{|d|}} f(\varepsilon) V = \operatorname{tanh}\left(f(\varepsilon)\frac{\sqrt{|d|}}{2}\right) \widetilde{V}.$$

Thus, applying Proposition 2.2 (iv) and (v) yields

$$\begin{split} \|\widetilde{V}_{\varepsilon} - \widetilde{V}\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} &\leq C \| \tanh\left(f(\varepsilon)\frac{\sqrt{|d|}}{2}\right) - 1\|_{W^{1}_{\infty}(\Sigma)} \|\widetilde{V}\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} \\ &\leq C \| \tanh\left(f(\varepsilon)\frac{\cdot}{2}\right) - 1\|_{W^{1}_{\infty}((d_{\inf},\infty))} \|\sqrt{|d|}\|_{W^{1}_{\infty}(\Sigma)} \|\widetilde{V}\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})} \\ &\leq C \| \tanh\left(f(\varepsilon)\frac{\cdot}{2}\right) - 1\|_{W^{1}_{\infty}((d_{\inf},\infty))}. \end{split}$$

The expression  $\|\tanh(f(\varepsilon)\frac{1}{2}) - 1\|_{W^1_{\infty}((d_{\inf},\infty))}$  is smaller than

$$\|\tanh\left(f(\varepsilon)\frac{\cdot}{2}\right) - 1\|_{L^{\infty}((d_{\inf},\infty))} + \frac{f(\varepsilon)}{2}\|\tanh'\left(f(\varepsilon)\frac{\cdot}{2}\right)\|_{L^{\infty}((d_{\inf},\infty))}.$$

Next, we estimate the two summands separately. We start with the first one. For  $t \in (d_{inf}, \infty)$  the inequality

$$|\tanh(f(\varepsilon)\frac{t}{2}) - 1| = \frac{2}{1 + e^{f(\varepsilon)t}} \le 2e^{-f(\varepsilon)t} < 2e^{-f(\varepsilon)d_{\inf}}$$

is valid. Hence,  $\|\tanh(f(\varepsilon)\frac{1}{2}) - 1\|_{L^{\infty}((d_{\inf},\infty))} \leq 2e^{-f(\varepsilon)d_{\inf}}$ . For the second term we obtain

$$|\tanh'(f(\varepsilon)\frac{t}{2})| = \frac{1}{\cosh(f(\varepsilon)t/2)^2} \le e^{-f(\varepsilon)t} < e^{-f(\varepsilon)d_{\inf}} \qquad \forall t \in (d_{\inf},\infty).$$

Consequently,  $\| \tanh(f(\varepsilon)_{\frac{1}{2}}) - 1 \|_{W^1_{\infty}((d_{\inf},\infty))} \leq f(\varepsilon) e^{-f(\varepsilon)d_{\inf}}$  and, in turn,

$$\|\widetilde{V}_{\varepsilon} - \widetilde{V}\|_{W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})} \le Cf(\varepsilon)e^{-f(\varepsilon)d_{\inf}}.$$

In particular, if  $\varepsilon_{\text{conf},1} \in (0, \varepsilon_{\text{tub}})$  is small enough, then Proposition 2.2 (iii) and Proposition 3.8 (i) give us for  $\varepsilon \in (0, \varepsilon_{\text{conf},1})$ 

$$\begin{aligned} \| (I + \mathcal{C}_{z}\widetilde{V})^{-1} \|_{H^{1/2}(\Sigma;\mathbb{C}^{N}) \to H^{1/2}(\Sigma;\mathbb{C}^{N})} \\ \cdot \| \mathcal{C}_{z}(\widetilde{V}_{\varepsilon} - \widetilde{V}) \|_{H^{1/2}(\Sigma;\mathbb{C}^{N}) \to H^{1/2}(\Sigma;\mathbb{C}^{N})} \leq C \| \widetilde{V}_{\varepsilon} - \widetilde{V} \|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N})} \leq \frac{1}{2}. \end{aligned}$$

Moreover, Proposition 2.28 yields for  $I + C_z \widetilde{V}_{\varepsilon} = I + C_z \widetilde{V} + C_z (\widetilde{V}_{\varepsilon} - \widetilde{V})$ 

$$\|(I + C_z \widetilde{V}_{\varepsilon})^{-1}\|_{H^{1/2}(\Sigma; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)} \leq 2\|(I + C_z \widetilde{V})^{-1}\|_{H^{1/2}(\Sigma; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)}.$$
 (7.5)

Using the estimates from above, Proposition 3.6 (i) and (7.4) yield

$$\begin{split} \| (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} - (H_{\widetilde{V}_{\varepsilon}} - z)^{-1} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ & \leq \| \Phi_{z}(\widetilde{V}_{\varepsilon} - \widetilde{V})(I + \mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1} \Phi_{z}^{*} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ & + \| \Phi_{z}\widetilde{V}_{\varepsilon} \left( (I + \mathcal{C}_{z}\widetilde{V})^{-1} - (I + \mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1} \right) \Phi_{z}^{*} \|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ & \leq C \| \widetilde{V}_{\varepsilon} - \widetilde{V} \|_{W_{\infty}^{1}(\Sigma;\mathbb{C}^{N \times N})} \\ & + \| (I + \mathcal{C}_{z}\widetilde{V})^{-1} - (I + \mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1} \|_{H^{1/2}(\Sigma;\mathbb{C}^{N}) \to H^{1/2}(\Sigma;\mathbb{C}^{N})} \\ & \leq Cf(\varepsilon)e^{-f(\varepsilon)d_{\inf}} \\ & + \| (I + \mathcal{C}_{z}\widetilde{V})^{-1}\mathcal{C}_{z}(\widetilde{V}_{\varepsilon} - \widetilde{V})(I + \mathcal{C}_{z}\widetilde{V}_{\varepsilon})^{-1} \|_{H^{1/2}(\Sigma;\mathbb{C}^{N}) \to H^{1/2}(\Sigma;\mathbb{C}^{N})} \\ & \leq C(f(\varepsilon)e^{-f(\varepsilon)d_{\inf}} + \| \widetilde{V}_{\varepsilon} - \widetilde{V} \|_{W_{\infty}^{1}(\Sigma;\mathbb{C}^{N \times N})}) \\ & \leq Cf(\varepsilon)e^{-f(\varepsilon)d_{\inf}}, \end{split}$$

which completes the proof.

 $\Box$ 

**Lemma 7.2.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (7.1),  $f(\varepsilon)$  be as in (7.2),  $\varepsilon_{\text{conf},1}$  be as in the previous lemma,  $z \in \mathbb{C} \setminus \mathbb{R}$ , and  $B_0(z)$  and  $C_0(z)$  be the operators from (4.26) and (4.21). Then,

$$\begin{split} \left\| f(\varepsilon)Vq(I+B_0(z)f(\varepsilon)Vq)^{-1}C_0(z) \right\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to 1/2} &\leq Cf(\varepsilon)^2, \\ \left\| f(\varepsilon)Vq(I+B_0(z)f(\varepsilon)Vq)^{-1}C_0(z) \right\|_{L^2(\mathbb{R}^\theta;\mathbb{C}^N)\to 0} &\leq Cf(\varepsilon), \end{split}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf},1})$ .

*Proof.* The proof of both results is very similar. Thus, we only verify the first norm estimate. We give some remarks on how to show the second result at the end of the proof. By Proposition 3.15 (iii), Proposition 4.13 and Lemma 7.1, the operator  $I + B_0(z)f(\varepsilon)Vq$  is continuously invertible in  $\mathcal{B}^{1/2}(\Sigma)$  and from (4.48) we get

$$f(\varepsilon)Vq(I+B_0(z)f(\varepsilon)Vq)^{-1}C_0(z) = f(\varepsilon)Vq\cos\left((\alpha\cdot\nu)\frac{f(\varepsilon)V}{2}\right)^{-1}\exp(-i(\alpha\cdot\nu)f(\varepsilon)VQ)\mathfrak{J}(I+\mathcal{C}_z\widetilde{V}_\varepsilon)^{-1}\Phi_{\overline{z}}^*,$$

with  $\widetilde{V}_{\varepsilon}$  from Lemma 7.1 and  $Q(t) = -\frac{1}{2} + \int_{-1}^{t} q(t) dt$  for  $t \in (-1, 1)$ ; see (4.37). We know from Proposition 3.6 that  $\Phi_{\overline{z}}^*$  acts as a bounded operator from  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $H^{1/2}(\Sigma; \mathbb{C}^N)$ . Moreover, from the proof of the previous lemma, see (7.5), we also know that  $\|(I + \mathcal{C}_z \widetilde{V}_{\varepsilon})^{-1}\|_{H^{1/2}(\Sigma; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)}$  is uniformly bounded with respect to  $\varepsilon \in (0, \varepsilon_{\text{conf},1})$ . Hence, using the boundedness of  $\mathfrak{J}$  acting as an operator from  $H^{1/2}(\Sigma; \mathbb{C}^N)$  to  $\mathcal{B}^{1/2}(\Sigma)$ , see (2.10), Proposition 2.2 (iii) and Proposition 2.19 gives us

$$\begin{aligned} \left\| f(\varepsilon)Vq(I+B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z) \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})\to 1/2} \\ &\leq C \left\| f(\varepsilon)Vq\cos\left((\alpha\cdot\nu)\frac{f(\varepsilon)V}{2}\right)^{-1}\exp(-i(\alpha\cdot\nu)f(\varepsilon)VQ) \right\|_{1/2\to 1/2} \\ &= Cf(\varepsilon)\mathrm{ess\,sup}_{t\in(-1,1)} \left\| Vq(t)\cos\left((\alpha\cdot\nu)\frac{f(\varepsilon)V}{2}\right)^{-1} \\ &\quad \cdot\exp(-i(\alpha\cdot\nu)f(\varepsilon)VQ(t)) \right\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})\to H^{1/2}(\Sigma;\mathbb{C}^{N})} \\ &\leq Cf(\varepsilon)\mathrm{ess\,sup}_{t\in(-1,1)} \left\| \cos\left((\alpha\cdot\nu)\frac{f(\varepsilon)V}{2}\right)^{-1} \\ &\quad \cdot\exp(-i(\alpha\cdot\nu)f(\varepsilon)VQ(t)) \right\|_{W^{1}_{\infty}(\Sigma;\mathbb{C}^{N\times N})}. \end{aligned}$$
(7.6)

Next, we fix  $t \in (-1, 1)$ . The identity  $((\alpha \cdot \nu)V)^2 = dI_N = -|d|I_N$  and  $f(\varepsilon) > 0$  lead to

$$\cos\left((\alpha \cdot \nu)\frac{f(\varepsilon)V}{2}\right)^{-1} \exp(-i(\alpha \cdot \nu)f(\varepsilon)VQ(t)) = \frac{\cosh(f(\varepsilon)\sqrt{|d|}Q(t)) + (\alpha \cdot \nu)\frac{V}{\sqrt{d}}\sinh(f(\varepsilon)\sqrt{|d|}Q(t))}{\cosh(f(\varepsilon)\frac{\sqrt{|d|}}{2})}.$$
(7.7)

Applying Proposition 2.2 (v) and using  $\nu \in C_b^1(\Sigma; \mathbb{R}^{\theta}) \subset W_{\infty}^1(\Sigma; \mathbb{R}^{\theta})$  lets us estimate

$$\begin{split} \left\| \frac{\cosh(f(\varepsilon)\sqrt{|d|}Q(t)) + (\alpha \cdot \nu)\frac{V}{\sqrt{d}}\sinh(f(\varepsilon)\sqrt{|d|}Q(t))}{\cosh(f(\varepsilon)\frac{\sqrt{|d|}}{2})} \right\|_{W_{\infty}^{1}(\Sigma;\mathbb{C}^{N\times N})} \\ \leq \left\| \frac{\cosh(f(\varepsilon)\sqrt{|d|}Q(t))}{\cosh(f(\varepsilon)\frac{\sqrt{|d|}}{2})} \right\|_{W_{\infty}^{1}(\Sigma;\mathbb{C}^{N\times N})} + C \left\| \frac{\sinh(f(\varepsilon)\sqrt{|d|}Q(t))}{\cosh(f(\varepsilon)\frac{\sqrt{|d|}}{2})} \right\|_{W_{\infty}^{1}(\Sigma;\mathbb{C}^{N\times N})} \\ \leq C \| f(\varepsilon)\sqrt{|d|}\|_{W^{1}(\Sigma)} \left\| \left( \left\| \frac{\cosh((\cdot)Q(t))}{\cosh(\frac{1}{2})} \right\|_{W_{\infty}^{1}((0,\infty))} + \left\| \frac{\sinh((\cdot)Q(t))}{\cosh(\frac{1}{2})} \right\|_{W_{\infty}^{1}((0,\infty))} \right) \\ \leq C f(\varepsilon) \left( \left\| \frac{\cosh((\cdot)Q(t))}{\cosh(\frac{1}{2})} \right\|_{W_{\infty}^{1}((0,\infty))} + \left\| \frac{\sinh((\cdot)Q(t))}{\cosh(\frac{1}{2})} \right\|_{W_{\infty}^{1}((0,\infty))} \right). \end{split}$$
(7.8)

Furthermore,  $|Q(t)| \leq \frac{1}{2}$  for  $t \in (-1, 1)$  since  $q \geq 0$  a.e. and  $\int_{-1}^{1} q(s) ds = 1$  by (5.1). Thus,

$$\left\|\frac{\cosh((\cdot)Q(t))}{\cosh(\frac{1}{2})}\right\|_{W^1_{\infty}((0,\infty))} + \left\|\frac{\sinh((\cdot)Q(t))}{\cosh(\frac{1}{2})}\right\|_{W^1_{\infty}((0,\infty))} < \infty.$$

Hence, using (7.7) and plugging (7.8) into (7.6) yields the first inequality of the claim.

Now, we shortly comment on how to prove the second inequality. According to Proposition 3.15 (iii) and Proposition 4.13,  $I + B_0(z)f(\varepsilon)Vq$  is continuously invertible in  $\mathcal{B}^0(\Sigma)$ . Moreover, the same formula for  $f(\varepsilon)Vq(I + B_0(z)f(\varepsilon)Vq)^{-1}C_0(z)$  is valid. Since one deals with the  $\mathcal{B}^0(\Sigma)$  case now, one may estimate the norm of this expression by

$$Cf(\varepsilon) \operatorname{ess\,sup}_{t \in (-1,1)} \left\| \cos\left( (\alpha \cdot \nu) \frac{f(\varepsilon)V}{2} \right)^{-1} \exp(-i(\alpha \cdot \nu)f(\varepsilon)VQ(t)) \right\|_{L^{\infty}(\Sigma; \mathbb{C}^{N \times N})},$$

which, in turn, can be estimated with the help of (7.7) by  $Cf(\varepsilon)$ .

**Lemma 7.3.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (7.1),  $f(\varepsilon)$  be as in (7.2),  $z \in \mathbb{C} \setminus \mathbb{R}$ ,  $B_{\varepsilon}(z)$  be as in (4.8) and assume  $m \operatorname{sign}(\tau) > 0$ . Then, there exists an  $\varepsilon_{\operatorname{conf},2} \in (0, \varepsilon_{ABC})$ , where  $\varepsilon_{ABC} > 0$  is chosen as in (4.19), such that  $(I + B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1}$  is continuously invertible in  $\mathcal{B}^0(\Sigma)$  and

$$\|f(\varepsilon)Vq(I+B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1}\|_{0\to 0} \le Cf(\varepsilon)^2$$

for all  $\varepsilon \in (0, \varepsilon_{\operatorname{conf}, 2})$ .

*Proof.* We start by mentioning  $0 \in \rho(H) = \mathbb{C} \setminus ((-\infty, -|m|] \cup [|m|, \infty))$  as  $m \neq 0$ . Thus, the operator  $\widetilde{B}_{\varepsilon}(0) = B_{\varepsilon}(0)M_{\varepsilon}^{-1}$ ,  $\varepsilon \in (0, \varepsilon_{ABC})$ , with  $M_{\varepsilon}$  from (4.20), which was introduced in (4.27), is well-defined and because of (4.36) also self-adjoint in  $\mathcal{B}^0(\Sigma)$ . Now, let us explain the strategy of the proof. We split the proof into three steps. In *Step 1* we show the estimate

$$\left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-1} \right\|_{0\to 0} \le f(\varepsilon) \quad \forall \varepsilon \in (0, \varepsilon_{\varepsilon_{ABC}}),$$
(7.9)

where

$$D = \sqrt{q} \operatorname{diag}(\sqrt{|\eta + \tau|} I_{N/2}, \sqrt{|\eta - \tau|} I_{N/2}).$$

In Step 2, we use Step 1 and  $Vq = \operatorname{sign}(\tau)D\beta D$ , which is valid since  $q \ge 0$  a.e and  $d(x_{\Sigma}) = \eta(x_{\Sigma})^2 - \tau(x_{\Sigma})^2 < 0$  for all  $x_{\Sigma} \in \Sigma$  by assumption, to show that the expression  $\|f(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0\to 0}$  is bounded by  $Cf(\varepsilon)$ . Lastly, in Step 3, we perform the change from 0 to z in order to prove (7.9).

Step 1. We begin by calculating the square

$$(f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{2} = f(\varepsilon)^{-2}I + f(\varepsilon)^{-1}\operatorname{sign}(\tau)D(\beta\widetilde{B}_{\varepsilon}(0) + \widetilde{B}_{\varepsilon}(0)\beta)D + (D\widetilde{B}_{\varepsilon}(0)D)^{2}.$$

It is clear that  $(D\widetilde{B}_{\varepsilon}(0)D)^2$  is a nonnegative operator. We claim that the operator  $\operatorname{sign}(\tau)D(\beta\widetilde{B}_{\varepsilon}(0)+\widetilde{B}_{\varepsilon}(0)\beta)D$  is also nonnegative. Using (4.8) and (4.27) yields

$$\beta \widetilde{B}_{\varepsilon}(0) + \widetilde{B}_{\varepsilon}(0)\beta = \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}(\beta R(0) + R(0)\beta) U_{\varepsilon}^{*} \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} M_{\varepsilon}^{-1}$$

with R(0) being the inverse of the free Dirac operator H. Moreover, from Definition 3.2 one easily concludes

$$\beta R(0) + R(0)\beta = R(0)H\beta R(0) + R(0)\beta = R(0)\beta(-H + 2m\beta)R(0) + R(0)\beta = 2mR(0)^2.$$

Consequently, we get with (4.8) and (4.25)

$$\beta \widetilde{B}_{\varepsilon}(0) + \widetilde{B}_{\varepsilon}(0)\beta = 2mC_{\varepsilon}(0)(C_{\varepsilon}(0))^*.$$

Therefore, the assumption  $m \operatorname{sign}(\tau) > 0$  guarantees that  $\operatorname{sign}(\tau) D(\beta \widetilde{B}_{\varepsilon}(0) + \widetilde{B}_{\varepsilon}(0)\beta) D$ is a nonnegative operator in  $\mathcal{B}^0(\Sigma)$ . Thus, the norm of  $(f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-2}$ is bounded by  $f(\varepsilon)^2$ . Furthermore, since  $f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D$  is self-adjoint, we also have

$$\begin{split} \left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-1} \right\|_{0 \to 0} \\ &= \left( \left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-2} \right\|_{0 \to 0} \right)^{1/2} \le f(\varepsilon) \end{split}$$

proving (7.9).

Step 2. Next, we choose  $\varepsilon_{\text{conf},2} \in (0, \varepsilon_{ABC})$  sufficiently small such that

$$\|D(\widetilde{B}_{\varepsilon}(0) - B_{\varepsilon}(0))D\|_{0\to 0} = \|DB_{\varepsilon}(0)(M_{\varepsilon}^{-1} - I)D\|_{0\to 0} \le \frac{1}{2f(\varepsilon)}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf},2})$ . This is possible according to Lemma 4.7 and Proposition 4.10. Then, by Proposition 2.28 the operator  $f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)DB_{\varepsilon}(0)D$  is also continuously invertible in  $\mathcal{B}^{0}(\Sigma)$  and the estimate

$$\begin{split} \left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)DB_{\varepsilon}(0)D)^{-1} \right\|_{0\to 0} \\ &\leq \frac{\left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-1} \right\|_{0\to 0}}{1 - \left\| (f(\varepsilon)^{-1}\beta + \operatorname{sign}(\tau)D\widetilde{B}_{\varepsilon}(0)D)^{-1} \right\|_{0\to 0} \|\operatorname{sign}(\tau)D(B_{\varepsilon}(0) - \widetilde{B}_{\varepsilon}(0))D\|_{0\to 0}} \\ &\leq \frac{f(\varepsilon)}{1 - \frac{f(\varepsilon)}{2f(\varepsilon)}} = 2f(\varepsilon) \quad \forall \varepsilon \in (0, \varepsilon_{\operatorname{conf}, 2}) \end{split}$$

is valid. Therefore,  $I + f(\varepsilon) \operatorname{sign}(\tau) \beta DB_{\varepsilon}(0) D = f(\varepsilon) \beta (f(\varepsilon)^{-1} \beta + \operatorname{sign}(\tau) \beta DB_{\varepsilon}(0) D)$ is also continuously invertible in  $\mathcal{B}^{0}(\Sigma)$  and its inverse is uniformly bounded by 2. Hence, by Proposition 2.29 and  $Vq = \operatorname{sign}(\tau) D\beta D$  the operator  $I + B_{\varepsilon}(0) f(\varepsilon) Vq$  is also continuously invertible in  $\mathcal{B}^{0}(\Sigma)$  and

$$\begin{split} \left\| f(\varepsilon) Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1} \right\|_{0 \to 0} \\ &= \left\| f(\varepsilon) D(I + f(\varepsilon)\mathrm{sign}(\tau)\beta DB_{\varepsilon}(0)D)^{-1}\mathrm{sign}(\tau)\beta D \right\|_{0 \to 0} \qquad (7.10) \\ &\leq Cf(\varepsilon) \end{split}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf},2})$ .

Step 3. In this step we perform the change from 0 to  $z \in \mathbb{C} \setminus \mathbb{R}$ . We start with the following observation

$$B_{\varepsilon}(z) - B_{\varepsilon}(0) = \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}(R(z) - R(0)) U_{\varepsilon}^{*} \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon}$$
  
=  $z \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon} R(z) R(0) U_{\varepsilon}^{*} \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} = z C_{\varepsilon}(z) A_{\varepsilon}(0)$ 

where we used R(z) - R(0) = zR(z)R(0) and (4.8). Next, we calculate

$$(I + B_{\varepsilon}(z)f(\varepsilon)Vq)(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}$$

$$= (I + (B_{\varepsilon}(0) + zC_{\varepsilon}(z)A_{\varepsilon}(0))f(\varepsilon)Vq)(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}$$

$$= I + zC_{\varepsilon}(z)A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}$$

$$= I + z(C_{\varepsilon}(z) - C_{0}(z))A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}$$

$$+ zC_{0}(z)A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}.$$
(7.11)

Now, we aim to apply Proposition 2.30 for

$$\begin{split} \mathcal{A} &:= I + B_{\varepsilon}(z)f(\varepsilon)Vq,\\ \mathcal{A}_{0} &:= I + B_{0}(z)f(\varepsilon)Vq,\\ \mathcal{T} &:= (I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1},\\ \mathcal{K}_{1} &:= z(C_{\varepsilon}(z) - C_{0}(z))A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1},\\ \mathcal{K}_{2} &:= zC_{0}(z)A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}. \end{split}$$

Rewriting (7.11) in terms of these newly defined operators yields  $\mathcal{AT} = I + \mathcal{K}_1 + \mathcal{K}_2$ . Moreover, we can estimate  $\mathcal{K}_1$  and  $(\mathcal{A}_0 - \mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2$  with Proposition 4.8, Proposition 4.9, Proposition 4.10 and (7.10) by

$$\begin{aligned} \|\mathcal{K}_1\|_{0\to 0} &= \|z(C_{\varepsilon}(z) - C_0(z))A_{\varepsilon}(0))f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0\to 0} \\ &\leq Cf(\varepsilon)\|C_{\varepsilon}(z) - C_0(z)\|_{L^2(\mathbb{R}^{\theta};\mathbb{C}^N)\to 0}\|A_{\varepsilon}(0)\|_{0\to L^2(\mathbb{R}^{\theta};\mathbb{C}^N)} \\ &\leq Cf(\varepsilon)\varepsilon^{1/2-r} \end{aligned}$$

and

$$\begin{aligned} \|(\mathcal{A}_{0} - \mathcal{A})\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\|_{0 \to 0} \\ &= \|z(B_{0}(z) - B_{\varepsilon}(z))f(\varepsilon)Vq(I + B_{0}(z)f(\varepsilon)Vq)^{-1} \\ &\cdot C_{0}(z)A_{\varepsilon}(0)f(\varepsilon)Vq(I + B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0 \to 0} \\ &\leq Cf(\varepsilon)\|B_{\varepsilon}(z) - B_{0}(z)\|_{1/2 \to 0} \\ &\cdot \|f(\varepsilon)Vq(I + B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to 1/2}\|A_{\varepsilon}(0)\|_{0 \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq Cf(\varepsilon)^{3}\varepsilon^{1/2 - r} \end{aligned}$$

for  $\varepsilon \in (0, \varepsilon_{\text{conf},2})$  as  $\varepsilon_{\text{conf},2}$  is chosen such that  $\varepsilon_{\text{conf},2} \in (0, \varepsilon_{ABC})$ . In particular, since  $f(\varepsilon)^4 \varepsilon^{1/2-r} \to 0$  for  $\varepsilon \to 0$ , see (7.2), we have  $\|\mathcal{K}_1 + (\mathcal{A}_0 - \mathcal{A})\mathcal{A}_0^{-1}\mathcal{K}_2\|_{0\to 0} \leq \frac{1}{2}$  for all  $\varepsilon \in (0, \varepsilon_{\text{conf},2})$  if  $\varepsilon_{\text{conf},2} \in (0, \varepsilon_{ABC})$  is sufficiently small. In turn, Proposition 2.30 implies that  $I + B_{\varepsilon}(z)f(\varepsilon)Vq$  has the bounded right inverse

$$\left(\mathcal{T}-\mathcal{A}_{0}^{-1}\mathcal{K}_{2}
ight)\left(I+\mathcal{K}_{1}+\left(\mathcal{A}_{0}-\mathcal{A}
ight)\mathcal{A}_{0}^{-1}\mathcal{K}_{2}
ight)^{-1},$$

which is its unique inverse according to Proposition 4.1 (i). Hence, Proposition 2.28,

Proposition 4.9, Lemma 7.2 and (7.10) yield

$$\begin{split} \|f(\varepsilon)Vq(I+B_{\varepsilon}(z)Vq)^{-1}\|_{0\to0} &\leq \|f(\varepsilon)Vq(\mathcal{T}-\mathcal{A}_{0}^{-1}\mathcal{K}_{2})\|_{0\to0} \\ &\quad \cdot \left\| \left(I+\mathcal{K}_{1}+(\mathcal{A}_{0}-\mathcal{A})\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\right)^{-1}\right\|_{0\to0} \\ &\leq \frac{\|f(\varepsilon)Vq(\mathcal{T}-\mathcal{A}_{0}^{-1}\mathcal{K}_{2})\|_{0\to0}}{1-\|\mathcal{K}_{1}+(\mathcal{A}_{0}-\mathcal{A})\mathcal{A}_{0}^{-1}\mathcal{K}_{2}\|_{0\to0}} \\ &\leq 2\|f(\varepsilon)Vq(\mathcal{T}-\mathcal{A}_{0}^{-1}\mathcal{K}_{2})\|_{0\to0} \\ &= 2\|f(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1} \\ &\quad -zf(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}C_{0}(z)\mathcal{A}_{\varepsilon}(0)f(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0\to0} \\ &\leq C\left(1+|z|\|f(\varepsilon)Vq(I+B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\mathcal{A}_{\varepsilon}(0)\|_{0\to0}\right) \\ &\quad \cdot \|f(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0\to0} \\ &\leq C\left(1+\|f(\varepsilon)Vq(I+B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})\to0}\right) \\ &\quad \cdot \|f(\varepsilon)Vq(I+B_{\varepsilon}(0)f(\varepsilon)Vq)^{-1}\|_{0\to0} \\ &\leq C(1+f(\varepsilon))f(\varepsilon) \\ &\leq Cf(\varepsilon)^{2}. \end{split}$$

After providing all these preliminary results, we are ready to prove the main theorem of Section 7.1.

**Theorem 7.4.** Let q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (7.1),  $V_{\varepsilon}$  be defined by (4.3),  $f(\varepsilon)$  be as in (7.2) (with  $r \in (0, 1/2)$ ),  $d_{\inf}$  be as in Lemma 7.1 and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, set  $\widetilde{V} = \frac{2}{\sqrt{|d|}}V$ , where  $d = \eta^2 - \tau^2$ . Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and there exists an  $\varepsilon_{\text{conf}} > 0$  and a C > 0 such that

$$\left\| (H_{f(\varepsilon)V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C \left( f(\varepsilon)^{4} \varepsilon^{1/2 - r} + f(\varepsilon) e^{-d_{\inf}f(\varepsilon)} \right)$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf}})$ . In particular,  $H_{f(\varepsilon)V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* Before we start, let us mention that we can w.l.o.g. assume  $\operatorname{sign}(\tau)m > 0$  according to Proposition 2.24. Let  $\varepsilon_{\operatorname{conf}} := \min\{\varepsilon_{\operatorname{conf},1}, \varepsilon_{\operatorname{conf},2}\}$  with  $\varepsilon_{\operatorname{conf},1} > 0$  and  $\varepsilon_{\operatorname{conf},2} > 0$  from Lemma 7.1 and Lemma 7.3, respectively. According to Lemma 7.1  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and it suffices to show

$$\left\| (H_{f(\varepsilon)V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}_{\varepsilon}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq Cf(\varepsilon)^{4} \varepsilon^{1/2-r},$$

where  $\widetilde{V}_{\varepsilon} = \operatorname{tanc}(\frac{\sqrt{d_{\varepsilon}}}{2})f(\varepsilon)V$  with  $d_{\varepsilon} = \eta_{\varepsilon}^2 - \tau_{\varepsilon}^2$  as in (7.3). Similarly as in the proof of Theorem 4.15, applying Proposition 4.8, Proposition 4.9, Proposition 4.10, Lemma 7.2 and Lemma 7.3 lets us estimate

$$\begin{split} \left\| (H_{f(\varepsilon)V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}_{\varepsilon}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq \|A_{\varepsilon}(z)f(\varepsilon)Vq(I + B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1}(C_{\varepsilon}(z) - C_{0}(z))\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \|A_{\varepsilon}(z)f(\varepsilon)Vq(I + B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1} \\ &\cdot (B_{\varepsilon}(z) - B_{0}(z))f(\varepsilon)Vq(I + B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &+ \|(A_{\varepsilon}(z) - A_{0}(z))f(\varepsilon)Vq(I + B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \\ &\leq C\varepsilon^{1/2-r} \left(\|f(\varepsilon)Vq(I + B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1}\|_{0\to 0} \\ &+ \|f(\varepsilon)Vq(I + B_{\varepsilon}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to 1/2} \\ &+ \|f(\varepsilon)Vq(I + B_{0}(z)f(\varepsilon)Vq)^{-1}C_{0}(z)\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to 0} \right) \\ &\leq C\varepsilon^{1/2-r}(f(\varepsilon)^{2} + f(\varepsilon)^{4} + f(\varepsilon)) \\ &\leq C\varepsilon^{1/2-r}(f(\varepsilon)^{2} + f(\varepsilon)^{4} + f(\varepsilon)) \\ &\leq C\varepsilon^{1/2-r}f(\varepsilon)^{4} \end{split}$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf}})$ .

We are now in a position to answer the question posed in the beginning of the current section, namely, whether there is a way to approximate a given Dirac operator with  $\delta$ -shell potential which induces confinement by Dirac operators with strongly localized potentials.

**Corollary 7.5.** Let q be as in (5.1),  $\widetilde{V} = \widetilde{\eta}I_N + \widetilde{\tau}\beta$  with  $\widetilde{\eta}, \widetilde{\tau} \in C_b^1(\Sigma; \mathbb{R})$  satisfy  $\widetilde{d} = \widetilde{\eta}^2 - \widetilde{\tau}^2 = -4$  and  $\operatorname{sign}(\widetilde{\tau}) = \operatorname{const.}, f(\varepsilon)$  be as in (7.2) (with  $r \in (0, 1/2)$ ) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Moreover, set  $V = v\widetilde{V}$  for a  $v \in C_b^1(\Sigma; \mathbb{R})$  with  $\inf_{x_{\Sigma} \in \Sigma} v(x_{\Sigma}) > 0$ , and let  $V_{\varepsilon}$  be defined by (4.3) and d<sub>inf</sub> be as in Lemma 7.1. Then, the operator  $H_{\widetilde{V}\delta_{\Sigma}}$  is self-adjoint and there exists an  $\varepsilon_{\operatorname{conf}} > 0$  and a C > 0 such that

$$\left\| (H_{f(\varepsilon)V_{\varepsilon}} - z)^{-1} - (H_{\widetilde{V}\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C \left( f(\varepsilon)^{4} \varepsilon^{1/2 - r} + f(\varepsilon)e^{-d_{\inf}f(\varepsilon)} \right)$$

for all  $\varepsilon \in (0, \varepsilon_{\text{conf}})$ . In particular,  $H_{f(\varepsilon)V_{\varepsilon}}$  converges to  $H_{\widetilde{V}\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* The conditions in (7.1) are fulfilled by definition. Moreover, using  $\tilde{d} = -4$  and  $d = \eta^2 - \tau^2 = v^2 (\tilde{\eta}^2 - \tilde{\tau}^2) = v^2 \tilde{d}$  yields

$$\frac{2}{\sqrt{|d|}}V = \frac{2}{v\sqrt{|\widetilde{d}|}}v\widetilde{V} = \widetilde{V}.$$

Therefore, the assertion follows from Theorem 7.4.

# 7.2 Spectra of Dirac operators with strongly localized potentials

In this section we investigate the discrete and essential spectrum of  $H_{V_{\varepsilon}}$ . Since many spectral results depend on the parameter  $m \in \mathbb{R}$ , we write in this section H(m),  $H_{\tilde{V}\delta_{\Sigma}}(m)$  and  $H_{V_{\varepsilon}}(m)$  instead of H,  $H_{\tilde{V}\delta_{\Sigma}}$  and  $H_{V_{\varepsilon}}$ , see Definition 3.2, Definition 3.12 and (4.4), to emphasize the *m*-dependence of these operators.

We start by presenting results for the case that  $\Sigma$  is bounded. The resolvent difference of  $H_{V_{\varepsilon}}(m) = H(m) + V_{\varepsilon}$  and H(m) is for  $z \in \rho(H(m)) \cap \rho(H_{V_{\varepsilon}}(m))$  given by

$$(H(m) - z)^{-1} - (H_{V_{\varepsilon}}(m) - z)^{-1} = (H_{V_{\varepsilon}}(m) - z)^{-1} V_{\varepsilon}(H(m) - z)^{-1}$$

In the case that  $\Sigma$  is bounded,  $\Omega_{\varepsilon}$ , see Definition 2.7, is also bounded. Thus,  $V_{\varepsilon}$  is compactly supported and, in turn, induces a compact operator from  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$ ; cf. [54, Theorem 3.27 (ii)]. Moreover,  $(H(m) - z)^{-1}$  is a bounded operator from  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta}; \mathbb{C}^N)$ ; see Proposition 3.3 (iii). Thus, also the resolvent difference is compact for  $z \in \rho(H(m)) \cap \rho(H_{V_{\varepsilon}}(m))$ . Therefore,

$$\sigma_{\rm ess}(H_{V_{\varepsilon}}(m)) = \sigma_{\rm ess}(H) = \sigma(H) = (-\infty, -|m|] \cup [|m|, \infty)$$

and  $\sigma_{\text{disc}}(H_{V_{\varepsilon}}(m)) \subset (-|m|, |m|)$ . Next, we focus on special cases which allow us to make more precise statements concerning the discrete spectrum. Here, we are particularly interested in cases which guarantee the existence of discrete eigenvalues.

**Proposition 7.6.** Let  $\theta = 3$ ,  $\Sigma \subset \mathbb{R}^3$  be the boundary of a bounded  $C^{\infty}$ -smooth domain,  $m \in \mathbb{R}$ ,  $V = \tau I_4$  with  $\tau \in \mathbb{R}$  and  $H_{V_{\varepsilon}}(m)$  be as in the beginning of Section 7.2. Then, the following holds:

- (i) If  $\tau m \geq 0$  and  $\delta \in (0, |m|)$ , then  $\sigma_{\text{disc}}(H_{V_{\varepsilon}}(m)) \cap [-|m| + \delta, |m| \delta] = \emptyset$  for  $\varepsilon > 0$  sufficiently small.
- (ii) If  $\tau \neq 0$  and  $M \in \mathbb{N}$ , then there exists for sufficiently large  $-\operatorname{sign}(\tau)m > 0$  an  $\varepsilon_m > 0$  such that for all  $\varepsilon \in (0, \varepsilon_m)$  the operator  $H_{V_{\varepsilon}}(m)$  has at least M discrete eigenvalues counted with multiplicities.

Proof. If  $V = \tau I_4$ , then  $d = -\tau^2 \leq 0$  and therefore it follows from Theorem 5.20 that  $H_{V_{\varepsilon}}(m)$  converges for  $\varepsilon \to 0$  to  $H_{\widetilde{V}\delta_{\Sigma}}(m)$  in the norm resolvent sense, where  $\widetilde{V} = \widetilde{\tau}\beta$  with

$$\widetilde{\tau} = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)\tau = \operatorname{tanc}\left(\frac{\sqrt{-\tau^2}}{2}\right)\tau = \frac{2\operatorname{tanh}\left(\frac{|\tau|}{2}\right)}{|\tau|}\tau = 2\operatorname{tanh}\left(\frac{\tau}{2}\right)$$

Having established the convergence of  $H_{V_{\varepsilon}}(m)$  we are able to prove (i) and (ii). We start with (i). If  $\tau m \geq 0$ , then also  $\tilde{\tau} m \geq 0$  and hence according to [37, Proposition 3.6 b)] the discrete spectrum of  $H_{\tilde{V}\delta_{\Sigma}}(m)$  is empty. Moreover, since  $\Sigma$  is bounded we have  $\sigma_{\text{ess}}(H_{V_{\varepsilon}}(m)) = (-\infty, -|m|] \cup [|m|, \infty)$  and thus by Proposition 2.25 (ii)  $\sigma(H_{\widetilde{V}\delta_{\Sigma}}(m)) = \sigma_{\text{ess}}(H_{\widetilde{V}\delta_{\Sigma}}(m)) = (-\infty, -|m|] \cup [|m|, \infty)$ . Consequently, Proposition 2.27 (i) implies

$$\sigma_{\text{disc}}(H_{V_{\varepsilon}}(m)) \cap [-|m| + \delta, |m| - \delta] = \sigma(H_{V_{\varepsilon}}(m)) \cap [-|m| + \delta, |m| - \delta] = \emptyset$$

for  $\varepsilon > 0$  sufficiently small.

Next, we consider (ii). We obtain from [37, Theorem 2.3 (e) and Corollary 4.4] that  $H_{\tilde{V}\delta_{\Sigma}}(m)$  has at least M discrete eigenvalues (counted with multiplicities) for sufficiently large  $-\text{sign}(\tau)m = -\text{sign}(\tilde{\tau})m > 0$ . Now, applying Proposition 2.26 yields assertion (ii).

After considering purely Lorentz scalar interaction strengths in Proposition 7.6, we consider purely electrostatic interaction strengths in the next statement.

**Proposition 7.7.** Let  $\theta = 3$ ,  $\Sigma \subset \mathbb{R}^3$  be the unit sphere, m = 1,  $V = \eta I_4$  with  $\eta \in (-\frac{\pi}{2}, \frac{\pi}{2})$ . Then, there holds the following:

- (i) If  $|\eta| < 2 \arctan(\sqrt{5} 2)$  and  $\delta \in (0, 1)$ , then  $\sigma_{\text{disc}}(H_{V_{\varepsilon}}(1)) \cap [-1 + \delta, 1 \delta] = \emptyset$ for  $\varepsilon > 0$  sufficiently small.
- (ii) If  $|\eta| > 2 \arctan(\sqrt{5} 2)$ , then  $\sigma_{\text{disc}}(H_{V_{\varepsilon}}(1)) \neq \emptyset$  for  $\varepsilon > 0$  sufficiently small.

*Proof.* The assumption  $\eta \in \left(-\frac{\pi}{2}, \frac{\pi}{2}\right)$  implies  $d = \eta^2 < \frac{\pi^2}{4}$  and therefore Theorem 5.20 guarantees that  $H_{V_{\varepsilon}}(1)$  converges for  $\varepsilon \to 0$  to  $H_{\widetilde{V}\delta_{\Sigma}}(1)$  in the norm resolvent sense, where  $\widetilde{V} = \widetilde{\eta}I_4$  with

$$\widetilde{\eta} = \operatorname{tanc}\left(\frac{\sqrt{d}}{2}\right)\eta = \operatorname{tanc}\left(\frac{\sqrt{\eta^2}}{2}\right)\eta = \frac{2\operatorname{tan}\left(\frac{|\eta|}{2}\right)}{|\eta|}\eta = 2\operatorname{tan}\left(\frac{\eta}{2}\right).$$

If  $|\eta| < 2 \arctan(\sqrt{5} - 2)$ , then  $|\tilde{\eta}| < 2\sqrt{5} - 4$  and hence [6, Theorem 1.1 and Lemma 5.2 (iii)] show that the discrete spectrum of  $H_{\tilde{V}\delta_{\Sigma}}(1)$  is empty. Now, the same arguments as in the proof of Proposition 7.6 (i) yield assertion (i). If the inequality  $|\eta| > 2 \arctan(\sqrt{5} - 2)$  holds, then  $|\tilde{\eta}| > 2\sqrt{5} - 4$  and thus  $\sigma_{\text{disc}}(H_{\tilde{V}\delta_{\Sigma}}(1)) \neq \emptyset$  by [5, (4.35)–(4.36) and the text below]. Consequently, Proposition 2.26 implies (ii).

Next, we consider the case where  $\Sigma$  is unbounded. In particular, we assume that  $\Sigma$  is a non-self-intersecting  $C^{\infty}$ -smooth curve in  $\mathbb{R}^2$  which coincides outside of a compact set with a broken line with opening angle  $2\omega$ ,  $\omega \in (0, \frac{\pi}{2})$ , which is given by

$$\Gamma_{\omega} = \{r(\cos(\omega), \sin(\omega)) : r > 0\} \cup \{r(\cos(\omega), -\sin(\omega)) : r > 0\}.$$

In this setting the discrete eigenvalues of  $H_{\tilde{V}\delta_{\Sigma}}(m)$  were studied in [10, Section 2.4]. In the upcoming two statements we use Theorem 5.20 to transfer these results from  $H_{\tilde{V}\delta_{\Sigma}}(m)$  to  $H_{V_{\varepsilon}}(m)$ . This allows us to provide conditions under which  $H_{V_{\varepsilon}}(m)$  admits discrete eigenvalues. **Proposition 7.8.** Let  $\theta = 2$  and  $\Sigma$  be a non-self-intersecting  $C^{\infty}$ -curve which coincides outside of a compact set with  $\widetilde{\Gamma}_{\omega}$  for a  $\omega \in (0, \frac{\pi}{2})$ , a < 0,  $\eta = \tau = \frac{a}{m}$  and  $V = V(m) = \eta I_2 + \tau \beta$  for m > 0. Then, the following holds:

- (i) If  $\delta \in (0, m \frac{m^2 a^2}{m^2 + a^2})$ , then  $\sigma_{\text{ess}}(H_{V_{\varepsilon}}(m)) \cap [-m + \delta, m \frac{m^2 a^2}{m^2 + a^2} \delta] = \emptyset$  for sufficiently small  $\varepsilon > 0$ .
- (ii) If m > 0 is sufficiently large, then there exists an  $\varepsilon_m > 0$  such that for  $\varepsilon \in (0, \varepsilon_m)$  the discrete spectrum of  $H_{V_{\varepsilon}}(m)$  is not empty.

*Proof.* Under the assumptions of this proposition we have  $d = \eta^2 - \tau^2 = \frac{a^2}{m^2} - \frac{a^2}{m^2} = 0$ and therefore  $(\tilde{\eta}, \tilde{\tau}) = \operatorname{tanc}(\frac{\sqrt{d}}{2})(\eta, \tau) = (\eta, \tau)$ . Thus, Theorem 5.20 implies that  $H_{V_{\varepsilon}}(m)$  converges for  $\varepsilon \to 0$  to  $H_{V\delta_{\Sigma}}(m)$  in the norm resolvent sense. Moreover, [10, Theorem 2.3 (ii) (c)] and rescaling the results from [10, Theorem 2.7] yields

$$\sigma_{\mathrm{ess}}(H_{V\delta_{\Sigma}}(m)) = (-\infty, m] \cup [m\frac{m^2 - a^2}{m^2 + a^2}, \infty)$$

and  $\sigma_{\text{disc}}(H_{V\delta_{\Sigma}}(m)) \neq \emptyset$  for sufficiently large m > 0, respectively. Combining these results with Proposition 2.27 and Proposition 2.26 concludes the proof of (i) and (ii), respectively.

**Proposition 7.9.** Let  $\theta = 2$  and  $\Sigma$  be a non-self-intersecting  $C^{\infty}$ -curve which coincides outside of a compact set with  $\widetilde{\Gamma}_{\omega}$  for a  $\omega \in (0, \frac{\pi}{2}), \tau < 0, V = \tau\beta$  and  $\widetilde{\tau} = 2 \tanh(\tau/2)$ . Moreover, assume that  $\tau m < 0$ . Then, the following holds:

- (i) If  $\delta \in (0, |m| \frac{4-\tilde{\tau}^2}{4+\tilde{\tau}^2})$ , then  $\sigma_{\text{ess}}(H_{V_{\varepsilon}}(m)) \cap [-|m| \frac{4-\tilde{\tau}^2}{4+\tilde{\tau}^2} + \delta, |m| \frac{4-\tilde{\tau}^2}{4+\tilde{\tau}^2} \delta] = \emptyset$  for sufficiently small  $\varepsilon > 0$ .
- (ii) For every  $M \in \mathbb{N}$  exists an  $\omega_M \in (0, \frac{\pi}{2})$  and an  $\varepsilon_M > 0$  such that for  $\varepsilon \in (0, \varepsilon_M)$  the operator  $H_{V_{\varepsilon}}(m)$  has at least M discrete eigenvalues with multiplicities taken into account.

Proof. If  $V = \tau I_4$ , then it follows from Theorem 5.20 that  $H_{V_{\varepsilon}}(m)$  converges for  $\varepsilon \to 0$ to  $H_{\widetilde{V}\delta_{\Sigma}}(m)$  in the norm resolvent sense, where  $\widetilde{V} = \widetilde{\tau}\beta$  with  $\widetilde{\tau} = 2 \tanh\left(\frac{\tau}{2}\right) \in (-2, 2)$ . This also shows that the assumption  $\tau m < 0$  implies  $\widetilde{\tau}m < 0$ . Hence, according to [10, Corollary 2.5] the essential spectrum of  $H_{\widetilde{V}\delta_{\Sigma}}(m)$  is given by

$$\sigma_{\mathrm{ess}}(H_{V_{\varepsilon}}(m)) = (-\infty, -|m|_{\frac{4-\tilde{\tau}^2}{4+\tilde{\tau}^2}}] \cup [|m|_{\frac{4-\tilde{\tau}^2}{4+\tilde{\tau}^2}}, \infty)$$

and therefore Proposition 2.27 (ii) implies assertion (i). By [10, Theorem 2.8] and [37, Theorem 2.3 (e)] there exists an  $\omega_M \in (0, \frac{\pi}{2})$  such that  $H_{\tilde{V}\delta_{\Sigma}}(m)$  has at least M discrete eigenvalues. Thus, assertion (ii) follows from Proposition 2.26

We conclude this section by considering the case where the interaction matrix has the form  $f(\varepsilon)V$  with  $V = \eta I_N + \tau \beta$ ,  $\eta, \tau \in \mathbb{R}$ , such that  $d = \eta^2 - \tau^2 < 0$  and f is the monotonically decreasing function from (7.2). Recall that in this case the conditions of Theorem 7.4 are met and therefore the operator  $H_{f(\varepsilon)V_{\varepsilon}}(m)$  converges in the norm resolvent sense to  $H_{\tilde{V}\delta_{\Sigma}}(m)$  with  $\tilde{V} = \frac{2}{\sqrt{|d|}}V$ , which is a Dirac operator that induces confinement; cf. Section 7.1.

**Proposition 7.10.** Let  $\theta = 3$ ,  $\Sigma \subset \mathbb{R}^3$  be the boundary of a bounded  $C^{\infty}$ -smooth domain,  $V = \eta I_4 + \tau \beta$  with  $\eta, \tau \in \mathbb{R}$  fulfil  $d = \eta^2 - \tau^2 < 0$  and  $m \neq 0$ . Then, the following holds:

- (i) If  $m\tau > 0$  and  $\delta \in (0, |m|)$ , then  $\sigma_{\text{disc}}(H_{f(\varepsilon)V_{\varepsilon}}(m)) \cap [-|m| + \delta, |m| \delta] = \emptyset$  for sufficiently small  $\varepsilon > 0$ .
- (ii) If  $\eta = 0$  and  $M \in \mathbb{N}$ , then for sufficiently large  $-\operatorname{sign}(\tau)m > 0$  exists an  $\varepsilon_m > 0$  such that for  $\varepsilon \in (0, \varepsilon_m)$  the operator  $H_{f(\varepsilon)V_{\varepsilon}}(m)$  has at least M discrete eigenvalues counted with multiplicities.

Proof. According to Theorem 7.4  $H_{f(\varepsilon)V_{\varepsilon}}(m)$  converges for  $\varepsilon \to 0$  to  $H_{\widetilde{V}\delta_{\Sigma}}(m)$  in the norm resolvent sense with  $\widetilde{V} = \widetilde{\eta}I_4 + \widetilde{\tau}\beta$ , where  $(\widetilde{\eta}, \widetilde{\tau}) = \frac{2}{\sqrt{|d|}}(\eta, \tau)$ . Hence, if  $\tau > 0$ , then  $\widetilde{\tau} > 0$  and therefore [11, Proposition 2.2] implies that the discrete spectrum of  $H_{\widetilde{V}\delta_{\Sigma}}$  is empty if m > 0. Thus, for m > 0 assertion (i) follows from Proposition 2.27; cf. the proof of Proposition 7.6 (i) for a analogous more detailed proof. It remains to consider the case m < 0. The anticommutation rules from (3.1) imply  $H_{f(\varepsilon)V_{\varepsilon}}(m) = -\beta H_{-f(\varepsilon)V_{\varepsilon}}(-m)\beta$  and thus this case can be reduced to the case m > 0. Now, let us consider (ii). Under the set of assumptions in (ii), [3, Corollary 1.15] and [37, Remark 2.1, Theorem 2.3 (e)] imply that  $H_{\widetilde{V}\delta_{\Sigma}}(m)$  has at least M discrete eigenvalues for sufficiently large  $-\operatorname{sign}(\tau)m$ . Consequently, (ii) follows from Proposition 2.26.

### 8 Convergence of Dirac operators with semilocal potentials

In this chapter we approximate Dirac operators with  $\delta$ -shell potentials by Dirac operators with so-called semilocal potentials. Before we define these semilocal potentials, we motivate their definition by results regarding one-dimensional Dirac operators.

Recall that  $\delta_0$  denotes the  $\delta$ -potential supported in the point  $\{0\}$ . Furthermore, let  $h \in L^{\infty}(\mathbb{R};\mathbb{R}) \cap L^1(\mathbb{R};\mathbb{R})$  with  $\int_{\mathbb{R}} h(x) dx = 1$  and  $h_{\varepsilon} := \frac{1}{\varepsilon} h(\frac{\varepsilon}{\varepsilon})$  for  $\varepsilon > 0$ . Then, the multiplication operator induced by  $h_{\varepsilon}$  and the projection operator defined by  $u \mapsto h_{\varepsilon}(u,h_{\varepsilon})_{L^2(\mathbb{R})}$  converge to  $\delta_0$  viewed as operators from  $C_0^{\infty}(\mathbb{R})$  to  $(C_0^{\infty}(\mathbb{R}))'$ , i.e.

$$\begin{split} \lim_{\varepsilon \to 0} |(C_0^{\infty}(\mathbb{R}))' \langle h_{\varepsilon} u, v \rangle_{C_0^{\infty}(\mathbb{R})} &= |(C_0^{\infty}(\mathbb{R}))' \langle \delta_0 u, v \rangle_{C_0^{\infty}(\mathbb{R})} \\ &= u(0)v(0) \\ &= \lim_{\varepsilon \to 0} |(C_0^{\infty}(\mathbb{R}))' \langle h_{\varepsilon}, u \rangle_{C_0^{\infty}(\mathbb{R})} |(C_0^{\infty}(\mathbb{R}))' \langle h_{\varepsilon}, v \rangle_{C_0^{\infty}(\mathbb{R})} \\ &= \lim_{\varepsilon \to 0} |(C_0^{\infty}(\mathbb{R}))' \langle h_{\varepsilon}(u, h_{\varepsilon})_{L^2(\mathbb{R})}, v \rangle_{C_0^{\infty}(\mathbb{R})} \end{split}$$

for all  $u, v \in C_0^{\infty}(\mathbb{R})$ . Similarly as in the multidimensional case,  $H + Vh_{\varepsilon}$  converges to  $H_{\tilde{V}\delta_{\Sigma}}$  in the norm (or strong) resolvent sense for  $V = V^* \in \mathbb{C}^{2\times 2}$ ; see [40, 41, 42, 67, 72] and Section 1.2. Surprisingly, it was shown in [34, Section 4] and [67] that  $H + Vh_{\varepsilon}(\cdot, h_{\varepsilon})_{L^2(\mathbb{R})}$  converges to  $H_{V\delta_{\Sigma}}$  in the norm resolvent sense, i.e. for these kinds of nonlocal potentials there is no rescaling necessary. In [35] this idea was picked up and used as an inspiration for the two and three-dimensional setting. In the mentioned paper they considered (generalizations of) Dirac operators of the type  $H + q_{\varepsilon}V(\cdot, q_{\varepsilon})_{L^2(\mathbb{R}^{\theta})}$ , where  $V = V^* \in \mathbb{C}^{N \times N}$  and

$$q_{\varepsilon}(x) := \begin{cases} \frac{1}{\varepsilon}q\left(\frac{t}{\varepsilon}\right), & x = x_{\Sigma} + t\nu(x_{\Sigma}) \in \Omega_{\varepsilon}, \\ 0, & x \notin \Omega_{\varepsilon}, \end{cases}$$

with q as in (4.1). It turns out that this family of operators converges to the operator which is formally given by  $H + V\delta_{\Sigma} C_0^{\infty}(\mathbb{R}^{\theta}) \langle \delta_{\Sigma}, \cdot \rangle_{C_0^{\infty}}(\mathbb{R}^{\theta})$  and can be realized as an unperturbed Dirac operator on  $\Omega_+ \cup \Omega_-$  and via nonlocal transmission conditions on  $\Sigma$ , i.e. transmission conditions which involve the integral over  $\Sigma$ ; see also [33]. Hence, on the one hand no renormalization of V is necessary but on the other hand the limit operator is not the operator  $H_{V\delta_{\Sigma}}$ , which is the operator we aim to approximate. This leads us to the definition of so-called semilocal potentials. They behave locally with respect to the surface  $\Sigma$  and nonlocally with respect to the normal direction of  $\Sigma$  and allow us to approximate  $H_{V\delta_{\Sigma}}$  without any rescaling.

Let us recall our general setting from the beginning of Chapter 4. Let  $\Sigma = \partial \Omega_{\pm} \subset \mathbb{R}^{\theta}$ be a special  $C^2$ -surface as in Definition 2.1,  $\Omega_{\varepsilon}$  and  $\iota$  be as in Definition 2.7, and  $\varepsilon_{\text{tub}} \in (0, \infty)$  be chosen as in Proposition 2.12. Moreover, recall from (4.1) and (4.2) that  $q \in L^{\infty}((-1, 1); \mathbb{R})$  with  $\int_{-1}^{1} q(t) dt = 1$  and  $V = V^* \in W^1_{\infty}(\Sigma; \mathbb{C}^N)$ . In this setting we define for  $\varepsilon \in (0, \varepsilon_{\text{tub}})$ 

$$\frac{V_{\varepsilon}}{V_{\varepsilon}}: L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}),$$

$$\underline{V_{\varepsilon}}u(x) := \begin{cases}
\frac{1}{\varepsilon}q\left(\frac{t}{\varepsilon}\right)V(x_{\Sigma})\int_{-1}^{1}u(\iota(x_{\Sigma}, \varepsilon s))q(s) & \\ & \cdot \det(I - s\varepsilon W(x_{\Sigma}))\,ds, \quad x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}, \\
0, & x \notin \Omega_{\varepsilon}.
\end{cases}$$
(8.1)

Note that the expression det $(I - s \in W(x_{\Sigma}))$ , which is defined below Definition 2.11 and stems from the usage of the tubular coordinates  $(x_{\Sigma}, t)$ , plays a secondary role as it converges to one for  $\varepsilon \to 0$ ; see Proposition 2.12. However, the term is necessary to guarantee the self-adjointness of  $V_{\varepsilon}$ .

#### 8.1 General interactions

In this section we consider  $H_{\underline{V_{\varepsilon}}} = H + \underline{V_{\varepsilon}}$  for general  $V = V^* \in W^1_{\infty}(\Sigma; \mathbb{C}^{N \times N})$ . It turns out that we can represent the resolvent of  $H_{\underline{V_{\varepsilon}}}$  in a similar way as the resolvent of  $H_{V_{\varepsilon}}$  in Proposition 4.1. This allows us to transfer the convergence results from the local to the semilocal case.

We use the operators  $\mathfrak{J}, \mathcal{I}_{\varepsilon}, \mathcal{S}_{\varepsilon}, U_{\varepsilon}$  and  $M_{\varepsilon}$ , see (2.10), (4.5)–(4.7) and (4.20), to express  $V_{\varepsilon}u$  for  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and  $x = \iota(x_{\Sigma}, t) \in \Omega_{\varepsilon}$  by

$$\begin{split} \underline{V}_{\varepsilon}u(x) &= \frac{1}{\varepsilon}q\left(\frac{t}{\varepsilon}\right)V(x_{\Sigma})\int_{-1}^{1}u(\iota(x_{\Sigma},\varepsilon s))q(s)\det(I-s\varepsilon W(x_{\Sigma}))\,ds\\ &= \frac{1}{\varepsilon}q\left(\frac{t}{\varepsilon}\right)V(x_{\Sigma})\int_{-1}^{1}q(s)\left(M_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)(s)(x_{\Sigma})\,ds\\ &= \frac{1}{\varepsilon}q\left(\frac{t}{\varepsilon}\right)\left(V\mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)(x_{\Sigma})\\ &= \frac{1}{\varepsilon}\left(q\mathfrak{J}V\mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)\left(\frac{t}{\varepsilon}\right)(x_{\Sigma})\\ &= \frac{1}{\sqrt{\varepsilon}}\left(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}q\mathfrak{J}V\mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)(x)\\ &= \left(U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}q\mathfrak{J}V\mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)(x). \end{split}$$

For  $u \in L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  and  $x \notin \Omega_{\varepsilon}$  we trivially have

$$\underline{V_{\varepsilon}}u(x) = 0 = \left(U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}q\mathfrak{J}V\mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}u\right)(x)$$

Thus,

$$\underline{V_{\varepsilon}} = U_{\varepsilon}^* \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} q \mathfrak{J} V \mathfrak{J}^* q M_{\varepsilon} \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}.$$

This representation implies that  $\underline{V}_{\varepsilon}$  is a bounded operator in  $L^2(\mathbb{R}^{\theta}, \mathbb{C}^N)$  and with the help of  $(\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon})^* = M_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}$ , see the lines above (4.25), one gets that  $\underline{V}_{\varepsilon}$  is selfadjoint. Hence, similarly as in Section 4.1, we define for  $z \in \rho(H)$ ,  $R(z) = (H - z)^{-1}$ and  $\varepsilon \in (0, \varepsilon_{\text{tub}})$ 

$$\underline{A_{\varepsilon}}(z) := R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}q\mathfrak{J}: L^{2}(\Sigma; \mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}),$$
  

$$\underline{B_{\varepsilon}}(z) := \mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z)U_{\varepsilon}^{*}\mathcal{I}_{\varepsilon}\mathcal{S}_{\varepsilon}q\mathfrak{J}: L^{2}(\Sigma; \mathbb{C}^{N}) \to L^{2}(\Sigma; \mathbb{C}^{N}),$$
  

$$\underline{C_{\varepsilon}}(z) := \mathfrak{J}^{*}qM_{\varepsilon}\mathcal{S}_{\varepsilon}^{-1}\mathcal{I}_{\varepsilon}^{-1}U_{\varepsilon}R(z): L^{2}(\mathbb{R}^{\theta}; \mathbb{C}^{N}) \to L^{2}(\Sigma; \mathbb{C}^{N}).$$

By definition, the identities

$$\underline{A}_{\varepsilon}(z) = A_{\varepsilon}(z)q\mathfrak{J}, \quad \underline{B}_{\varepsilon}(z) = \mathfrak{J}^*qM_{\varepsilon}B_{\varepsilon}(z)q\mathfrak{J}, \quad \underline{C}_{\varepsilon}(z) = \mathfrak{J}^*qM_{\varepsilon}C_{\varepsilon}(z),$$

are valid, where  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$  and  $C_{\varepsilon}(z)$  are the operators from (4.8). Applying Proposition 3.11 for  $P_L = U_{\varepsilon}^* \mathcal{I}_{\varepsilon} \mathcal{S}_{\varepsilon} q \mathfrak{J} V$ ,  $P_R = \mathfrak{J}^* q M_{\varepsilon} \mathcal{S}_{\varepsilon}^{-1} \mathcal{I}_{\varepsilon}^{-1} U_{\varepsilon}$  and  $\underline{V_{\varepsilon}} = P_L P_R$  yields the following proposition.

**Proposition 8.1.** Let q and V be as in (4.1) and (4.2),  $\underline{V_{\varepsilon}}$  be defined by (8.1) for  $\varepsilon \in (0, \varepsilon_{tub}), z \in \rho(H)$  and  $R(z) = (H - z)^{-1}$ , where H is the free Dirac operator introduced in Definition 3.2. Then,  $H_{V_{\varepsilon}}$  is self-adjoint and the following holds:

(i) 
$$z \in \sigma_p(H_{V_{\varepsilon}}) \iff -1 \in \sigma_p(B_{\varepsilon}(z)V).$$

(ii) If  $-1 \in \rho(VB_{\varepsilon}(z))$ , then  $z \in \rho(H_{V_{\varepsilon}})$  and

$$(H_{\underline{V_{\varepsilon}}}-z)^{-1} = R(z) - \underline{A_{\varepsilon}}(z)V(I + \underline{B_{\varepsilon}}(z)V)^{-1}\underline{C_{\varepsilon}}(z).$$

Proposition 8.1 shows that the resolvent of  $H_{\underline{V}_{\varepsilon}}$  has a similar structure as the resolvent of  $H_{V_{\varepsilon}}$ ; cf. Proposition 4.1 (ii). Moreover, the operators  $\underline{A}_{\varepsilon}(z)$ ,  $\underline{B}_{\varepsilon}(z)$  and  $\underline{C}_{\varepsilon}(z)$  are strongly connected to the operators  $A_{\varepsilon}(z)$ ,  $B_{\varepsilon}(z)$  and  $\overline{C}_{\varepsilon}(z)$ , respectively. We use this connection to transfer the convergence results from the local operators to the semilocal operators. Before we do so, we introduce the limit operators

$$\underline{A_0}(z) := \Phi_z : L^2(\Sigma; \mathbb{C}^N) \to L^2(\mathbb{R}^\theta; \mathbb{C}^N),$$
  

$$\underline{B_0}(z) := \mathcal{C}_z : L^2(\Sigma; \mathbb{C}^N) \to L^2(\Sigma; \mathbb{C}^N),$$
  

$$C_0(z) := \Phi_{\overline{z}}^* : L^2(\mathbb{R}^\theta; \mathbb{C}^N) \to L^2(\Sigma; \mathbb{C}^N).$$

**Proposition 8.2.** Let  $z \in \rho(H)$  and  $\varepsilon \in (0, \varepsilon_{ABC})$  with  $\varepsilon_{ABC}$  given by (4.19). Then, the families of operators  $(\underline{A}_{\varepsilon}(z))_{\varepsilon \in (0,\varepsilon_{ABC})}$ ,  $(\underline{B}_{\varepsilon}(z))_{\varepsilon \in (0,\varepsilon_{ABC})}$  and  $(\underline{C}_{\varepsilon}(z))_{\varepsilon \in (0,\varepsilon_{ABC})}$  are uniformly bounded and for  $r \in (0, 1/2)$  holds

$$\begin{split} & \left\|\underline{A}_{\varepsilon}(z) - \underline{A}_{0}(z)\right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\Sigma;\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}, \\ & \left\|\underline{B}_{\varepsilon}(z) - \underline{B}_{0}(z)\right\|_{H^{1/2}(\Sigma;\mathbb{C}^{N}) \to L^{2}(\Sigma;\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}, \\ & \left\|\underline{C}_{\varepsilon}(z) - \underline{C}_{0}(z)\right\|_{L^{2}(\Sigma;\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}. \end{split}$$

Furthermore, the operators  $\underline{C}_{\varepsilon}(z)$ ,  $\varepsilon \in (0, \varepsilon_{ABC})$ , are also well-defined acting as operators from  $L^2(\mathbb{R}^{\theta}; \mathbb{C}^N)$  to  $\overline{H^{1/2}}(\Sigma; \mathbb{C}^N)$  and the corresponding operator norms are uniformly bounded.

*Proof.* The assertions are direct consequences of Lemma 4.7, Proposition 4.8, Proposition 4.9 and Proposition 4.10 if

$$\underline{A}_0(z) = A_0(z)q\mathfrak{J}, \quad \underline{B}_0(z) = \mathfrak{J}^* q B_0(z) q\mathfrak{J}, \quad \underline{C}_0(z) = \mathfrak{J}^* q C_0(z).$$
(8.2)

According to (4.24) we have  $A_0(z) = \Phi_z \mathfrak{J}^*$ . Hence,  $A_0(z)q\mathfrak{J} = \Phi_z \mathfrak{J}^*q\mathfrak{J}$ . Furthermore,  $\int_{-1}^1 q(t) dt = 1$ , (2.10) and (2.11) give us for  $\psi \in L^2(\Sigma; \mathbb{C}^N)$ 

$$\mathfrak{J}^* q \mathfrak{J} \psi = \int_{-1}^1 q(t) (\mathfrak{J} \psi)(t) \, dt = \int_{-1}^1 q(t) \, dt \, \psi = \psi, \tag{8.3}$$

which shows  $A_0(z)q\mathfrak{J} = \Phi_z = \underline{A_0}(z)$ . Moreover, (4.21) and (8.3) also imply

$$\mathfrak{J}^*qC_0(z) = \mathfrak{J}^*q\mathfrak{J}\Phi_{\overline{z}}^* = \Phi_{\overline{z}}^* = \underline{C_0}(z).$$

The representation for  $B_0(z)$  in (4.38) and (8.3) yield

$$\mathfrak{J}^*qB_0(z)q\mathfrak{J} = \mathfrak{J}^*qT(\alpha \cdot \nu)q\mathfrak{J} + \mathfrak{J}^*q\mathfrak{J}\mathcal{C}_z\mathfrak{J}^*q\mathfrak{J} = \mathfrak{J}^*qT(\alpha \cdot \nu)q\mathfrak{J} + \mathcal{C}_z.$$

Thus, it remains to show  $\mathfrak{J}^*qT(\alpha \cdot \nu)q\mathfrak{J} = 0$ . This follows from

$$\begin{aligned} \mathfrak{J}^* q T(\alpha \cdot \nu) q \mathfrak{J} \psi &= \int_{-1}^1 q(t) \frac{i}{2} \int_{-1}^1 \operatorname{sign}(t-s) (\alpha \cdot \nu) q(s) \psi \, ds \, dt \\ &= \int_{-1}^1 \int_{-1}^1 \operatorname{sign}(t-s) q(t) q(s) \, ds \, dt \frac{i}{2} (\alpha \cdot \nu) \psi = 0 \quad \forall \psi \in L^2(\Sigma; \mathbb{C}^N). \end{aligned}$$

Having stated these preliminary results, which were essentially consequences of Section 4.3, we are able to present the main theorem of this section, which states conditions under which  $H_{V_{\varepsilon}}$  converges to  $H_{V\delta_{\Sigma}}$  in the norm resolvent sense.

**Theorem 8.3.** Let q and V be as in (4.1) and (4.2),  $\varepsilon_{ABC} > 0$  be as in (4.19) and assume that for some  $z \in \rho(H)$  the following conditions are fulfilled:

- (i) There exists an  $\underline{\varepsilon_{\text{conv}}} \in (0, \varepsilon_{ABC}]$  such that  $(I + \underline{B_{\varepsilon}}(z)Vq)^{-1}$  exists for  $\varepsilon \in (0, \underline{\varepsilon_{\text{conv}}})$ and is uniformly bounded in  $L^2(\Sigma; \mathbb{C}^N)$ .
- (ii) The operator  $I + \mathcal{C}_z V$  is bijective in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ .

Then, the operator  $H_{V\delta_{\Sigma}}$  is self-adjoint,  $z \in \rho(H_{V\delta_{\Sigma}}) \cap \rho(H_{\underline{V}_{\varepsilon}})$  for all  $\varepsilon \in (0, \underline{\varepsilon_{\text{conv}}})$ and for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{\underline{V_{\varepsilon}}} - z)^{-1} - (H_{V\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}$$

for  $\varepsilon \in (0, \underline{\varepsilon_{\text{conv}}})$ . In particular,  $H_{\underline{V_{\varepsilon}}}$  converges to  $H_{V\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

*Proof.* Since  $I + \mathcal{C}_z V$  is continuously invertible in  $H^{1/2}(\Sigma; \mathbb{C}^N)$ , it follows from Proposition 3.14 that  $H_{V\delta_{\Sigma}} - z$  is invertible and

$$(H_{V\delta_{\Sigma}} - z)^{-1} = R(z) - \Phi_z V (I + \mathcal{C}_z V)^{-1} \Phi_{\overline{z}}^*$$
  
=  $R(z) - \underline{A_0}(z) V (I + \underline{B_0}(z) V)^{-1} \underline{C_0}(z).$ 

The rest of the proof can be shown in exactly the same way as Theorem 4.15 by applying Proposition 8.1, Proposition 8.2 and using the spaces  $H^r(\Sigma; \mathbb{C}^N)$  instead the spaces  $\mathcal{B}^r(\Sigma)$ .

# 8.2 An explicit condition for electrostatic and Lorentz scalar interactions

Similarly as in Chapter 5, we find an explicit convergence condition for the norm resolvent convergence of  $H_{\underline{V}_{\varepsilon}}$  if  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  in this section. Our goal is to show that the conditions (5.1) and

$$\sup_{x_{\Sigma}\in\Sigma} d(x_{\Sigma}) < 4, \quad d = \eta^2 - \tau^2, \tag{8.4}$$

guarantees norm resolvent convergence. If (8.4) holds, then Proposition 3.15 (iii) shows that (ii) of Theorem 8.3 is fulfilled for  $z \in \mathbb{C} \setminus \mathbb{R}$ . Thus, we only have to consider (i) of Theorem 8.3. We proceed as follows: We show that if  $\Sigma$  is a rotated  $C_b^2$ -graph, then the operator  $(I + V\underline{B}_{\varepsilon}(z))^{-1}$  exists and is uniformly bounded in  $L^2(\Sigma; \mathbb{C}^N)$  with respect to  $\varepsilon$ . Then, (ii) of Theorem 8.3 is fulfilled and therefore  $H_{V_{\varepsilon}}$ converges in the norm resolvent sense to  $H_{V\delta_{\Sigma}}$  for  $\varepsilon \to 0$ . If  $\Sigma$  is a special  $C^2$ -surface as in Definition 2.1 one uses again a partition of unity to prove the main result of this section which is Theorem 8.9. Recall from Section 5.1.2 that if  $\Sigma$  is a a rotated  $C_b^2$ -graph, then there exists a function  $\zeta \in C_b^2(\mathbb{R}^{\theta}; \mathbb{R})$  and a rotation matrix  $\kappa \in SO(\mathbb{R}^{\theta})$  such that

$$\Sigma = \Sigma_{\zeta,\kappa} = \{\kappa(x',\zeta(x')) : x' \in \mathbb{R}^{\theta-1}\}.$$

As in Section 8.1, we make extensive use of the results known from the case of local potentials; cf. Section 5.1. Thereby, we use the following notation: If O is one of the operators introduced in (5.11) or (5.48), which act as bounded operators in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ , then we define the semilocal version of O through

$$\underline{O} = \mathfrak{J}^* q O q \mathfrak{J} : L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N).$$
(8.5)

If O also acts as a bounded operator from  $\mathcal{B}^{r}(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^{r'}(\mathbb{R}^{\theta-1})$  with  $r, r' \in [0, 1]$ , then the properties of  $\mathfrak{J}$  and q immediately imply

$$\|\underline{O}\|_{H^{r}(\Sigma;\mathbb{C}^{N})\to H^{r'}(\Sigma;\mathbb{C}^{N})} \leq 2\|q\|_{L^{\infty}((-1,1))}^{2}\|O\|_{r\to r'}.$$
(8.6)

We aim to prove that  $(I + \underline{B}_{\varepsilon}(z)V)^{-1}$  is uniformly bounded in  $L^2(\Sigma; \mathbb{C}^N)$  if  $\varepsilon$  is sufficiently small. We show this in the same way as we proved the uniform boundedness of  $(I + B_{\varepsilon}(z)Vq)^{-1}$  in Section 5.1. As almost all the steps are identical, with the sole exception of Proposition 8.5, we keep this section as short as possible and refer to Section 5.1 for details. To shorten notation and to emphasize the connection to Section 5.1 we also write  $\|\cdot\|_{\underline{r}\to\underline{r'}}$  instead of  $\|\cdot\|_{H^r(\mathbb{R}^{\theta-1};\mathbb{C}^N)\to H^{r'}(\mathbb{R}^{\theta-1};\mathbb{C}^N)}$  in the current section.

**Proposition 8.4.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta$  and  $\kappa$  be as in (5.42),  $\zeta_{x'_0}$  be as in (5.46), q be as in (4.1),  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ ,  $Q_{\eta,\tau}^{\zeta,\kappa}$  be as in (5.18),  $z \in \rho(H)$ ,  $\psi \in C_b^1(\mathbb{R}^{\theta-1})$ ,  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  with  $\varepsilon_{\text{gr},1}$  chosen as in Lemma 5.12 and  $a_{\varepsilon} = \varepsilon^{1/6}$ . Then, the operators  $\underline{E}_{\varepsilon}(z)$  and  $[\underline{D}_{\varepsilon}^{\zeta_{x'_0},\kappa}(z), \psi]$  act as bounded operators from  $L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and

$$\begin{split} \|\underline{E}_{\varepsilon}(z)\|_{\underline{0}\to\underline{1}} &\leq C\frac{1+|\log(\varepsilon)|}{a_{\varepsilon}},\\ \|[\underline{D}_{\varepsilon}^{\zeta_{x_{0}'},\kappa}(z),\psi]\|_{\underline{0}\to\underline{1}} &\leq C\|\psi\|_{W^{1}_{\infty}(\mathbb{R}^{\theta-1})}(1+|\log(\varepsilon)|),\\ \|\chi_{B(x_{0}',3a_{\varepsilon})}(\underline{D}_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}-\underline{D}_{\varepsilon}^{\zeta_{x_{0}'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x_{0}'))\chi_{B(x_{0}',3a_{\varepsilon})}\|_{\underline{0}\to\underline{0}} &\leq Ca_{\varepsilon}(1+|\log(\varepsilon)|) \end{split}$$

where C > 0 does not depend on  $x'_0$  and  $\varepsilon$ .

*Proof.* The statement follows from Proposition 5.14 and (8.6).

In Proposition 8.5 we show that  $(I + \underline{D}_{\varepsilon}^{\zeta_{x'_0},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x'_0))^{-1}$  is uniformly bounded with respect to  $\varepsilon$  and  $x'_0 \in \mathbb{R}^{\theta-1}$ . This is the only proposition in the current section where we apply a different proof strategy than in the analogous local statement given by Proposition 5.15 in Section 5.1.

**Proposition 8.5.** Let  $x'_0 \in \mathbb{R}^{\theta-1}$ ,  $\zeta$  and  $\kappa$  be as in (5.42),  $\zeta_{x'_0}$  be as in (5.46), q be as in (5.1),  $\eta, \tau \in C^1_b(\Sigma; \mathbb{R})$ ,  $d = \eta^2 - \tau^2$  satisfy (8.4),  $Q^{\zeta,\kappa}_{\eta,\tau}$  be as in (5.18) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\underline{\varepsilon_{\text{gr},1}} > 0$  such that the operators  $(I + \underline{D}^{\zeta_{x'_0},\kappa}_{\varepsilon}(z)Q^{\zeta,\kappa}_{\eta,\tau}(x'_0))^{-1}$  are uniformly bounded with respect to  $\varepsilon \in (0, \varepsilon_{\text{gr},1})$  and  $x'_0 \in \mathbb{R}^{\theta-1}$ .

*Proof.* Let us first assume that  $\zeta_{x'_0} \equiv y_0 \in \mathbb{R}$ . Then, we get from (8.5), the comments below (5.20) and Proposition 5.2 for  $\xi' \in \mathbb{R}^{\theta-1}$ 

$$(\mathcal{F}\underline{D_{\varepsilon}^{y_0,\kappa}}(z)\mathcal{F}^{-1}f)(\xi') = (\mathcal{F}\underline{D_{\varepsilon}^{0,\kappa}}(z)\mathcal{F}^{-1}f)(\xi') = \frac{\widetilde{\alpha}'\cdot\xi' + m\beta + zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}}\omega(\xi',\varepsilon),$$

where  $\omega(\xi',\varepsilon) = i \int_{-1}^{1} \int_{-1}^{1} q(t)q(s) e^{|\varepsilon(t-s)|i\sqrt{z^2-m^2-|\xi'|^2}} ds dt$  and  $\widetilde{\alpha}' \cdot \xi'$  is defined as in (5.22). Hence, the operator

$$I + \mathcal{F}\underline{D}_{\varepsilon}^{\zeta_{x_0'},\kappa}(z)\mathcal{F}^{-1}Q_{\eta,\tau}^{\zeta,\kappa}(x_0')$$

is a matrix multiplication operator induced by the matrix-valued function

$$\mathbb{R}^{\theta-1} \ni \xi' \mapsto I_N + \frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}} \omega(\xi', \varepsilon) Q_{\eta, \tau}^{\zeta, \kappa}(x_0').$$

We define the function

$$\mathbb{R}^2 \times \mathbb{R}^{\theta-1} \times (0,\infty) \ni (\widehat{\eta},\widehat{\tau},\xi',\varepsilon) \\ \mapsto p(\widehat{\eta},\widehat{\tau},\xi',\varepsilon) := 1 + (\widehat{\eta}^2 - \widehat{\tau}^2) \frac{\omega(\xi',\varepsilon)^2}{4} + (\widehat{\eta}z + \widehat{\tau}m) \frac{\omega(\xi',\varepsilon)}{\sqrt{z^2 - m^2 - |\xi'|^2}}.$$

Now, using the rules from (5.23) and  $Q_{\eta,\tau}^{\zeta,\kappa}(x_0') = \eta(\varkappa_{\zeta,\kappa}(x_0'))I_N + \tau(\varkappa_{\zeta,\kappa}(x_0'))\beta$  yields

$$\left( I_N + \frac{\widetilde{\alpha}' \cdot \xi' + m\beta + zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}} \omega(\xi', \varepsilon) \left( \eta(\varkappa_{\zeta,\kappa}(x'_0))I_N + \tau(\varkappa_{\zeta,\kappa}(x'_0))\beta \right) \right) \cdot \left( I_N - \left( \eta(\varkappa_{\zeta,\kappa}(x'_0))I_N - \tau(\varkappa_{\zeta,\kappa}(x'_0))\beta \right) \frac{\widetilde{\alpha}' \cdot \xi' + m\beta - zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}} \omega(\xi', \varepsilon) \right) = p(\eta(\varkappa_{\zeta,\kappa}(x'_0)), \tau(\varkappa_{\zeta,\kappa}(x'_0)), \xi', \varepsilon)I_N.$$

Hence, if  $p(\eta(\varkappa_{\zeta,\kappa}(x'_0)), \tau(\varkappa_{\zeta,\kappa}(x'_0)), \xi', \varepsilon) \neq 0$ , then  $I_N + \frac{\tilde{\alpha}' \cdot \xi' + m\beta - zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}} \omega(\xi', \varepsilon) Q_{\eta,\tau}^{\zeta,\kappa}(x'_0)$ is invertible and by applying (5.24),  $\eta, \tau \in C_b^2(\Sigma; \mathbb{R})$  and  $\zeta \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{R})$  we are able to estimate

$$\left| \left( I_N + \frac{\widetilde{\alpha}' \cdot \xi' + m\beta - zI_N}{2\sqrt{z^2 - m^2 - |\xi'|^2}} \omega(\xi', \varepsilon) Q_{\eta, \tau}^{\zeta, \kappa}(x'_0) \right)^{-1} \right| \le \frac{C}{|p(\eta(\varkappa_{\zeta, \kappa}(x'_0)), \tau(\varkappa_{\zeta, \kappa}(x'_0)), \xi', \varepsilon)|},$$

where C > 0 does not depend on  $\xi'$ ,  $x'_0$ ,  $\kappa$  and  $\varepsilon$ . Moreover, there exists an  $\underline{\varepsilon_{\text{gr},1}} > 0$  such that

$$p_{\inf} = \inf_{(\widehat{\eta},\widehat{\tau})\in\overline{\operatorname{ran}(\eta,\tau)}, \xi'\in\mathbb{R}^{\theta-1}, \varepsilon\in(0,\underline{\varepsilon_{\operatorname{gr},1}})} |p(\widehat{\eta},\widehat{\tau},\xi',\varepsilon)| > 0.$$
(8.7)

We postpone the proof of (8.7) to the end of the proof. This estimate and Proposition 2.19 imply that for  $\varepsilon \in (0, \underline{\varepsilon}_{\text{gr},1})$  the operator  $I + \underline{D}_{\varepsilon}^{y_0,\kappa}Q_{\eta,\tau}^{\zeta,\kappa}(x'_0)$  is invertible and the norm of its inverse is bounded by  $\frac{C}{p_{\text{inf}}}$ , which does not depend on  $\kappa$ ,  $x'_0$ ,  $y_0$  and  $\varepsilon$ . Thus, the assertion is true if  $\zeta_{x'_0} \equiv y_0 \in \mathbb{R}$ . If this is not the case, then as  $\Sigma_{\zeta_{x'_0},\kappa}$ is an affine hyperplane in  $\mathbb{R}^{\theta}$ , there exists an  $\tilde{y}_0(x'_0) \in \mathbb{R}$  and a  $\tilde{\kappa}(x'_0) \in \text{SO}(\theta)$  such that  $\Sigma_{\zeta_{x'_0},\kappa} = \Sigma_{\tilde{y}_0(x'_0),\tilde{\kappa}(x'_0)}$ ; cf (5.57). Hence, one gets in the same way as in the proof of Proposition 5.15

$$\left\| (I + \underline{D_{\varepsilon}^{\zeta_{x_0'},\kappa}}Q_{\eta,\tau}^{\zeta,\kappa}(x_0'))^{-1} \right\|_{\underline{0}\to\underline{0}} = \left\| (I + \underline{D_{\varepsilon}^{\widetilde{y}_0(x_0'),\widetilde{\kappa}(x_0')}}Q_{\eta,\tau}^{\zeta,\kappa}(x_0'))^{-1} \right\|_{\underline{0}\to\underline{0}} \le \frac{C}{p_{\inf}}$$

It remains to prove (8.7). We start by introducing

$$d_{\max} := \max_{(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}} \widehat{\eta}^2 - \widehat{\tau}^2 \quad \text{and} \quad \widetilde{\omega}(\xi', \varepsilon) = i \int_{-1}^1 \int_{-1}^1 q(s)q(t) e^{-\varepsilon |t-s||\xi'|} \, ds \, dt.$$

Then, (8.4) implies  $d_{\max} < 4$ . Moreover,  $|\widetilde{\omega}(\xi', \varepsilon)| \leq 1$  and  $\widetilde{\omega}(\xi', \varepsilon)^2 < 0$  since  $q \geq 0$  a.e. on (-1, 1) by (5.1). Thus,

$$\left|1 + (\widehat{\eta}^2 - \widehat{\tau}^2)\frac{\widetilde{\omega}(\xi',\varepsilon)^2}{4}\right| \ge \min\left\{1 - \frac{d_{\max}}{4}, 1\right\} > 0$$
(8.8)

for all  $\xi' \in \mathbb{R}^{\theta-1}$ ,  $\varepsilon > 0$  and  $(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ . Furthermore, (5.32), (5.33) and  $q \in L^{\infty}((-1, 1))$  yield

$$|\omega(\xi',\varepsilon) - \widetilde{\omega}(\xi',\varepsilon)| \le \frac{C}{1+|\xi'|}$$

for all  $\xi' \in \mathbb{R}^{\theta-1}$  and  $\varepsilon > 0$ . Thus, since  $\overline{\operatorname{ran}(\eta, \tau)}$  is bounded and  $|\omega(\xi', \varepsilon)| \leq 1$ , there exists an R > 0 such that for all  $|\xi'| \geq R$ ,  $\varepsilon > 0$  and  $(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ 

$$\left| (\widehat{\eta}^2 - \widehat{\tau}^2) \frac{\omega(\xi', \varepsilon)^2 - \widetilde{\omega}(\xi', \varepsilon)^2}{4} + (\widehat{\eta}z + \widehat{\tau}m) \frac{\omega(\xi', \varepsilon)}{\sqrt{z^2 - m^2 - |\xi'|^2}} \right| \le \frac{\min\left\{1 - \frac{d_{\max}}{4}, 1\right\}}{2}.$$

This and (8.8) imply for all  $|\xi'| \ge R$ ,  $\varepsilon > 0$  and  $(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ 

$$\begin{split} |p(\widehat{\eta},\widehat{\tau},\xi',\varepsilon)| &\geq \left|1 + (\widehat{\eta}^2 - \widehat{\tau}^2)\frac{\widetilde{\omega}(\xi',\varepsilon)^2}{4}\right| \\ &- \left|(\widehat{\eta}^2 - \widehat{\tau}^2)\frac{\omega(\xi',\varepsilon)^2 - \widetilde{\omega}(\xi',\varepsilon)^2}{4} + (\widehat{\eta}z + \widehat{\tau}m)\frac{\omega(\xi',\varepsilon)}{\sqrt{z^2 - m^2 - |\xi'|^2}}\right| \\ &> \frac{\min\left\{1 - \frac{d_{\max}}{4}, 1\right\}}{2}; \end{split}$$

i.e.

$$\inf_{(\widehat{\eta},\widehat{\tau})\in\overline{\operatorname{ran}(\eta,\tau)},|\xi'|\geq R,\varepsilon>0}|p(\widehat{\eta},\widehat{\tau},\xi',\varepsilon)|>\frac{\min\left\{1-\frac{d_{\max}}{4},1\right\}}{2}>0.$$
(8.9)

Thus, it remains to consider the case  $|\xi'| \leq R$ . We use  $\int_{-1}^{1} q(t) dt = 1$  to get

$$|\omega(\xi',\varepsilon)-i| = \left|\int_{-1}^{1}\int_{-1}^{1}q(t)q(s)\left(e^{|\varepsilon(t-s)|i\sqrt{z^2-m^2-|\xi'|^2}}-1\right)ds\,dt\right| \le C\varepsilon R \quad \forall |\xi'| \le R,$$

where C > 0 does only depend on q, z and m. Hence,

$$\left| p(\widehat{\eta}, \widehat{\tau}, \xi', \varepsilon) - \left( 1 - \frac{\widehat{\eta}^2 - \widehat{\tau}^2}{4} + \frac{i(\widehat{\eta}z + \widehat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}} \right) \right| \le C\varepsilon R \tag{8.10}$$

for all  $|\xi'| \leq R$ ,  $\varepsilon > 0$  and  $(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ . Next, we show

$$\inf_{(\hat{\eta},\hat{\tau})\in\bar{\mathrm{ran}}\,(\eta,\tau),|\xi'|\leq R} \left|1 - \frac{\hat{\eta}^2 - \hat{\tau}^2}{4} + \frac{i(\hat{\eta}z + \hat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}}\right| > 0.$$
(8.11)

Since  $\overline{B(0,R)} \times \overline{\operatorname{ran}(\eta,\tau)}$  is compact, it suffices to show

$$1 - \frac{\widehat{\eta}^2 - \widehat{\tau}^2}{4} \neq -\frac{i(\widehat{\eta}z + \widehat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}} \quad \forall (\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}, |\xi'| \le R.$$

Squaring the equation, multiplying with  $z^2 - m^2 - |\xi'|^2$  and setting  $\hat{d} = \hat{\eta}^2 - \hat{\tau}^2$  yields

$$\left(1-\frac{\widehat{d}}{4}\right)^{2}(|\xi'|^{2}+m^{2}-z^{2})\neq\widehat{\eta}^{2}z^{2}+2\widehat{\eta}\widehat{\tau}mz+\widehat{\tau}^{2}m^{2}\quad\forall(\widehat{\eta},\widehat{\tau})\in\overline{\operatorname{ran}\left(\eta,\tau\right)},|\xi'|\leq R.$$
(8.12)

By the solution formula for quadratic equations, this is equivalent to

$$z \neq \frac{-\widehat{\eta}\widehat{\tau}m \pm \sqrt{(\widehat{\eta}\widehat{\tau}m)^2 + ((1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2)((1 - \frac{\widehat{d}}{4})^2(|\xi'|^2 + m^2) - \widehat{\tau}^2m^2)}}{(1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2} \\ = \frac{-\widehat{\eta}\widehat{\tau}m \pm \sqrt{((1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2)(1 - \frac{\widehat{d}}{4})^2(|\xi'|^2 + m^2) - (1 - \frac{\widehat{d}}{4})^2\widehat{\tau}^2m^2}}{(1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2} \\ = \frac{-\widehat{\eta}\widehat{\tau}m \pm (1 - \frac{\widehat{d}}{4})\sqrt{((1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2)(|\xi'|^2 + m^2) - \widehat{\tau}^2m^2}}{(1 - \frac{\widehat{d}}{4})^2 + \widehat{\eta}^2}$$
(8.13)

for all  $|\xi'| \leq R$  and  $(\hat{\eta}, \hat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ . Using the identity  $(1 - \frac{\hat{d}}{4})^2 + \hat{\eta}^2 = (1 + \frac{\hat{d}}{4})^2 + \hat{\tau}^2$  shows that the right-hand side of the above inequality equals

$$\begin{aligned} \frac{-\widehat{\eta}\widehat{\tau}m \pm (1-\frac{\widehat{d}}{4})\sqrt{((1+\frac{\widehat{d}}{4})^2 + \widehat{\tau}^2)(|\xi'|^2 + m^2) - \widehat{\tau}^2m^2}}{(1+\frac{\widehat{d}}{4})^2 + \widehat{\tau}^2} \\ &= \frac{-\widehat{\eta}\widehat{\tau}m \pm (1-\frac{\widehat{d}}{4})\sqrt{((1+\frac{\widehat{d}}{4})^2 + \widehat{\tau}^2)|\xi'|^2 + (1+\frac{d}{4})^2m^2}}{(1+\frac{\widehat{d}}{4})^2 + \widehat{\tau}^2} \end{aligned}$$

Thus, the expression in the square root is nonnegative and therefore the right-hand side in (8.13) is a real. Moreover, z is assumed to be in  $\mathbb{C} \setminus \mathbb{R}$ . Hence, (8.13) is valid, and therefore also (8.12) and (8.11) are true. Consequently, according to (8.10) we can choose  $\varepsilon_{\text{gr},1} > 0$  sufficiently small such that

$$\begin{split} |p(\widehat{\eta},\widehat{\tau},\xi',\varepsilon)| &\geq \left|1 - \frac{\widehat{\eta}^2 - \widehat{\tau}^2}{4} + \frac{i(\widehat{\eta}z + \widehat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}}\right| - C\varepsilon R\\ &\geq \inf_{(\widehat{\eta},\widehat{\tau})\in\overline{\operatorname{ran}\left(\eta,\tau\right)},|\xi'|\leq R} \left|1 - \frac{\widehat{\eta}^2 - \widehat{\tau}^2}{4} + \frac{i(\widehat{\eta}z + \widehat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}}\right| - C\varepsilon R\\ &\geq \frac{\inf_{(\widehat{\eta},\widehat{\tau})\in\overline{\operatorname{ran}\left(\eta,\tau\right)},|\xi'|\leq R} \left|1 - \frac{\widehat{\eta}^2 - \widehat{\tau}^2}{4} + \frac{i(\widehat{\eta}z + \widehat{\tau}m)}{\sqrt{z^2 - m^2 - |\xi'|^2}}\right|}{2} > 0 \end{split}$$

for all  $|\xi'| \leq R$ ,  $\varepsilon \in (0, \underline{\varepsilon_{\text{gr},1}})$  and  $(\widehat{\eta}, \widehat{\tau}) \in \overline{\operatorname{ran}(\eta, \tau)}$ . Combined with (8.9) this gives us (8.7), which completes the proof.

In the next statements we use similarly as in Section 5.1.2 the functions  $\phi_{n'}^a$  and  $\vartheta_{n'}^a$ , where  $a \in (0, \infty)$  and  $n' \in \mathbb{Z}^{\theta-1}$ , from Corollary A.3. They allow us to construct a uniformly bounded right inverse of  $I + D_{\varepsilon}^{\zeta,\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}$  with the help of the operators  $(I + \underline{D}_{\varepsilon}^{\zeta_{x_0'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(x_0'))^{-1}$ ,  $x_0' \in \mathbb{R}^{\theta-1}$ ; cf. the text between Proposition 5.15 and Lemma 5.16.

**Lemma 8.6.** Let  $z \in \rho(H)$ ,  $\varepsilon \in (0, \varepsilon_{tub})$  and  $a_{\varepsilon} = \varepsilon^{1/6}$ . Then,

$$(1 - \vartheta_{n'}^{a_{\varepsilon}})\underline{E_{\varepsilon}}(z)\phi_{n'}^{a_{\varepsilon}} = (1 - \vartheta_{n'}^{a_{\varepsilon}})\underline{D_{\varepsilon}}^{\zeta,\kappa}(z)\phi_{n'}^{a_{\varepsilon}}$$

*Proof.* This follows from Lemma 5.16 and the fact that  $\mathfrak{J}$  and  $\mathfrak{J}^*$  commute with  $\vartheta_{n'}^{a_{\varepsilon}}$  and  $\phi_{n'}^{a_{\varepsilon}}$ .

**Proposition 8.7.** Let  $\zeta$  and  $\kappa$  be as in (5.42), q be as in (5.1),  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$ ,  $d = \eta^2 - \tau^2$  satisfy (8.4),  $Q_{\eta,\tau}^{\zeta,\kappa}$  be defined by (5.18) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\varepsilon_{\text{gr},2} \in (0, \varepsilon_{ABC}]$ , with  $\varepsilon_{ABC} > 0$  chosen according to (4.19), such that  $I + D_{\varepsilon}^{\zeta,\kappa}(z) Q_{\eta,\tau}^{\zeta,\kappa}$  has a right inverse which is uniformly bounded in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  with respect to  $\varepsilon \in (0, \varepsilon_{\text{gr},2})$ . *Proof.* We define for  $\varepsilon \in (0, \min\{\varepsilon_{ABC}, \varepsilon_{gr,1}\})$  with  $\varepsilon_{gr,1}$  chosen as in Proposition 8.5

$$\begin{split} \underline{R_{\varepsilon}} &: L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N), \\ \underline{R_{\varepsilon}} &:= \sum_{n' \in \mathbb{Z}} \phi_{n'}^{a_{\varepsilon}} \underline{R_{n',\varepsilon}} \vartheta_{n'}^{a_{\varepsilon}}, \end{split}$$

with  $a_{\varepsilon} = \varepsilon^{1/6}$  and  $\underline{R_{n',\varepsilon}} := (I + \underline{D_{\varepsilon}^{\zeta_{a\varepsilon n'},\kappa}}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n'))^{-1}$ . The equality  $\vartheta_{n'}^{a_{\varepsilon}}\phi_{n'}^{a_{\varepsilon}} = \phi_{n'}^{a_{\varepsilon}}$ , implies  $\underline{R_{\varepsilon}} = \sum_{n' \in \mathbb{Z}^{\theta-1}} \vartheta_{n'}^{a_{\varepsilon}}\phi_{n'}^{a_{\varepsilon}} \underline{R_{n',\varepsilon}}\vartheta_{n'}^{a_{\varepsilon}}$  and therefore Proposition 8.5 and Corollary C.4 show that  $\underline{R_{\varepsilon}}$  is well-defined and uniformly bounded by

$$\left\|\underline{R}_{\varepsilon}\right\|_{\underline{0}\to\underline{0}} \leq 11^{\theta-1} \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\| (I + \underline{D}_{\varepsilon}^{\zeta_{a_{\varepsilon}n'},\kappa}(z)Q_{\eta,\tau}^{\zeta,\kappa}(a_{\varepsilon}n'))^{-1} \right\|_{\underline{0}\to\underline{0}} \leq C,$$

where C > 0 does not depend on  $\varepsilon$ . In the exact same way as in the *Steps 1–3* of Proposition 5.17 one shows with the help of Proposition 8.4, Proposition 8.5 and Lemma 8.6 that

$$(I + \underline{D_{\varepsilon}^{\zeta,\kappa}}(z)Q_{\eta,\tau}^{\zeta,\kappa})\underline{R_{\varepsilon}} = I + \underline{K_{\varepsilon}} + \underline{L_{\varepsilon}},$$
(8.14)

where  $\underline{K_{\varepsilon}}$  acts as a bounded operator from  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  and the norm estimates

$$\left\|\underline{K_{\varepsilon}}\right\|_{\underline{0}\to\underline{1}} \leq C \frac{1+|\log(\varepsilon)|}{a_{\varepsilon}^{2}} \quad \text{and} \quad \left\|\underline{L_{\varepsilon}}\right\|_{\underline{0}\to\underline{0}} \leq C a_{\varepsilon}(1+|\log(\varepsilon))$$

Moreover, (8.14) shows that the operator  $\underline{K_{\varepsilon}} + \underline{L_{\varepsilon}}$  is uniformly bounded in  $L^{2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N})$ with respect to  $\varepsilon \in (0, \min\{\varepsilon_{ABC}, \varepsilon_{\text{gr},1}\})$ . According to Proposition 3.15 (iii) the operator  $I + \mathcal{C}_{z}V = I + \underline{B_{0}}(z)V$  (with  $V = \eta I_{N} + \tau\beta$ ) is continuously invertible in  $L^{2}(\Sigma; \mathbb{C}^{N})$  and  $H^{1/2}(\Sigma; \mathbb{C}^{N})$ . Thus, by (5.11), (5.18), (8.2) and (8.5) the operator  $(I + \underline{D_{0}^{\zeta,\kappa}}(z)Q_{\eta,\tau}^{\zeta,\kappa})^{-1}$  is continuously invertible in  $L^{2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N})$  and  $H^{1/2}(\mathbb{R}^{\theta-1}; \mathbb{C}^{N})$ . Using this observation as well as the properties of  $\underline{L_{\varepsilon}}$  and  $\underline{K_{\varepsilon}}$  one proves the assertion of the proposition in the same way as in *Step 4* of Proposition 5.17.

**Proposition 8.8.** Let  $\Sigma$  be a is  $C_b^2$ -graph as described in the beginning of Section 5.1.2, q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (8.4) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, there exists an  $\underline{\varepsilon_{\text{conv}}} \in (0, \varepsilon_{ABC}]$ , with  $\varepsilon_{ABC} > 0$  from (4.19), such that  $I + B_{\varepsilon}(z)V$  has a inverse which is uniformly bounded in  $L^2(\Sigma; \mathbb{C}^N)$  with respect to  $\varepsilon \in (0, \underline{\varepsilon_{\text{conv}}})$ .

*Proof.* The statement can be proven in the same way as Proposition 5.18.  $\Box$ 

Finally, we are able to provide the main theorem of this section.

**Theorem 8.9.** Let  $\Sigma$  be a special  $C^2$ -boundary as in Definition 2.1, q be as in (5.1),  $V = \eta I_N + \tau \beta$  with  $\eta, \tau \in C_b^1(\Sigma; \mathbb{R})$  satisfy (8.4) and  $z \in \mathbb{C} \setminus \mathbb{R}$ . Then, the operator  $H_{V\delta_{\Sigma}}$  is self-adjoint and there exists an  $\underline{\varepsilon_{\text{conv}}} > 0$  such that for any  $r \in (0, \frac{1}{2})$  exists a C > 0 such that

$$\left\| (H_{\underline{V_{\varepsilon}}} - z)^{-1} - (H_{V\delta_{\Sigma}} - z)^{-1} \right\|_{L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N}) \to L^{2}(\mathbb{R}^{\theta};\mathbb{C}^{N})} \leq C\varepsilon^{1/2-r}$$

for  $\varepsilon \in (0, \underline{\varepsilon_{\text{conv}}})$ . In particular,  $H_{\underline{V_{\varepsilon}}}$  converges to  $H_{V\delta_{\Sigma}}$  in the norm resolvent sense as  $\varepsilon \to 0$ .

Proof. In the case that  $\Sigma$  is a rotated  $C_b^2$ -graph the assertion follows from Theorem 8.3, the text below (8.4) and Proposition 8.8. In the general case one can use the  $C^1$ -partition of unity  $\hat{\varphi}_1, \ldots, \hat{\varphi}_p$  from Corollary A.5 and the same strategy as in the proof of Theorem 5.20 in order to prove the assertion. We remark that in the current semilocal case the choice of the partition of unity is crucial since the property  $\hat{\varphi}_l(x_{\Sigma}) = \hat{\varphi}_l(x)$  for  $x = x_{\Sigma} + t\nu(x_{\Sigma})$  with  $(x_{\Sigma}, t) \in \Sigma \times (-\frac{\varepsilon_{\text{tub}}}{2}, \frac{\varepsilon_{\text{tub}}}{2})$ , which the  $C^1$ partition of unity from Corollary A.5 possesses, guarantees that  $\underline{V}_{\varepsilon}$  and  $\hat{\varphi}_l$  commute for  $\varepsilon \in (0, \frac{\varepsilon_{\text{tub}}}{2})$ . Making use of this property allows one to prove formula (5.66) in the semilocal case in a very similar manner, which, in turn, ensures that the proof strategy from Theorem 5.20 is successful.

## Appendix A. Partitions of unity

In this chapter, which contains results from [15], we construct various useful partitions of unity. Similarly as in [54, Chapter 3], we define a partition of unity as follows.

**Definition A.1.** We call a sequence of functions  $(\varphi_j)_{j \in J}$ , where J is a countable (finite or infinite) index set and  $\varphi_j \in C^{\infty}(\mathbb{R}^n)$ ,  $n \in \mathbb{N}$ ,  $(\varphi_j \in C^k(\mathbb{R}^n), k, n \in \mathbb{N})$  for all  $j \in J$ , a partition ( $C^k$ -partition) of unity for a set  $S \subset \mathbb{R}^n$  if the following three conditions are met:

- (i)  $\phi_j(x) \ge 0$  for  $x \in \mathbb{R}^n$  and  $j \in J$ .
- (ii) Every  $x \in \mathbb{R}^n$  has a neighbourhood that intersects supp  $\phi_j$  for only finitely many j's.
- (iii)  $\sum_{j \in J} \varphi_j(x) = 1$  for all  $x \in S$ .

Moreover, if  $(W_j)_{j \in J}$  is an open cover of S and  $\operatorname{supp} \varphi_j \subset W_j$  for all  $j \in J$ , then we call  $(\varphi_j)_{j \in J}$  a partition ( $C^k$ -partition) of unity for S subordinate to the open cover  $(W_j)_{j \in J}$ .

In this thesis we often use partitions of unity in settings where derivatives are involved; cf. Section 8.1 and the proofs of Proposition 5.17, Theorem 5.20 and Theorem 6.7. Thus, it is important to construct partitions of unity with uniformly bounded derivatives. We show in several situations that such a choice is possible, starting with the case  $S = \mathbb{R}^n$ ,  $n \in \{1, 2, 3\}$ .

**Proposition A.2.** Let  $n \in \{1, 2, 3\}$ . Then, there exists a partition of unity  $(\phi_j)_{j \in \mathbb{Z}^n}$  for  $\mathbb{R}^n$  subordinate to the open cover  $(B(j, 1))_{j \in \mathbb{Z}^n}$  of  $\mathbb{R}^n$  and a sequence of smooth functions  $(\vartheta_j)_{j \in \mathbb{N}}$  which have the following properties:

(i)  $\sup_{j\in\mathbb{Z}^n} \max\{\|\phi_j\|_{W^1_\infty(\mathbb{R}^n)}, \|\vartheta_j\|_{W^1_\infty(\mathbb{R}^n)}\} < \infty.$ 

(*ii*) supp  $\vartheta_j \subset B(j,3)$ ,  $0 \leq \vartheta_j \leq 1$  and  $\vartheta_j = 1$  on B(j,2) for all  $j \in \mathbb{Z}^n$ .

Proof. Note that since  $\theta \in \{1, 2, 3\}$ , the family  $(B(j, 7/8))_{j \in \mathbb{Z}^n}$  is also an open cover of  $\mathbb{R}^n$ . We start by choosing a function  $\phi \in C^{\infty}(\mathbb{R}^n)$  such that  $0 \leq \phi \leq 1, \phi = 1$ on B(0, 7/8) and supp  $\phi \subset B(0, 1)$ . Furthermore, we set  $\tilde{\phi}_j := \phi(\cdot - j)$  for  $j \in \mathbb{Z}^n$ . Then,  $0 \leq \tilde{\phi}_j \leq 1, \tilde{\phi}_j = 1$  on B(j, 7/8) and supp  $\tilde{\phi}_j \subset B(j, 1)$ . Next, we fix a bijection  $\mathcal{Z} : \mathbb{N} \to \mathbb{Z}^n$  and set  $\phi_{\mathcal{Z}(1)} := \tilde{\phi}_{\mathcal{Z}(1)}$  and  $\phi_{\mathcal{Z}(l)} := (1 - \tilde{\phi}_{\mathcal{Z}(1)}) \cdots (1 - \tilde{\phi}_{\mathcal{Z}(l-1)}) \tilde{\phi}_{\mathcal{Z}(l)}$  for  $l \in \mathbb{N} \setminus \{1\}$ . Then,  $\operatorname{supp} \phi_j \subset \operatorname{supp} \widetilde{\phi}_j$ ,  $0 \leq \widetilde{\phi}_j \leq 1$  for  $j \in \mathbb{Z}^n$  and one gets via induction for  $l \in \mathbb{N}$ 

$$\sum_{k=1}^{l} \phi_{\mathcal{Z}(k)} = 1 - \prod_{k=1}^{l} (1 - \widetilde{\phi}_{\mathcal{Z}(k)}).$$

This implies  $\sum_{j \in \mathbb{Z}^n} \phi_j(x) = \sum_{l=1}^{\infty} \phi_{\mathcal{Z}(l)}(x) = 1$  for  $x \in \mathbb{R}^n$ . Thus,  $(\phi_j)_{j \in \mathbb{Z}^n}$  is a partition of unity for  $\mathbb{R}^n$  subordinate to  $(B(j,1))_{j \in \mathbb{Z}^n}$ .

Furthermore, let  $l \in \mathbb{N}$ ,  $w \in \{1, \ldots, n\}$  and  $x \in \mathbb{R}^n$ . We estimate

$$\begin{aligned} |\partial_w \phi_{\mathcal{Z}(l)}(x)| &= \left| \partial_w (\widetilde{\phi}_{\mathcal{Z}(l)}(x)) \prod_{k=1}^{l-1} (1 - \widetilde{\phi}_{\mathcal{Z}(k)}(x)) \\ &- \sum_{k=1}^{l-1} \widetilde{\phi}_{\mathcal{Z}(l)}(x) (\partial_w \widetilde{\phi}_{\mathcal{Z}(k)}(x)) \prod_{r=1, r \neq k}^{l-1} (1 - \widetilde{\phi}_{\mathcal{Z}(r)}(x)) \right| \\ &\leq \sum_{k=1}^{l} \left| \partial_w \widetilde{\phi}_{\mathcal{Z}(k)}(x) \right| = \sum_{k=1, x \in B(\mathcal{Z}(k), 1)}^{l} \left| \partial_w \widetilde{\phi}_{\mathcal{Z}(k)}(x) \right| \\ &\leq 2^n \| \partial_w \phi \|_{L^{\infty}(\mathbb{R}^n)}, \end{aligned}$$

where we used that  $x \in \mathbb{R}^n$  can be contained in at most  $2^n$  balls of the type B(j, 1)with  $j \in \mathbb{Z}^n$ . This shows that the derivatives of the  $\phi_j$ 's are uniformly bounded by  $2^n \|\phi\|_{W^1_{\infty}(\mathbb{R}^n)}$ . Next, we construct the sequence  $(\vartheta_j)_{j\in\mathbb{Z}^n}$ . To do so, we choose  $\vartheta \in C^{\infty}(\mathbb{R}^n)$  such that  $0 \le \vartheta \le 1$ ,  $\theta = 1$  on B(0, 2) and  $\operatorname{supp} \vartheta \subset B(0, 3)$ . Then, we define  $\vartheta_j := \vartheta(\cdot - j)$ . The constructed sequence has the claimed properties.  $\Box$ 

A useful consequence of this proposition is the following corollary.

**Corollary A.3.** Let  $n \in \{1, 2, 3\}$  and b > 0. Then, for all  $a \in (0, b)$  exists a partition of unity  $(\phi_j^a)_{j \in \mathbb{Z}^n}$  for  $\mathbb{R}^n$  subordinate to the open cover  $(B(ja, a))_{j \in \mathbb{Z}^n}$  of  $\mathbb{R}^n$  and a sequence of smooth functions  $(\vartheta_j^a)_{j \in \mathbb{Z}^n}$  which have the following properties:

- (i)  $\sup_{j\in\mathbb{Z}^n} \max\{\|\phi_j^a\|_{W^1_{\infty}(\mathbb{R}^n)}, \|\vartheta_j^a\|_{W^1_{\infty}(\mathbb{R}^n)}\} < \frac{C}{a}$ , where C > 0 does not depend on  $a \in (0, b)$ .
- (*ii*) supp  $\vartheta_j^a \subset B(ja, 3a)$ ,  $0 \leq \vartheta_j^a \leq 1$  and  $\vartheta_j^a = 1$  on B(ja, 2) for all  $j \in \mathbb{Z}^n$ .

*Proof.* Define  $\phi_j^a := \phi_j(\frac{\cdot}{a})$  and  $\vartheta_j^a := \vartheta_j^a(\frac{\cdot}{a})$  for  $j \in \mathbb{Z}^n$ . Then, all the claims follow directly from Proposition A.4.

Next, we also find suitable partitions of unity for special  $C^2$ -surfaces defined in Definition 2.1.

**Proposition A.4.** Let  $\Sigma \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , be a special  $C^2$ -surface as in Definition 2.1 and  $W_1, \ldots, W_p$  be the corresponding open cover of  $\Sigma$ . Then, there exists a partition of unity  $\varphi_1, \ldots, \varphi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  for  $\Sigma$  subordinate to the open cover  $W_1, \ldots, W_p$  of  $\Sigma$ . Moreover, there exist functions  $\chi_1, \ldots, \chi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  such that  $\operatorname{supp} \chi_l \subset W_l$  and  $\varphi_l \chi_l = \varphi_l$  for all  $l \in \{1, \ldots, p\}$ .

Proof. According to [69, Appendix 1, Lemma 1.2 and Lemma 1.3] there exists a sequence  $(x_n)_{n\in\mathbb{N}} \subset \mathbb{R}^{\theta}$ ,  $M \in \mathbb{N}$ ,  $0 < \delta < \frac{\varepsilon_{\Sigma}}{4}$ , with  $\varepsilon_{\Sigma}$  from Definition 2.1, and a sequence of real-valued  $C^{\infty}$ -functions  $(\phi_n)_{n\in\mathbb{N}}$  such that  $(B(x_n, \delta))_{n\in\mathbb{N}}$  is an open cover of  $\mathbb{R}^{\theta}$ ,  $(\phi_n)_{n\in\mathbb{N}}$  is a partition of unity for  $\mathbb{R}^{\theta}$ ,  $\supp \phi_n \subset B(x_n, \delta)$  for all  $n \in \mathbb{N}$ , every point  $x \in \mathbb{R}^{\theta}$  is contained in at most M of the sets  $B(x_n, \delta)$ , and the derivatives (of any order) of the functions  $\phi_n$  are uniformly bounded. Next, we define the set  $Y := \{x_n : B(x_n, 2\delta) \cap \Sigma \neq \emptyset\}$ . Note that for all  $x_n \in Y$  there exists an  $l \in \{1, \ldots, p\}$  such that  $B(x_n, 2\delta) \subset W_l$ . In fact, as  $B(x_n, 2\delta) \cap \Sigma \neq \emptyset$ , there exists  $y_{\Sigma} \in B(x_n, 2\delta) \cap \Sigma$  and thus, Definition 2.1 implies  $B(y_{\Sigma}, \varepsilon_{\Sigma}) \subset W_l$  for an  $l \in \{1, \ldots, p\}$ . Hence, for any  $y \in B(x_n, 2\delta)$  one has

$$|y - y_{\Sigma}| \le |y - x_n| + |x_n - y_{\Sigma}| < 4\delta < \varepsilon_{\Sigma},$$

which shows  $B(x_n, 2\delta) \subset W_l$ . Define the sets  $I_1 := \{n : x_n \in Y, B(x_n, 2\delta) \subset W_l\}$ and  $I_l := \{n : x_n \in Y, B(x_n, 2\delta) \subset W_l, B(x_n, 2\delta) \not\subset W_k, k \in \{1, \ldots, l-1\}\}$  for  $l \in \{2, \ldots, p\}$ . Then, it is not difficult to see that

$$\varphi_l = \sum_{n \in I_l} \phi_n \qquad \Box$$

is a partition of unity having the claimed properties. Moreover, the construction of  $\varphi_l$ ,  $l \in \{1, \ldots, p\}$ , also implies  $\operatorname{supp} \varphi_l + B(0, \delta) \subset W_l$ . Thus, [54, Theorem 3.6] guarantees the existence of the functions  $\chi_1, \ldots, \chi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  with  $\operatorname{supp} \chi_l \subset W_l$ and  $\varphi_l \chi_l = \varphi_l$  for  $l \in \{1, \ldots, p\}$ .

**Corollary A.5.** Let  $\Sigma \subset \mathbb{R}^{\theta}$ ,  $\theta \in \{2,3\}$ , be a special  $C^2$ -surface as in Definition 2.1,  $W_1, \ldots, W_p$  be the corresponding open cover of  $\Sigma$ ,  $\Omega_{\varepsilon}$  be as in Definition 2.7 and  $\varepsilon_{\text{tub}} > 0$  be chosen as in Proposition 2.12. Then, there exists a  $C^1$ -partition of unity  $\widehat{\varphi}_1, \ldots, \widehat{\varphi}_p \in C_b^1(\mathbb{R}^{\theta})$  for  $\Omega_{\underline{\varepsilon}_{\text{tub}}}$  and a function  $\varpi \in C^1(\mathbb{R})$  with  $0 \leq \varpi \leq 1$  such that  $\widehat{\varphi}_l(x_{\Sigma} + t\nu(x_{\Sigma})) = \widehat{\varphi}_l(x_{\Sigma})\overline{\omega}(t)$  for  $(x_{\Sigma}, t) \in \Sigma \times (-\varepsilon_{\text{tub}}, \varepsilon_{\text{tub}})$ ,  $\operatorname{supp} \widehat{\varphi}_l \cap \Sigma \subset W_l$  and  $\operatorname{supp} \widehat{\varphi}_l \subset \Omega_{\varepsilon_{\text{tub}}}$  for all  $l \in \{1, \ldots, p\}$ .

*Proof.* Let  $\varphi_1, \ldots, \varphi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  be the partition of unity from Proposition A.4 and define for  $l \in \{1, \ldots, p\}$  the function  $\widehat{\varphi}_l$  as the extension of  $\varphi_l \upharpoonright \Sigma$  given by Lemma 4.3. Then, the functions  $\widehat{\varphi}_1, \ldots, \widehat{\varphi}_p$  have the claimed properties because of the way they are constructed in Lemma 4.3.

### Appendix B. Proof of the estimate (4.34)

Before we start with the proof of (4.34), let us mention that this section can be found in [14, Appendix B]. Let us shortly recall the problem setting. Let  $z \in \rho(H)$  and  $\widetilde{B}_{\varepsilon}(z) : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{0}(\Sigma)$  and  $\overline{B}_{\varepsilon}(z) : \mathcal{B}^{1/2}(\Sigma) \to \mathcal{B}^{1/2}(\Sigma)$  be the operators defined by (4.27) and (4.29), respectively. In this chapter we prove (4.34), i.e. we show that  $\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)$  can be extended to a bounded operator from  $\mathcal{B}^{0}(\Sigma)$  to  $\mathcal{B}^{1/2}(\Sigma)$  and that

$$\|\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)\|_{0 \to 1/2} \le C\varepsilon^{1/2} (1 + |\log(\varepsilon)|)^{1/2}$$
(B.1)

for some C > 0, which is used in (4.34) in *Step 3* in the proof of Proposition 4.10. With (4.28) and (4.32) one obtains for  $f \in \mathcal{B}^{1/2}(\Sigma)$ 

$$\left( \widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z) \right) f(t)(x_{\Sigma}) = \int_{-1}^{1} \int_{\Sigma} \left( G_{z}(x_{\Sigma} - y_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - \varepsilon s\nu(y_{\Sigma})) - G_{z}(x_{\Sigma} - y_{\Sigma} + \varepsilon(t - s)\nu(x_{\Sigma})) \right) f(s)(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, ds$$

$$(B.2)$$

for a.e.  $t \in (-1, 1)$  and for  $\sigma$ -a.e.  $x_{\Sigma} \in \Sigma$ , where  $G_z$  is the integral kernel of  $R_z = (H - z)^{-1}$ ; cf. (3.3)–(3.4). Thus, in order to show (B.1), we proceed as follows: We prove in Proposition B.2 that for fixed  $t \neq s \in (-1, 1)$  the operator formally acting on  $\psi \in L^2(\Sigma; \mathbb{C}^N)$  as

$$b_{t,s,\varepsilon}(z)\psi(x_{\Sigma}) = \int_{\Sigma} \left( G_z(x_{\Sigma} - y_{\Sigma} + \varepsilon t\nu(x_{\Sigma}) - \varepsilon s\nu(y_{\Sigma})) - G_z(x_{\Sigma} - y_{\Sigma} + \varepsilon(t - s)\nu(x_{\Sigma})) \right) \psi(y_{\Sigma}) \, d\sigma(y_{\Sigma}), \tag{B.3}$$

 $x_{\Sigma} \in \Sigma$ , gives rise to a bounded operator from  $L^{2}(\Sigma; \mathbb{C}^{N})$  to  $H^{1/2}(\Sigma; \mathbb{C}^{N})$  and we prove an estimate for its operator norm. Then, we show in Lemma B.3 that the map  $(s,t) \mapsto b_{t,s,\varepsilon}(z)$  is measurable and use (B.2) to transfer the results from  $b_{t,s,\varepsilon}(z)$  to  $\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)$ .

In the following, we always assume  $\varepsilon \in (0, \varepsilon_{ABC})$  with  $\varepsilon_{ABC} > 0$  satisfying (4.19). Recall that  $\varepsilon_{tub}$  and  $\varepsilon_{\iota}$  are the numbers that are specified in Proposition 2.12 and Proposition 2.9, respectively. Since  $\varepsilon_{tub} < \varepsilon_{\iota}$ , see the proof of Proposition 2.12, we conclude from (4.19) that  $\varepsilon_{ABC} < \frac{\varepsilon_{\iota}}{2}$ . We define for  $t \neq s \in (-1, 1)$  and  $x_{\Sigma}, y_{\Sigma} \in \Sigma$ 

$$\Delta G_z(x_{\Sigma}, y_{\Sigma}, t, s) := G_z(x_{\Sigma} - y_{\Sigma} + \varepsilon t \nu(x_{\Sigma}) - \varepsilon s \nu(y_{\Sigma})) - G_z(x_{\Sigma} - y_{\Sigma} + \varepsilon (t - s) \nu(x_{\Sigma})).$$
(B.4)

Moreover, we introduce for  $t \neq s \in (-1, 1)$  and  $x_{\Sigma}, y_{\Sigma} \in \Sigma$  the quantities

$$\begin{aligned} z_0(x_{\Sigma}, y_{\Sigma}, t, s) &:= x_{\Sigma} - y_{\Sigma} + \varepsilon t \nu(x_{\Sigma}) - \varepsilon s \nu(y_{\Sigma}) = \iota(x_{\Sigma}, \varepsilon t) - \iota(y_{\Sigma}, \varepsilon s), \\ z_1(x_{\Sigma}, y_{\Sigma}, t, s) &:= x_{\Sigma} - y_{\Sigma} + \varepsilon (t - s) \nu(x_{\Sigma}) = \iota(x_{\Sigma}, \varepsilon (t - s)) - \iota(y_{\Sigma}, 0), \\ z_{\mu}(x_{\Sigma}, y_{\Sigma}, t, s) &:= \mu z_0(x_{\Sigma}, y_{\Sigma}, t, s) + (1 - \mu) z_1(x_{\Sigma}, y_{\Sigma}, t, s) \\ &= \iota(x_{\Sigma}, \mu \varepsilon t + (1 - \mu) \varepsilon (t - s)) - \iota(y_{\Sigma}, \mu \varepsilon s) \quad \text{for } \mu \in (0, 1), \\ L(x_{\Sigma}, y_{\Sigma}, t, s) &:= |x_{\Sigma} - y_{\Sigma}| + |\varepsilon (t - s)|. \end{aligned}$$

Then,  $\Delta G_z(x_{\Sigma}, y_{\Sigma}, t, s) = G_z(z_0(x_{\Sigma}, y_{\Sigma}, t, s)) - G_z(z_1(x_{\Sigma}, y_{\Sigma}, t, s))$ . It follows from Proposition 2.9 (ii) that for  $\mu \in [0, 1]$  the inequalities

$$C_{\iota,2}^{-1}L(x_{\Sigma}, y_{\Sigma}, t, s) \le |z_{\mu}(x_{\Sigma}, y_{\Sigma}, t, s)| \le C_{\iota,2}L(x_{\Sigma}, y_{\Sigma}, t, s)$$
(B.5)

hold. To shorten notation we also set

$$c := \frac{C_{G,2}}{C_{\iota,2}} > 0 \tag{B.6}$$

with  $C_{G,2}$  from Proposition 3.4. Furthermore, until Lemma B.3 we fix  $t \neq s \in (-1, 1)$ and hence omit the arguments t, s in the functions  $L, \Delta G_z, z_0, z_{\mu}$  and  $z_1$ .

**Lemma B.1.** Let  $G_z$  be the integral kernel of  $R_z$  in (3.3)–(3.4),  $\Delta G_z$  as in (B.4),  $l \in \{1, \ldots, p\}$ , and  $\varkappa_l$  as in (2.1). Then, the following is true:

(i) There exists C > 0 which does not depend on  $\varepsilon$ , t, and s such that

$$|\Delta G_z(x_{\Sigma}, y_{\Sigma})| \le C \varepsilon L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})}$$

for all  $x_{\Sigma}, y_{\Sigma} \in \Sigma$ .

(ii) There exists C > 0 which does not depend on  $\varepsilon$ , t, and s such that

$$\left|\frac{d}{dx'_k}\Delta G_z(\varkappa_l(x'), y_{\Sigma})\right| \le C\varepsilon L(\varkappa_l(x'), y_{\Sigma})^{-\theta} e^{-cL(\varkappa_l(x'), y_{\Sigma})}$$

for all  $k \in \{1, \ldots, \theta - 1\}$ ,  $y_{\Sigma} \in \Sigma$ , and  $x' \in \varkappa_l^{-1}(\Sigma)$ .

*Proof.* Before we prove (i) and (ii), we show a useful estimate of the difference

 $z_0(x_{\Sigma}, y_{\Sigma}) - z_1(x_{\Sigma}, y_{\Sigma})$ . Since  $\varepsilon_{ABC} < \frac{\varepsilon_{\iota}}{2}$ , it follows from Proposition 2.9 (ii) that

$$\begin{aligned} |z_{0}(x_{\Sigma}, y_{\Sigma}) - z_{1}(x_{\Sigma}, y_{\Sigma})| &= |\varepsilon t \nu(x_{\Sigma}) - \varepsilon s \nu(y_{\Sigma}) - \varepsilon(t - s) \nu(x_{\Sigma})| \\ &= \varepsilon |s| |\nu(x_{\Sigma}) - \nu(y_{\Sigma})| \\ &\leq \frac{\varepsilon}{\varepsilon_{ABC}} |\varepsilon_{ABC} \nu(x_{\Sigma}) - \varepsilon_{ABC} \nu(y_{\Sigma})| \\ &\leq \frac{\varepsilon}{\varepsilon_{ABC}} (|x_{\Sigma} - y_{\Sigma}| + |x_{\Sigma} + \varepsilon_{ABC} \nu(x_{\Sigma}) - y_{\Sigma} - \varepsilon_{ABC} \nu(y_{\Sigma})|) \\ &= \frac{\varepsilon}{\varepsilon_{ABC}} (|x_{\Sigma} - y_{\Sigma}| + |\iota(x_{\Sigma}, \varepsilon_{ABC}) - \iota(y_{\Sigma}, \varepsilon_{ABC})|) \\ &\leq \frac{1 + C_{\iota,2}}{\varepsilon_{ABC}} \varepsilon |x_{\Sigma} - y_{\Sigma}| \\ &\leq \frac{1 + C_{\iota,2}}{\varepsilon_{ABC}} \varepsilon L(x_{\Sigma}, y_{\Sigma}) \end{aligned}$$
(B.7)

for all  $x_{\Sigma}, y_{\Sigma} \in \Sigma$ .

(i) Applying Lemma 2.8, Proposition 3.4, (B.5) and (B.7) yields

$$\begin{aligned} |\Delta G_z(x_{\Sigma}, y_{\Sigma})| &= |G_z(z_0(x_{\Sigma}, y_{\Sigma})) - G_z(z_1(x_{\Sigma}, y_{\Sigma}))| \\ &\leq C \sup_{\mu \in [0,1], j \in \{1, \dots, \theta\}} |\partial_j G_z(z_\mu(x_{\Sigma}, y_{\Sigma}))| |z_0(x_{\Sigma}, y_{\Sigma}) - z_1(x_{\Sigma}, y_{\Sigma})| \\ &\leq C \sup_{\mu \in [0,1], j \in \{1, \dots, \theta\}} |\partial_j G_z(z_\mu(x_{\Sigma}, y_{\Sigma}))| \varepsilon L(x_{\Sigma}, y_{\Sigma}) \\ &\leq C \sup_{\mu \in [0,1]} |z_\mu(x_{\Sigma}, y_{\Sigma})|^{-\theta} e^{-C_{G,2}|z_\mu(x_{\Sigma}, y_{\Sigma})|} \varepsilon L(x_{\Sigma}, y_{\Sigma}) \\ &\leq C \varepsilon L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} \end{aligned}$$

for all  $x_{\Sigma}, y_{\Sigma} \in \Sigma$ , where c is defined in (B.6) and C > 0 is a constant which does not depend on  $\varepsilon$ , t, and s. Hence, the claim in (i) is shown.

(ii) For  $k \in \{1, \ldots, \theta - 1\}, y_{\Sigma} \in \Sigma$ , and  $x' \in \varkappa_l^{-1}(\Sigma)$  we compute

$$\begin{split} &\frac{d}{dx'_{k}}\Delta G_{z}(\varkappa_{l}(x'),y_{\Sigma}) = \frac{d}{dx'_{k}} \Big( G_{z}(z_{0}(\varkappa_{l}(x'),y_{\Sigma})) - G_{z}(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \Big) \\ &= \sum_{j=1}^{\theta} (\partial_{j}G_{z})(z_{0}(\varkappa_{l}(x'),y_{\Sigma})) \frac{d}{dx'_{k}}(z_{0}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &- \sum_{j=1}^{\theta} (\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \frac{d}{dx'_{k}}(z_{1}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &= \sum_{j=1}^{\theta} \left( (\partial_{j}G_{z})(z_{0}(\varkappa_{l}(x'),y_{\Sigma})) - (\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \right) \frac{d}{dx'_{k}}(z_{0}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &+ \sum_{j=1}^{\theta} (\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \frac{d}{dx'_{k}}(z_{0}(\varkappa_{l}(x'),y_{\Sigma}) - z_{1}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &= \sum_{j=1}^{\theta} \left( (\partial_{j}G_{z})(z_{0}(\varkappa_{l}(x'),y_{\Sigma})) - (\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \right) \frac{d}{dx'_{k}}(z_{0}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &+ \sum_{j=1}^{\theta} \varepsilon_{\delta}(\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) - (\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma}))) \frac{d}{dx'_{k}}(z_{0}(\varkappa_{l}(x'),y_{\Sigma}))[j] \\ &+ \sum_{j=1}^{\theta} \varepsilon_{\delta}(\partial_{j}G_{z})(z_{1}(\varkappa_{l}(x'),y_{\Sigma})) \frac{d}{dx'_{k}}(\nu_{l}(x'))[j], \end{split}$$

where the convention  $\nu_l(x') = \nu(\varkappa_l(x'))$  was used in the last step. To estimate the second sum we use (B.5), Proposition 3.4 and  $\zeta_l \in C_b^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$  and obtain

$$\begin{split} \left| \sum_{j=1}^{\theta} \varepsilon s(\partial_j G_z)(z_1(\varkappa_l(x'), y_{\Sigma})) \frac{d}{dx'_k}(\nu_l(x'))[j] \right| \\ &\leq C \varepsilon \sup_{j \in \{1, \dots, \theta\}} |(\partial_j G_z)(z_1(\varkappa_l(x'), y_{\Sigma}))| \|D\nu_l\|_{L^{\infty}(\mathbb{R}^{\theta-1}; \mathbb{R}^{\theta \times (\theta-1)})} \\ &\leq C \varepsilon L(\varkappa_l(x'), y_{\Sigma})^{-\theta} e^{-cL(\varkappa_l(x'), y_{\Sigma})}. \end{split}$$

For the remaining part given by

$$\sum_{j=1}^{\theta} \left( (\partial_j G_z)(z_0(\varkappa_l(x'), y_{\Sigma})) - (\partial_j G_z)(z_1(\varkappa_l(x'), y_{\Sigma})) \right) \frac{d}{dx'_k} (z_0(\varkappa_l(x'), y_{\Sigma}))[j] \quad (B.8)$$

we proceed in the same way as in the proof of (i). Using Lemma 2.8 as well as  $\zeta_l \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$  one can show that the absolute value of the expression in (B.8) is bounded by the term

$$C \sup_{\mu \in [0,1], n, j \in \{1,\dots,\theta\}} |\partial_n \partial_j G_z(z_\mu(\varkappa_l(x'), y_\Sigma))| \varepsilon L(\varkappa_l(x'), y_\Sigma),$$

which, in turn, is according to Proposition 3.4 and (B.5) also bounded by

$$C\varepsilon L(\varkappa_l(x'), y_{\Sigma})^{-\theta} e^{-cL(\varkappa_l(x'), y_{\Sigma})}.$$

To estimate the operator norm of  $b_{t,s,\varepsilon}(z)$  in Proposition B.2 below we make use of a partition of unity  $\varphi_1, \ldots, \varphi_p$  for  $\Sigma$  subordinate to  $W_1, \ldots, W_p$  with the additional property that the derivatives are uniformly bounded; the existence of such a partition of unity is shown in Proposition A.4.

**Proposition B.2.** Let  $t \neq s \in (-1,1)$  and  $\varepsilon \in (0, \varepsilon_{ABC})$ . Then, (B.3) gives rise to a bounded operator  $b_{t,s,\varepsilon}(z) : L^2(\Sigma; \mathbb{C}^N) \to H^{1/2}(\Sigma; \mathbb{C}^N)$  and there exists a C > 0which does not depend on  $\varepsilon$ , t, and s such that

$$\|b_{t,s,\varepsilon}(z)\|_{L^{2}(\Sigma;\mathbb{C}^{N})\to H^{1/2}(\Sigma;\mathbb{C}^{N})} \leq C\left(\varepsilon(1+|\log(\varepsilon|t-s|)|)\right)^{1/2}\frac{1}{|t-s|^{1/2}}.$$
 (B.9)

*Proof.* We split this proof into four steps. In *Step 1* we verify the preliminary estimate

$$\sup_{x_{\Sigma}\in\Sigma} \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{j} e^{-cL(x_{\Sigma}, y_{\Sigma})} d\sigma(y_{\Sigma}) \le C \begin{cases} 1 + |\log(\varepsilon|t - s|)|, & j = 1 - \theta, \\ \frac{1}{\varepsilon|t - s|}, & j = -\theta, \end{cases}$$
(B.10)

which will be used in Step 2 and Step 3 to obtain bounds for  $b_{t,s,\varepsilon}(z)$  viewed as an operator from  $L^2(\Sigma; \mathbb{C}^N)$  to  $L^2(\Sigma; \mathbb{C}^N)$  and from  $L^2(\Sigma; \mathbb{C}^N)$  to  $H^1(\Sigma; \mathbb{C}^N)$ , respectively. Finally, we conclude with an interpolation argument (B.9) in Step 4.

Step 1. Let  $x_{\Sigma} \in \Sigma$  and  $j \in \{1 - \theta, -\theta\}$ . Recall that  $\Sigma$  satisfies Definition 2.1 and let  $\varphi_1, \ldots, \varphi_p \in C_b^{\infty}(\mathbb{R}^{\theta})$  be the partition of unity from Proposition A.4. Using the definition of the boundary integral, we can write

$$\int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{j} e^{-cL(x_{\Sigma}, y_{\Sigma})} d\sigma(y_{\Sigma})$$
  
= 
$$\sum_{n=1}^{p} \int_{\varkappa_{n}^{-1}(\Sigma)} L(x_{\Sigma}, \varkappa_{n}(y'))^{j} e^{-cL(x_{\Sigma}, \varkappa_{n}(y'))} \varphi_{n}(\varkappa_{n}(y')) \sqrt{1 + |\nabla \zeta_{n}(y')|^{2}} dy'.$$

Hence,  $0 \leq \varphi_n \leq 1$ ,  $\zeta_n \in C_b^2(\mathbb{R}^{\theta-1};\mathbb{R})$ , and  $\varkappa_n^{-1}(\Sigma) \subset \mathbb{R}^{\theta-1}$  yield

$$\int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{j} e^{-cL(x_{\Sigma}, y_{\Sigma})} d\sigma(y_{\Sigma}) \leq C \max_{n \in \{1, \dots, p\}} \int_{\mathbb{R}^{\theta-1}} L(x_{\Sigma}, \varkappa_{n}(y'))^{j} e^{-cL(x_{\Sigma}, \varkappa_{n}(y'))} dy'$$
$$\leq C \max_{n \in \{1, \dots, p\}} \int_{\mathbb{R}^{\theta-1}} c^{j} L(x_{\Sigma}, \varkappa_{n}(y'))^{j} e^{-cL(x_{\Sigma}, \varkappa_{n}(y'))} dy',$$

where C > 0 does not depend on  $x_{\Sigma}$ , t, s, and  $\varepsilon$ . Next, let  $n \in \{1, \ldots, p\}$  and fix  $x'_n \in \mathbb{R}^{\theta-1}$  such that  $|x_{\Sigma} - \varkappa_n(x'_n)| = \min_{y' \in \mathbb{R}^{\theta-1}} |x_{\Sigma} - \varkappa_n(y')|$ . With this choice we obtain for all  $y' \in \mathbb{R}^{\theta-1}$ 

$$\begin{aligned} \frac{1}{2} |x'_n - y'| &\leq \frac{1}{2} |\varkappa_n(x'_n) - \varkappa_n(y')| \\ &\leq \frac{1}{2} (|x_{\Sigma} - \varkappa_n(y')| + |x_{\Sigma} - \varkappa_n(x'_n)|) \\ &\leq |x_{\Sigma} - \varkappa_n(y')|. \end{aligned}$$

This implies for any  $y' \in \mathbb{R}^{\theta-1}$ 

$$cL(x_{\Sigma}, \varkappa_n(y')) = c\big(|x_{\Sigma} - \varkappa_n(y')| + \varepsilon|t - s|\big) \ge \frac{c}{2}|x'_n - y'| + c\varepsilon|t - s|.$$

Moreover,  $a \mapsto a^j e^{-a}$ , a > 0,  $j \in \{1 - \theta, -\theta\}$ , is a monotonically decreasing function and therefore we get with  $\rho(x'_n, y') := \frac{c}{2}|x'_n - y'|$ 

$$\begin{split} \int_{\Sigma} c^{j} L(x_{\Sigma}, y_{\Sigma})^{j} e^{-cL(x_{\Sigma}, y_{\Sigma})} \, d\sigma(y_{\Sigma}) \\ &\leq C \max_{n \in \{1, \dots, p\}} \int_{\mathbb{R}^{\theta-1}} (\rho(x'_{n}, y') + c\varepsilon |t-s|)^{j} e^{-\rho(x'_{n}, y') - c\varepsilon |t-s|} \, dy' \\ &\leq C \int_{0}^{\infty} (\rho + c\varepsilon |t-s|)^{j} e^{-\rho - c\varepsilon |t-s|} \rho^{\theta-2} d\rho \\ &\leq C \int_{0}^{\infty} (\rho + c\varepsilon |t-s|)^{j+\theta-2} e^{-\rho - c\varepsilon |t-s|} \, d\rho \\ &= C \int_{c\varepsilon |t-s|}^{\infty} \rho^{j+\theta-2} e^{-\rho} \, d\rho \\ &\leq C \begin{cases} 1 + |\log(\varepsilon |t-s|)|, \quad j = 1-\theta, \\ \frac{1}{\varepsilon |t-s|}, \qquad j = -\theta, \end{cases} \end{split}$$

where C > 0 does not depend on  $x_{\Sigma}$ , t, s, and  $\varepsilon$ . This proves (B.10).

Step 2. In this step we verify the estimate

$$\|b_{t,s,\varepsilon}(z)\psi\|_{L^2(\Sigma;\mathbb{C}^N)} \le C\varepsilon(1+|\log(\varepsilon|t-s|)|)\|\psi\|_{L^2(\Sigma;\mathbb{C}^N)}, \quad \psi \in L^2(\Sigma;\mathbb{C}^N).$$
(B.11)

In fact, with the help of the Cauchy-Schwarz inequality, Lemma B.1 (i), and (B.10)
we obtain for  $\psi \in L^2(\Sigma; \mathbb{C}^N)$  and  $x_{\Sigma} \in \Sigma$ 

$$\begin{aligned} \left| b_{t,s,\varepsilon}(z)\psi(x_{\Sigma}) \right|^{2} &= \left| \int_{\Sigma} \Delta G_{z}(x_{\Sigma}, y_{\Sigma})\psi(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \right|^{2} \\ &\leq \int_{\Sigma} \left| \Delta G_{z}(x_{\Sigma}, y_{\Sigma}) \right| \, d\sigma(y_{\Sigma}) \int_{\Sigma} \left| \Delta G_{z}(x_{\Sigma}, y_{\Sigma}) \right| \left| \psi(y_{\Sigma}) \right|^{2} \, d\sigma(y_{\Sigma}) \\ &\leq C\varepsilon^{2} \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} \, d\sigma(y_{\Sigma}) \\ &\quad \cdot \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}) \\ &\leq C\varepsilon^{2} (1 + \left| \log(\varepsilon |t - s|) \right|) \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}). \end{aligned}$$
(B.12)

Now, Fubini's theorem and (B.10) show

$$\begin{split} &\int_{\Sigma} |b_{t,s,\varepsilon}(z)\psi(x_{\Sigma})|^2 \, d\sigma(x_{\Sigma}) \\ &\leq C\varepsilon^2 (1+|\log(\varepsilon|t-s|)|) \int_{\Sigma} \int_{\Sigma} L(x_{\Sigma},y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma},y_{\Sigma})} |\psi(y_{\Sigma})|^2 \, d\sigma(y_{\Sigma}) \, d\sigma(x_{\Sigma}) \\ &= C\varepsilon^2 (1+|\log(\varepsilon|t-s|)|) \int_{\Sigma} \int_{\Sigma} L(x_{\Sigma},y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma},y_{\Sigma})} \, d\sigma(x_{\Sigma}) |\psi(y_{\Sigma})|^2 \, d\sigma(y_{\Sigma}) \\ &\leq C\varepsilon^2 (1+|\log(\varepsilon|t-s|)|)^2 \int_{\Sigma} |\psi(y_{\Sigma})|^2 \, d\sigma(y_{\Sigma}), \end{split}$$

which yields (B.11).

Step 3. Next, we prove the estimate

$$\|b_{t,s,\varepsilon}(z)\psi\|_{H^1(\Sigma;\mathbb{C}^N)} \le C \frac{1}{|t-s|} \|\psi\|_{L^2(\Sigma;\mathbb{C}^N)}, \quad \psi \in L^2(\Sigma;\mathbb{C}^N).$$
(B.13)

Let  $\psi \in L^2(\Sigma; \mathbb{C}^N)$  and  $\varkappa_l(x') = x_{\Sigma} \in \Sigma$  with  $x' \in \mathbb{R}^{\theta-1}$ . By Proposition A.4 the function  $\varphi_l$  and its derivatives are bounded. Thus, with  $\widetilde{\varphi}_l := \varphi_l \circ \varkappa_l$  we have

$$\begin{aligned} \left| \frac{d}{dx'_{k}} \Big( \widetilde{\varphi}_{l}(x')(b_{t,s,\varepsilon}(z)\psi)(\varkappa_{l}(x')) \Big) \right|^{2} \\ &\leq 2 |\widetilde{\varphi}_{l}(x') \frac{d}{dx'_{k}}(b_{t,s,\varepsilon}(z)\psi)(\varkappa_{l}(x'))|^{2} + 2 \Big| \Big( \frac{d}{dx'_{k}} \widetilde{\varphi}_{l}(x') \Big) b_{t,s,\varepsilon}(z)\psi(\varkappa_{l}(x')) \Big|^{2} \\ &\leq C \Big( \Big| \frac{d}{dx'_{k}}(b_{t,s,\varepsilon}(z)\psi)(\varkappa_{l}(x')) \Big|^{2} + |b_{t,s,\varepsilon}(z)\psi(\varkappa_{l}(x'))| \Big). \end{aligned}$$

Using the dominated convergence theorem and the properties of  $\Delta G_z$  stated in Lemma B.1 one obtains

$$\frac{d}{dx'_k}(b_{t,s,\varepsilon}(z)\psi)(\varkappa_l(x')) = \int_{\Sigma} \frac{d}{dx'_k} \Delta G_z(\varkappa_l(x'), y_{\Sigma})\psi(y_{\Sigma}) \, d\sigma(y_{\Sigma}).$$

Hence, we get with the Cauchy-Schwarz inequality, Lemma B.1 (ii),  $x_{\Sigma} = \varkappa_l(x')$ , and (B.10)

$$\begin{split} \left| \frac{d}{dx'_{k}} (b_{t,s,\varepsilon}(z)\psi)(\varkappa_{l}(x')) \right|^{2} &= \left| \int_{\Sigma} \frac{d}{dx'_{k}} \Delta G_{z}(\varkappa_{l}(x'), y_{\Sigma})\psi(y_{\Sigma}) \, d\sigma(y_{\Sigma}) \right|^{2} \\ &\leq \int_{\Sigma} \left| \frac{d}{dx'_{k}} \Delta G_{z}(\varkappa_{l}(x'), y_{\Sigma}) \right| \, d\sigma(y_{\Sigma}) \int_{\Sigma} \left| \frac{d}{dx'_{k}} \Delta G_{z}(\varkappa_{l}(x'), y_{\Sigma}) \right| |\psi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}) \\ &\leq C \varepsilon^{2} \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} \, d\sigma(y_{\Sigma}) \\ &\quad \cdot \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}) \\ &\leq C \frac{\varepsilon}{|t-s|} \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}). \end{split}$$

According to (B.12) we can estimate

$$|b_{t,s,\varepsilon}(z)\psi(\varkappa_l(x'))|^2 \leq C\varepsilon^2 (1+|\log(\varepsilon|t-s|)|) \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^2 \, d\sigma(y_{\Sigma}),$$
  
where  $x_{\Sigma} = \varkappa_l(x')$ . Moreover,  $1+|\log(a)| \leq C\frac{1}{a}$  for  $a \in (0, 2\varepsilon_{ABC})$  yields

$$|b_{t,s,\varepsilon}(z)\psi(\varkappa_l(x'))|^2 \le C \frac{\varepsilon}{|t-s|} \int_{\Sigma} L(x_{\Sigma}, y_{\Sigma})^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma})} |\psi(y_{\Sigma})|^2 \, d\sigma(y_{\Sigma}).$$

Thus,

$$\begin{split} \int_{\varkappa_{l}^{-1}(\Sigma)} & \left| \frac{d}{dx'_{k}} \Big( \varphi_{l}(\varkappa_{l}(x'))b_{t,s,\varepsilon}(z)\psi \Big)(x') \right|^{2} dx' \\ & \leq C \frac{\varepsilon}{|t-s|} \int_{\varkappa_{l}^{-1}(\Sigma)} \int_{\Sigma} (L(\varkappa_{l}(x'),y_{\Sigma})^{1-\theta} + L(\varkappa_{l}(x'),y_{\Sigma})^{-\theta}) \\ & \cdot e^{-cL(\varkappa_{l}(x'),y_{\Sigma})} |\psi(y_{\Sigma})|^{2} d\sigma(y_{\Sigma}) dx' \\ & \leq C \frac{\varepsilon}{|t-s|} \int_{\varkappa_{l}^{-1}(\Sigma)} \int_{\Sigma} (L(\varkappa_{l}(x'),y_{\Sigma})^{1-\theta} + L(\varkappa_{l}(x'),y_{\Sigma})^{-\theta}) \\ & \cdot e^{-cL(\varkappa_{l}(x'),y_{\Sigma})} |\psi(y_{\Sigma})|^{2} d\sigma(y_{\Sigma}) \sqrt{1+|\nabla\zeta_{l}(x')|^{2}} dx' \\ & \leq C \frac{\varepsilon}{|t-s|} \int_{\Sigma} \int_{\Sigma} (L(\varkappa_{l}(x'),y_{\Sigma})^{1-\theta} + L(x_{\Sigma},y_{\Sigma})^{-\theta}) e^{-cL(x_{\Sigma},y_{\Sigma})} |\psi(y_{\Sigma})|^{2} d\sigma(y_{\Sigma}) d\sigma(x_{\Sigma}). \end{split}$$

Therefore, Fubini's theorem, (B.10) and using  $1 + |\log(a)| \le C_a^1$  for  $a \in (0, 2\varepsilon_{ABC})$  again yield

$$\int_{\varkappa_l^{-1}(\Sigma)} \left| \frac{d}{dx'_k} \Big( \varphi_l(\varkappa_l(x')) b_{t,s,\varepsilon}(z) \psi \Big)(x') \right|^2 dx' \le C \frac{1}{|t-s|^2} \|\psi\|_{L^2(\Sigma;\mathbb{C}^N)}^2.$$

This estimate, the definition of the norm in  $H^1(\Sigma; \mathbb{C}^N)$ , see (2.2), and (B.11) imply (B.13).

Step 4. By Proposition 2.2 (i), we have  $H^{1/2}(\Sigma; \mathbb{C}^N) = [L^2(\Sigma; \mathbb{C}^N), H^1(\Sigma; \mathbb{C}^N)]_{1/2}$ and using that also  $L^2(\Sigma; \mathbb{C}^N) = [L^2(\Sigma; \mathbb{C}^N), L^2(\Sigma; \mathbb{C}^N)]_{1/2}$  we conclude from the bounds (B.11) and (B.13) together with (xiii) from Section 2.1 that

$$\begin{aligned} \|b_{t,s,\varepsilon}(z)\|_{L^{2}(\Sigma;\mathbb{C}^{N})\to H^{1/2}(\Sigma;\mathbb{C}^{N})} \\ &\leq C\|b_{t,s,\varepsilon}(z)\|_{L^{2}(\Sigma;\mathbb{C}^{N})\to L^{2}(\Sigma;\mathbb{C}^{N})}^{1/2}\|b_{t,s,\varepsilon}(z)\|_{L^{2}(\Sigma;\mathbb{C}^{N})\to H^{1}(\Sigma;\mathbb{C}^{N})}^{1/2} \\ &\leq C\big(\varepsilon(1+|\log(\varepsilon|t-s|)|)\big)^{1/2}\frac{1}{|t-s|^{1/2}}. \end{aligned}$$

This completes the proof of Proposition B.2.

**Lemma B.3.** Let  $\varepsilon \in (0, \varepsilon_{ABC})$ . Then, the operator-valued function

$$F: (-1,1)^2 \to \mathcal{L}(L^2(\Sigma; \mathbb{C}^N), H^{1/2}(\Sigma; \mathbb{C}^N)),$$
$$F(t,s) = \begin{cases} b_{t,s,\varepsilon}(z), & \text{if } t \neq s, \\ 0, & \text{if } t = s, \end{cases}$$

is measurable.

Proof. It suffices to prove that  $(F(\cdot, \cdot)\varphi, \psi)_{H^{1/2}(\Sigma;\mathbb{C}^N)}$  is measurable on  $(-1, 1)^2$  for all  $\varphi \in L^2(\Sigma;\mathbb{C}^N)$  and  $\psi \in H^{1/2}(\Sigma;\mathbb{C}^N)$ ; cf. Definition 2.13. For this, we prove that the function  $(F(\cdot, \cdot)\varphi, \psi)_{H^{1/2}(\Sigma;\mathbb{C}^N)}$  is continuous on  $\mathcal{O} := (-1, 1)^2 \setminus \{(t, t) : t \in (-1, 1)\}$ . Let  $(t, s) \in \mathcal{O}$  be fixed and let us consider the case t > s. We choose a sequence  $((t_n, s_n))_{n \in \mathbb{N}}$  in  $\mathcal{O}$  which converges to (t, s). It is no restriction to assume that  $\frac{3}{2}(t_n - s_n) > t - s > \frac{1}{2}(t_n - s_n)$  holds for all  $n \in \mathbb{N}$ . Then,

$$L(x_{\Sigma}, y_{\Sigma}, t_n, s_n) = |x_{\Sigma} - y_{\Sigma}| + \varepsilon |t_n - s_n| < |x_{\Sigma} - y_{\Sigma}| + 2\varepsilon |t - s| \le 2L(x_{\Sigma}, y_{\Sigma}, t, s)$$

and in a similar way

$$L(x_{\Sigma}, y_{\Sigma}, t_n, s_n)^{1-\theta} \le \left(\frac{2}{3}\right)^{1-\theta} L(x_{\Sigma}, y_{\Sigma}, t, s)^{1-\theta}.$$

Moreover, as  $|t_n - s_n| \ge 0 = |t - s| - |t - s| \ge |t - s| - 2$ , one has

$$e^{-cL(x_{\Sigma}, y_{\Sigma}, t_n, s_n)} \leq e^{-c(|x_{\Sigma} - y_{\Sigma}| + \varepsilon |t-s|) + 2c\varepsilon} \leq e^{2c\varepsilon_{ABC}} e^{-cL(x_{\Sigma}, y_{\Sigma}, t, s)}$$

Combining Lemma B.1 (i) with the latter three displayed formulas yields the existence of a constant C > 0 which is independent of  $x_{\Sigma}, y_{\Sigma}, t, s, t_n, s_n$ , and  $\varepsilon$  such that

$$\begin{aligned} |\Delta G_z(x_{\Sigma}, y_{\Sigma}, t_n, s_n)| &\leq C \varepsilon L(x_{\Sigma}, y_{\Sigma}, t_n, s_n)^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma}, t_n, s_n)} \\ &\leq C \varepsilon L(x_{\Sigma}, y_{\Sigma}, t, s)^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma}, t, s)}. \end{aligned}$$
(B.14)

We claim that  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  converges weakly to  $b_{t,s,\varepsilon}(z)\varphi$  in  $L^2(\Sigma;\mathbb{C}^N)$ . Let the function  $\gamma \in L^2(\Sigma;\mathbb{C}^N)$  be fixed. Then,

$$\begin{aligned} &((b_{t_n,s_n,\varepsilon}(z) - b_{t,s,\varepsilon}(z))\varphi,\gamma)_{L^2(\Sigma;\mathbb{C}^N)} \\ &= \int_{\Sigma} \int_{\Sigma} \left\langle (\Delta G_z(x_{\Sigma}, y_{\Sigma}, t_n, s_n) - \Delta G_z(x_{\Sigma}, y_{\Sigma}, t, s))\varphi(y_{\Sigma}), \gamma(x_{\Sigma}) \right\rangle d\sigma(y_{\Sigma}) \, d\sigma(x_{\Sigma}). \end{aligned}$$

The integrand on the right-hand side converges pointwise almost everywhere to zero, as  $n \to \infty$ . Moreover, (B.14) shows that the integrand is bounded by  $M(x_{\Sigma}, y_{\Sigma})|\varphi(y_{\Sigma})||\gamma(x_{\Sigma})|$  with

$$M(x_{\Sigma}, y_{\Sigma}) := C \varepsilon L(x_{\Sigma}, y_{\Sigma}, t, s)^{1-\theta} e^{-cL(x_{\Sigma}, y_{\Sigma}, t, s)}$$

Applying the Cauchy-Schwarz inequality twice, Fubini's theorem, and the symmetry relation  $M(x_{\Sigma}, y_{\Sigma}) = M(y_{\Sigma}, x_{\Sigma})$  yields

$$\begin{split} \left( \int_{\Sigma} \int_{\Sigma} M(x_{\Sigma}, y_{\Sigma}) |\varphi(y_{\Sigma})| |\gamma(x_{\Sigma})| \, d\sigma(y_{\Sigma}) \, d\sigma(x_{\Sigma}) \right)^{2} \\ &\leq \int_{\Sigma} \left( \int_{\Sigma} M(x_{\Sigma}, y_{\Sigma}) |\varphi(y_{\Sigma})| \, d\sigma(y_{\Sigma}) \right)^{2} \, d\sigma(x_{\Sigma}) \|\gamma\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} \\ &\leq \int_{\Sigma} \int_{\Sigma} M(x_{\Sigma}, y_{\Sigma}) |\varphi(y_{\Sigma})|^{2} \, d\sigma(y_{\Sigma}) \int_{\Sigma} M(x_{\Sigma}, y_{\Sigma}) \, d\sigma(y_{\Sigma}) \, d\sigma(x_{\Sigma}) \|\gamma\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} \\ &\leq C \left( \sup_{x_{\Sigma} \in \Sigma} \int_{\Sigma} M(x_{\Sigma}, y_{\Sigma}) \, d\sigma(y_{\Sigma}) \right)^{2} \|\varphi\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} \|\gamma\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2}. \end{split}$$

Furthermore, with (B.10) we see that

$$\sup_{x_{\Sigma}\in\Sigma}\int_{\Sigma}M(x_{\Sigma},y_{\Sigma})\,d\sigma(y_{\Sigma})\leq C\varepsilon(1+|\log(\varepsilon|t-s|)|)<\infty.$$

Hence,  $((b_{t_n,s_n,\varepsilon}(z) - b_{t,s,\varepsilon}(z))\varphi,\gamma)_{L^2(\Sigma;\mathbb{C}^N)} \to 0$  for  $n \to \infty$  by applying the dominated convergence theorem. Since  $\gamma \in L^2(\Sigma;\mathbb{C}^N)$  was arbitrary, we conclude that  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  converges weakly to  $b_{t,s,\varepsilon}(z)\varphi$  in  $L^2(\Sigma;\mathbb{C}^N)$ .

Next, we show that  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  converges weakly to  $b_{t,s,\varepsilon}(z)\varphi$  in the space  $H^{1/2}(\Sigma;\mathbb{C}^N)$ , which shows the claimed continuity. For this, we note that Proposition B.2 and  $\frac{3}{2}(t_n - s_n) > t - s > 0$  imply that  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  is a bounded sequence in  $H^{1/2}(\Sigma;\mathbb{C}^N)$ . Let us assume that  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  does not converge weakly to  $b_{t,s,\varepsilon}(z)\varphi$  in  $H^{1/2}(\Sigma;\mathbb{C}^N)$ . Then, the  $H^{1/2}$ -boundedness implies that there exists a weakly convergent subsequence  $(b_{t_{nk},s_{nk},\varepsilon}(z)\varphi)_{k\in\mathbb{N}}$  which converges to some  $\varphi' \in H^{1/2}(\Sigma;\mathbb{C}^N)$  with  $\varphi' \neq b_{t,s,\varepsilon}(z)\varphi$ . However, in this case  $(b_{t_{nk},s_{nk},\varepsilon}(z)\varphi)_{k\in\mathbb{N}}$  would also converge weakly to  $\varphi'$  in  $L^2(\Sigma;\mathbb{C}^N)$  which contradicts the first part of the proof. Hence,  $(b_{t_n,s_n,\varepsilon}(z)\varphi)_{n\in\mathbb{N}}$  converges weakly to  $b_{t,s,\varepsilon}(z)\varphi$  in  $H^{1/2}(\Sigma;\mathbb{C}^N)$  and therefore,  $((b_{t_n,s_n,\varepsilon}(z)\varphi,\psi)_{H^{1/2}(\Sigma;\mathbb{C}^N)})_{n\in\mathbb{N}}$  converges to  $(b_{t,s,\varepsilon}(z)\varphi,\psi)_{H^{1/2}(\Sigma;\mathbb{C}^N)}$  for all  $\psi \in H^{1/2}(\Sigma;\mathbb{C}^N)$ .

After all these preliminary considerations we are prepared to prove (4.34).

*Proof of* (4.34). Let  $f \in \mathcal{B}^0(\Sigma)$  be fixed. Using Proposition B.2, the Cauchy-Schwarz inequality, and Fubini's theorem we obtain

$$\begin{split} &\int_{-1}^{1} \left( \int_{-1}^{1} \|b_{t,s,\varepsilon}(z)f(s)\|_{H^{1/2}(\Sigma;\mathbb{C}^{N})} \, ds \right)^{2} dt \\ &\leq C \int_{-1}^{1} \left( \int_{-1}^{1} \left( \varepsilon(1+|\log(\varepsilon|t-s|)|) \right)^{1/2} \frac{1}{|t-s|^{1/2}} \|f(s)\|_{L^{2}(\Sigma;\mathbb{C}^{N})} \, ds \right)^{2} dt \\ &\leq C \int_{-1}^{1} \left( \int_{-1}^{1} \left( \varepsilon(1+|\log(\varepsilon|t-s|)|) \right)^{1/2} \frac{1}{|t-s|^{1/2}} \, ds \\ &\quad \cdot \int_{-1}^{1} \left( \varepsilon(1+|\log(\varepsilon|t-s|)|) \right)^{1/2} \frac{1}{|t-s|^{1/2}} \|f(s)\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} \, ds \right) dt \\ &\leq C \sup_{s \in (-1,1)} \left( \int_{-1}^{1} \left( \varepsilon(1+|\log(\varepsilon|t-s|)|) \right)^{1/2} \frac{1}{|t-s|^{1/2}} \, dt \right)^{2} \int_{-1}^{1} \|f(s)\|_{L^{2}(\Sigma;\mathbb{C}^{N})}^{2} \, ds \\ &\leq C \varepsilon(1+|\log(\varepsilon)|) \|f\|_{0}^{2}, \end{split}$$

where we used that for  $\varepsilon > 0$  sufficiently small one has

$$\begin{split} \sup_{s \in (-1,1)} \int_{-1}^{1} \left( \varepsilon (1 + |\log(\varepsilon|t - s|)|) \right)^{1/2} \frac{1}{|t - s|^{1/2}} \, dt \\ &\leq \int_{-2}^{2} \left( \varepsilon (1 + |\log(\varepsilon|\tau|)|) \right)^{1/2} \frac{1}{|\tau|^{1/2}} \, d\tau \\ &\leq C \varepsilon^{1/2} (1 + |\log(\varepsilon)|)^{1/2} \int_{-2}^{2} (1 + |\log(|\tau|)|)^{1/2} \frac{1}{|\tau|^{1/2}} \, d\tau \\ &\leq C \varepsilon^{1/2} (1 + |\log(\varepsilon)|)^{1/2} \int_{-2}^{2} (1 + |\log(|\tau|)|)^{1/2} \frac{1}{|\tau|^{1/2}} \, d\tau \end{split}$$

Combined with Lemma B.3, (2.8) and Proposition 2.15 this shows that the Bochner integral  $\int_{-1}^{1} b_{t,s,\varepsilon}(z)f(s) ds \in H^{1/2}(\Sigma; \mathbb{C}^N)$  exists for a.e.  $t \in (-1, 1)$  and that the function  $t \mapsto \int_{-1}^{1} b_{t,s,\varepsilon}(z)f(s) ds \in H^{1/2}(\Sigma; \mathbb{C}^N)$  is measurable. Hence, the mapping

$$\mathbb{B}_{\varepsilon}(z) : \mathcal{B}^{0}(\Sigma) \to \mathcal{B}^{1/2}(\Sigma),$$
$$\mathbb{B}_{\varepsilon}(z)f(t) := \int_{-1}^{1} b_{t,s,\varepsilon}(z)f(s) \, ds,$$
(B.15)

is well-defined, bounded, and  $\|\mathbb{B}_{\varepsilon}(z)\|_{0\to 1/2} \leq C\varepsilon^{1/2}(1+|\log(\varepsilon)|^{1/2})$ . By (B.2), (B.3), and Proposition 2.18 (iii) we also have

$$(\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z))f(t) = \int_{-1}^{1} b_{t,s,\varepsilon}(z)f(s) \, ds = \mathbb{B}_{\varepsilon}(z)f(t) \tag{B.16}$$

for all  $f \in \mathcal{B}^{1/2}(\Sigma)$ . Therefore,  $\widetilde{B}_{\varepsilon}(z) - \overline{B}_{\varepsilon}(z)$  can be extended to a bounded operator from  $\mathcal{B}^0(\Sigma)$  to  $\mathcal{B}^{1/2}(\Sigma)$  and (4.34) is true.  $\Box$ 

### Appendix C. Additional results for Section 5.1.2 and Section 8.2

In this chapter, which is based on [15], we provide results which are used in Section 5.1.2 and Section 8.2. We begin by stating a fitting version of the Schur test.

**Lemma C.1.** Let k be a measurable function in  $\mathbb{R}^{\theta-1} \times \mathbb{R}^{\theta-1}$  with values in  $\mathbb{C}^{N \times N}$ and  $\widetilde{k} \in L^1(\mathbb{R}^{\theta-1})$  such that

$$|k(x',y')| \le \widetilde{k}(x'-y')$$
 for a.e.  $x',y' \in \mathbb{R}^{\theta-1}$ .

Then, k induces an integral operator  $K : L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N) \to L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ , which is bounded by  $\|\widetilde{k}\|_{L^1(\mathbb{R}^{\theta-1})}$ . Moreover, if  $k \in C^1(\mathbb{R}^{\theta-1} \times \mathbb{R}^{\theta-1}; \mathbb{C}^{N \times N})$  and

$$\sum_{l=1}^{\theta-1} \left| \frac{d}{dx'_l} k(x', y') \right| \le \widetilde{k}(x' - y') \quad \text{for a.e. } x', y' \in \mathbb{R}^{\theta-1},$$

then the induced operator K also acts as bounded operator from  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  and the corresponding operator norm is bounded by  $C\|\widetilde{k}\|_{L^1(\mathbb{R}^{\theta-1})}$ , where C > 0 does not depend on k or  $\widetilde{k}$ .

*Proof.* The first assertion is an immediate consequence of the Schur test; see for instance [44, Chapter III, Example 2.4]. Next, let us prove the second assertion. We start by choosing  $g \in C_0^{\infty}(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ . Our assumptions and dominated convergence show that in this case, Kg is differentiable and

$$\partial_l(Kg)(x') = \int_{\mathbb{R}^{\theta-1}} \frac{d}{dx'_l} k(x', y') g(y') \, dy'.$$

Hence, applying the Schur test shows

$$||Kg||_{H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)} \le C ||k||_{L^1(\mathbb{R}^{\theta-1})} ||g||_{L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)}.$$

The fact that  $C_0^{\infty}(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  is dense in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ , see [54, the text above eq. (3.22)], the completeness of  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  and the  $L^2$ -continuity of K imply that the estimate is also valid for  $g \in L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ .

Our next goal is to use the functions  $\vartheta_{n'}^a$ ,  $n' \in \mathbb{Z}$ , from Corollary A.3 (for  $n = \theta - 1$ ) to construct operators based on a uniformly bounded sequence of operators. We start by providing a useful variant of the Cotlar-Stein lemma.

**Lemma C.2.** Let  $\mathcal{H}$  and  $\mathcal{G}$  be Hilbert spaces, let  $(\mathcal{A}_{n'})_{n'\in\mathbb{Z}^{\theta-1}}$  be a family of uniformly bounded operators acting from  $\mathcal{H}$  to  $\mathcal{G}$ . Moreover, assume that there exists a number  $M \in \mathbb{N}$  such that for every  $n' \in \mathbb{Z}^{\theta-1}$  exist at most M indices  $m' \in \mathbb{Z}^{\theta-1}$  such that  $\mathcal{A}_{n'}^*\mathcal{A}_{m'}$  and  $\mathcal{A}_{n'}\mathcal{A}_{m'}^*$  are nonzero operators. Then, the family  $(\mathcal{A}_{n'})_{n'\in\mathbb{Z}^{\theta-1}}$  is strongly summable (in the sense of (v) of Section 2.1) and for  $\mathcal{A} = \sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \mathcal{A}_{n'}$  the estimate

$$\left\|\mathcal{A}\right\|_{\mathcal{H}\to\mathcal{G}} \le M \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\|\mathcal{A}_{n'}\right\|_{\mathcal{H}\to\mathcal{G}}$$

is valid.

*Proof.* Our assumptions guarantee

$$\sup_{n'\in\mathbb{Z}^{\theta-1}}\sum_{m'\in\mathbb{Z}^{\theta-1}} \|\mathcal{A}_{n'}\mathcal{A}_{m'}^*\|_{\mathcal{G}\to\mathcal{G}}^{1/2} \leq M \sup_{n'\in\mathbb{Z}^{\theta-1}} \|\mathcal{A}_{n'}\|_{\mathcal{H}\to\mathcal{G}},$$
$$\sup_{n'\in\mathbb{Z}^{\theta-1}}\sum_{m'\in\mathbb{Z}^{\theta-1}} \|\mathcal{A}_{n'}^*\mathcal{A}_{m'}\|_{\mathcal{H}\to\mathcal{H}}^{1/2} \leq M \sup_{n'\in\mathbb{Z}^{\theta-1}} \|\mathcal{A}_{n'}\|_{\mathcal{H}\to\mathcal{G}}.$$

Hence, the assertions follow from the Cotlar-Stein lemma; see [38, Lemma 18.6.5].□

**Proposition C.3.** Let  $a \in (0,b)$  for  $a \ b > 0$ ,  $(\vartheta_{n'}^a)_{n' \in \mathbb{Z}^{\theta-1}}$  be the sequence from Corollary A.3 (for  $n = \theta - 1$ ) and  $(A_{n'})_{n' \in \mathbb{Z}^{\theta-1}}$  be a sequence of uniformly bounded operators in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . Then,

$$A = \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\text{st.}} \vartheta_{n'}^a A_{n'} \vartheta_{n'}^a$$

is a well-defined operator in  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$  which is bounded by  $11^{\theta-1} \sup_{n' \in \mathbb{Z}^{\theta-1}} ||A_{n'}||_{0 \to 0}$ . Moreover, if  $(A_{n'})_{n' \in \mathbb{Z}^{\theta-1}}$  is also a family of uniformly bounded operators acting from  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^{1}(\mathbb{R}^{\theta-1})$ , then A also acts as a bounded operator from  $\mathcal{B}^{0}(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^{1}(\mathbb{R}^{\theta-1})$  and  $||A||_{0 \to 1} \leq \frac{C}{a} \sup_{n' \in \mathbb{Z}^{\theta-1}} ||A_{n'}||_{0 \to 1}$ , where C > 0 does not depend on  $a \in (0, b)$ .

Proof. Let us start by proving the assertions where we consider A and  $A_{n'}, n' \in \mathbb{Z}^{\theta-1}$ , as operators acting from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$ . We set  $\mathcal{A}_{n'} := \vartheta^a_{n'} \mathcal{A}_{n'} \vartheta^a_{n'}$ . Since a fixed ball B(an', 3a) overlaps with at most  $11^{\theta-1}$  balls of the type B(am', 3a),  $m' \in \mathbb{Z}^{\theta-1}$ , there exist for every  $n' \in \mathbb{Z}^{\theta-1}$  at most  $M = 11^{\theta-1}$  indices  $m' \in \mathbb{Z}^{\theta-1}$  such that  $\mathcal{A}_{n'}\mathcal{A}^*_{m'} \neq 0$  and  $\mathcal{A}^*_{n'}\mathcal{A}_{m'} \neq 0$ . Moreover, for all  $n' \in \mathbb{Z}^{\theta-1}$  we have

$$\|\mathcal{A}_{n'}\|_{0\to 0} \le \|\vartheta_{n'}^{a}\|_{0\to 0} \|A_{n'}\|_{0\to 0} \|\vartheta_{n'}^{a}\|_{0\to 0} \le \|A_{n'}\|_{0\to 0}.$$

Thus, by Lemma C.2 the assertions where A and  $A_{n'}$ ,  $n' \in \mathbb{Z}^{\theta-1}$ , are considered as bounded operators from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  are true. Next, we assume that  $A_{n'}$ ,  $n' \in \mathbb{Z}^{\theta-1}$ , act as uniformly bounded operators from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^1(\mathbb{R}^{\theta-1})$ . Using again the fact that a fixed ball B(an', 3a) overlaps with at most  $11^{\theta-1}$  balls of the type B(am', 3a),  $m' \in \mathbb{Z}^{\theta-1}$ , shows that there exist for every  $n' \in \mathbb{Z}^{\theta-1}$  at most  $M = 11^{\theta-1}$  indices  $m' \in \mathbb{Z}^{\theta-1}$  such that  $\mathcal{A}_{n'}\mathcal{A}_{m'}^{0*1} \neq 0$  and  $\mathcal{A}_{n'}^{0*1}\mathcal{A}_{m'} \neq 0$ , where the expressions  $\mathcal{A}_{n'}^{0*1}$  and  $\mathcal{A}_{m'}^{0*1}$  denote the adjoint operators of  $\mathcal{A}_{n'}$  and  $\mathcal{A}_{m'}$ , respectively, considered as operators mapping from  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  to  $\mathcal{B}^1(\mathbb{R}^{\theta-1})$ . Furthermore, applying Corollary A.3 gives us for all  $n' \in \mathbb{Z}^{\theta-1}$ 

$$\begin{aligned} \|\mathcal{A}_{n'}\|_{0\to 1} &\leq \|\vartheta_{n'}^{a}\|_{1\to 1} \|A_{n'}\|_{0\to 1} \|\vartheta_{n'}^{a}\|_{0\to 0} \\ &\leq \|\vartheta_{n'}^{a}\|_{1\to 1} \|A_{n'}\|_{0\to 1} \\ &\leq C \|\vartheta_{n'}^{a}\|_{W_{\infty}^{1}(\mathbb{R}^{\theta-1})} \|A_{n'}\|_{0\to 1} \\ &\leq \frac{C}{a} \|A_{n'}\|_{0\to 1} \\ &\leq \frac{C}{a} \|A_{n'}\|_{0\to 1}. \end{aligned}$$

Applying Lemma C.2 again concludes the proof.

As a corollary we obtain a result for series of operators in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ .

**Corollary C.4.** Let  $a \in (0, b)$  for  $a \ b > 0$ ,  $(\vartheta_{n'})_{n' \in \mathbb{Z}^{\theta-1}}$  be the sequence from Corollary A.3 (for  $n = \theta - 1$ ) and  $(\underline{A_{n'}})_{n' \in \mathbb{Z}^{\theta-1}}$  be a sequence of uniformly bounded operators in  $L^2(\mathbb{R}^{\theta-1}; \mathbb{C}^N)$ . Then,

$$\underline{A} = \sum_{n' \in \mathbb{Z}^{\theta-1}}^{\mathrm{st.}} \vartheta^a_{n'} \underline{A_{n'}} \vartheta^a_{n'}$$

is a well-defined operator in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  which is bounded by

$$11^{\theta-1} \sup_{n' \in \mathbb{Z}^{\theta-1}} \|A_{n'}\|_{L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N) \to L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)}$$

Moreover, if  $(\underline{A}_{n'})_{n'\in\mathbb{Z}^{\theta-1}}$  is also a family of uniformly bounded operators acting from  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ , then  $\underline{A}$  also acts as a bounded operator from  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  to  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  and

$$\left\|\underline{A}\right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})} \leq \frac{C}{a} \sup_{n'\in\mathbb{Z}^{\theta-1}} \left\|\underline{A}_{n'}\right\|_{L^{2}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})\to H^{1}(\mathbb{R}^{\theta-1};\mathbb{C}^{N})},$$

where C > 0 does not depend on  $a \in (0, b)$ .

*Proof.* With the help of the operators  $\mathfrak{J}$  and  $\mathfrak{J}^*$  defined in (2.10) and (2.11) (for  $\mathcal{O} = (-1, 1)$ ), respectively, we are able to write

$$\sum_{n'\in\mathbb{Z}^{\theta-1}}^{\mathrm{st.}}\vartheta_{n'}^{a}\underline{A_{n'}}\vartheta_{n'}^{a}=\mathfrak{J}^{*}\Big(\sum_{n'\in\mathbb{Z}^{\theta-1}}\vartheta_{n'}^{a}\big(\frac{1}{4}\mathfrak{J}\underline{A_{n'}}\mathfrak{J}^{*}\big)\vartheta_{n'}^{a}\Big)\mathfrak{J}.$$

Here, we used  $\mathfrak{J}^*\mathfrak{J} = 2I$  and the fact that  $\vartheta_{n'}^a$  commutes with  $\mathfrak{J}$  and  $\mathfrak{J}^*$ , if one considers  $\vartheta_{n'}^a$  once as a multiplication operator in  $\mathcal{B}^0(\mathbb{R}^{\theta-1})$  (or  $\mathcal{B}^1(\mathbb{R}^{\theta-1})$ ) and once as a multiplication operator in  $L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)$  (or  $H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)$ ). Thus, the result follows from applying Proposition C.3 to  $(A_{n'})_{n'\in\mathbb{Z}^{\theta-1}} = (\frac{1}{2}\mathfrak{J}\underline{A_{n'}}\mathfrak{J}^*)_{n'\in\mathbb{Z}^{\theta-1}}$  and

$$\begin{aligned} \|\mathfrak{J}\|_{L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)\to 0} &= \|\mathfrak{J}^*\|_{0\to L^2(\mathbb{R}^{\theta-1};\mathbb{C}^N)} \\ &= \|\mathfrak{J}\|_{H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)\to 1} = \|\mathfrak{J}^*\|_{1\to H^1(\mathbb{R}^{\theta-1};\mathbb{C}^N)} = \sqrt{2}. \end{aligned}$$

- Г		

### References

- S. Albeverio, F. Gesztesy, R. Høegh-Krohn, and H. Holden. Solvable Models in Quantum Mechanics. AMS Chelsea Publishing, Providence, RI, second edition, 2005. With an appendix by Pavel Exner.
- [2] J.-P. Antoine, F. Gesztesy, and J. Shabani. Exactly solvable models of sphere interactions in quantum mechanics. J. Phys. A 20: 3687–3712, 1987.
- [3] N. Arrizabalaga, L. Le Treust, and N. Raymond. On the MIT bag model in the non-relativistic limit. *Comm. Math. Phys.* 354(2): 641–669, 2017.
- [4] N. Arrizabalaga, A. Mas, and L. Vega. Shell interactions for Dirac operators. J. Math. Pures Appl. (9) 102(4): 617–639, 2014.
- [5] N. Arrizabalaga, A. Mas, and L. Vega. Shell interactions for Dirac operators: on the point spectrum and the confinement. SIAM J. Math. Anal. 47(2): 1044– 1069, 2015.
- [6] N. Arrizabalaga, A. Mas, and L. Vega. An isoperimetric-type inequality for electrostatic shell interactions for Dirac operators. *Comm. Math. Phys.* 344(2): 483–505, 2016.
- [7] J. Behrndt, P. Exner, M. Holzmann, and V. Lotoreichik. Approximation of Schrödinger operators with δ-interactions supported on hypersurfaces. *Math. Nachr.* 290(8-9): 1215–1248, 2017.
- [8] J. Behrndt, P. Exner, M. Holzmann, and V. Lotoreichik. On the spectral properties of Dirac operators with electrostatic  $\delta$ -shell interactions. J. Math. Pures Appl. (9) 111: 47–78, 2018.
- [9] J. Behrndt, P. Exner, M. Holzmann, and V. Lotoreichik. On Dirac operators in  $\mathbb{R}^3$  with electrostatic and Lorentz scalar  $\delta$ -shell interactions. *Quantum Studies* 6(3): 295–314, 2019.
- [10] J. Behrndt, P. Exner, M. Holzmann, and M. Tušek. On two dimensional Dirac operators with δ-shell interactions supported on unbounded curves with straight ends. To appear in the volume "Singularities, Asymptotics, and Limiting Models" of the Springer-INdAM series.

- [11] J. Behrndt, D. Frymark, M. Holzmann, and C. Stelzer-Landauer. Nonrelativistic Limit of Generalized MIT Bag Models and Spectral Inequalities. *Math. Phys. Anal. Geom.* 36(2): 12 (30 pages), 2024.
- [12] J. Behrndt and M. Holzmann. On Dirac operators with electrostatic  $\delta$ -shell interactions of critical strength. J. Spectral Theory 10(1): 147–184, 2020.
- [13] J. Behrndt, M. Holzmann, T. Ourmieres-Bonafos, and K. Pankrashkin. Twodimensional Dirac operators with singular interactions supported on closed curves. J. Funct. Anal. 279(8): 108700 (47 pages), 2020.
- [14] J. Behrndt, M. Holzmann, and C. Stelzer. Approximation of Dirac operators with  $\delta$ -shell potentials in the norm resolvent sense. Preprint: arXiv:2308.13344.
- [15] J. Behrndt, M. Holzmann, and C. Stelzer-Landauer. Approximation of Dirac operators with  $\delta$ -shell potentials in the norm resolvent sense, II. Quantitative results. Preprint: In preparation.
- [16] J. Behrndt, M. Holzmann, C. Stelzer-Landauer, and G. Stenzel. Boundary triples and Weyl functions for Dirac operators with singular interactions. *Rev. Math. Phys.* 36(2): 2350036 (65 pages), 2024.
- [17] J. Behrndt, M. Holzmann, and M. Tušek. Spectral transition for Dirac operators with electrostatic  $\delta$ -shell potentials supported on the straight line. Integral Equations Operator Theory 94(3): 33 (13 pages), 2022.
- [18] J. Behrndt, M. Holzmann, and M. Tušek. Two-dimensional Dirac operators with general  $\delta$ -shell interactions supported on a straight line. J. Phys. A 56(4): 045201 (29 pages), 2023.
- [19] B. Benhellal. Spectral properties of the Dirac operator coupled with  $\delta$ -shell interactions. Lett. Math. Phys. 112(3): 52 (52 pages), 2022.
- [20] B. Benhellal and K. Pankrashkin. Curvature contribution to the essential spectrum of Dirac operators with critical shell interactions. *Pure Appl. Anal.* 6(1): 237–252, 2024
- [21] K. Brewster, D. Mitrea, I. Mitrea, and M. Mitrea. Extending Sobolev functions with partially vanishing traces from locally ( $\varepsilon, \delta$ )-domains and applications to mixed boundary problems. J. Funct. Anal. 266(7): 4314–4421, 2014.
- [22] M. G. Calkin, D. Kiang, and Y. Nogami. Proper treatment of the delta function potential in the one-dimensional Dirac equation. Am. J. Phys. 55(8): 737–739, 1987.
- [23] M. G. Calkin, D. Kiang, and Y. Nogami. Nonlocal separable potential in the one-dimensional Dirac equation. Phys. Rev. C 38(2): 1076–1077, 1988.

- [24] B. Cassano, V. Lotoreichik, A. Mas, and M. Tušek. General  $\delta$ -shell interactions for the two-dimensional Dirac operator: self-adjointness and approximation. *Rev. Mat. Iberoam.* 39(4): 1443–1492, 2023.
- [25] S. N. Chandler-Wilde, D. P. Hewett, and A. Moiola. Interpolation of Hilbert and Sobolev spaces: Quantitative estimates and counterexamples. *Mathematika* 61(2): 414–443, 2015.
- [26] S. Chandler-Wilde, D. P. Hewett, and A. Moiola. Corrigendum: Interpolation of Hilbert and Sobolev spaces: Quantitative estimates and counterexamples. *Mathematika* 68(4): 1393–1400, 2022.
- [27] T. R. Chow. A spectral theory for direct integrals of operators. Math. Ann. 188: 285–303, 1970.
- [28] J. Dittrich, P. Exner, and P. Šeba. Dirac operators with a spherically symmetric  $\delta$ -shell interaction. J. Math. Phys. 30(12): 2875–2882, 1989.
- [29] L. C. Evans and R. F. Gariepy. Measure Theory and Fine Properties of Functions. Studies in Advanced Mathematics, CRC Press, Inc. 1992.
- [30] P. Exner and T. Ichinose. Geometrically induced spectrum in curved leaky wires. J. Phys. A 34(7): 1439–1450. 2001.
- [31] P. Exner and S. Kondej. Bound states due to a strong  $\delta$  interaction supported by a curved surface. J. Phys. A 36(2): 443–457, 2003.
- [32] F. Gesztesy, Y. Latushkin, M. Mitrea, and M. Zinchenko. Non-self-adjoint operators, infinite determinants, and some applications. *Russ. J. Math. Phys.* 12(4): 443–471, 2005.
- [33] L. Heriban, M. Holzmann, C. Stelzer-Landauer, G. Stenzel, and M. Tušek. Twodimensional Schrödinger operators with non-local singular potentials. J. Math. Anal. Appl. 549 (2): 129498 (39 pages), 2025.
- [34] L. Heriban and M. Tušek. Non-self-adjoint relativistic point interaction in one dimension. J. Math. Anal. Appl. 516(2): 126536 (28 pages), 2022.
- [35] L. Heriban and M. Tušek. Non-local relativistic  $\delta$ -shell interactions. Lett. Math. Phys. 114(3): 79 (18 pages), 2024.
- [36] A. Holderrieth. Matrix Multiplication Operators Generating One Parameter Semigroups. Semigroup Forum 42(2): 155–166, 1991.
- [37] M. Holzmann, T. Ourmières-Bonafos, and K. Pankrashkin. Dirac operators with Lorentz scalar shell interactions. *Rev. Math. Phys.* 30(5): 1850013 (46 pages), 2018.

- [38] L. Hörmander. Analysis of Partial Differential Operators III: Pseudodifferential Operators. Classics in Mathematics. Springer-Verlag, Berlin, 2007. Reprint of the 1994 edition.
- [39] J. S. Howland. On the Weinstein–Aronszajn formula. Arch. Rat. Mech. Anal. 39: 323–339, 1970.
- [40] R. J. Hughes. Renormalization of the Relativistic Delta Potential in One Dimension. Lett. Math. Phys. 34(4): 395–406, 1995.
- [41] R. J. Hughes. Relativistic point interactions: approximation by smooth potentials. *Rep. Math. Phys.* 39(3): 425–432, 1997.
- [42] R. J. Hughes. Finite-rank perturbations of the Dirac operator. J. Math. Anal. Appl. 238(1): 67–81, 1999.
- [43] T. Hytönen, J. v. Neerven, M. Veraar, and L. Weis. Analysis in Banach Spaces: Volume I: Martingales and Littlewood-Paley Theory, volume 48 of A Series of Modern Surveys in Mathematics. Springer International Publishing, Cham, 2016.
- [44] T. Kato. Perturbation Theory for Linear Operators. Classics in Mathematics. Springer-Verlag, Berlin, 1995. Reprint of the 1980 edition.
- [45] T. Kato. Wave operators and similarity for some non-selfadjoint operators. Math. Ann. 162: 258–279, 1966.
- [46] R. Konno and S. T. Kuroda. On the finiteness of perturbed eigenvalues. J. Fac. Sci. Univ. Tokyo Sect. I 13: 55–63, 1966.
- [47] W. Kühnel. Differential Geometry. Curves-Surfaces-Manifolds. American Mathematical Society, Providence, RI, 2002.
- [48] J. Marschall. The trace of Sobolev-Slobodeckij spaces on Lipschitz domains. Manuscripta Math. 58(1-2): 47–65, 1987.
- [49] A. Mas. Dirac operators, shell interactions and discontinuous gauge functions across the boundary. J. Math. Phys. 58(2): 022301 (14 pages), 2017.
- [50] A. Mas and F. Pizzichillo. The relativistic spherical  $\delta$ -shell interaction in  $\mathbb{R}^3$ : spectrum and approximation. J. Math. Phys. 58(8): 082102 (14 pages), 2017.
- [51] A. Mas and F. Pizzichillo. Klein's Paradox and the relativistic δ-shell interaction in R<sup>3</sup>. Anal. & PDE 11(3): 705–744, 2018.
- [52] B. H. J. McKellar and G. J. Stephenson. Relativistic quarks in one-dimensional periodic structures. *Phys. Rev. C* 35(6): 2262–2271, 1987.

- [53] B. H. J. McKellar and G. J. Stephenson. Klein paradox and the Dirac-Kronig-Penney model. Phys. Rev. A 36(6): 2566–2569, 1987.
- [54] W. McLean. Strongly Elliptic Systems and Boundary Integral Equations. Cambridge University Press, Cambridge, 2000.
- [55] F. W. J. Olver, A. B. Olde Daalhuis, D. W. Lozier, B. I. Schneider, R. F. Boisvert, C. W. Clark, B. R. Miller, B. V. Saunders, H. S. Cohl, and M. A. McClain, eds. *NIST Digital Library of Mathematical Functions*. https://dlmf.nist.gov, Release 1.2.2 of 2024-09-15.
- [56] T. Ourmières-Bonafos and L. Vega. A strategy for self-adjointness of Dirac operators: Application to the MIT bag model and  $\delta$ -shell interactions. *Publ.* Mat. 62(2): 397–437, 2018.
- [57] T. Palmer. Banach Algebras and the General Theory of \*-Algebras, Vol. 1. Cambridge University Press, Cambridge, 1994.
- [58] I. Yu. Popov. The operator extension theory, semitransparent surface and short range potential. Math. Proc. Cambridge Phil. Soc. 118(3): 555–563, 1995.
- [59] V. Rabinovich. Two-Dimensional Dirac Operators with Interactions on Unbounded Smooth Curves. Russ. J. Math. Phys. 28(4): 524–542, 2021.
- [60] V. Rabinovich. Dirac Operators with Delta-Interactions on Smooth Hypersurfaces in R<sup>n</sup>. J. Fourier Anal. Appl. 28(2): 20, 2022.
- [61] V. Rabinovich. Fredholm property and essential spectrum of 3-D Dirac operators with regular and singular potentials. *Complex Var. Elliptic Equ.* 67(4): 938–961, 2022.
- [62] M. Reed and B. Simon. Methods of Modern Mathematical Physics I. Functional Analysis. Academic Press, 1972.
- [63] M. Reed and B. Simon. Methods of Modern Mathematical Physics II. Fourier Analysis, Self-adjointness. Academic Press, 1975.
- [64] M. Reed and B. Simon. Methods of Modern Mathematical Physics IV. Analysis of Operators. Academic Press, 1977.
- [65] V. Růžek. One-dimensional relativistic point interactions approximations by regular potentials, application to models of the Dirac materials. Master's thesis, https://dspace.cvut.cz/bitstream/handle/10467/95493/ F4-DP-2021-Ruzek-Vaclav-dp\_mf\_21\_ruzek.pdf?sequence=-1&isAllowed= y, 2021
- [66] K. Schmüdgen. Unbounded Self-adjoint Operators on Hilbert Space. Graduate Texts in Mathematics. Springer Netherlands, 2012.

- [67] P. Šeba. Klein's paradox and the relativistic point interaction. Lett. Math. Phys. 18(1): 77–86, 1989.
- [68] S. Shimada. The approximation of the Schrödinger operators with penetrable wall potentials in terms of short range Hamiltonians. J. Math. Kyoto Univ. 32: 583–592, 1992.
- [69] M. A. Shubin. Spectral theory of elliptic operators on non-compact manifolds. Astérisque 207: 35–108, 1992.
- [70] E. Stein. Singular Integrals and Differentiability Properties of Functions. Monographs in harmonic analysis. Princeton University Press, 1970.
- [71] B. Thaller. The Dirac Equation. Texts and Monographs in Physics. Springer-Verlag, Berlin, 1992.
- [72] M. Tušek. Approximation of one-dimensional relativistic point interactions by regular potentials revised. Lett. Math. Phys. 110(10): 2585–2601, 2020.
- [73] J. Weidmann. Lineare Operatoren in Hilberträumen. Teil I Grundlagen. Mathematische Leitfäden. B. G. Teubner, Stuttgart, 2000.
- [74] M. Zreik. On the approximation of the  $\delta$ -shell interaction for the 3-D Dirac operator. *Cubo* 26(3): 489–505, 2024.

Vol. 1	Steffen Alvermann
	Effective Viscoelastic Behavior of Cellular Auxetic Materials
	2008
	ISBN 978-3-902465-92-4
Vol. 2	Sendy Fransiscus Tantono
	The Mechanical Behaviour of a Soilbag
	under Vertical Compression
	2008
	ISBN 978-3-902465-97-9
Vol. 3	Thomas Rüberg
	Non-conforming FEM/BEM Coupling in Time Domain
	2008
	ISBN 978-3-902465-98-6
Vol. 4	Dimitrios E. Kiousis
	Biomechanical and Computational Modeling
	of Atherosclerotic Arteries
	2008
	ISBN 978-3-85125-023-7
Vol. 5	Lars Kielhorn
	A Time-Domain Symmetric Galerkin BEM
	for Viscoelastodynamics
	2009
	ISBN 978-3-85125-042-8

Vol. 6	Gerhard Unger
	Analysis of Boundary Element Methods
	for Laplacian Eigenvalue Problems
	2009
	ISBN 978-3-85125-081-7
Vol. 7	Gerhard Sommer
	Mechanical Properties of Healthy and
	Diseased Human Arteries
	2010
	ISBN 978-3-85125-111-1
Vol. 8	Mathias Nenning
	Infinite Elements for Elasto- and Poroelastodynamics
	2010
	ISBN 978-3-85125-130-2
Vol. 9	Thanh Xuan Phan
	Boundary Element Methods for Boundary Control Problems
	2011
	ISBN 978-3-85125-149-4
Vol. 10	Loris Nagler
	Simulation of Sound Transmission through
	Poroelastic Plate-like Structures
	2011
	ISBN 978-3-85125-153-1

Vol. 11	Markus Windisch
	Boundary Element Tearing and Interconnecting
	Methods for Acoustic and Electromagnetic Scattering
	2011
	ISBN 978-3-85125-152-4
Vol. 12	Christian Walchshofer
	Analysis of the Dynamics at the Base of a Lifted Strongly
	<b>Buoyant Jet Flame Using Direct Numerical Simulation</b>
	2011
	ISBN 978-3-85125-185-2
Vol. 13	Matthias Messner
	Fast Boundary Element Methods in Acoustics
	2012
	ISBN 978-3-85125-202-6
Vol. 14	Peter Urthaler
	Analysis of Boundary Element Methods for Wave
	Propagation in Porous Media
	2012
	ISBN 978-3-85125-216-3
Vol. 15	Peng Li
	Boundary Element Method for Wave Propagation
	in Partially Saturated Poroelastic Continua
	2012
	ISBN 978-3-85125-236-1

Vol. 16	Andreas Jörg Schriefl
	Quantification of Collagen Fiber Morphologies
	in Human Arterial Walls
	2013
	ISBN 978-3-85125-238-5
Vol. 17	Thomas S. E. Eriksson
	Cardiovascular Mechanics
	2013
	ISBN 978-3-85125-277-4
Vol. 18	Jianhua Tong
	Biomechanics of Abdominal Aortic Aneurysms
	2013
	ISBN 978-3-85125-279-8
Vol. 19	Jonathan Rohleder
	Titchmarsh–Weyl Theory and Inverse Problems
	for Elliptic Differential Operators
	2013
	ISBN 978-3-85125-283-5
Vol. 20	Martin Neumüller
	Space-Time Methods
	2013
	ISBN 978-3-85125-290-3

Vol. 21	Michael J. Unterberger
	Microstructurally-Motivated Constitutive Modeling of
	Cross-Linked Filamentous Actin Networks
	2013
	ISBN 978-3-85125-303-0
Vol. 22	Vladimir Lotoreichik
	Singular Values and Trace Formulae for Resolvent
	Power Differences of Self-Adjoint Elliptic Operators
	2013
	ISBN 978-3-85125-304-7
Vol. 23	Michael Meßner
	A Fast Multipole Galerkin Boundary Element Method
	for the Transient Heat Equation
	2014
	ISBN 978-3-85125-350-4
Vol. 24	Lorenz Johannes John
	Optimal Boundary Control in Energy Spaces
	2014
	ISBN 978-3-85125-373-3
Vol. 25	Hannah Weisbecker
	Softening and Damage Behavior of Human Arteries
	2014
	ISBN 978-3-85125-370-2

Vol. 26	Bernhard Kager		
	Efficient Convolution Quadrature based Boundary		
	Element Formulation for Time-Domain Elastodynamics		
	2015		
	ISBN 978-3-85125-382-5		
Vol. 27	Christoph M. Augustin		
	Classical and All-floating FETI Methods with		
	Applications to Biomechanical Models		
	2015		
	ISBN 978-3-85125-418-1		
Vol. 28	Elias Karabelas		
	Space-Time Discontinuous Galerkin Methods for		
	Cardiac Electromechanics		
	2016		
	ISBN 978-3-85125-461-7		
Vol. 29	Thomas Traub		
	A Kernel Interpolation Based Fast Multipole Method		
	for Elastodynamic Problems		
	2016		
	ISBN 978-3-85125-465-5		
Vol. 30	Matthias Gsell		
	Mortar Domain Decomposition Methods for		
	Quasilinear Problems and Applications		
	2017		
	ISBN 978-3-85125-522-5		

Vol. 31	Christian Kühn
	Schrödinger operators and singular infinite rank perturbations
	2017
	ISBN 978-3-85125-551-5
Vol. 32	Michael H. Gfrerer
	Vibro-Acoustic Simulation of Poroelastic Shell Structures
	2018
	ISBN 978-3-85125-573-7
Vol. 33	Markus Holzmann
	Spectral Analysis of Transmission and Boundary
	Value Problems for Dirac Operators
	2018
	ISBN 978-3-85125-642-0
Vol. 34	Osman Gültekin
	Computational Inelasticity of Fibrous Biological Tissues
	with a Focus on Viscoelasticity, Damage and Rupture
	2019
	ISBN 978-3-85125-655-0
Vol. 35	Justyna Anna Niestrawska
	Experimental and Computational Analyses of Pathological
	Soft Tissues – Towards a Better Understanding of the
	Pathogenesis of AAA
	2019
	ISBN 978-3-85125-678-9

Vol. 36	Marco Zank
	Inf-Sup Stable Space-Time Methods for Time-Dependent
	Partial Differential Equations
	2020
	ISBN 978-3-85125-721-2
Vol. 37	Christoph Irrenfried
	Convective turbulent near wall heat transfer
	at high Prandtl numbers
	2020
	ISBN 978-3-85125-724-3
Vol. 38	Christopher Albert
	Hamiltonian Theory of Resonant Transport Regimes
	in Tokamaks with Perturbed Axisymmetry
	2020
	ISBN 978-3-85125-746-5
Vol. 39	Daniel Christopher Haspinger
	Material Modeling and Simulation of Phenomena at
	the Nano, Micro and Macro Levels in Fibrous Soft Tissues
	of the Cardiovascular System
	2021
	ISBN 978-3-85125-802-8
Vol. 40	Markus Alfons Geith
	Percutaneous Coronary Intervention
	2021
	ISBN 978-3-85125-801-1

Vol. 41	Dominik Pölz		
	Space-Time Boundary Elements for		
	Retarded Potential Integral Equations		
	2021		
	ISBN 978-3-85125-811-0		
Vol. 42	Douglas Ramalho Queiroz Pacheco		
	Stable and stabilised finite element methods		
	for incompressible flows of generalised Newtonian fluids		
	2021		
	ISBN 978-3-85125-856-1		
Vol. 43	Peter Schlosser		
	Superoscillations and their Schrödinger time evolution		
	2022		
	ISBN 978-3-85125-930-8		
Vol. 44	Raphael Watschinger		
	Fast space-time boundary element methods		
	for the heat equation		
	2023		
	ISBN 978-3-85125-949-0		
Vol. 45	Ishan Gupta		
	Modelling Growth and Formation of Thrombus:		
	A Multiphasic Approach		
	2023		
	ISBN 978-3-85125-964-3		

# Monographic Series TU Graz

Computation in Engineering and Science

Vol. 46Mario GobrialSpace-time Finite Element Methodsfor the Eddy Current Problem and Applications20252025/SBN 978-3-99161-048-9Vol. 47Christian Stelzer-LandauerApproximation of Dirac Operators withDelta-Shell Potentials in the Norm Resolvent Sense2025

ISBN 978-3-99161-052-6

Approximation of Dirac Operators with Delta-Shell Potentials in the Norm Resolvent Sense

The present thesis is devoted to the approximation of Dirac operators with  $\delta$ -shell potentials supported on the boundary of a two or three-dimensional  $C^2$ -domain. These singular potentials are used as idealized replacements for potentials which are strongly localized in a neighbourhood of the support of the  $\delta$ -shell potential and they often simplify the spectral analysis. To justify the usage of such potentials it is essential to prove that Dirac operators with  $\delta$ -shell potentials can be approximated by Dirac operators with strongly localized potentials in a way which transfers the spectral properties. The most important contribution of this thesis is the establishment of conditions for the convergence of Dirac operators with strongly localized potentials in the norm resolvent sense. This type of convergence implies that the spectrum of the Dirac operator with  $\delta$ -shell potential can be completely characterized by the spectra of the approximating operators and vice versa. In the special case of electrostatic and Lorentz scalar  $\delta$ -shell potentials an explicit convergence condition is provided. Furthermore, counterexamples which imply the sharpness of this condition are also presented.

## MONOGRAPHIC SERIES TU GRAZ COMPUTATION IN ENGINEERING AND SCIENCE

Verlag der Technischen Universität Graz www.tugraz-verlag.at

ISBN 978-3-99161-052-6 ISSN 1990-357X

